Examining the Role and Neurophysiological Mechanisms of Social Support in Pain Experience

Xianwei Che

Master of Psychology

Thesis submitted in total fulfilment of the requirements for the degree of

Doctor of Philosophy



Monash Alfred Psychiatry Research Centre

Central Clinical School

Faculty of Medicine, Nursing and Health Sciences

Monash University

March 2019

Copyright notice

© Xianwei Che (2019)

I certify that I have made all reasonable efforts to secure copyright permissions for third-party content included in this thesis and have not knowingly added copyright content to my work without the owner's permission.

Table of Contents

ABSTRACT	i
LIST OF PUBLICATIONS	iv
DECLARATION	vii
ACKNOWLEDGEMENT	xii
GLOSSARY OF ABBREVIATIONS	xv
CHAPTER ONE: GENERAL INTRODUCTION AND THESIS OVERVIEW	1
1.1. Pain and Pain Mechanisms	1
1.2. Pain Modulation	4
1.2.1. Attentional Modulation of Pain	5
1.2.2. Emotional Modulation of Pain	5
1.2.3. Beliefs and Expectations' Influence on Pain	6
1.3. Social Support and Pain	7
1.4. Neural and Physiological Techniques in Social Support and Pain	9
1.4.1. Functional Magnetic Resonance Imaging (fMRI)	9
1.4.2. Magnetoencephalography (MEG)	
1.4.3 Laser-evoked Potentials (LEPs)	10
1.4.4. Assessments of the Autonomic Nervous System	11
1.5. Promising Neurophysiological Methods to Better Understand the Connection between S Support and Pain	ocial 12
1.5.1. Electroencephalography (EEG)	12
1.5.2. Transcranial Magnetic Stimulation (TMS)	14
1.6. The Clinical Importance of Understanding the Influence of Social Support on Pain	15
1.7. Summary of the Literature	15
1.8. Aims and Overview of Chapters	16
CHAPTER TWO: EFFECTS AND SOCIAL CONTEXT	
Preamble to systematic review and meta-analysis	
Paper 1	20
CHAPTER THREE: POTENTIAL PROCESSES AND MECHANISMS	
Preamble to systematic review	
Paper 2	

CHAPTER FOUR: THE THREAT OF PAIN	55
Preamble to empirical paper	55
Paper 3	57
CHAPTER FIVE: NEURAL OSCILLATIONS AND PARASYMPATHETIC ACTIVITY	68
Preamble to empirical paper	68
Paper 4	70
CHAPTER SIX: NEURAL PLASTICITY AND CONNECTIVITY	103
Preamble to empirical paper	103
Paper 5	105
CHAPTER SEVEN: GENERAL DISCUSSION	145
7.1. Summary of Main Findings	145
7.1.1. Chapter 2	145
7.1.2. Chapter 3	146
7.1.3. Chapter 4	147
7.1.4. Chapter 5	148
7.1.5. Chapter 6	150
7.2. Implications	151
7.2.1. Understanding the Effects and Mechanisms of Social Support on Pain	152
7.2.2. Novel Neuroimaging Techniques to Reveal the Mechanistic Evidence	153
7.2.3. Translational Implications for Social Support Strategies in the Management of Pain	155
7.3. Methodological Considerations	157
7.3.1. EEG Analysis	157
7.3.2. Sampling Considerations	158
7.3.3. Social Support Paradigms	159
7.3.4. Relationship Quality	160
7.3.5. Pain Induction	161
7.3.6. Potential Additional Modulating Factors	161
7.4. Future Directions	
7.4.1. Using Hypescanning to Reveal the Synchronisation between Social Partners	163
7.4.2. Oxytocin in the Social Modulation of Pain	164
7.4.3. Social Network Analysis	165

7.5. Conclusions	165
REFERENCES:	167
APPENDIX A	187
APPENDIX B	217
APPENDIX C	221

ABSTRACT

Pain is a highly subjective experience determined not only by sensation but also by an individual's psychological state and social environment. Indeed, previous research has identified the key role of cognitive and emotional factors in pain perception and the neurobiological mechanisms that may underpin these processes. More recently, a growing body of research has indicated the significance of social factors contributing to pain perception, particularly social support demonstrating a protective influence on pain. If social support is influential in modulating the experience of pain, social support may be fundamental for therapeutic strategies in the management of pain. However, optimizing social support strategies is limited by our understanding of the social contexts in which social support has an analgesic influence, the magnitude of the effect, and the neurophysiological mechanisms through which social support may have an effect on pain.

The focus of this thesis was to investigate the role of social support in pain experience and the neurophysiological mechanisms that may facilitate this effect. This was addressed through a total of five manuscripts including two review articles and three experimental studies.

Through a systematic review and meta-analysis, the first manuscript identified the social contexts within which social support may reduce pain. The results of this investigation identified that social support may have the most protective effect on pain when it is clearly communicated (e.g. via verbal support). The results further indicated the particular importance of intimate relationships in pain reduction through touching or visual representation of an intimate other. This study also quantified the effects of social support on pain-related physiological arousal whereby conclusions were highly limited by the small number of studies in this space.

i

In the second study, I undertook a systematic review to investigate the potential psychological and behavioural processes as well as neurobiological mechanisms underpinning the influence of social support on pain reduction. This review aimed to address whether evidence best supports the main or the buffering effect of social support on pain experience. The results indicated that social support may reduce pain through a buffering effect on the threat of pain, in which social support may modulate stress appraisal and attenuate neurobiological stress systems.

In study three, the buffering effect of social support was directly examined in the context of the threat of pain. An experimental protocol was used to introduce continuous threat of pain leading up to painful stimulation during which participants held the hand of a significant other, a stranger, or not at all. Electroencephalogram (EEG) and electrocardiogram (ECG) were used to evaluate neural and physiological threat of pain. The results demonstrated reduced neural (i.e. frontal theta power, 4-8 Hz) and autonomic (i.e. heart rate) response to the threat of pain under the social support (significant other) condition, which were further associated with subsequent pain reduction. In addition, neural changes in theta power were source localized to the insular cortex and the anterior cingulate cortex, regions commonly observed in the processing of threat and pain.

Study four investigated the effects of social support on the neural dynamics and parasympathetic activity to prolonged pain using EEG and ECG. A dynamic pain protocol was used to induce tonic pain perception during which visual representation of an intimate other or a stranger was provided. The results indicated reduced pain perception in the context of an intimate other, which was associated with increased frontocentral alpha activity and decreased central gamma activity. The ECG data demonstrated increased high-frequency heart-rate variability in the intimate other condition. Findings of this investigation may indicate the role of social support in modulating the gating and integration of nociception.

ii

The medial prefrontal cortex (mPFC) is thought to play a key role in the analgesic influence of social support. Study five investigated the causal role of the mPFC in orchestrating the behavioural and brain connectivity effects of social support on pain. Across sessions the activity in the mPFC was increased or decreased (or no change as in the Sham condition) by facilitatory intermittent Theta Burst Stimulation (iTBS) or suppressive continuous TBS (cTBS) respectively. The findings suggested that iTBS over the mPFC has the capacity to causally modulate pain perception and network configuration in a context-dependent manner. Specifically, visual representation of stranger increased pain and connectivity between central regions and frontoparietal regions. In contrast, visual representation of an intimate other increased connectivity only between frontal and occipital regions and did not modulate pain perception. This study also identified neuroplastic changes as evaluated by TMS-EEG and the association with social modulation of pain.

The combined results of this thesis suggest: (1) the effects of social support on pain are contextdependent; (2) social support may reduce pain by modifying stress appraisal and neurobiological stress systems prior to pain onset; (3) social support may modulate the gating and integration of nociceptive information; (4) the mPFC may play a causal role in mediating the influence of social support on pain. Together, this series of studies are an important contribution to understanding and neural mechanisms underpinning the impact of social support on pain which may have therapeutic implications.

iii

LIST OF PUBLICATIONS

Published Manuscripts Directly Relevant to this Thesis

Che, X., Cash, R., Chung, S., Fitzgerald, P., &, Fitzgibbon B. (2018). Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis. *Neuroscience & Biobehavioral Reviews*, *92*, 437-452.

Che, X., Cash, R., Ng, S., Fitzgerald, P., & Fitzgibbon B. (2018). A systematic review of the processes underlying the main and the buffering effect of social support on the experience of pain. *Clinical Journal of Pain 34*, 1061-1076.

Che, X., Cash, R., Fitzgerald, P., &, Fitzgibbon B. (2018). The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat. *Journal of Pain*, *19*, 496-505.

Submitted Manuscripts Directly Relevant to this Thesis

Che, X., Cash, R., Fitzgerald, P., &, Fitzgibbon B. Distinct neural dynamics and autonomic activity in response to viewing a romantic other compared to a stranger during pain: An EEG and HRV investigation. Under review.

Che, X., Cash, R., Fitzgerald, P., &, Fitzgibbon B. The medial prefrontal cortex as a flexible hub mediating behavioral as well as local and distributed neural effects of social support on pain: a theta burst stimulation and TMS-EEG Study. Under review.

Peer-reviewed Manuscripts Unrelated to the Thesis during Candidature

Roebuck, G.S., Urquhart, D.M., **Che, X.,** Knox, L., Fitzgerald, P.B., Cicuttini, F.M., Lee, S., Segrave, R., and Fitzgibbon, B.M. Ultra-marathon runners' psychological resilience, emotion regulation strategies and personality traits: an exploratory self-report and psychophysiological study. Under review

Li, X., Li, B., Jin, L., **Che, X.,** & Lu, M. Alexithymia and aggression: The moderating role of sostile attribution bias. Under review.

Luo, X., Qiao, L., & **Che, X.** (2018). Self-compassion modulates heart rate variability and negative affect to experimentally induced stress. *Mindfulness*, *9*, 1522-1528.

Qiao, L., Xu. L., **Che, X.,** Zhang, L., Li, Y., Xue G., Chen, A. (2018). The motivation-based promotion of proactive control: the role of salience network. *Frontiers in Human Neuroscience*, *12*, e328.

Zhang, Y., Luo, X., **Che, X**., Duan, W. (2016). Self-compassion buffers the effect of examination stress on negative affect in college students. *Frontiers in Psychology*, *7*, e1802.

Li, B., Cheng, G., Zhang, D., Wei, D., Qiao, L., Wang, X. & **Che, X**. (2016). Emotional type modulates regional brain responses to infant faces in nulliparous women. *Plos One, 11*, e0166860.

Che, X., Luo, X., Tong, D., Fitzgibbon, B., & Yang, J. (2015). Habitual suppression relates to difficulty in regulating emotion with cognitive reappraisal. *Biological Psychology*, *112*, 20-26.

Oral Presentations

Che, X., Cash, R., Fitzgerald, P., &, Fitzgibbon B. The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat. *Australasian Society for Social & Affective Neuroscience (AS4SAN)*, Melbourne, Australia, 2017.

Poster presentations

Che, X., Cash, R., Fitzgerald, P., &, Fitzgibbon B. The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat. *Australian Pain Society*, Sydney, Australia, 2018.

Che, X., Cash, R., Chung, S., Fitzgerald, P., &, Fitzgibbon B. (2018). Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis. *17th World Congress on Pain*, Boston, USA, 2018.

Awards

IASP Financial Aid for the 17th World Congress on Pain, Boston, USA, 2018.

Monash Travel Grant for the 38th Australian Pain Society Annual Meeting, Sydney, Australia, 2018

Full scholarship for overseas PhD student from China Scholarship Council (CSC) (2015-2019)

DECLARATION

I hereby declare that this thesis contains no material which has been accepted for the award of any other degree or diploma at any university or equivalent institution and that, to the best of my knowledge and belief, this thesis contains no material previously published or written by another person, except where due reference is made in the text of the thesis.

This thesis includes three (3) original papers published/accepted in peer reviewed journals and two (2) unpublished publications. The core theme of the thesis is to investigate the role and neurophysiological mechanisms of social support in pain experience. The ideas, development and writing up of all the papers in the thesis were the principal responsibility of myself, the candidate, working within the Central Clinical School under the supervision of Dr Bernadette Fitzgibbon, Dr Robin Cash, and Professor Paul Fitzgerald.

The inclusion of co-authors reflects the fact that the work came from active collaboration between researchers and acknowledges input into team-based research.

In the case of Chapters 2, 3, 4, 5, and 6, my contribution to the work involved the following:

Project design (in consultation with my supervisors and co-authors); review of appropriate literature; securing ethics approval; recruitment of participants; data collection; conducting data analysis; writing of papers. Supervisors and co-authors provided input into completed manuscript drafts.

Thesis	Publication Title	Publication	Nature and % of	Co-Author	Co-
Chapter		Status	Candidate's	Names,	Authors
			Contribution	Nature and %	Monash
				of Contribution	Student?

2	Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis	Published Neuroscience and Biobehavioral Reviews	80%. Manuscript conceptualisation, review of relevant literature, data analysis, manuscript preparation.	 Dr Bernadette Fitzgibbon. Review of manuscript, supervisory input (10%). Dr Robin Cash. Review of manuscript, supervisory input (5%). Prof Paul B. Fitzgerald. Review of manuscript, supervisory input (3%). Dr Sung Wook Chung. Screening of included studies, review of manuscript (2%). 	No (all co- authors)
3	A Systematic Review of the Processes Underlying the Main and the Buffering Effect of Social Support on the Experience of Pain	Published Clinical Journal of Pain	80%. Manuscript conceptualisation, review of relevant literature, manuscript preparation.	 Dr Bernadette Fitzgibbon. Review of manuscript, supervisory input (10%). Dr Robin Cash. Review of manuscript, supervisory input (5%). Prof Paul B. Fitzgerald. Review of manuscript, 	No

				supervisory input (3%). Mrs Sin-Ki Ng. Screening of included studies, review of manuscript (2%).	Yes
4	The social regulation of pain: autonomic and neurophysiologi cal changes associated with perceived threat	Published Journal of Pain	80%. Study conceptualisation and protocol design, securing of ethical approval, participant recruitment and testing, data collection and analysis, manuscript preparation	1) Dr Bernadette Fitzgibbon. Review of manuscript, supervisory input (10%). 2) Dr Robin Cash. Review of manuscript, supervisory input (6%). 3) Prof Paul B. Fitzgerald. Review of manuscript, supervisory input (4%).	No (all co- authors)
5	Distinct neural dynamics and autonomic activity in response to viewing a romantic other compared to a stranger during pain: An EEG and HRV investigation	Under review European Journal of Pain	80%. As above	 Dr Bernadette Fitzgibbon. Review of manuscript, supervisory input (10%). Dr Robin Cash. Review of manuscript, supervisory input (6%). 	No (all co- authors)

				3) Prof Paul B. Fitzgerald. Review of manuscript, supervisory input (4%).	
6	The Medial Prefrontal Cortex as a Flexible Hub Mediating Behavioral as well as Local and Distributed Neural Effects of Social Support on Pain: a Theta Burst Stimulation and TMS-EEG Study	Under review NeuroImage	80%. As above	 Dr Bernadette Fitzgibbon. Review of manuscript, supervisory input (8%). Dr Robin Cash. Review of manuscript, supervisory input (6%). Prof Paul B. Fitzgerald. Review of manuscript, supervisory input (2%). Dr Neil Bailey. Technical assistance (2%). Dr Sung Wook Chung. Technical assistance (2%). 	No (all co- authors)

I have renumbered sections of submitted or published papers in order to generate a consistent presentation within the thesis.

Student signature:

Date: 10/03/2019

The undersigned hereby certifies that the above declaration correctly reflects the nature and extent of the student's and co-authors' contributions to this work. In instances where I am not the responsible author, I have consulted with the responsible author to agree on the respective contributions of the authors.

Main Supervisor signature:

Date: 10/03/2019

ACKNOWLEDGEMENT

I would like to thank my main supervisor, Dr Bernadette Fitzgibbon, who offered me the opportunity to study at Monash University. Coming out of a poor country, I was looking for a PhD position from a top university. Bernie replied to my email quickly and, following discussion, offered me a PhD position. During the past three to four years you have provided enormous help to grow my knowledge, skills, confidence, and enthusiasm to science. I have learnt so much from your research expertise and you have encouraged me to think about the most important questions in science: why we need scientific research and how best to conduct it. You have always been extremely patient, especially when I made mistakes or I rushed into things. Under your supervision, I have become a more confident, skilled, and competent researcher. I am truly thankful for the great opportunity and the incredible supervision you gave to me. You have provided me a wonderful gift that I will benefit from for many years and I am not sure if you realize it! I will graduate soon but this is definitely not the endpoint of our relationship or collaboration. I will keep learning from you and let's contribute more to science.

I also want to thank Dr Robin Cash, another PhD supervisor. Robin came on board right after he joined MAPrc and truly completed the supervision. I still remember the day when Bernie suggested to add you as another supervisor because of your skills in MRI, TMS, and EMG. Throughout the candidature you have provided excellent suggestions and advice on every aspect of my PhD, spanning from research ideas and study designs to data analysis and writing up. I have always been amazed by your capabilities in elevating my data analysis and formulating the conclusions. I truly appreciate your research expertise and the enormous help you have provided to improve my research. Sometimes I felt bad that my research and publications will not help to build up your track record because we are working on different research questions. Yet, you always helped me anyway

xii

no matter how busy you are. I hope we could collaborate more in the future and contribute to the growing research in brain stimulation.

To my final supervisor, Professor Paul Fitzgerald, I would like to thank you for your patience and supervision throughout my PhD. I remember the first time Bernie and I were discussing PhD plans with you and I was disappointed in myself as I did not know much on this topic. I understand that you are super busy as a team leader and a psychiatrist, and I always appreciated that you responded to my requests quickly and kindly. You are able to point out the issues and concerns that were missed by all three of us. The most important things I have learnt from you are enthusiasm towards science, the potential for this to lead to new therapies and your unique way to lead a team which, above all else, encourages scientific endeavor, mutual respect, and team work. I am really honored to have you as a supervisor.

A special thank is also needed to my fellow PhDs and friends, Dr Sung Wook Chung, Dr Aron Hill, and Mr Caley Sullivan. After we first met, you accepted me as a member of the group. I cannot remember how much laughter we have had while we were coding or randomly talking about science and multi-culture. You are all passionate to science and are truly the pioneers in preparing the scripts for complicated analysis in the team. I have learnt so much from you and you have elevated my programming skills and my scientific thinking skills. It's impossible for me to have had a successful PhD without the tremendous help provided by you. I know some of you are going to leave MAPrc, including myself, but I sincerely wish that this is just one of the chapters of our friendship and there are many more to come.

I would also like to thank other friends and team members in the Therapeutic Brain Stimulation team. Thank you Neil, you introduced yourself to me in my first day and trained me to set up EEG studies and connectivity analysis. Thank you Melanie, Karyn, David, Hannah, Caitlyn, Sin-Ki, Laura,

xiii

Kirsten, Lenore, Megan, Manreena, and Greg, who have been really nice to me and treated me with care and support. I also want to thank the new PhDs, Aleks, Robert, Marie-Claire, and Andrea, I really enjoy working with you.

In addition, I am grateful for the economic support from the China Scholarship Council. I did not need to worry about money with the funding so that I could focus on my research. This award is wholly responsible for my ability to study abroad and pursue higher education.

Last but not least, I would like to thank my wife, my parents and my sister. You have provided all the emotional support I needed to achieve a doctoral degree. Studying on the other side of the globe I sometimes felt distressed and homesick. It has been a blessing for me to have your support, and your encouragement in believing in my potential and work ethic. You encouraged me to chase my goals, and you shared my happiness and sorrows along this journey. I can't wait to spend more time with you.

GLOSSARY OF ABBREVIATIONS

ACC	Anterior Cingulate Cortex
СВТ	Cognitive-behavioural Therapies
СРТ	Cold Pressor Test
cTBS	Continuous Theta Burst Stimulation
ECG	Electrocardiogram
EEG	Electroencephalography
fMRI	Functional Magnetic Resonance Imaging
HF-HRV	High-frequency HRV Heart Rate Variability
HRV	Heart Rate Variability
IASP	International Association for the Study of Pain
iTBS	Intermittent Theta Burst Stimulation
LEPs	Laser-evoked Potentials
MEG	Magnetoencephalography
mPFC	Medial Prefrontal Cortex
NCF	Nucleus Cuneiformis
OFC	Orbital Frontal Cortex
PAG	Periaqueductal Grey
РВ	Parabrachial Nucleus
PNS	Parasympathetic Nervous System
RVM	Ventromedial Medulla
S1	Primary Somatosensory Cortex
S2	Secondary Somatosensory Cortex
SNA	Social Network Analysis
SNS	Sympathetic Nervous System

- TEPs TMS-evoked Potentials
- TBS Theta Burst Stimulation
- TMS Transcranial Magnetic Stimulation
- VMPFC Ventromedial Prefrontal Cortex

CHAPTER ONE

General Introduction and Thesis Overview

This chapter provides an overview on pain mechanisms and pain modulation. This chapter will focus on the influence of social support on pain. It will also review neuroscience methods to investigate how social support may impact pain, as relevant to the thesis, and briefly summarises the potential of this knowledge to clinical practice. A critical review of the field is provided in chapters 2 to 5.

1.1. Pain and Pain Mechanisms

The International Association for the Study of Pain (IASP) defines pain as "an unpleasant sensory and emotional experience associated with actual or potential tissue damage, or described in terms of such damage" (IASP, 1994). Thus, to experience pain does not require nociception or injury, nor does the magnitude of an insult equate to an individual perceived pain. Rather, pain is multi-dimensional, whereby a sensory component provides information on the intensity, location, and duration of the stimulus, and an affective component drives the unpleasant or aversive feeling of pain (Haggard et al., 2013; Melzack and Casey, 1968). Pain is therefore a subjective experience, differing between individuals as a result of a complex and dynamic interaction between 'bottom-up' signalling and 'top-down' modulation as a result of environmental influence, psychosocial factors and neurobiological circuits (Ploner et al., 2017).

There are different functional meanings for acute versus chronic pain. Acute pain is caused by external or internal injury and has a distinct warning function. It drives protective behaviours to

avoid bodily injury and therefore is of particular importance for survival and evolution. In contrast, chronic pain, usually defined at lasting longer than 3 months (IASP, 1994; Loeser, 2001), is not only lacking obvious behavioural benefits but also heavily impairs quality of life (Hodges and Tucker, 2011). The fact that chronic pain has a major adverse impact on human well-being has driven numerous studies to investigate the neural mechanisms of pain with the prospect to develop more effective treatments (Mouraux and Jannetti, 2018).

Advances in neuroimaging techniques revolutionized our understanding of the central nervous representation of pain (for a review see Derbyshire, 2000; Martucci et al., 2014; Morton et al., 2016; Mouraux and Iannetti, 2018; Tracey, 2008). The 'neuromatrix' perspective (Melzack, 1999), commonly referred to as the 'pain matrix' (Tracey and Mantyh, 2007), indicates that painful stimuli activate a distributed brain network including the primary (S1) and secondary (S2) somatosensory cortex, thalamus, insular cortex, as well as the anterior cingulate cortex (ACC) and prefrontal cortex (Apkarian et al., 2005; Peyron et al., 1999; Tracey et al., 2000). This theoretical perspective suggests that the 'pain matrix' is pain-specific, i.e. mediating pain experience itself (Ploghaus et al., 1999). Moreover, some studies have indicated that brain structures constituting the 'pain matrix' may have a distinct role in mediating different components of pain (Ingvar and Hsieh, 1999). For example, the sensory component is thought to be independently and specifically represented in the S1 and S2, while the affective aspect of pain is represented in medial brain structures such as the ACC (Albe-Fessar et al., 1985; Avenanti et al., 2005). However, brain regions implicated in the 'pain matrix' are not solely activated by painful stimuli, but also by non-painful stimuli, such as auditory and tactile stimuli (for a review, see Legrain et al., 2011). Thus, other adaptions have been proposed to explain the central modulation of pain, including the 'interoception' and the 'salience network' perspective.

The 'interoception' perspective views pain as a homeostatic emotion (Craig, 2002, 2003a, b). Homeostasis of the human body is described as a dynamic and ongoing process in order to maintain

an optimal balance in the physiological condition of the body (Cannon, 1939; Craig, 2003a). In this view, pain is both a distinct sensation and a motivation that requires a behavioural response, akin to other sensations such as temperature, itch, hunger and thirst (Craig, 2003a). Moreover, pain and these body sensations are suggested to be mediated by a recently discovered lamina I 'spinothalamocortical' pathway (Craig, 2002, 2003a). Specifically, the small-diameter primary afferent fibres (e.g. A\delta and C fibre) that carry the information of the physiological status of the body (e.g. nociceptors, thermoreceptors) terminate on projection neurons in lamina I of the spinal dorsal horn. Modality specific lamina I neurons then project to the brainstem sites (e.g. the parabrachial nucleus, PB, the periaqueductal gray, PAG), which provide the central afferent pathway for homeostasis. In primates, lamina I neurons also project to the thalamus, which relays information to both the insular cortex to represent interoception, and to the ACC and orbital frontal cortex (OFC) to produce behavioural drive (Craig, 2003a, b). This argument provides a rational explanation for the association between pain, temperature, itch and other feelings from the body.

Another more recent theoretical perspective argues that brain regions in the 'pain matrix' may form part of a 'salience network' which is involved in detecting, orienting attention to, and reacting to salient sensory events (Cauda et al., 2012; Legrain et al., 2011; Seeley et al., 2007). A key concept in this argument is the 'salience' of a given stimulus, generally defined as its ability to stand out from the surrounding stimuli (Yantis, 2008). From a neural network perspective, the ACC and insular cortex are suggested to be the key nodes of the 'salience network' (Seeley et al., 2007). Similar to the homeostatic perspective, the salience system acts regardless of whether a salient stimulus is conveyed through nociceptive pathways. However, this does not mean that the salience system does not contribute to the experience of pain, but rather highlights the significance of the nociceptive system in detecting and reacting to salient changes (Legrain et al., 2011).

There is also evidence surrounding neural mechanisms in chronic pain and possible differences with acute pain. People with chronic pain have undergone both structural and functional brain changes, most consistently in the insula and prefrontal cortex (Baliki et al., 2008; Geha et al., 2008; Rodriguez-Raecke et al., 2009). Interestingly, one study demonstrated that fluctuations of spontaneous pain in chronic pain patients were uniquely associated with increased activity in the mPFC, whereby experimentally-induced thermal pain was strongly associated with increased insula activity. This study indicated the role of the mPFC in differentiating fluctuations of spontaneous pain from experimentally-induced pain in people with chronic pain (Baliki et al., 2006).

Overall, these mechanistic investigations have improved our understanding of how the brain represents pain perception and drives protective behavioral responses. These studies have also provided the groundwork for mechanistic explorations supporting top-down pain modulation. Studies have begun to reveal a cerebral network mediating top-down pain modulation, i.e. the descending pain modulatory system, which commonly includes the frontal cortex, ACC, insular cortex, amygdala, hypothalamus, PAG, nucleus cuneiformis (NCF), and rostral ventromedial medulla (RVM) (Tracey and Mantyh, 2007). This descending pain system is suggested to mediate the modulation of pain by a variety of psychosocial factors (for a review see Bushnell et al., 2013).

1.2. Pain Modulation

It is well accepted that pain is a highly subjective experience and varies across individuals and contexts. Indeed, a direct correlation between the intensity of noxious stimuli and pain perception is not always apparent, suggesting that the experience of pain is influenced by contextual processes in the top-down modulation of pain (Ossipov et al., 2010). Among other influential factors, literature to date has revealed the influence of attention (Ploner et al., 2010; Tracey et al., 2002), emotion (Hampton et al., 2015; Lapate et al., 2012), as well as beliefs and expectations (Wager et al., 2004;

Wiech et al., 2006) in modulating pain perception. Understanding these contextual factors may shed a light on the role of social support in pain which can prime positive affect (Younger et al., 2010) and may distract attention from pain (Montoya et al., 2004). Moreover, while there is now greater understanding of the factors that contribute to variability of pain experience, in both acute and chronic contexts, mechanisms implicated in the risk or resilience factors to pain remain to be determined (Bushnell et al., 2013; Ossipov et al., 2014).

1.2.1. Attentional Modulation of Pain

Numerous studies have demonstrated the role of cognition in pain perception, in particular attention (Brooks et al., 2002; Dunckley et al., 2007; Frankenstein et al., 2001; Kong et al., 2013; Seminowicz et al., 2004). In general, paying attention to pain increases pain perception whilst distraction reduces pain. In studies using neuroimaging techniques, research suggests that the descending pain modulatory system, especially the PAG, mediates these attentional effects on pain (Bantick et al., 2002; Petrovic et al., 2000; Tracey et al., 2002; Valet et al., 2004). For example, distraction away from pain can increase PAG activity and reduce pain perception compared to attending to pain (Tracey et al., 2002). Further, distraction increases the top-down gating influence of cingulo-frontal cortex on PAG and posterior thalamus activity (Valet et al., 2004). These studies together provide mechanistic evidence on how attention could affect pain.

1.2.2. Emotional Modulation of Pain

It is well-established that pain may lead to negative mood and emotion. Conversely, there is also evidence that negative mood and emotion can exacerbate pain experience (Loggia et al., 2008; Wiech and Tracey, 2009). It is thought that emotional modulation of pain alters the perceived unpleasantness of pain (Loggia et al., 2008; Villemure et al., 2003). This is contrast to the attentional effects whereby pain intensity appears to be affected more consistently (Loggia et al., 2008; Royl et

al., 2011; Villemure et al., 2003). However, it is acknowledged that research protocols used in attention and emotion may not be able to completely isolate one from the other (Bushnell et al., 2013). One study was designed to dissociate the emotional and attentional circuits in the modulation of pain, which found ACC and the insular cortex to be associated with the emotional and attentional effects on pain respectively (Villemure and Bushnell, 2009). In addition to the induction of emotion, some studies have also demonstrated the efficacy of emotion regulation strategies (e.g. reappraisal, emotion suppression) in pain reduction (Hampton et al., 2015; Lapate et al., 2012), suggesting behavioural interventions targeting emotional processes to have an effect on pain.

1.2.3. Beliefs and Expectations' Influence on Pain

Pain experience can also be modulated by pain-related cognitions such as beliefs and expectations towards pain. A prime example of how beliefs and expectations influence pain can be seen in the socalled "placebo" (or expectation) effect, in which the mere belief that one is receiving an effective treatment can induce pain relief (Amanzio and Benedetti, 1999; Benedetti et al., 1999; Price et al., 1999). Advances in placebo research have revealed the descending pathway via brainstem structures that may mediate the placebo analgesia (Petrovic et al., 2002; Wager et al., 2004; Zubieta et al., 2005). For example, administration of the opioid antagonist naloxone decreased placebo analgesia and activity in the descending pain modulatory system including the hypothalamus, PAG, RVM, which demonstrated the role of the descending pathway in placebo analgesia (Eippert et al., 2009a). Further evidence has extended the descending pain modulatory system to the spinal level in which decreased spinal cord responses were observed in placebo analgesia (Eippert et al., 2009b).

Perceived controllability is another relevant pain-related belief which refers to the extent to which one believes that pain is under control (Salomons et al., 2004). Perceived controllability has been shown to reduce pain (Wiech et al., 2006) and increase the ability to cope with pain (Jensen and Karoly, 1991; Jensen et al., 2001). There is evidence that perceived controllability may reduce painevoked activations in the ACC, insular cortex, and the somatosensory cortex (Salomons et al., 2004). The analgesic influence of perceived controllability was further shown to rely on the functioning of the prefrontal cortex in changing the threatening quality of pain (Salomons et al., 2007; Wiech et al., 2006). Together, this body of research suggests pain-specific cognitions, such as beliefs and expectations of pain, influence the activity of pain-related brain regions and the experience of pain.

1.3. Social Support and Pain

The effects of social relationships on our physiological and psychological well-being has long been observed clinically (Uchino, 2006; Uchino et al., 1996). A body of research is beginning to emerge providing experimental evidence for the role of social factors in pain modulation (Krahé et al., 2013). Indeed, the growing recognition of the importance of socials factors is reflected in the recent re-definition of pain proposed to include a social component, arguing that *"social environments determine exposure to pain, thoughts, and feelings when in pain, communication of distress to others, and others' experience and responses"* (Williams and Craig, 2016).

Among these social factors the role social support in pain experience has received the most attention (Brown et al., 2003; Goldstein et al., 2016; Roberts et al., 2015). Social support generally refers to the perception or experience that one is loved and cared for by others (Wills, 1991), although there is some debate around its definition. It is commonly accepted that social support encapsulates both 'perceived' and 'received' support, in which perceived support refers to the perception that supportive resources would be available should they be needed whereas received support means the reported exchange of supportive resources (Dunkel-Schetter and Bennett, 1990; Uchino, 2009). Based on the supportive resources, social support is also believed to have multiple types, including informational, socioemotional, and instrumental support (House, 1983; Turner, 1983).

The effects of social support on pain perception have been mixed. Brown and colleagues provided the first line of evidence that verbal communication with a friend or even a stranger reduced experimental pain (Brown et al., 2003). A series of studies then demonstrated an analgesic influence of social support in the context of touching (Goldstein et al., 2016; Master et al., 2009) or viewing (Eisenberger et al., 2011; Master et al., 2009) a significant other. However, there is also evidence showing that social support may have no effect (Modić Stanke and Ivanec, 2010) or even an adverse impact (Gallant and Hadjistavropoulos, 2017; Karmann et al., 2014; Vervoort et al., 2011) on pain.

Notably, however, different variations of social support have been manipulated across these studies, such as social presence, verbal support, social touch, and mental representation of social support. In the cases of social presence, the social partner is physically present without any verbal or physical contact with the participant (Edwards et al., 2017). In the studies of verbal support or social touch, the social partner verbally interacts with (Roberts et al., 2015) or holds the hand of the participant (Master et al., 2009). In the cases of mental representation of social support, participants view an image of a social partner without the social partner being physically present (Eisenberger et al., 2011). Perhaps, different types of social support, as well as other considerations such as the specific relationship and relationship quality, impact on whether or not social support has an effect on pain.

In order to address the discrepancy in the literature to date, Krahé and colleagues systematically reviewed evidence surrounding the influence of social support on experimental pain (Krahé et al., 2013). The authors found that the effects of social support on pain were largely influenced by the degree to which participants could perceive the social partner as helpful and the pre-existing social relationships. In general, social support is able to reduce pain when it is explicitly expressed (e.g. verbal communication, social touch) by a significant other (e.g. a romantic partner) (Krahé et al., 2013). However, given that this is a burgeoning field, the results of this analysis were limited by the

number of studies. Personality factors were also found to modulate the impact of social support on pain, notably pain catastrophizing (Sullivan et al., 2004), adult attachment style (Krahé et al., 2015), and the motivation to help by the social partner (Kindt et al., 2018a; Kindt et al., 2018b).

Beyond the experimental evidence, several behavioural models were proposed to explain the association of social support and pain experience in chronic pain conditions. The operant model suggests that positive responses from social partners (e.g. reassure positive concerns) reinforce pain behaviours (Fordyce, 1977). The intimacy model posits that communication of pain may serve to enhance intimacy between couples (Cano and Williams, 2010). In another model, the communal coping model of pain catastrophizing, pain behaviours serve to attract social support, which in turn triggers or maintains exaggerated pain expression (Sullivan et al., 2001).

1.4. Neural and Physiological Techniques in Social Support and Pain

In contrast to the growing literature on the 'behavioural effects' of social support on pain, there has been less investigation into the potential 'mechanisms' underpinning the analgesic influence of social support. In this section, I will review the neural and physiological effects of social support on pain, based on the method applied.

1.4.1. Functional Magnetic Resonance Imaging (fMRI)

Functional Magnetic Resonance Imaging measures brain activity by detecting changes associated with blood flow. It has excellent spatial resolution especially for deep brain structures but has relatively poor temporal resolution. There have been two studies providing mechanistic evidence on how social support influences pain (Eisenberger et al., 2011; Younger et al., 2010). In one study, viewing images of a romantic partner relative to an acquaintance resulted in less pain in healthy participants. Moreover, reduced pain was associated with decreased pain-related activation (ACC,

insular cortex) but with increased reward activation (caudate nucleus, nucleus accumbens, OFC) in the romantic partner relative to the acquaintance condition (Younger et al., 2010). In another study, viewing images of a romantic partner relative to control images produced less pain perception, lower activation in pain-related regions (ACC, insular cortex) but higher activation in a safety-related region (ventromedial prefrontal cortex, VMPFC). Further, greater activation in the VMPFC was associated with lower pain ratings and pain-related brain activation when viewing a romantic partner relative to control images (Eisenberger et al., 2011). Therefore, the findings from these two studies suggest a mechanism in which social support may prime feelings of attachment and/or reward to reduce pain perception and brain response.

1.4.2. Magnetoencephalography (MEG)

Magnetoencephalography measures brain activity by recording magnetic fields produced by electrical activity of neurons. It has excellent temporal resolution and good spatial resolution. One study has used MEG in the investigation of social support effects on pain. In this study, participants with a diagnosis of fibromyalgia, a chronic pain disorder defined by widespread pain and tenderness, reported less pain in response to thermal stimuli when they were accompanied by a romantic partner compared to when they were alone. MEG data further showed reduced amplitude of somatosensory evoked fields elicited by elbow stimulation in the presence of a romantic partner. The authors suggested that decreased pain and somatosensory activity may be related to social support buffering the threat of pain or distracting attention away from pain (Montoya et al., 2004).

1.4.3 Laser-evoked Potentials (LEPs)

LEPs measure evoked brain responses time-locked to transient, noxious thermal stimulation. Two types of LEPs are commonly observed in responding to noxious stimuli. The first is an early negative deflection, termed as N1, which is believed to reflect early sensory processing preceding the

conscious experience (Lee et al., 2009). The second type of LEPs comprises a biphasic complex, termed as N2–P2, which may underlie the conscious experience of the sensory processing captured by N1 (Lee et al., 2009). Using LEPs, one study found that the presence of a romantic partner compared to the absence affected N2 and P2 local peak amplitudes, but this effect was modulated by attachment style (Krahé et al., 2015).

1.4.4. Assessments of the Autonomic Nervous System

In addition to the central nervous activity, there is also evidence on the autonomic mechanisms associated with the influence of social support on pain (McClelland and McCubbin, 2008; Roberts et al., 2015). As discussed earlier, pain is suggested to be mediated by a 'spinothalamocortical' pathway (Craig, 2002, 2003a). This circuit has also been found to regulate the balance of the sympathetic ('SNS') and parasympathetic ('PNS') nervous system in pain (for a review, see Cortelli et al., 2013). SNS activity in the context of social support and pain has been evaluated using heart rate (Sambo et al., 2010), blood pressure (McClelland and McCubbin, 2008), skin response (Platow et al., 2007), and cortisol levels (Roberts et al., 2015). Although some studies indicated that social support reduced sympathetic arousal to pain (Roberts et al., 2015; Sambo et al., 2010), the number of studies is highly limited and the evidence is to a large extent confounded by social contexts (McClelland and McCubbin, 2008; Roberts et al., 2015).

In contrast, no study has evaluated the PNS response in the social modulation of pain. The PNS is suggested to be associated with the regulatory control over sympathetic arousal (Thayer et al., 2009). The PNS response may therefore provide a means by which social support can increase physiological control over pain-related arousal. Heart rate variability (HRV), which is the physiological variation in time intervals between heartbeats, has been widely used to evaluate the balance between the SNS and PNS activity (Cardiology, 1996). A large body of evidence

demonstrates that a high HRV may represent an important index of an effective sympathovagal balance and of cardiac health (Bootsma et al., 1994; Thayer et al., 2010). There are different ways to measure HRV, ranging from time-domain to frequency-domain measures (Cardiology, 1996), among which high-frequency HRV (HF-HRV) is suggested to be strongly associated with cardiac vagal tone (i.e. parasympathetic tone) (Koenig et al., 2014).

1.5. Promising Neurophysiological Methods to Better Understand the

Connection between Social Support and Pain

Electroencephalography (EEG), Transcranial Magnetic Stimulation (TMS), and TMS-EEG are promising and complimentary neurophysiological methods to better understanding how social support acts to influence pain.

1.5.1. Electroencephalography (EEG)

EEG provides a measure of rhythmic fluctuations in electrical activity produced by large populations of neurons (Buzsáki, 2004; Buzsáki and Draguhn, 2004; Le Van Quyen and Bragin, 2007). EEG has excellent temporal resolution (i.e. in the millisecond range), which is highly valuable for detecting transient changes in brain state. Oscillatory frequencies are conventionally divided into delta (<4 Hz), theta (4 – 7 Hz), alpha (8 – 12 Hz), beta (13 – 30 Hz) and gamma (>30 Hz) bands (Buzsaki, 2006; Hirsch and Brenner, 2011). In addition to rhythmic oscillation, EEG also provides means for evaluating neural coupling over distance and across frequencies (Sauseng and Klimesch, 2008). Neurons in the human brain are to a large extent interconnected, whereby the whole brain can be seen as a huge network consisting of millions of sub-networks ranging from micro-level to largescale connections (Varela et al., 2001). It is widely accepted that human behaviour arises from the communication between neurons within and between neural networks (Fuster, 1997; Varela et al., 2001).

To the best of our knowledge, no study has investigated the oscillatory mechanisms associated with social modulation of pain. But there are many studies investigating neural oscillations in the context of pain (Chang et al., 2002; Hauck et al., 2015; Nickel et al., 2017). Although the frequency dynamics of pain are not entirely understood (Ploner et al., 2017), most of the studies found decreased activity in alpha band (sometimes extending to beta band) but increased activity in gamma band (Nickel et al., 2017; Peng et al., 2014; Schulz et al., 2015; Tiemann et al., 2010; Zhang et al., 2012). Changes in alpha and gamma activity are thought to underlie nociceptive attention and subjective experience of pain respectively (Hauck et al., 2015; Nickel et al., 2007; Peng et al., 2016). Nuclei et al., 2007; Peng et al., 2000), but its functional relevance in pain remains to be determined. In terms of EEG connectivity, only one study has explored phase synchronization in response to painful stimuli and found increased theta range connectivity between central and parietal regions (Taesler and Rose, 2016).

In the field of social support, one study found that viewing images of a romantic partner was associated with higher delta activity compared to images of an unknown person or a known and appreciated person. The authors indicated delta activity to be associated with feelings of attachment or love (Başar et al., 2008). Additionally, one study has looked at EEG connectivity in the context of social touch and pain by simultaneously recording EEG in both individuals in a romantic relationship. Social touch by a romantic partner decreased pain perception but increased alpha band coherence between the couples. Alpha band coupling between couples was suggested to be associated with the integration of information from tactile, visual, and nociceptive inputs (Goldstein et al., 2018).

1.5.2. Transcranial Magnetic Stimulation (TMS)

Transcranial Magnetic Stimulation (TMS) is a non-invasive approach which can modulate brain activity and thereby provide unique causal insights into brain-behaviour relationships. Specifically, TMS can generate weak electric currents via rapid changes in magnetic field and trigger depolarization of the neurons under the coil (Siebner and Rothwell, 2003; Wagner et al., 2007). When TMS is applied repetitively (repetitive TMS, rTMS), the excitability of stimulated cortical region can be altered (Maeda et al., 2000). Theta Burst Stimulation (TBS) is one of the most established rTMS protocols and has the capacity to modulate neural excitability (Chung et al., 2017; Huang et al., 2005; Ni et al., 2014). TBS delivers pulses in bursts of three at high frequency (50 Hz) with an interburst interval at low frequency (5 Hz) for a total of 600 pulses. TBS can be used to increase or decrease cortical excitability depending on whether intermittent (iTBS, a 2s train of TBS repeated every 10s) or continuous (cTBS, 20 or 40 s of TBS without any interruption) stimulation, respectively, is employed (Huang et al., 2005). Thus, TBS provides a means to modulate brain activity and explore the effects and mechanisms of social support on pain but has not been used for this purpose to date.

In addition, changes in neural activity induced by TBS can be measured using single-pulse TMS and concurrent EEG (TMS-EEG) (Cash et al., 2017; Chung et al., 2017). EEG responses induced by a single-pulse TMS can be illustrated with waveforms and topographic representation of TMS-evoked potentials (TEPs) (Rogasch et al., 2014). TEPs are suggested to reflect the shifts in the inhibition-excitation balance in cortical circuits following a single TMS pulse (Du et al., 2018; Rogasch and Fitzgerald, 2013). TEPs following stimulation of the motor cortex and non-motor context (e.g. prefrontal cortex) consist of a series of negative and positive peaks (Chung et al., 2017; Mäki and Ilmoniemi, 2010; Rogasch and Fitzgerald, 2013). Among these peaks, N100 is considered to be the most robust TEP component with the greatest signal-to-noise ratio (Cash et al., 2017; Noda et al., 2016) and to be most reliably modulated by TBS (Chung et al., 2017; Chung et al., 2018b). Although the physiological origin and functional significance of N100 component has not been fully elucidated,

several studies have suggested the N100 component to be linked to cortical inhibitory processes (Bender et al., 2005; Bonnard et al., 2009; Chung et al., 2017; Farzan et al., 2013; Rogasch et al., 2012, 2015).

1.6. The Clinical Importance of Understanding the Influence of Social Support

on Pain

While it is not the focus of this PhD, understanding the effects and mechanisms of social support on pain is important for improving the application of social support strategies in the management of pain. Indeed, a number of cognitive-behavioural therapies (CBT) have included a significant other (mostly the spouse/partner) in the management of chronic pain and these studies have demonstrated add-on effects of social support (Abbasi et al., 2012; Keefe et al., 2004; Keefe et al., 1996, 1999; Martire et al., 2003; Martire et al., 2008; Radojevic et al., 1992). Better understanding in which contexts social support has an analgesic influence and the potential mechanisms will allow for optimising social support strategies in pain management.

1.7. Summary of the Literature

The role of social support in pain experience is a growing area of research. Overall, the literature suggests a potential role of social support strategies in the management of pain. Indeed, some interventional trials have demonstrated the effectiveness of social support-assisted therapies for the management of chronic pain (Abbasi et al., 2012; Keefe et al., 1996, 1999). However, studies have demonstrated mixed effects of social support and it has not been fully elucidated in which contexts social support has a protective effect on pain and the magnitude of this effect. Moreover, research is still in the early stage to understand the neural and physiological mechanisms through which social
support infuences pain. Overall, the utilization of social support strategies in pain management is challenged by limitations in the current research evidence.

1.8. Aims and Overview of Chapters

The overall aim of this thesis is to examine the role of social support in pain experience and neurophysiological mechanisms. This was addressed through a total of two review articles and three experimental studies (Chapter 2 to Chapter 6, three published and two under review).

In chapter 1, a brief introduction and overview of the thesis are provided.

Chapter 2 contains a published systematic review and meta-analysis (Neuroscience & Biobehavioral Review) which characterized the contexts in which social support has an analgesic effect and quantified the magnitude of the effect.

Chapter 3 contains a published systematic review (Clinical Journal of Pain) which assessed the potential psychological and behavioural processes as well as neurobiological mechanisms associated with the effect of social support on pain reduction.

Chapter 4 contains the first published empirical paper (Journal of Pain), which explicitly manipulated the perceived threat of pain to examine whether or not social support can reduce pain through a buffering effect on the neurophysiological makers of threat of pain.

Chapter 5, currently under review, investigated the changes in neural oscillations and parasympathetic activity that may be associated with the influence of social support on prolonged pain.

Chapter 6, currently under review, describes the use of TBS to the medial prefrontal cortex and TMS-EEG to investigate neural plasticity and brain connectivity changes that may mediate the influence of social support on pain.

Chapter 7 contains the summary of the study chapters and the implications of the results. In addition, limitations and future directions are included in this section. The thesis closes with a brief conclusion.

CHAPTER TWO

Effects and Social Context

Manuscript

Che X., Cash R., Chung S.W., Fitzgerald P., & Fitzgibbon B.M. (2018). Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis. *Neuroscience & Biobehavioral Review*, *92*, 437-452.

Preamble to systematic review and meta-analysis

Social support can have mixed effects on pain experience (Brown et al., 2003; Gallant and Hadjistavropoulos, 2017; Goldstein et al., 2016; Karmann et al., 2014; Modić Stanke and Ivanec, 2010). A previous systematic review highlighted the contribution of potential covariates in the social modulation of pain, such as the perceived intention of the social partner and the pre-existing social relationships (Krahé et al., 2013). However, no study has characterized the magnitude of the influence of social support. Moreover, a number of relevant additional studies have been published since the 2013 systematic review (Edwards et al., 2017; Goldstein et al., 2016; Roberts et al., 2015; Shaygan et al., 2017). In addition, how social support may influencepain-related physiological arousal has not previously been systematically assessed.

This chapter systematically characterises the contexts in which social support has an analgesic effect and quantifies the magnitude of the effect using a meta-analysis. We also provide a detailed examination of the role of social relationships in the social modulation of pain. In addition to pain experience, this chapter also demonstrates the influence of social support on pain-related

physiological arousal. The presence of publication bias is also considered.

Contents lists available at ScienceDirect



Neuroscience and Biobehavioral Reviews

journal homepage: www.elsevier.com/locate/neubiorev

Review article

Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis



Xianwei Che^{a,*}, Robin Cash^a, Sungwook Chung^a, Paul B. Fitzgerald^{a,b}, Bernadette M. Fitzgibbon^a

^a Monash Alfred Psychiatry Research Centre (MAPrc), The Alfred and Central Clinical School, Monash University, Melbourne, Australia ^b Epworth Clinic, Epworth Healthcare, Camberwell, VIC, Australia

ARTICLE INFO	A B S T R A C T
<i>Keywords:</i> Social support Pain Physiology Arousal Meta-analysis	Social support is demonstrated to have mixed effects on both pain and related physiological arousal. In this study, a meta-analysis was conducted to characterise these effects. A total of 2416 studies were identified in a systematic search, among which 21 were eligible for the quantitative review. The mere presence of another person was not sufficient to modulate pain perception. However, the presence of a stranger was identified to decrease pain-related arousal (SMD = -0.31), and the presence of a significant other increased facial expression of pain (SMD = 0.21). We further found verbal support to decrease pain (SMD = -0.69) and arousal (SMD = -0.69) and we demonstrated moderate to large analgesic effects of intimate relationships through touching (SMD = -0.95) and viewing (SMD = -0.60) of a romantic partner. Finally, we presented evidence of publication bias for pain-related arousal but not for behavioural pain outcomes. Together, our findings suggest that the impact of social support on pain is context-dependent with the verbal communication of support and in-timate relationships being of particular importance.

1. Introduction

Under the right conditions, having supportive relationships with others provides benefits to the individual (Cohen and Wills, 1985; Uchino, 2006). These benefits may arise through the forms of receiving supportive resources from others (e.g. emotional, economical, informational) and the perception that supportive resources are available should they be needed (Dunkel-Schetter and Bennett, 1990; Uchino, 2009). In the context of pain research, social support may be associated with the reduction of pain (Brown et al., 2003; Eisenberger et al., 2011) and related physiological arousal (Roberts et al., 2015; Sambo et al., 2010). For instance, holding the hand of a significant other or providing social support without actual presence (i.e. primed support) can reduce the intensity and/or unpleasantness of pain and heart rate increase evoked by painful stimuli (Che et al., 2017; Master et al., 2009).

In contrast, social support also has the potential to increase pain experience (Hurter et al., 2014; McClelland and McCubbin, 2008). Indeed, pain behaviours are in some instances suggested to be used to trigger sympathy or attention (Williams, 2002), and/or to avoid social responsibilities (Glenton, 2003). A recent review highlighted the contribution of potential covariates in the social modulation of pain. Specifically, social support in general decreases pain when it is clearly communicated (e.g. verbal support, 'social support variations'), or in cases where it is provided by a significant other (e.g. romantic partner, 'social relationships') (Krahé et al., 2013). However, no study has characterised the magnitude of the influence of social support nor the modulating impact of potential covariates. Moreover, since the last systematic review (Krahé et al., 2013) there have been a number of additional studies published in this area (Edwards et al., 2017; Gallant and Hadjistavropoulos, 2017; Goldstein et al., 2016; Karmann et al., 2014; Roberts et al., 2015), with some of these studies also examining pain-related arousal which has not previously been assessed in a systematic review.

A growing body of research indicates the effectiveness of supportassisted therapies for the management of chronic pain (e.g., spouseassisted coping skills training) (Keefe et al., 1996, 1999). However, in order to optimise these approaches, a better understanding of the context(s) in which social support has an analgesic effect is necessary. To this end, a meta-analysis was conducted to quantify the influence of social support on pain. Studies were limited to those that provided a nosupport baseline condition relative to stranger or close other support. This helps to control the variability induced by baseline conditions (e.g. a stranger condition) and clarify the role of pre-existing social connections. Moreover, we quantified physiological changes evoked by painful stimuli in the context of social support. Physiological arousal is closely associated with health outcomes (Cohen and Wills, 1985;

https://doi.org/10.1016/j.neubiorev.2018.07.005

Available online 10 July 2018

^{*} Corresponding author at: Monash Alfred Psychiatry Research Centre (MAPrc), Monash University Level 4, 607St Kilda Road, Melbourne, VIC 3004, Australia. *E-mail address*: (X. Che).

Received 8 February 2018; Received in revised form 9 July 2018; Accepted 9 July 2018

^{0149-7634/ © 2018} Elsevier Ltd. All rights reserved.

Uchino, 2006), and it may suggest ways in which social support modulates pain experience (Goldstein et al., 2017). This meta-analysis may help to summarise and quantify the complex and multivariate influence of social support on experimental pain and is intended to help direct future research and assist in the optimisation of treatment strategies.

2. Methods

2.1. Protocol and registration

This review was conducted in accordance with the Preferred Reporting Items for Systematic Reviews and Meta-Analysis (PRISMA) (Moher et al., 2009). The protocol was registered in the database of International Prospective Register of Systematic Reviews (PROSPERO, registration number: CRD42017076667).

2.2. Search strategy

A comprehensive electronic literature search was performed in PubMed, PsycINFO, The Cochrane Library and EMBASE to the end of August 2017. The keywords used for the search were 'pain' AND ('interpersonal' OR 'social support' OR 'social presence' OR 'social interaction' OR 'social modulation' OR 'social context' OR 'attachment' OR 'social influence' OR 'social touch' OR 'empathy'). Full search records can be found in *Supplementary Material S1*.

Two reviewers (XC and SC) independently assessed the titles and abstracts of the initial search results against the inclusion criteria (see Table 1). Full-text versions were examined in instances where it was unclear from the summary data alone whether a particular study met inclusion criteria. Full-text articles of potentially eligible studies were then screened. Discrepancies between the reviewers were solved by consensus. Reference lists of full-text potentially eligible studies were also checked for missing studies.

2.3. Selection criteria

The same inclusion and exclusion criteria were used for the systematic review and meta-analysis, as outlined in Table 1. Specifically, studies were included if they matched all five *a priori* criteria: studies were included if (1) pain was induced in experimental settings (e.g. cold pain) but were excluded if they used clinical procedure or clinical pain

Table 1

Inclusion and exclusion criteria.

	Inclusion	Exclusion
Interventions	Experimentally induced pain	Clinical procedure Clinical pain assessment
Comparisons	Presence and absence of social support	No control condition of social support
Outcomes	Behavioural pain outcomes: Pain intensity Pain unpleasantness Facial expression of pain Pain threshold Pain tolerance Physiological studies were included if they also reported behavioural pain outcomes	No behavioural pain outcome
Data reported	Data that enables analysis and estimation of effect size	Unpublished data Data without Mean Data without SD/SEM Data without N
Publication type	Peer-reviewed journal Written in English	Review article Case report Grey literature Non-English article

SD-standard deviation; SEM-standard errorof mean; N-sample size.

assessment; (2) both the presence and absence of social support were manipulated (regardless of a within- or between-group design), but were excluded if there was no baseline condition of support; (3) behavioural pain outcomes were reported, but were excluded if they did not report any behavioural pain measure; (4) sufficient data were available to compute effect size using Hedge's adjusted g (i.e. mean, standard deviation (SD), and sample size) but were excluded if data were not available; (5) the article was published in peer-reviewed journals and written in English.

2.4. Outcome measures

Based on the literature on social support and pain, behavioural pain measures included pain intensity, pain unpleasantness, facial expression of pain, pain threshold, and pain tolerance. Physiological outcomes included heart rate, blood pressure (both diastolic and systolic), skin response, and cortisol levels.

2.5. Data extraction

The mean, SD, and sample size were extracted for the outcome measures in each experimental condition or group (i.e. support, no-support). In cases where numerical values were not available, data were extracted directly from relevant figures using Plot Digitizer software (Huwaldt, 2010). We also contacted the corresponding authors for additional data when they were not directly available from the article.

In accordance with meta-analyses in pain research (Boerner et al., 2014; Thompson et al., 2016), data were averaged in certain instances using the formula from the Cochrane Handbook for Systematic Reviews of Interventions (Formula 1, m = mean; n = sample size; SD = standard deviation) (Higgins and Green, 2011). Specifically, data were averaged: (1) when a study applied more than one type of pain-induction technique (Edwards et al., 2017; Montoya et al., 2004); (2) when painful stimuli were applied to different sites of the body (Montoya et al., 2004); (3) when a particular pain outcome (e.g. pain intensity) was assessed with more than one measurement (Edwards et al., 2017; Roberts et al., 2015); or (4) when results were reported in different genders (Edwards et al., 2017; Jackson et al., 2005; McClelland and McCubbin, 2008). Moreover, in cases where painful stimuli were delivered at different intensity levels (Eisenberger et al., 2011; Kleck et al., 1976), data of the highest intensity were extracted because it was most likely to reliably evoke pain. In addition, if a study assessed pain outcomes in different time points, data were extracted from the last time point (Brown et al., 2003; Sullivan et al., 2004). This was done as most of the included studies assessed pain at the end of painful stimulation. Finally, data from the first paper were extracted when the same dataset was reported in multiple papers (Goldstein et al., 2017).

Formula 1:

$$Mean = \frac{n1*m1 + n2*m2}{n1 + n2}$$

SD

1

$$=\sqrt{\frac{(n1-1)^* SD1^2 + (n2-1)^* SD2^2 + \frac{n1^*n2}{n1+n2} * (m1^2 + m2^2 - 2*m1*m2)}{n1+n2-1}}$$

2.6. Methodological study appraisal

We adapted the assessment tool developed by Lautenbacher et al. (2017), to assess the quality of psychophysical studies on experimental pain among healthy controls. This tool was developed based on the pain literature (e.g. Tesarz et al., 2012) and the Newcastle-Ottowa Scale (Wells et al., 2010). This assessment tool includes six items and was used to examine age effect on pain perception. We therefore adapted

this tool to our dataset by replacing one of the items with social support blinding. Specifically, the items were: (1) reported blinding processes of study purpose to participants; (2) similar gender distribution across all condition/group; (3) specification of stimulus location; (4) specification of physical type of stimulus; (5) report on psychophysical method of threshold determination; and (6) the extent to which the study population represents the true population. Two reviewers (XC and SC) independently assessed each study and discrepancies were addressed by consensus. More details of this assessment tool can be found in *Supplementary Material S2*.

2.7. Meta-analysis

2.7.1. Calculating effect sizes

Continuous outcome measures were used in the meta-analysis. Extracted data were entered into the MIX 2.0 computer program (Bax, 2011), which allows for the calculation of statistical significance of differences between means with 95% confidence intervals (CIs). The standardised mean difference (SMD) calculated using Hedge's adjusted g was estimated for the effect size. Hedge's adjusted g is similar to Cohen's d but it adjusts for the small sample bias (Hedges and Olkin, 1985). For SMDs, values of 0.2 were considered small, 0.5 as medium and 0.8 as large (Cohen, 1988). It is noted that an SMD below 0.2 (with $p \leq 0.05$) was only considered as trivial. In cases where standard error (SE) values were reported, SD values were estimated using the formula $SD = SE^* \sqrt{n}$ (n = sample size) (Higgins and Green, 2011).

It is noted that the SMD method does not account for the differences in the scale directions. In this case, pain threshold and pain tolerance have opposite directions compared to other behavioural pain measures. Based on the Cochrane Handbook for Systematic Reviews of Interventions (Higgins and Green, 2011), we thus multiplied the mean values of pain threshold and pain tolerance by -1 to correct for the difference in scale direction.

2.7.2. Subgroup analyses

In the subgroup analyses, we examined the main effect and the interaction effect between social support variations and social relationships.

There were four variations of social support based on the included studies: social presence, verbal support, social touch, and primed support. In the cases of social presence, the social partner was physically present without any verbal communication or eye contact with the participant (Edwards et al., 2017). In the studies of verbal support the social partner verbally interacted with the participant. In some studies, the social partner verbally encouraged the participant to engage in the pain task (Brown et al., 2003; Roberts et al., 2015), while in another study the social partner answered the questions of the participant in an empathetic fashion (Jackson et al., 2005). In the condition of social touch, the social partner held the hand of the participant without verbal communication (Master et al., 2009). In the cases of primed support, the social partner was not physically present. Social support was manipulated through viewing an image of the social partner (Eisenberger et al., 2011), or verbally interacting with the social partner before the pain task (Platow et al., 2007).

We further distinguished between different types of social relationships (Krahé et al., 2013), based on whether the social partner was a stranger, a parent, a friend, or a romantic partner of the participant. According to the history of social interactions, social relationships were categorised as 'close other' (i.e. romantic partner, parent, friend), and 'stranger'.

In the supplementary analysis, we also investigated the gender difference as some studies have shown that the social modulation of pain was most prominent in female pain recipients (Chambers et al., 2002; Jackson et al., 2005; McClelland and McCubbin, 2008). Data of two genders were thus analysed separately from relevant studies.

2.7.3. Test of heterogeneity

In systematic reviews and meta-analyses, conclusions are less clear when the results vary across studies. A statistical test of heterogeneity is therefore commonly reported to establish whether studies are consistent (Higgins et al., 2003). In line with other meta-analyses in experimental pain (Kim et al., 2017; Thompson et al., 2016), heterogeneity between studies was evaluated using the I^2 statistics in this study. I^2 measures the percentage of total variance across included studies that is due to heterogeneity. It is expressed as $I^2 = 100\% \times (Q - Q)$ df) / Q, where df is the degree of freedom, and Q is computed by summing the squared deviations of each study's estimate from the overall meta-analytic estimate (Higgins et al., 2003). I^2 ranges from 0% to 100%, with 0% indicating no observed heterogeneity, and > 50%representing moderate heterogeneity. Moreover, Galbraith plots were included to assess and illustrate the extent of heterogeneity between studies. These plots indicate the proportion of studies whose results are positioned within two standard errors of the effect estimates (i.e. metaanalytic estimate divided by its standard error). If at least 95% of the included studies lie within two standard errors of the population effect, the effect is interpreted as being consistent across studies (Thompson, 1994). As of the variations in important study characteristics (e.g. pain modality, support variations, pain outcomes), we anticipated heterogeneity among studies and thus random-effect models were employed. A random-effect model allows for statistical control of heterogeneity and permits generalisation of results beyond those included in the metaanalysis.

2.7.4. Publication bias

Publication bias may adversely affect the reliability of the conclusions of a meta-analysis. A number of methods were therefore employed to examine publication bias. The selectivity funnel plot is a straightforward method to visually inspect publication bias (Light and Pillemer, 1984). This provides a graphical representation of effect estimates against sample size, which is computed as the standard error of the effect estimates. Results from small studies will be positioned at the bottom of the graph, with the spread narrowing among larger studies (Sterne et al., 2011). In the absence of bias, the scatter will resemble a symmetrical inverted funnel. However, the funnel plot provides only an informal illustration, which should be further examined by more objective statistics (Egger et al., 1997). Egger's regression test and the Begg-Mazumdar Kendall τ are most commonly used. Egger's regression test uses a linear regression approach to measure publication bias. It is expressed as Standardized Effect Size = $a + b \times$ Inverse Standard Error, where intercept 'a' provides a measure of publication bias (Egger et al., 1997). The Begg-Mazumdar test is based on correlating the standardized effect estimates with the variance of the effect estimates using a rank correlation test (i.e. Kendall's tau) (Begg and Mazumdar, 1994). In addition, a Bayesian approach was used to detect and mitigate the effects of publication bias. This approach assigns weights to a set of four models under different publication bias assumptions and averages the results from these models. It can yield the posterior distribution of a mitigated test statistic which can be used to make inferences about the existence of an effect (Guan and Vandekerckhove, 2016).

3. Results

3.1. Selection of studies and characteristics

Online database searches identified a total of 2416 records (Fig. 1). After duplicates were removed, 1836 studies remained. Initial screening of the title and abstract was performed against the inclusion and exclusion criteria. After excluding 1788 records from the initial screening, full-text versions of 48 studies were screened for eligibility. A total of 21 studies were included in the systematic review and meta-analysis, among which 21 were appropriate for behavioural outcome analysis and 6 for physiological outcome analysis.



Fig. 1. Flow diagram of selected studies (n = number of articles).

Selected studies included both within-group (i.e. participants underwent both social support and no-support manipulation, e.g. Eisenberger et al., 2011), and between-group (i.e. participants underwent either a social support or a no-support manipulation, e.g. Brown et al., 2003) designs. A total of 81 contrasts were identified for behavioural pain assessment. All participants (n = 1379) reported behavioural pain measures, among which 830 were females. Participants were mostly health adults except for fibromyalgia and migraine patients in one study (Montoya et al., 2004), and child-parent dyads in another (Vervoort et al., 2011). The experimental pain was induced by a variety of methods including thermal (both cold and heat), pressure, electrical, and laser stimuli. It is noted that most of the studies examined social presence (62 contrasts), while relatively fewer studies investigated verbal support (6 contrasts), social touch (5 contrasts), or primed support (8 contrasts). All the studies demonstrated accepted qualities of 4 as suggested by Lautenbacher et al. (2017) (Table 2). Moreover, 22 contrasts of physiological arousal were identified which were conducted among 348 healthy adults (180 females).

3.2. Behavioural pain outcomes

Table 2 summarises the characteristics of the studies that examined the effects of social support on behavioural and physiological pain outcomes. Studies were separated into different datasets that included multiple experiments, and/or contrasts, and/or outcome measures.

3.2.1. Effects of social support on behavioural pain outcomes

We first pooled data extracted from all experiments, contrasts, and behavioural outcomes (i.e. intensity, unpleasantness, facial expression of pain, tolerance, and threshold). Overall, social support showed no effect on behavioural pain with a pooled SMD of -0.10 (95% CI: [-0.16, -0.03], p = 0.003) (Fig. 2). Moreover, there was no effect of social support on each behavioural pain measures (with no SMD \ge 0.2, and $p \le 0.05$).

3.2.2. Publication bias in studies examining behavioural outcomes

Galbraith plot indicated heterogeneity in the dataset of behavioural outcomes, with more than 5% of the dots beyond two standard errors of the population effect (Fig. 3A). The test of heterogeneity was significant (Q = 124.60, p = 0.001, $I^2 = 35.79\%$). There was a slight asymmetry in the shape of the selectivity funnel plot (Fig. 3B), with each line in the funnel representing various levels of significance (0.01, 0.05, and 0.1). Begg's test (tau = -0.15, p = 0.07) and Egger's regression test (t = -0.72, p = 0.08) indicated no evidence of publication bias (Fig. 3C). In addition, the Bayesian analysis yielded similar effect size (-0.09) compared to the estimated effect size (-0.10) (Fig. 3D). These combined analyses suggested a minimal possibility of publication bias in the dataset of behavioural outcomes.

3.2.3. Subgroup analyses on behavioural outcomes

A series of subgroup analyses were performed to examine the main

Table 2Characteristics of included studies.

Study	Sample	Gender ratio	Age (range)	Pain modality	Conditions and contrasts	Behavioural outcomes	Physiological outcomes	Quality score
Borsook and MacDonald (2010)	Healthy participants (N = 45)	16M:29F	(18-30)	Pressure pain	1. Stranger positive interaction vs. no interaction (pre-post	 Intensity Unpleasantness 		5.5
Brown et al. (2003)	Healthy participants (N = 101)	52M:49F	(17-21)	Cold pain	 Random interaction Random interaction vs. alone Active support vs. alone Social presence vs. alone 	 Intensity Intensity Intensity 		5
Edwards et al. (2017)	Experiment 1: Healthy participants (N = 48) Experiment 2: Healthy participants (N = 48) Experiment 3: Healthy participants (N = 48)	24M:24F 24M:24F 24M:24F	M: 26.62 \pm 8.14; F: 22.75 \pm 3.54 M: 24.21 \pm 7.6; F: 19.67 \pm 2.35 M: 25.42 \pm 5.11; F: 23.92 \pm 4.23	Cold pain; Pressure pain Cold pain; Pressure pain Cold pain; Pressure pain	 Friend presence vs. alone Stranger presence vs. alone Male friend presence vs. alone Female friend presence vs. alone Partner presence vs. alone Friend presence vs. alone 	 Intensity Tolerance Threshold 		4 4 4
Eisenberger et al. (2011)	Healthy participants (N = 17)	0M:17F	23.4 ± 3.8	Heat pain	 Partner image- viewing vs. object image-viewing Stranger image- viewing vs. object image-viewing 	1. Unpleasantness 1. Unpleasantness		4
Fishman et al. (1995)	Healthy participants $(N = 60)$	30M:30F	-	Cold pain	1. Stranger social touch vs. alone	1. Intensity	1. HR 2(1). SBP 2(2). DBP	4
Gallant and Hadjistavropoulos (2017)	Healthy participants (N = 97)	42M:55F	(60-89)	Heat pain	 Family member presence vs. alone Stranger presence vs. alone 	 Intensity Unpleasantness Facial expression Tolerance Threshold Intensity Unpleasantness Facial Tolerance Thershold 		4.5
Goldstein et al. (2016) ^a	Healthy participants (N = 23)	0M:23F	(19-40)	Heat pain	 Stranger hand- holding vs. alone Partner hand- holding vs. alone Partner presence vs. alone 	 Intensity Intensity Intensity Intensity 		4.5
Jackson et al. (2005) ^a	Healthy participants (N = 88)	33M:55F	(18-47)	Cold pain	 Transaction with empathetic stranger vs. coping alone 	 Intensity Tolerance 		4.5
Karmann et al. (2014)	Healthy participants (N = 126)	63M:63F	39.9 ± 13.5	Heat pain	 Partner presence vs. alone Stranger presence vs. alone 	 Intensity Facial Intensity Facial 		6
Kleck et al. (1976)	Experiment 1: Healthy participants (N = 20) Experiment 2: Healthy male participants (N = 40)	20M:0F 40M:0F	-	Electrical shock Electrical shock	 Stranger observer vs. alone Stranger observer vs. alone 	 Intensity Facial Intensity Facial 	1. SCR 1. SCR	4 4.5
Krahé et al. (2015)	Healthy participants (N = 39)	0M:39F	25.87 ± 5.17	Laser pain	 Partner presence (participant focus) vs. alone Partner presence (other focus) vs. alone 	1. Intensity 1. Intensity		5.5

(continued on next page)

Table 2 (continued)

Study	Sample	Gender ratio	Age (range)	Pain modality	Conditions and contrasts	Behavioural outcomes	Physiological outcomes	Quality score
Master et al. (2009)	Healthy participants (N = 25)	0M:25F	-	Heat pain	 Partner hand- holding vs. object hand-holding Stranger hand- holding vs. object hand-holding Partner image- viewing vs. object image-viewing Stranger image- viewing vs. object image-viewing 	 Unpleasantness Unpleasantness Unpleasantness Unpleasantness Unpleasantness 		4
McClelland and McCubbin (2008)	Healthy participants $(N = 68)$	36M:32F	19.97 ± 2.75	Cold pain	1. Friend presence vs. alone	 Intensity Tolerance 	1. HR 2(1). SBP 2(2). DBP	4
Modić Stanke and Ivanec (2010)	Healthy participants (N = 43)	0M:43F	(19-33)	Heat pain	 Stranger presence at 0.5 m vs. alone Stranger presence at 1.5 m vs. alone 	 Intensity Unpleasantness Tolerance Threshold Intensity Unpleasantness Tolerance Thershold 		5.5
Montoya et al. (2004)	Fibromyalgia patients (FIB) (N = 16); Migraine patients (MIG) (N = 16)	FIB : – MIG : –	FIB : (36-68) MIG : (36-68)	Thermal pain (hot and cold)	1. Partner presence vs. alone	 Intensity^b Threshold^c 		4.5
Platow et al. (2007)	Healthy participants (N = 54)	32M:22F	(18-35)	Cold pain	 In-group stranger reassurance vs. no- reassurance Out-group stranger reassurance vs. no- reassurance 	1. Tolerance 1. Tolerance	1. SCR 1. SCR	5
Roberts et al. (2015)	Healthy participants (N = 76)	0M:76F	(18-21)	Cold pain	 Stranger verbal support vs. alone Stranger presence vs. alone 	 Intensity Unpleasantness Intensity Unpleasantness 	1. HR 2(1). SBP 2(2). DBP 3. Cortisol ^d 1. HR 2(1). SBP 2(2). DBP 3. Cortisol	5
Sambo et al. (2010)	Healthy participants (N = 30)	10M:20F	29.1 ± 7.3	Heat pain	 High-empathy stranger presence vs. alone Low-empathy stranger presence vs. alone 	1. Intensity 1. Intensity	1. HR 2. SCR 1. HR 2. SCR	4.5
Sullivan et al. (2004)	Healthy participants (N = 64)	26M:38F	(17-30)	Cold pain	1. Stranger presence vs. alone	 Intensity^e Facial^f 		5
Vervoort et al. (2011)	Healthy children and parents (N = 38)	16M:22F	14.5 ± 2.52	Cold pain	1. Believed parent observation vs. alone	 Intensity Facial 		5
Vlaeyen et al. (2009)	Healthy participants $(N = 149)$	61M:88F	M: 33.3 ± 9.3; F: 29.6 ± 11.3	Cold pain	1. Stranger presence vs. alone	1. Intensity ⁸ 2. Facial ⁸		5.5

M-male; F-female; HR-heart rate; DBP-diastolic blood pressure; SBP-systolic blood pressure; SCR-skin conductance response; FIB-fibromyalgia; MIG-migraine.

^b data were extracted from neutral image condition.

^c data were averaged across people with fibromyalgia and migraine.

^d data were extracted from the time point right after pain.

^e data were averaged across high and low catastrophizers.

^f data were the total pain behaviours.

^g data were averaged across threat and no-threat conditions.

and the interaction effect of social support variations and social relationships on behavioural pain.

3.2.3.1. Social presence. Overall, social presence showed no effect on behavioural outcomes (SMD = -0.06, 95% CI: [-0.11, -0.004], p = 0.03). Similarly, no effect was observed on each behavioural pain

measure (with no SMD \geq 0.2, and $p \leq$ 0.05) (see *Supplementary Material S3*).

Further analysis revealed that close other presence increased facial expression of pain with a small effect size (SMD = 0.21, 95% CI: [0.03, 0.38], p = 0.02). It was demonstrated to have no effect on other behavioural pain measures (with no SMD ≥ 0.2 , and $p \leq 0.05$) (Fig. 4A).



Fig. 2. Forest plot of the Hedge's adjusted g analysis for pooled effects of social support on behavioural pain outcomes.

There was no effect of stranger presence on behavioural pain, either in the pooled analysis (SMD = -0.09, 95% CI: [-0.17, -0.02], p = 0.01), or in the analyses of separate behavioural measures (with no SMD ≥ 0.2 , and $p \le 0.05$) (Fig. 4B).

3.2.3.2. Verbal support. Overall, verbal support decreased behavioural pain with a medium effect size (SMD = -0.69, 95% CI: [-1.30, -0.08], p = 0.03). However, verbal support showed no effect on separate measures of pain intensity or pain tolerance (with no SMD \ge 0.2, and $p \le 0.05$) (Fig. 5A).

Moreover, stranger verbal support resulted in a decrease in behavioural pain with a large effect size (SMD = -0.92, 95% CI: [-1.71, -0.13], p = 0.02). Similarly, this effect was not observed on pain intensity or pain tolerance separately (with no SMD ≥ 0.2 , and $p \le 0.05$) (Fig. 5B).

No study has exclusively examined the effect of verbal support by a close other on behavioural pain ratings. Brown and colleagues examined verbal support from a friend but data were averaged across friend and stranger verbal support (Brown et al., 2003).

3.2.3.3. Social touch. Results showed no effect of social touch on behavioural pain, either in the pooled analysis (SMD = -0.30, 95% CI: [-0.80, -0.21], p = 0.25), or in the analyses of different behavioural measures (with no SMD ≥ 0.2 , and $p \le 0.05$) (Fig. 6A).

We did observe decreased behavioural pain in close other (all from romantic partner) social touch with a large effect size (SMD = -0.95, 95% CI: [-1.37, -0.52], p = 0.00001). However, stranger social touch showed no effect on behavioural pain outcomes (SMD = 0.07, 95% CI: [-0.27, 0.41], p = 0.70) (Fig. 6B).

3.2.3.4. *Primed support*. Overall, primed support demonstrated no effect on behavioural pain outcomes (SMD = -0.01, 95% CI: [-0.25, 0.22], p = 0.91). Similarly, no effect was observed on separate pain measures (with no SMD ≥ 0.2 , and $p \leq 0.05$) (Fig. 7A).

Further analysis showed decreased behavioural pain in primed support from a close other (all from romantic partner image-viewing) with a medium effect size (SMD = -0.60, 95% CI: [-1.04, -0.16], p = 0.008). But primed support from a stranger demonstrated no effect on behavioural pain (SMD = 0.15, 95%CI: [-0.06, 0.36], p = 0.17) (Fig. 7B).

3.2.3.5. Gender effect. Among the studies that reported findings for each gender, the majority examined the effect of social presence, but we pooled data from all variations of support. In male participants, social support showed no effect on the overall pain experience (SMD = -0.13, 95% CI: [-0.29, 0.03], p = 0.12), or on the separate pain measures (with no SMD ≥ 0.2 , and $p \le 0.05$). Similarly, no significant effect was observed among female participants in either the pooled analysis (SMD = -0.05, 95% CI: [-0.21, 0.11], p = 0.53) or the separate analyses of measures (with no SMD ≥ 0.2 , and $p \le 0.2$, and $p \le 0.05$) (see *Supplementary Material S4*).

3.3. Physiological outcomes

3.3.1. Effects of social support on physiological outcomes

As shown in Fig. 1, physiological measures included heart rate (4 studies, 6 contrasts), blood pressure (3 studies, 8 contrasts), skin conductance (3 studies, 6 contrasts), and cortisol level (1 study, 2 contrasts). As each measure reflects certain aspects of physiological arousal, they were pooled together to investigate overall pain-related arousal.

Overall, social support decreased physiological response to painful stimulation with a small effect size (SMD = -0.25, 95% CI: [-0.44, -0.06], p = 0.008). Further analyses showed that this effect was only observed in skin response (SMD = -0.28, 95% CI: [-0.51, -0.05], p = 0.01), but not in other physiological measures (with no SMD \ge 0.2, and $p \le 0.05$) (Fig. 8).

3.3.2. Publication bias in studies examining physiological outcomes

Galbraith plot suggested heterogeneity in the pooled dataset of the physiological outcomes, in which more than 5% of the points were beyond two standard errors of the population effect (Fig. 9A). The test



Fig. 3. Series of tests for heterogeneity and publication bias for the effects of social support on behavioural pain outcomes. (A) Galbraith plot suggested heterogeneity with more than 5% of the data beyond two standard errors of the population effect. (B) Selectivity funnel plot indicated slight publication bias with a few studies showing asymmetry towards decreased pain. (C) Regression plot suggested no publication bias with the intercept showing small deviation from 0 (p > 0.05). (D) Bayesian triplot indicated no publication bias with the likelihood effect size (-0.09) comparable as the estimated effect size (-0.10).

of heterogeneity was also significant (Q = 63.14, p = 0.001, $I^2 = 66.74\%$). There was an asymmetry in the shape of the selectivity funnel plot (Fig. 9B). Begg's test (tau = -0.32, p = 0.03) and Egger's regression test (t = -3.25, p = 0.05) indicated evidence of publication bias (Fig. 9C). Moreover, the Bayesian analysis yielded a smaller effect size (-0.20) compared to the original outcome (-0.25) (Fig. 9D). Therefore, these combined analyses suggested the possibility of publication bias in physiological outcomes.

3.3.3. Subgroup analyses on physiological outcomes

3.3.3.1. Social presence. There was no effect of social presence on pooled physiological outcomes (SMD = -0.12, 95% CI: [-0.34, 0.10], p = 0.29), however, it was shown to decrease skin response with a small effect size (SMD = -0.38, 95% CI: [-0.64, -0.12], p = 0.004) (Fig. 10A).

Further analyses demonstrated that stranger presence decreased physiological outcomes (SMD = -0.31, 95% CI: [-0.48, -0.13], p = 0.0004), while close other presence (a friend) increased physiological responses (SMD = 0.44, 95% CI: [0.16, 0.71], p = 0.002), both with small effect sizes (Fig. 10B).

3.3.3.2. Verbal support. Verbal support decreased physiological

responses from one study with a large effect size (SMD = -0.99, 95% CI: [-1.29, -0.69], p = 0.00001) (Fig. 11A).

3.3.3.3. Social touch. Social touch was shown to have no effect on physiological responses from one study (SMD = -0.09, 95% CI: [-0.30, 0.11], p = 0.37) (Fig. 11B).

4. Discussion

The present study aimed to quantify the influence of social support on pain and related physiological arousal. Built on the literature to date, we further examined two covariates, i.e. social support variations and social relationships, which have been largely reported to modulate social support effect on pain. Our data demonstrated that social presence alone did not affect pain perception. However, the presence of a significant other resulted in an increase of facial expression of pain, and the presence of a stranger was associated with decreased physiological arousal. Meanwhile, verbal support, mainly from a stranger, decreased pain and arousal. We also demonstrated an analgesic effect of intimate relationships occurred in response to touching and viewing a romantic other. Finally, we found evidence of publication bias for pain-related arousal but not for behavioural pain outcomes.

A. Close other	presence: pain outcomes			B. Stranger p	resence: pain outcomes		
Author: Contrast: Outcome		Hedge's g (95% Cl)	Weight %	Author: Contrast: Outcome	• .	Hedge's g (95% CI)	Weight 1
Edwards et al. (2017):1:1	Pain intensity	0.01 (-0.55; 0.58)	1.77%	Edwards et al. (2017):2:1	Pain intensity	0.08 (-0.49; 0.65)	1.73%
Edwards et al. (2017):3:1	•	0.04 (-0.52; 0.61)	1.77%	Gallant et al. (2017):2:1		-0.1 (-0.4; 0.2)	6.21%
Edwards et al. (2017):4:1	_	0.02 (-0.55; 0.58)	1.77%	Karmannetal (2014):2:1	+	-0.09 (-0.34; 0.16)	9.1%
Edwards et al. (2017):5:1	.	-0.04 (-0.6; 0.53)	1.77%	Kleck et al. (1976):1:1		-0.24 (-0.86; 0.38)	1.43%
Edwards et al. (2017):6:1		-0.01 (-0.57; 0.56)	1.77%	Kleck et al. (1976):2:1		-0.56 (-1; -0.11)	2.78%
Gallart et al. (2017):1:1		-0.01 (-0.31: 0.29)	6.36%	Modić Stanke & Ivanec (2010):1:	1	0.16 (-0.43; 0.75)	1.58%
Goldstein et al. (2016):3:1		-0.66 (-1.26: -0.07)	1.61%	Modić Stanke & Ivanec (2010):2:	1	0.17 (-0.43; 0.78)	1.51%
Karmannetal (2014):1:1		-0.13 (-0.38: 0.12)	9.29%	Roberts et al (2015):2:1	•	-0.12 (-0.68; 0.43)	1.8%
Krahé et al (2015)-1-1		0.1(-0.35-0.54)	2 884	Sambo et al (2010):1:1		0.06 (-0.45; 0.56)	2.17%
Kenhá et al (2015):2:1		0.0210.42:045	2 004	Sambo et al (2010):2:1		-0.01 (-0.52; 0.5)	2.17%
Mallelland R. Machabia (2009):1:1		0.02 (0.22, 0.40)	2.00%	Sullivanet al. (2004):1:1	!	0.04 (-0.45; 0.53)	2.3%
Meeteriand & Matcobilin (2006).1.1		0.24 (0.25, 0.72)	1 174	Viaeyen et al. (2009):1:1		-0.24 (-0.56; 0.08)	5.34%
Montoya et al. (2004).1.1		0.32 (1.02, 0.37)	1.175	Subtotal		-0.10 (-0.22, 0.01)	38.12%
vervoort et al.(2011):1:1		-0.31 (-0.76; 0.14)	2.77%	Gallant et al. (2017):2:2		-0.16 (-0.46; 0.13)	1 5 5 12
Subtotal	\sim	-0.07 (-0.19, 0.05)	38.34%	Modić Stanke & Ivane (2010):12		0.20(0.55,0.05)	1.5675
Gallart et al. (2017):1:2	Pain unpleasantness	-0.08 (-0.38; 0.22)	6.35%	Roberts et al. (2015)-2-2		0.03(-0.51;0.7)	1.5275
Subtotal		-0.08 (-0.38, 0.22)	6.35%	Subtotal		-0.06 (-0.28, 0.16)	11.09%
Gallart et al. (2017):1:3	Facial expression of pain	0.2 (-0.1; 0.5)	6.32%	Gallart et al (2017):2:3	Facial expression of pain	-0 23 (-0 53: 0 07)	6 17%
Karmannetal (2014):1:2		0.17 (-0.08; 0.42)	9.28%	Karmannetal (2014):2:2		-0.03 (-0.28: 0.22)	9.1%
Vervoort et al. (2011):1:2		0.36 (-0.09; 0.82)	2.76%	Kleck et al. (1976):1:2		-0.37 (-0.99: 0.26)	1.42%
Subtotal	\sim	0.21 (0.03, 0.38)	18.36%	Kleck et al. (1976):2:2	.	-0.54 (-0.98: -0.09)	2.79%
Edwards et al. (2017):1:2	Pain tolerance	-0.22 (-0.78; 0.35)	1.76%	Sullivan et al. (2004):1:2		0.33 (-0.16; 0.83)	2.27%
Edwards et al. (2017):3:2		-0.29 (-0.86; 0.28)	1.75%	Vlaeven et al. (2009):1:2		-0.01 (-0.33: 0.31)	5.38%
Edwards et al. (2017):4:2	·	-0.21 (-0.78; 0.36)	1.76%	Subtotal	$ \rightarrow $	-0.12 (-0.32, 0.07)	27.13%
Edwards et al. (2017):5:2	•	-0.11 (-0.68; 0.45)	1.77%	Edwards et al. (2017):2:2	Pain tolerance	-0.13 (-0.7; 0.43)	1.73%
Edwards et al. (2017):6:2		-0.1 (-0.66; 0.47)	1.77%	Gallart et al. (2017):2:4	-	-0.04 (-0.32; 0.24)	7.01%
Gallart et al. (2017):1:4	- _	0.05 (-0.23; 0.33)	7.17%	Modić Stanke & Ivane (2010):1:3	,	0.05 (-0.54; 0.64)	1.59%
McClelland & MacCubbin (2008):1:2	•	0.09 (-0.38; 0.56)	2.55%	Modić Stanke & Ivane (2010):2:3	,	0.19 (-0.41; 0.8)	1.51%
Subtotal	\sim	-0.06 (-0.23, 0.12)	18.53%	Subtotal	$ \rightarrow $	-0.01 (-0.23, 0.21)	11.84%
Edwards et al. (2017):1:3	Pain threshold	-0.1 (-0.67; 0.47)	1.77%	Edwards et al. (2017):2:3	Pain threshold	-0.06 (-0.63; 0.5)	1.73%
Edwards et al. (2017):3:3	.	-0.16 (-0.73; 0.41)	1.77%	Gallart et al. (2017):2:5	•++-	-0.22 (-0.5; 0.07)	6.97%
Edwards et al. (2017):4:3	•	-0.11 (-0.68; 0.45)	1.77%	Modić Stanke & Ivane (2010):1:4	, <u> </u>	0 (-0.59; 0.6)	1.59%
Edwards et al. (2017):5:3		-0.03 (-0.6; 0.53)	1.77%	Modić Stanke & Ivane (2010):2:4	 •	0.04 (-0.57; 0.64)	1.52%
Edwards et al. (2017):6:3		-0.01 (-0.57; 0.56)	1.77%	Subtotal	\rightarrow	-0.13 (-0.35, 0.08)	11.81%
Gallart et al. (2017):1:5		-0.06 (-0.35; 0.22)	7.17%	Synthesis	~	-0.09 (-0.17, -0.02)	100%
Montoya et al. (2004):1:2	•	-0.1 (-0.6; 0.39)	2.36%	-	1.5 -1 -0.5 0 0.5	1 1.5	
Subtotal	$ \rightarrow $	-0.08 (-0.25, 0.12)	18.38%		(Decrease pain) Effect size (Inc	reased pain)	
Synthesis	\diamond	-0.01 (-0.09, 0.06)	100%				
-							
-1.	5 -1 -0.5 0 0.5	1 1.5					

Fig. 4. Forest plot of the Hedge's adjusted g analysis for the influence of social presence on behavioural pain outcomes.

4.1. Social presence

We found significant distinctions in the effects of close other presence and stranger presence. Results are therefore discussed separately. Our results demonstrate that pain ratings (i.e. intensity, unpleasantness) and pain sensitivity (i.e. threshold, tolerance) did not vary as a function of the presence of a close other (Fig. 4A). These findings suggest that the mere presence of a significant other has no influence on pain perception. However, close other presence did increase facial expression of pain (Fig. 4A). Moreover, this result was consistently observed across various close relationships (i.e. parent, romantic partner, and family member) and age groups (i.e. children, young adult, and old adult) (Gallant and Hadjistavropoulos, 2017; Karmann et al., 2014; Vervoort et al., 2011, 2008). The facial expression of pain is a powerful means of communicating painful feelings which can easily capture the attention of the significant other, and would be expected to attract necessary support for the pain sufferer (Botvinick et al., 2005; Eritz and Hadjistavropoulos, 2011). In this regard, heightened facial display of pain may serve as a trigger for empathy and social support from significant others, and thus bears evolutionary significance (Williams, 2002).

We also found increased physiological responses to pain in the presence of a friend (Fig. 10B). This finding is consistent with the heightened facial expression of pain (Gallant and Hadjistavropoulos, 2017; Karmann et al., 2014; Vervoort et al., 2011, 2008). McClelland and McCubbin (2008) suggested that in the presence of a significant other, the pain recipient may increase physiological arousal to elicit desirable social responses. Their results indicated that the increase in arousal occurred in the absence of visual interactions with the social partner, which can otherwise play a critical role in eliciting an empathetic response (Williams, 2002). On the other hand, this finding is contradictory to evidence that physical presence of a friend can



A. Verbal support: pain outcomes

B. Stranger verbal support: pain outcomes

Fig. 5. Forest plot of the Hedge's adjusted g analysis for the effects of verbal support on behavioural pain outcomes.

role in the social presence effect of pain. There is evidence that stranger

presence may signal 'safety' in the reduction of pain and related threat.

But this effect was only observed in a high threatening context (Vlaeyen

et al., 2009). Further evidence has suggested that the effects of stranger

presence can be determined by attachment style. Specifically, in-

dividuals with high attachment avoidance reported more pain in the

presence of a stranger than in absence (Sambo et al., 2010). Avoidant

individuals tend to mistrust social relationships (Bowlby, 1969; Feeney

and Noller, 1990; Main, 2000). Their preference for independence may

be associated with increased anxiety in the presence of a total stranger,

which may in turn increases pain perception (Sambo et al., 2010).

Moreover, pain catastrophising may modulate the influence of social support. In the presence of a stranger, high pain catastrophisers ex-

pressed more pain and used fewer coping strategies that might mini-

mise pain (Sullivan et al., 2004). These findings highlight the com-

stranger presence (Fig. 10B), in contrast to the presence of a close other

which tended to increase arousal. Stranger presence has been re-

peatedly reported to decrease physiological arousal to painful stimuli

(Kleck et al., 1976; Sambo et al., 2010), as well as to other psychosocial

stressors (Fontana et al., 1999; Lepore, 1995; Lepore et al., 1993).

However, it is interesting to find that decreased arousal did not ne-

cessarily translate into pain reduction. One possibility is the lack of

perceived support in the presence of a stranger (Roberts et al., 2015).

Indeed, the effect of social support on pain is modulated by the extent to

which the participant could perceive the social partner as helpful

(Krahé et al., 2013). Therefore, knowing that someone is present is

sufficient to buffer physiological arousal; but pain reduction relies on

Interestingly, physiological responses were reduced in the context of

plexities of stranger presence in pain perception and display.



Fig. 6. Forest plot of the Hedge's adjusted g analysis for the influence of social touch on behavioural pain outcomes.

suppress physiological arousal to psychosocial stressors, for example, an arithmetic task or public speech (Fontana et al., 1999). It is possible that stress type plays a role in this discordance. Inserting hand into iced water for two minutes composes a direct threat to the body in which individuals may need help from the significant other (McClelland and McCubbin, 2008). Meanwhile, individuals may be less aroused and more capable of coping with a moderate social-evaluative threat with the presence of a close other (Fontana et al., 1999).

Our results demonstrate that the presence of a stranger did not influence pain perception or expression (Fig. 4B). Some studies have shown that the presence of a stranger is associated with decreased pain (Edwards et al., 2017; Gallant and Hadjistavropoulos, 2017; Karmann et al., 2014; Kleck et al., 1976). It has been proposed that individuals tend to display and report less pain to an unfamiliar other as pain expression could be 'inappropriate' and may indicate 'vulnerability' (Brody, 2000; Craig, 2009; Underwood et al., 1992). Moreover, individuals are reluctant to show excessive pain as they may not receive desirable responses from an unpredictable stranger (Gallant and Hadjistavropoulos, 2017; Williams, 2002). However, another line of research has demonstrated no impact of stranger presence on pain (Modić Stanke and Ivanec, 2010; Roberts et al., 2015). The authors argued that stranger presence is not sufficient to communicate social support (Roberts et al., 2015), and that decreased pain perception or display could be the outcome of socially desirable behaviour that was minimised in their protocol (Modić Stanke and Ivanec, 2010). Nevertheless, these findings are not mutually exclusive and they suggest the need for controlling social desirability bias in support and pain research.

In addition, other social and temperament factors may also play a



29

Fig. 7. Forest plot of the Hedge's adjusted g analysis for the effects of primed support on behavioural pain outcomes.



Fig. 8. Forest plot of the Hedge's adjusted g analysis for pooled effects of social support on physiological outcomes.

not only the physical presence but also the level of perceived support from the social partner.

4.2. Verbal support

In contrast to social presence, our data showed that verbal support decreased overall pain experience (Fig. 5A). In a systematic review, Krahé et al. (2013) suggested an analgesic influence of verbal communication when it is positive and structured (i.e. pre-determined verbal elements and interactions) (Brown et al., 2003; Chambers et al., 2002; Jackson, 2007; Roberts et al., 2015). However, uninstructed (i.e. not pre-determinant, random) verbal communications were related to increased pain (Brown et al., 2003). Here the majority of the data centred on positive and structured verbal support. Our results thus corroborate the findings of the systematic review (Krahé et al., 2013). Interestingly, most of the data were derived from verbal support from a stranger which reduced pain with a large effect size (SMD = -0.92) (Fig. 5B). Moreover, this effect was most prominent among females (Jackson et al., 2005; Roberts et al., 2015). Findings together suggest that verbal support, at least from a stranger, can communicate support and reduce pain, especially in female pain recipients.

Our results also suggest that verbal support may buffer physiological arousal in the reduction of pain (Fig. 11A). Although this finding is only based on one study, it did assess multiple physiological responses. Nevertheless, more studies are required in this area. Moreover, these results are consistent with recent evidence which emphasises the significance of intimacy and emotion regulation (Cano and Williams, 2010). Specifically, verbal communication and validations may serve to increase intimacy and/or healthy emotion regulation, which may in turn result in a decrease in pain and perceived threat (Corley et al., 2016; Leong et al., 2015). Moreover, the stress-buffering effect of verbal support is enhanced by oxytocin (Heinrichs et al., 2003), a neuropeptide that promotes intimacy and social interaction (Carter, 1998; McCarthy and Altemus, 1997; Uvnäs-Moberg, 1998). Overall, our findings suggest that the analgesic effects of verbal support may be modulated by availability or levels of oxytocin that buffers physiological arousal to pain.

It is noted that studies categorised under verbal support may provide somewhat different forms of support. In some studies, the social partner verbally encouraged the participant to engage in the pain task (Brown et al., 2003; Roberts et al., 2015), while in another study the social partner was not allowed to initiate any interaction but answer the questions of the participant in an empathetic fashion (Jackson et al., 2005). These types of social interaction may result in varied perceived social support by the one in pain and therefore have different effects on pain experience. Here we averaged them for the purpose of a metaanalysis. Future studies are warranted to investigate the effects of different aspects of verbal support.

4.3. Social touch

Social touch did not influence overall pain ratings (i.e. intensity, unpleasantness) (Fig. 6A). However, the effects of social touch were strongly influenced by pre-existing social connectedness. Analgesic effects of social touch were evident with touch by a romantic partner but not by a stranger (Fig. 6B). Although based on a small number of studies, these findings highlight the importance of intimacy in communicating support and reducing pain. Indeed, human touch communicates distinct emotions (Hertenstein et al., 2006). Holding the hand of a romantic partner is suggested to promote intimacy between couples



Fig. 9. Series of tests for heterogeneity and publication bias for the influence of social support on physiological outcomes. (A) Galbraith plot suggested heterogeneity with more than 5% of the data beyond two standard errors of the population effect. (B) Selectivity funnel plot indicated publication bias with studies showing asymmetry around the estimated effect. (C) Regression plot suggested publication bias with the intercept deviated from 0 (p = 0.05). (D) Bayesian triplot indicated publication bias with the likelihood effect size (-0.20) smaller than the estimated effect size (-0.25).



A. Social presence: physiological outcomes



A. Verbal support



B. Social touch



Fig. 11. Forest plot of the Hedge's adjusted g analysis for the influence of verbal support and social touch on physiological outcomes.

(Goldstein et al., 2016, 2017; Master et al., 2009). On the contrary, holding hand by a stranger may be somewhat socially uncomfortable for both partners and thus associated with mixed results (Krahé et al., 2013).

Recent advances in neurobiological research have enriched our understanding of the mechanisms behind social touch. Although physiological arousal to pain was decreased by stranger touch (Fishman et al., 1995), the effect size was small and did not reach significance (Fig. 11B). Meanwhile, social touch by the romantic partner attenuated bodily arousal and threat-related brain activation (e.g. ventral anterior cingulate cortex) to painful stimuli (Coan et al., 2017, 2006). Close other hand-holding is also associated with decreased heart rate and threat-related theta oscillation in high threatening contexts (Che et al., 2017). These findings together indicate the role of intimate social touch in buffering neural and physiological threatening responses to pain. Moreover, intimate social touch has been shown to increase the interpartner coupling of respiration and heart rate during pain, which was suggested to promote the communication of emotions (Goldstein et al., 2017). Therefore, social touch from a significant other may be associated with neurophysiological processes supporting emotional interaction, stress control, and pain reduction.

4.4. Primed support

In a similar manner to social touch, our data also demonstrated no changes in overall pain by primed support (e.g. image-viewing) (Fig. 7A). However, viewing images of a romantic partner was associated with pain relief in healthy individuals (Fig. 7B). Moreover, this finding has been extended into people with chronic pain, in which images of a loved one were rated the most pleasant and were associated with the greatest pain reduction (Shaygan et al., 2017). These studies suggest that the effects of primed support, where present, are strongly determined by relationship quality. Functional imaging studies have

explored the neural correlates of this effect. One study found that the analgesic effect of partner image-viewing was associated with rewardrelated neural activation (e.g. orbitofrontal cortex) (Younger et al., 2010). Moreover, partner images may be interpreted as safety stimuli (e.g. ventromedial prefrontal cortex) that in turn decrease pain and threat response (Eisenberger and Cole, 2012; Eisenberger et al., 2011). This argument has been corroborated by recent studies in which social support images could serve as prepared safety stimuli to attenuate conditioned fear response (i.e. skin response) (Hornstein and Eisenberger, 2017; Hornstein et al., 2016, 2017). Therefore, viewing an image of a significant other may prime safety or attachment feelings which have analgesic effects.

It is noted that partner hand-holding and image-viewing both decreased pain. Unlike the presence of a social partner, our results support that holding the hand or viewing images of an intimate partner can provide social support to an individual in pain. This is proposed to occur through an increase in attachment and emotional interactions (Eisenberger et al., 2011; Goldstein et al., 2017). However, limited conclusions can be made here as the findings are based on two studies under each category, which clearly indicates the need for more studies to determine the consistency of effects. Moreover, these two variants of support may rely on different mechanisms to decrease pain. Initial evidence suggests a mechanism of emotional communication and stress buffering in the context of intimate social touch (Che et al., 2017; Goldstein et al., 2017). A more recent study found that intimate social touch increased brain-to-brain coupling of alpha power mainly over the central cortical regions between romantic couples. The authors proposed that this may result from sensory integration from tactile, visual, and nociceptive inputs (Goldstein et al., 2018). In contrast, intimate image-viewing is thought to prime attachment feelings involved in the prefrontal cortex (Eisenberger et al., 2011; Younger et al., 2010). It would be interesting for future studies to directly compare and contrast the effects of these two types of social support during pain.

Personal and interpersonal variables may modulate the receipt and provision of social support in the influence on pain. Some characteristics of the support recipients, e.g. attachment style and pain catastrophising, have been demonstrated to modulate the social presence effect on pain (Sambo et al., 2010; Sullivan et al., 2004). However, it remains unknown whether other related characteristics also play a role in the social modulation of pain. There is evidence that individuals who are more prone to rely on others, or lower in neuroticism, tend to benefit more from social support (Park et al., 2013). Moreover, characteristics of the social partner may also influence the provision of support. For example, individuals with higher dispositional empathy may be more willing to provide social support to others (Trobst et al., 1994). Future investigations should therefore explore the characteristics of both the individual providing support and the person experiencing pain.

4.5. Publication bias

The presence of publication bias can affect the results of a metaanalysis. We therefore used several methods to test publication bias in this study. There was no clear evidence of publication bias in the behavioural dataset, although we observed a trend towards significance (Fig. 3). This is probably due to the 'true heterogeneity' (Sterne et al., 2011), in which a few studies (i.e. potential outliers) reported relatively larger analgesic effects of verbal support (Roberts et al., 2015) and partner hand-holding (Goldstein et al., 2016) among most of the studies of social presence. It suggests the different effects of social support variants and the need for more studies evaluating verbal support and intimate hand-holding. However, we did observe a possibility of publication bias in the physiological dataset (Fig. 9). It is possible that potential variables were confounding the results, e.g. sample sizes in each study, and the total number of studies. Moreover, publications from languages other than English were omitted here which might contribute to publication bias.

4.6. Limitations

Some limitations should be considered when interpreting the results of this study. Sample sizes varied in different analyses and they were small in certain subgroup analyses, e.g. close other social touch. This could influence the effect size and statistical significance. Further, the control condition in some studies may be different from the experimental condition in terms of multisensory integration beyond the proposed difference in social support, e.g. no touch versus social touch (Krahé et al., 2013). Moreover, some studies have suggested a gender difference in the social support effect of pain (Chambers et al., 2002; Jackson et al., 2005). Here we did not find an overall effect of gender on the social presence effect (see Supplementary Material S2). Due to the limited data, we did not statistically examine the gender differences in other types of social support. However, the analgesic influence of partner hand-holding (Fig. 6B), partner image-viewing (Fig. 7B), and stranger verbal support (Fig. 5B) was behaviourally more prominent in females. Future research is therefore warranted investigating gender effects in social support and pain. Some studies have also suggested other possible covariates beyond the support variations and relationship quality, e.g. attachment style (Krahé et al., 2015), and pain catastrophising (Sullivan et al., 2004). We did not exclusively quantify these covariates as of the limited data. In addition, neural imaging studies were not reviewed quantitatively due to the limited number of studies.

4.7. Future directions

Results of this study also provide insights for future studies. Oxytocin is suggested to increase the stress-buffering effect of verbal support (Heinrichs et al., 2003). However, no study has exclusively examined the role of oxytocin in the analgesic influence of social support, despite oxytocin being linked to both analgesia (González-Hernández et al., 2014; Paloyelis et al., 2016) and to social support (Heinrichs et al., 2003) independently. Moreover, while social touch is associated with higher level of oxytocin (Holt-Lunstad et al., 2008; Light et al., 2005), it remains to be determined if this effect is common across other forms of social support. Further research is needed to investigate the role of oxytocin and other stress hormones in support and pain research, e.g. glucocorticoids (Wittig et al., 2016), and epinephrine (Wirtz et al., 2006). Finally, two recent studies have assessed the interpartner coupling of neural and physiological responses that may be involved in communicating emotions and reducing pain (Goldstein et al., 2018, 2017). These studies extend the current investigations to concurrent recordings of both partners in the social modulation of pain. However, more studies are needed to validate and extend these interesting findings.

5. Conclusions and implications

Social support has been documented to modulate pain and related arousal for several decades. In this study, we systematically quantified these effects and possible covariates. The mere presence of another person, although it may impact physiological arousal, was not found to be sufficient to reduce pain. However, we did identify social support to decrease pain when it is more clearly expressed, e.g. verbal communication, hand-holding. Our findings also highlight the significance of intimate relationships in pain reduction. Together, we provide comprehensive findings in which the context of social support can modulate pain as well as the magnitude and possible mechanisms underlying this effect.

Our results may also provide insights for the support therapies for chronic pain management (Abbasi et al., 2012; Keefe et al., 2004, 1996, 1999). Verbal support showed a medium effect size and it would be particularly effective at highlighting the positive and structured interactions between couples (Krahé et al., 2013) as well as verbal validations from the partner (Leong et al., 2015). Verbal interactions characterised by encouragement, reinterpretation, and emotional validation could be involved in training programs. Moreover, our results support that spousal intimacy could be an area of therapeutic focus (Eisenberger et al., 2011), potentially through mutual activities between couples. It may also important to clearly express intimacy in support therapies through nonverbal interactions, e.g. touch. Beyond pain research, our results also suggest a protective influence of social support on threat response, i.e. pain (Che et al., 2018), which therefore has an implication in coping with traumatic experiences (Richmond et al., 2018).

Financial disclosures

None declared.

Conflicts of interest

There are no other conflicts.

Acknowledgments

XC is supported by the China Scholarship Council (201506990016). PBF is supported by a NHMRC Practitioner Fellowship (606907). BMF is supported by a NHMRC Early Career Fellowship.

Appendix A. Supplementary data

Supplementary material related to this article can be found, in the online version, at doi:https://doi.org/10.1016/j.neubiorev.2018.07. 005.

X. Che et al.

References

- Abbasi, M., Dehghani, M., Keefe, F., Jafari, H., Behtash, H., Shams, J., 2012.
- Spouse-assisted training in pain coping skills and the outcome of multidisciplinary pain management for chronic low back pain treatment: a 1-year randomized controlled trial. Eur. J. Pain 16, 1033–1043.
- Bax, L., 2011. MIX 2.0. Professional Software for Meta-Analysis in Excel. Available at: https://www.meta-analysis-made-easy.com/.
- Begg, C.B., Mazumdar, M., 1994. Operating characteristics of a rank correlation test for publication bias. Biometrics 50, 1088–1101.
- Boerner, K.E., Birnie, K.A., Caes, L., Schinkel, M., Chambers, C.T., 2014. Sex differences in experimental pain among healthy children: a systematic review and meta-analysis. Pain 155, 983–993.
- Borsook, T.K., MacDonald, G., 2010. Mildly negative social encounters reduce physical pain sensitivity. Pain 151, 372–377.
- Botvinick, M., Jha, A.P., Bylsma, L.M., Fabian, S.A., Solomon, P.E., Prkachin, K.M., 2005. Viewing facial expressions of pain engages cortical areas involved in the direct experience of pain. Neuroimage 25, 312–319.
- Bowlby, J., 1969. Attachment and Loss: Attachment 1 Basic Books, New York.
- Brody, L.R., 2000. The socialization of gender differences in emotional expression: display rules, infant temperament, and differentiation. In: Fischer, A.H. (Ed.), Gender and Emotion: social Psychological Perspectives. Cambridge University Press, New York, pp. 24–47.
- Brown, J.L., Sheffield, D., Leary, M.R., Robinson, M.E., 2003. Social support and experimental pain. Psychosom. Med. 65, 276–283.
- Cano, A., Williams, A.Cd.C., 2010. Social interaction in pain: reinforcing pain behaviors or building intimacy? Pain 149, 9–11.
- Carter, C.S., 1998. Neuroendocrine perspectives on social attachment and love. Psychoneuroendocrinology 23, 779–818.
- Chambers, C.T., Craig, K.D., Bennett, S.M., 2002. The impact of maternal behavior on children's pain experiences: an experimental analysis. J. Pediatr. Psychol. 27, 293–301.
- Che, X., Cash, R., Fitzgerald, P., Fitzgibbon, B.M., 2017. The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat. J. Pain 19, 496–505.
- Che, X., Cash, R., Ng, S.K., Fitzgerald, P., Fitzgibbon, B.M., 2018. A systematic review of the processes underlying the main and the buffering effect of social support on the experience of pain. Clin. J. Pain. https://doi.org/10.1097/AJP.00000000000624.
- Coan, J.A., Schaefer, H.S., Davidson, R.J., 2006. Lending a hand: social regulation of the neural response to threat. Psychol. Sci. 17, 1032–1039.
- Coan, J.A., Beckes, L., Gonzalez, M.Z., Maresh, E.L., Brown, C.L., Hasselmo, K., 2017. Relationship status and perceived support in the social regulation of neural responses to threat. Soc. Cogn. Affect. Neurosci. 12, 1574–1583.
- Cohen, J., 1988. Statistical Power Analysis for the Behavioral Sciences. Lawrence Earlbaum Associates, New Jersey.
- Cohen, S., Wills, T.A., 1985. Stress, social support, and the buffering hypothesis. Psychol. Bull. 98, 310–357.
- Corley, A.M., Cano, A., Goubert, L., Vlaeyen, J.W.S., Wurm, L.H., 2016. Global and situational relationship satisfaction moderate the effect of threat on pain in couples. Pain Med. 17, 1664–1675.
- Craig, K.D., 2009. The social communication model of pain. Can. Psychol./Psychologie Canadienne 50, 22–32.
- Dunkel-Schetter, C., Bennett, T.L., 1990. Differentiating the cognitive and behavioral aspects of social support. In: Sarason, B.R., Sarason, I.G., Pierce, G.R. (Eds.), Wiley Series on Personality Processes. Social Support: An Interactional View. John Wiley, Oxford, pp. 267–296.
- Edwards, R., Eccleston, C., Keogh, E., 2017. Observer influences on pain: an experimental series examining same-sex and opposite-sex friends, strangers, and romantic partners. Pain 158, 846–855.
- Egger, M., Smith, G.D., Schneider, M., Minder, C., 1997. Bias in meta-analysis detected by a simple, graphical test. BMJ: Br. Med. J. 315, 629–634.
- Eisenberger, N.I., Cole, S.W., 2012. Social neuroscience and health: neurophysiological mechanisms linking social ties with physical health. Nat. Neurosci. 15, 669–674.
- Eisenberger, N.I., Master, S.L., Inagaki, T.K., Taylor, S.E., Shirinyan, D., Lieberman, M.D., Naliboff, B.D., 2011. Attachment figures activate a safety signal-related neural region and reduce pain experience. Proc. Natl. Acad. Sci. 108, 11721–11726.
- Eritz, H., Hadjistavropoulos, T., 2011. Do informal caregivers consider nonverbal behavior when they assess pain in people with severe dementia? J. Pain 12, 331–339.
- Feeney, J.A., Noller, P., 1990. Attachment style as a predictor of adult romantic relationships. J. Pers. Soc. Psychol. 58, 281–291.
- Fishman, E., Turkheimer, E., DeGood, D.E., 1995. Touch relieves stress and pain. J. Behav. Med. 18, 69–79.
- Fontana, A.M., Diegnan, T., Villeneuve, A., Lepore, S.J., 1999. Nonevaluative social support reduces cardiovascular reactivity in young women during acutely stressful performance situations. J. Behav. Med. 22, 75–91.
- Gallant, N.L., Hadjistavropoulos, T., 2017. Experiencing pain in the presence of others: a structured experimental investigation of older adults. J. Pain 18, 456–467.
- Glenton, C., 2003. Chronic back pain sufferers—striving for the sick role. Soc. Sci. Med. 57, 2243–2252.
- Goldstein, P., Shamay-Tsoory, S.G., Yellinek, S., Weissman-Fogel, I., 2016. Empathy predicts an experimental pain reduction during touch. J. Pain 17, 1049–1057.
- Goldstein, P., Weissman-Fogel, I., Shamay-Tsoory, S.G., 2017. The role of touch in regulating inter-partner physiological coupling during empathy for pain. Sci. Rep. 7. https://doi.org/10.1038/s41598-017-03627-7.
- Goldstein, P., Weissman-Fogel, I., Dumas, G., Shamay-Tsoory, S.G., Goldstein, P.,

Weissman-Fogel, I., Dumas, G., Shamay-Tsoory, S.G., 2018. Brain-to-brain coupling during handholding is associated with pain reduction. Proc. Natl. Acad. Sci. 115, E2528–E2537.

- González-Hernández, A., Rojas-Piloni, G., Condés-Lara, M., 2014. Oxytocin and analgesia: future trends. Trends Pharmacol. Sci. 35, 549–551.
- Guan, M., Vandekerckhove, J., 2016. A bayesian approach to mitigation of publication bias. Psychon. Bull. Rev. 23, 74–86.
- Hedges, L., Olkin, I., 1985. Statistical Methods for Meta-Analysis. Academic Press, Orlando.
- Heinrichs, M., Baumgartner, T., Kirschbaum, C., Ehlert, U., 2003. Social support and oxytocin interact to suppress cortisol and subjective responses to psychosocial stress. Biol. Psychiatry 54, 1389–1398.

Hertenstein, M.J., Keltner, D., App, B., Bulleit, B.A., Jaskolka, A.R., 2006. Touch communicates distinct emotions. Emotion 6. pp. 528–533.

- Higgins, J.P., Green, S., 2011. Cochrane Handbook for Systematic Reviews of Interventions. Version 5.1.0. The Cochrane Collaboration. Available at:. http:// www.cochrane-handbook.org.
- Higgins, J.P., Thompson, S.G., Deeks, J.J., Altman, D.G., 2003. Measuring inconsistency in meta-analyses. BMJ: Br. Med. J. 327, 557–560.
- Holt-Lunstad, J., Birmingham, W.A., Light, K.C., 2008. Influence of a "warm touch" support enhancement intervention among married couples on ambulatory blood pressure, oxytocin, alpha amylase, and cortisol. Psychosom. Med. 70, 976–985.
- Hornstein, E.A., Eisenberger, N.I., 2017. Unpacking the buffering effect of social support figures: social support attenuates fear acquisition. PLoS One 12, e0175891.
- Hornstein, E.A., Fanselow, M.S., Eisenberger, N.I., 2016. A safe haven: investigating social-support figures as prepared safety stimuli. Psychol. Sci. 27, 1051–1060.
- Hornstein, E.A., Haltom, K.E., Shirole, K., Eisenberger, N.I., 2017. A unique safety Signal: social-support figures enhance rather than protect from fear extinction. Clin. Psychol. Sci. doi. https://doi.org/10.1177/2167702617743002.
- Hurter, S., Paloyelis, Y., Williams, A.Cd.C., Fotopoulou, A., 2014. Partners' empathy increases pain ratings: effects of perceived empathy and attachment style on pain report and display. J. Pain 15, 934–944.
- Huwaldt, J.A., 2010. Plot Digitizer, Open Source Software. available at. http:// plotdigitizer.sourceforge.net/.
- Jackson, T., 2007. Interpersonal transactions and responses to cold pressor pain among Australian women and men. Sex Roles 56, 55–62.
- Jackson, T., Iezzi, T., Chen, H., Ebnet, S., Eglitis, K., 2005. Gender, interpersonal transactions, and the perception of pain: an experimental analysis. J. Pain 6, 228–236.
- Karmann, A.J., Lautenbacher, S., Bauer, F., Kunz, M., 2014. The influence of communicative relations on facial responses to pain: does it matter who is watching? Pain Res. Manage. 19, 15–22.
- Keefe, F.J., Caldwell, D.S., Baucom, D., Salley, A., Robinson, E., Timmons, K., Beaupre, P., Weisberg, J., Helms, M., 1996. Spouse-assisted coping skills training in the management of osteoarthritic knee pain. Arthr. Rheumatol. 9, 279–291.
- Keefe, F.J., Caldwell, D.S., Baucom, D., Salley, A., Robinson, E., Timmons, K., Beaupre, P., Weisberg, J., Helms, M., 1999. Spouse-assisted coping skills training in the management of knee pain in osteoarthritis: long-term followup results. Arthritis Rheum. 12, 101–111.
- Keefe, F.J., Blumenthal, J., Baucom, D., Affleck, G., Waugh, R., Caldwell, D.S., Beaupre, P., Kashikar-Zuck, S., Wright, K., Egert, J., 2004. Effects of spouse-assisted coping skills training and exercise training in patients with osteoarthritic knee pain: a randomized controlled study. Pain 110, 539–549.
- Kim, H.J., Yang, G.S., Greenspan, J.D., Downton, K.D., Griffith, K.A., Renn, C.L., Johantgen, M., Dorsey, S.G., 2017. Racial and ethnic differences in experimental pain sensitivity: systematic review and meta-analysis. Pain 158, 194–211.
- Kleck, R.E., Vaughan, R.C., Cartwright-Smith, J., Vaughan, K.B., Colby, C.Z., Lanzetta, J.T., 1976. Effects of being observed on expressive, subjective, and physiological responses to painful stimuli. J. Pers. Soc. Psychol. 34, 1211–1218.
- Krahé, C., Springer, A., Weinman, J.A., Fotopoulou, A., 2013. The Social Modulation of Pain: Others as Predictive Signals of Salience–A Systematic Review 7. pp. 386.
- Krahé, C., Paloyelis, Y., Condon, H., Jenkinson, P.M., Williams, S.C., Fotopoulou, A., 2015. Attachment style moderates partner presence effects on pain: a laser-evoked potentials study. Soc. Cogn. Affect. Neurosci. 10, 1030–1037.
- Lautenbacher, S., Peters, J., Heesen, M., Scheel, J., Kunz, M., 2017. Age changes in pain perception: a systematic-review and meta-analysis of age effects on pain and tolerance thresholds. Neurosci. Biobehav. Rev. 75, 104–113.
- Leong, L.E., Cano, A., Wurm, L.H., Lumley, M.A., Corley, A.M., 2015. A perspectivetaking manipulation leads to Greater empathy and less pain during the cold pressor task. J. Pain 16, 1176–1185.
- Lepore, S.J., 1995. Cynicism, social support, and cardiovascular reactivity. Health Psychol. 14, 210–216.
- Lepore, S.J., Allen, K., Evans, G.W., 1993. Social support lowers cardiovascular reactivity to an acute stressor. Psychosom. Med. 55, 518–524.
- Light, R., Pillemer, D., 1984. Summing up: The Science of Research Reviewing. Harvard University Press, Cambridge.
- Light, K.C., Grewen, K.M., Amico, J.A., 2005. More frequent partner hugs and higher oxytocin levels are linked to lower blood pressure and heart rate in premenopausal women. Biol. Psychol. 69, 5–21.
- Main, M., 2000. The organized categories of infant, child, and adult attachment: flexible vs. Inflexible attention under attachment-related stress. J. Am. Psychoanal. Assoc. 48, 1055–1096.
- Master, S.L., Eisenberger, N.I., Taylor, S.E., Naliboff, B.D., Shirinyan, D., Lieberman, M.D., 2009. A picture's worth: partner photographs reduce experimentally induced pain. Psychol. Sci. 20, 1316–1318.
- McCarthy, M.M., Altemus, M., 1997. Central nervous system actions of oxytocin and modulation of behavior in humans. Mol. Med. Today 3, 269–275.

McClelland, L.E., McCubbin, J.A., 2008. Social influence and pain response in women and men. J. Behav. Med. 31, 413–420.

- Modić Stanke, K., Ivanec, D., 2010. Social context of pain perception: the role of other people's presence and physical distance. Rev. Psychol. 17, 69–74.
- Moher, D., Liberati, A., Tetzlaff, J., Altman, D.G., 2009. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. Ann. Intern. Med. 151, 264–269.
- Montoya, P., Larbig, W., Braun, C., Preissl, H., Birbaumer, N., 2004. Influence of social support and emotional context on pain processing and magnetic brain responses in fibromyalgia. Arthritis Rheum. 50, 4035–4044.
- Paloyelis, Y., Krahé, C., Maltezos, S., Williams, S., Howard, M., Fotopoulou, A., 2016. The analgesic effect of oxytocin in humans: a double-blind, placebo-controlled cross-over study using laser-evoked potentials. J. Neuroendocrinol. 28. https://doi.org/10. 1111/ine.12347.
- Park, J., Kitayama, S., Karasawa, M., Curhan, K., Markus, H.R., Kawakami, N., Miyamoto, Y., Love, G.D., Coe, C.L., Ryff, C.D., 2013. Clarifying the links between social support and health: culture, stress, and neuroticism matter. J. Health Psychol. 18, 226–235.
- Platow, M.J., Voudouris, N.J., Gilford, N., Jamieson, R., Najdovski, L., Papaleo, N., Pollard, C., Terry, L., 2007. In-group reassurance in a pain setting produces lower levels of physiological arousal: direct support for a self-categorization analysis of social influence. Eur. J. Soc. Psychol. 37, 649–660.
- Richmond, N.L., Meyer, M.L., Hollowell, A.G., Isenberg, E.E., Domeier, R.M., Swor, R.A., Hendry, P.L., Peak, D.A., Rathlev, N.K., Jones, J.S., 2018. Social support and pain outcomes after trauma exposure among older adults: a multicenter longitudinal study. Clin. J. pain 34, 366–374.
- Roberts, M.H., Klatzkin, R.R., Mechlin, B., 2015. Social support attenuates physiological stress responses and experimental pain sensitivity to cold pressor pain. Ann. Behav. Med. 49, 557–569.
- Sambo, C.F., Howard, M., Kopelman, M., Williams, S., Fotopoulou, A., 2010. Knowing you care: effects of perceived empathy and attachment style on pain perception. Pain 151, 687–693.
- Shaygan, M., Böger, A., Kröner-Herwig, B., 2017. Valence and arousal value of visual stimuli and their role in the mitigation of chronic pain: what is the power of pictures? J. Pain 18, 124–131.
- Sterne, J.A., Sutton, A.J., Ioannidis, J.P., Terrin, N., Jones, D.R., Lau, J., Carpenter, J., Rücker, G., Harbord, R.M., Schmid, C.H., 2011. Recommendations for examining and interpreting funnel plot asymmetry in meta-analyses of randomised controlled trials. BMJ: Br. Med. J. https://doi.org/10.1136/bmj.d4002.
- Sullivan, M.J.L., Adams, H., Sullivan, M.E., 2004. Communicative dimensions of pain catastrophizing: social cueing effects on pain behaviour and coping. Pain 107, 220–226.
- Tesarz, J., Schuster, A.K., Hartmann, M., Gerhardt, A., Eich, W., 2012. Pain perception in athletes compared to normally active controls: a systematic review with meta-

- analysis. Pain 153, 1253-1262.
- Thompson, S.G., 1994. Why sources of heterogeneity in meta-analysis should be investigated. BMJ: Br. Med. J. 309, 1351–1355.
- Thompson, T., Correll, C.U., Gallop, K., Vancampfort, D., Stubbs, B., 2016. Is pain perception altered in people with depression? A systematic review and meta-analysis of experimental pain research. J. Pain 17, 1257–1272.
- Trobst, K.K., Collins, R.L., Embree, J.M., 1994. The role of emotion in social support provision: gender, empathy and expressions of distress. J. Soc. Pers. Relat. 11, 45–62.
- Uchino, B.N., 2006. Social support and health: a review of physiological processes potentially underlying links to disease outcomes. J. Behav. Med. 29, 377–387.
- Uchino, B.N., 2009. Understanding the links between social support and physical health: a life-span perspective with emphasis on the separability of perceived and received support. Perspect. Psychol. Sci. 4, 236–255.
- Underwood, M.K., Coie, J.D., Herbsman, C.R., 1992. Display rules for anger and aggression in school-age children. Child Dev. 63, 366–380.
- Uvnäs-Moberg, K., 1998. Oxytocin may mediate the benefits of positive social interaction and emotions. Psychoneuroendocrinology 23, 819–835.
- Vervoort, T., Goubert, L., Eccleston, C., Verhoeven, K., De Clercq, A., Buysse, A., Crombez, G., 2008. The effects of parental presence upon the facial expression of pain: the moderating role of child pain catastrophizing. Pain 138, 277–285.
- Vervoort, T., Caes, L., Trost, Z., Sullivan, M., Vangronsveld, K., Goubert, L., 2011. Social modulation of facial pain display in high-catastrophizing children: an observational study in schoolchildren and their parents. Pain 152, 1591–1599.
- Vlaeyen, J.W., Hanssen, M., Goubert, L., Vervoort, T., Peters, M., van Breukelen, G., Sullivan, M.J., Morley, S., 2009. Threat of pain influences social context effects on verbal pain report and facial expression. Behav. Res. Ther. 47, 774–782.
- Wells, G., Shea, B., O'Connell, D., Peterson, J., Welch, V., Losos, M., Tugwell, P., 2010. The Newcastle-Ottawa Scale (NOS) For Assessing the Quality Of Nonrandomized Studies In Meta-Analysis. Available from:. http://www.ohri.ca/programs/clinical_ epidemiology/oxford.asp.
- Williams, A.Cd.C., 2002. Facial expression of pain: an evolutionary account. Behav. Brain Sci. 25, 439–455.
- Wirtz, P.H., von Känel, R., Mohiyeddini, C., Emini, L., Ruedisueli, K., Groessbauer, S., Ehlert, U., 2006. Low social support and poor emotional regulation are associated with increased stress hormone reactivity to mental stress in systemic hypertension. J. Clin. Endocrinol. Metab. 91, 3857–3865.
- Wittig, R.M., Crockford, C., Weltring, A., Langergraber, K.E., Deschner, T., Zuberbühler, K., 2016. Social support reduces stress hormone levels in wild chimpanzees across stressful events and everyday affiliations. Nat. Commun. 7. https://doi.org/10.1038/ ncomms13361.
- Younger, J., Aron, A., Parke, S., Chatterjee, N., Mackey, S., 2010. Viewing pictures of a romantic partner reduces experimental pain: involvement of neural reward systems. PLoS One 5, e13309.

CHAPTER THREE

Potential Processes and Mechanisms

Manuscript

Che X., Cash R., Ng S.K., Fitzgerald P., & Fitzgibbon B.M. (2018). A systematic review of the processes underlying the main and the buffering effect of social support on the experience of pain. *Clinical Journal of Pain*, *34*, 1061-1076.

Preamble to systematic review

The previous chapter provides a quantitative review of the social contexts and effect size in the social modulation of pain. However, the processes and mechanisms that link social support to pain reduction remain unknown. In a recent systematic review, Bernardes et al. (2017) argued that pain-related social support needs to be investigated within the scope of stress and coping process as social support has been shown to buffer the detrimental effects of stress on health outcomes.

This chapter aims to explore potential processes and mechanisms associated with social support effects on pain by systematically assessing the literature suggesting it holds a protective effect. Given the diversity of the literature (e.g. experimental versus clinical), a broad concept of "pain experience" is used which includes both physical (e.g. intensity, duration, disability) and psychosocial suffering (e.g. unpleasantness, negative mood, psychological, and social functioning). The literature is assessed in the context of the main and the buffering effect hypothesis, two primary models within general health literature to explain the association of social support and health outcomes. This chapter also provides a detailed examination of cognitive processes, behavioural responses, as well as neural and physiological activities derived from pain research both in experimental and clinical settings.

A Systematic Review of the Processes Underlying the Main and the Buffering Effect of Social Support on the Experience of Pain

Xianwei Che, MSc, Robin Cash, PhD, Sin Ki Ng, MSc, Paul Fitzgerald, PhD, and Bernadette M. Fitzgibbon, PhD

Objective: This review aimed to explore the processes that underlie the main and the buffering effect of social support on decreased pain experience.

Materials and Methods: The systematic review was conducted according to the PRISMA guidelines. Online databases of PubMed and PsycINFO were searched for peer-reviewed articles using keywords ("social support," OR "interpersonal," OR "social presence," OR "spouse," OR "couple," OR "marriage") AND "pain"). Articles were included if they examined the cognitive or behavioral processes linking social support to any aspects of reduced pain experience.

Results: The database search identified 38 studies, of which 33 were cognitive-behavioral studies and 5 were neurobiological. Cognitive-behavioral studies generated a total of 57 findings of the analgesic influence of social support. This effect was further categorized as social support decreasing the adverse influence of pain-related stress (28/44 findings), reappraising pain-related stress (7/9 findings), and facilitating coping attempts (2/4 findings). Of the 5 neurobiological studies, the influence of social support on pain reduction was associated with reduced neural and physiological stress systems in response to painful stimuli.

Discussion: This review presents evidence that the stress-buffering effect is more often able to account for the relationship between social support and pain experience. Moreover, findings suggest the critical significance of stress appraisal and attenuated stress systems in linking social support to aspects of reduced pain experience. Findings implicate the role of integrating perceived support and intimacy in support-oriented interventional trials for chronic pain.

Key Words: social support, pain, stress, main effect, buffering effect

(Clin J Pain 2018;34:1061-1076)

- X.C. is supported by the China Scholarship Council (201506990016), China. P.F. is supported by a NHMRC Practitioner Fellowship (606907), Australia. B.M.F. is supported by a NHMRC Early Career Fellowship, Australia. The authors declare no conflict of interest.
- Reprints: Xianwei Che, MSc, Monash Alfred Psychiatry Research Centre (MAPrc), Monash University, Level 4, 607 St Kilda Road, Melbourne, Vic. 3004, Australia (e-mail: xianwei.che@monash. edu).
- Supplemental Digital Content is available for this article. Direct URL citations appear in the printed text and are provided in the HTML and PDF versions of this article on the journal's website, www. clinicalpain.com.

P ain is by definition an unpleasant sensory and emotional experience that is associated with actual or potential tissue damage.¹ Pain is therefore not defined by a noxious stimulus alone but is shaped by a variety of factors including cognition such as beliefs and attention, as well as affective factors such as mood.^{2–5} In a recent proposed update, the definition of pain was suggested to further include a social component.⁶ This was done to reflect the influence of one's social environment on pain, such as its effect over the cognitive and affective components of pain.^{7,8} The proposed inclusion of a social dimension in part reflects the growing number of studies looking into the social modulation of pain, of which social support is one key factor.^{9–12}

Social support refers to the perception or experience that one is loved and cared for by others.¹³ Social support thus encapsulates both "perceived" and "received" social support. Perceived social support refers to the perception that supportive resources would be available should they be needed, whereas received social support means the reported exchange of supportive resources.^{14–16} A number of studies have reported that social support is associated with decreased pain experience in clinical settings.^{17–20} In experimental settings, the overwhelming body of research has shown that social support is associated with less pain and greater pain tolerance, regardless of whether the support comes from interacting with support providers, ^{9,12,21–24} the presence of others, ^{5,25–27} or simply reminding individuals experiencing pain of their social connections. ^{12,28–30} However, there is also evidence suggesting that social support can increase^{31,32} or provide no influence on pain, ^{10,33} suggesting that further clarity about how social support impacts on pain and under what circumstances is needed.

Recently, increasing evidence suggests the effectiveness of support-assisted pain management therapies (eg, spouse-assisted coping skills training).^{34,35} However, optimizing these approaches is highly limited by our knowledge of the processes translating social support to pain reduction. To date, several behavioral models have been proposed to explain the association of social support and pain experience, especially in the context of chronic pain. The operant model posits that positive responses of social partners (eg, reassurance, positive concerns) can serve as reinforcement of pain behaviors.³⁶ The communal coping model of pain catastrophizing claims that pain catastrophizers tend to engage in more pain behaviors to attract social support, which in turn may serve to trigger or maintain exaggerated pain expression.37 Another perspective, the intimacy model, argues that communication of pain may serve to enhance intimacy between couples.³⁸ According to the first 2 models, however, social support as a type of positive response would increase pain expression. Meanwhile, although the intimacy

Received for publication April 27, 2017; revised April 2, 2018; accepted April 18, 2018.

From the Monash Alfred Psychiatry Research Centre (MAPrc), The Alfred and Central Clinical School, Monash University, Melbourne, Australia.

Copyright © 2018 Wolters Kluwer Health, Inc. All rights reserved. DOI: 10.1097/AJP.00000000000624

model suggests decreased pain in the context of social support, it is more limited in verbal support and romantic couples. Therefore, it remain unclear the cognitive and behavioral variables linking a general overview of social support to decreased pain experience. This is particularly pertinent for the design of support-assisted therapies for chronic pain.

To address this issue, we systematically reviewed evidence surrounding social support and decreased pain experience by drawing on certain theoretical accounts from the general health literature. In particular, we examined evidence for current theories that aim to delineate the cognitive and behavioral processes behind the analgesic influence of social support. Specifically, the "main-effect" hypothesis suggests that social support has an overall beneficial effect on pain experience, irrespective of any influences of stress. Meanwhile, the "buffering-effect" hypothesis suggests that social support may decrease pain experience by effectively reducing the influence of stress. This may involve reevaluating the threatening quality or perceived ability to cope with stress ("stress appraisal" process), or positively modulating an individual's coping attempts ("response regulation" process).³⁹ These 2 hypotheses are therefore primarily differentiated by whether or not social support modulates pain directly or indirectly via the influence of stress.

Although the main and the buffering effect may not be mutually exclusive in their effects on pain experience,⁴⁰ this literature review aims to examine the potential mechanisms that explain each of these models and summarize the strength of evidence in favor of each. To provide a more general review of diverse pain conditions, the review covers both experimental induced pain in healthy individuals (but see Montoya et al²⁶) and people with chronic pain. Moreover, as chronic pain includes more than pain ratings, broader pain experiences were reviewed which included not only pain ratings but other pain experiences, for example, functional disability, depressed mood.⁴¹ Exploring the relative merits of these proposals bears practical importance as each has direct implications for the design of supportassisted pain management therapy.³⁹

MATERIALS AND METHODS

This review was conducted according to the PRISMA guidelines for systematic reviews.^{42,43} The study selection process is depicted in Figure 1.

Definition of Variables and Effects

To date, there has no consensus on the operational definition of "social support." In this review, we defined social support through measures of "social integration" (the number of social roles a person holds, frequency of interaction with members of their social network, and the interconnectedness of relationships between network members), and through "social function" (informational, instrumental, and emotional support).^{44,45} Pain is by definition a sensory and emotional experience. Here a broad concept of "pain experience" was used, which was defined as a multidimensional phenomenon including physical (eg, intensity, duration, disability) and psychosocial suffering (eg, unpleasantness, negative mood, psychological, and social functioning). We chose this broader definition as pain intensity ratings alone do not capture the dimensionality of pain experience and, perhaps, the domains in which social support has an impact. "Stress" was defined as undesirable changes in daily life that require substantial behavioral readjustment.46,47 For individuals with pain, stress events included physical suffering, functional limitations, financial burden, social isolation,



FIGURE 1. Flow diagram of selected studies (n = number of articles).

and critical remarks/response from their significant others, as well as other negative life events, all of which have been shown to influence pain experience.^{48–50} The purpose was not to provide a comprehensive typology of stress, but rather to represent those functions for review. We also defined cognitive-behavioral processes linking support to pain as variables related to perception of pain (eg, perceived threat of pain) and behavioral response to pain (eg, health care utilization). In addition, we examined neurobiological studies that delineate the activations of central and/or autonomic nervous systems in the context of support and pain.

Study results on the analgesic influence of social support were sorted according to whether it supported the main-effect or buffering-effect hypothesis. Specifically, we categorized data to support the main-effect hypothesis where social support was associated with reduced pain experience in the absence of any statistical interaction/ moderation or mediation effect with stress, stress appraisal, or stress coping variable. In contrast, we categorized data to support the buffering-effect hypothesis where social support showed a protective effect on pain experience through the influence on stress. Specifically, where the data supported an interaction/moderation model whereby the support×stress interaction term is related to decreased pain experience. Alternatively, where the data supported a mediation model where social support is linked to decreased stress that in turn reduces pain experience.5,20

Literature Search

We conducted a systematic search of the online database of PubMed and PsycINFO for peer-reviewed articles. Title and abstract were searched with ("social support," OR "interpersonal," OR "social presence," OR "spouse," OR "couple," OR "marriage," AND "pain").^{51,52}

Eligibility Criteria

Eligibility was defined according to 5 a priori criteria: (1) studies were included if they assessed social support from human beings and excluded if they looked into support from other species or objects⁵³; (2) studies were included if they assessed the experience related to physical pain and excluded if they investigated other types of pain (eg, social-exclusion pain⁵⁴); (3) studies were included if they reported protective effect and excluded if they reported adverse or no influence of social support on pain experience^{55,56}; (4) studies were included if they investigated cognitive and behavioral variable or neurobiological mechanisms that might link social support to pain experience, and were excluded if they only reported the benefits of support without examining the underlying mediator and mechanism^{9,12}; (5) studies were included if they were published in English peer-reviewed journal from 1980 to 2016 and were in full-text. As per PRISMA guidelines, this time frame was chosen to include studies most likely to directly address the theories are investigating.

Quality Assessment

We used the assessment criteria developed by Campbell et al.⁵⁷ The assessment criteria has been used to review the influence of social support on chronic pain.^{57,58} The criteria was developed by combining the guidance on quality assessment with a number of review articles in the area of chronic pain.^{59–61} The combined criteria cover research objective, participant recruitment, data analysis, and so on. For cohort studies, the criteria also evaluate the attrition rate, and the follow-up time period (see the checklist at Supplementary Material S1, Supplemental Digital Content 1, http://links.lww.com/CJP/A510). Two reviewers (X.C., S.K.N.) separately assessed each study using these criteria to avoid potential bias. Any discrepancies were subsequently resolved at a consensus meeting.

Data Collection Procedure

Data extraction followed a priori developed data extraction forms based on modified PICOS-criteria.⁴² Specifically, extracted data included (1) number of participants and their clinical characteristics (sample); (2) type of social support; (3) cognitive and behavioral variables underlying the main versus the buffering effect (cognitive and behavioral mediators), or pain induction technique in neurobiological studies; (4) pain experience; and (5) findings.

Studies were classified as either cognitive-behavioral or neurobiological studies. With regard to cognitive-behavioral studies, this review examined whether social support was related to decreased pain experience through modulating the influence of stress, appraisal of stress and coping resources, or coping attempts. With respect to neurobiological studies, it reported on the neural and physiological mechanisms that are associated with pain reduction in the context of social support.

Effect Size

We also reviewed the magnitude of the analgesic influence of social support where possible. Effect size was calculated in studies using analysis of variance. In cases where path analysis (including mediation) was performed, the path coefficient was calculated from the independent variable to the mediator and dependent variable. Standardized or unstandardized coefficients were also extracted from studies using multiple regression analysis.

RESULTS

Study Selection

The electronic database search delivered 5400 results (Fig. 1). We removed 1199 duplicates and screened 4201

remaining studies. Of these, 3869 studies were discarded as they did not meet the eligibility criteria after reviewing the title and abstract. The remaining 332 studies were full-text assessed and 294 did not meet the eligibility criteria and were consequently discarded. This systematic review thus contains 38 studies (33 cognitive-behavioral studies and 5 neurobiological studies).

Cognitive-behavioral Studies Examining the Main Effect Versus the Buffering Effect

All of the included 33 cognitive-behavioral studies examined both the main and the buffering effect, in which they assessed pain as well as pain-related stress (Table 1). Some studies examined > 1 type of support, stress, or pain experience, and reported evidence in favor of both the main and buffering effect in a single study.⁴⁰ Consequently, the number of findings, rather than the number of studies, was tallied to assess evidence in favor of these 2 effects. In total, 57 findings were reported in 33 studies, of which 37 (65%) supported the buffering effect. Of those, 28 findings showed the benefits of support on pain experience by reducing the adverse influence of pain-related stress ("Stress" sections in Table 1); 7 findings showed this effect by decreasing the threatening quality of stress or increasing perceived ability to cope with stress ("Stress appraisal" section in Table 1); and 2 reported this effect by facilitating stress coping ("Active coping" section in Table 1). Moreover, 20 findings (35%) supported the main effect without any interaction with stress ("Effect" section in Table 1).

Moreover, we also considered the findings according to whether they were a "cross-sectional" or "cohort" study (Table 1). Among the buffering-effect findings, 29/37 (78%) were cross-sectional (ie, data were collected at a specific timepoint), whereas 8 of them (22%) were cohort findings (ie, data were collected at intervals through a period of time). Moreover, of the main-effect findings, 15/20 (75%) were cross-sectional and 5/20 (25%) were cohort.

Quality Assessment Analysis

Reviewer agreement on the quality assessment was 92.7%. The authors who developed this assessment tool used a score of 73 to separate studies of low quality from others within their investigations.⁵⁷ This was performed to assign approximately same number of studies in each quality categories. Similarly, we chose to use a quality score of 77, which was close to 73 and was able to approximately split the included studies equally. Moreover, next scores in our data set (70 or 80) would result in too less (ie, 3) or too many (ie, 14) studies in the low-quality group. Specifically, studies with a score below 77 were classified as "low quality" (n=11), a score between 77 and 84 as "medium quality" (n = 10), and a score above 84 as "high quality" (n = 12). Nearly all the included studies (94% or above) reported clear research objective, population parameters, and appropriate sample size. Most studies (73% or above) offered sufficient assessment of variables (eg, measures are validated or measures at least 2 dimensions), recruitment procedure, and strong statistics. However, only 67% of the studies reported inclusion/exclusion criteria and only 33% of studies provided evidence of the ratio of recruitment versus participation. In the cohort studies, only 29% provided the attrition rate, 57% reported an attrition rate smaller than 20%, and 71% had a follow-up period longer than 6 months.

Of the 37 findings supporting the buffering effect 26 (70%) showed medium to high quality. Among the main-effect

References	Sample	Type of Social Support	Cognitive/ Behavioral Variable	Pain Experience	Findings	Study Design (CS/CH)	Quality	Effect
Stress (physical su Brown et al ¹⁷	ffering) Patients with RA (N = 233)	Emotional support	Pain	Depression	 The negative correlation of emotional support and depression was strengthened when pain was included. Moderating effect of emotional support was not 	CS CH	Medium Medium	Buffering Main
Revenson et al ⁶²	Patients with RA (N = 101)	Received positive support	Pain severity	Depression	 Received positive support predicted reduced depression without interacting with pain 	CS	High	Main
Penninx et al ⁴⁰	Older people without chronic disease (N = 719); with mild arthritis (N = 612); with severe arthritis (N = 359)	Presence of a partner; close social relationships; diffuse social relationships; emotional support	Arthritis	Psychological functioning; depressive symptoms	 seventy Presence of a partner had direct, favorable effect on psychological functioning without interacting with arthritis. Close social relationships had direct, favorable effect on psychological functioning without interacting with arthritis. Having diffuse social relationships interacted with arthritis pain to predict depressive symptoms in severe arthritis Emotional support interacted with arthritis pain to predict depressive symptoms in severe arthritis 	CS CS CS CS	Low Low Low	Main Main Buffering Buffering
Feldman et al ⁶³	Patients with reflex sympathetic dystrophy syndrome (N = 109)	Perceived social support	Pain	Depressed mood	 Previous day' support interacted with previous day' pain to predict present day' depressed mood 	СН	Low	Buffering
Telfair and Gardner ⁶⁴	Adolescents with sickle cell disease (N = 79)	Group satisfaction	Pain	Psychological well-being (anxiety; depression)	1. Group satisfaction interacted with high pain to predict high psychological well-being	CS	Medium	Buffering
Riemsma et al ⁵⁰	Patients with RA (N = 197)	Positive support	Pain	Depression	 Positive support predicted reduced depression without interacting with pain 	CS	High	Main
Cano et al ¹⁸	Chronic musculoskeletal pain patients (N = 110)	Marital satisfaction	Pain	Depressive symptoms	 Marital satisfaction predicted reduced depressive symptoms without interaction with pain 	CS	Low	Main
Ferreira and Sherman ⁶⁵	Older adults with osteoarthritis $(N = 73)$	Perceived social support	Pain	Depressive symptoms	1. Perceived social support mediated the impact of pain on depressive symptoms	CS	High	Buffering
Holtzman and DeLongis ⁶⁶	Patients with RA (N = 69)	Morning satisfaction with spousal response	Morning pain severity	Evening pain catastrophizing	 Morning satisfaction with spousal response interacted with morning pain severity to predict evening pain catastrophizing 	CS	Low	Buffering
López- Martínez et al ²⁰	Chronic pain patient (N = 117)	Perceived social support	Pain intensity	Depressed mood; functional impairment; functional status	 Decreased pain intensity mediated the relationship between perceived social support and decreased depression. Decreased pain intensity mediated the relationship between perceived social support and decreased functional impairment. Decreased pain intensity mediated the relationship between perceived social support and increased functional status 	CS CS CS	High High High	Buffering Buffering Buffering

(Continued)

TABLE 1. (continued)

References	Sample	Type of Social Support	Cognitive/ Behavioral Variable	Pain Experience	Findings	Study Design (CS/CH)	Quality	Effect
Morgan et al ⁶⁷	Cancer patients with pain	Partner support	Pain	Quality of life	1. Partner support mediated the negative impact of pain	CS	Medium	Buffering
Sturgeon et al ⁶⁸	(N = 177 dyads) Chronic pain patients (N = 675)	Satisfaction with social roles	Pain intensity	Depression and anger	 on quarty of life Satisfaction with social roles mediated the negative impact of pain intensity on depression. Satisfaction with social roles mediated the negative impact of pain intensity on anger 	CS CS	Medium Medium	Buffering Buffering
Lee et al ¹⁹	Chronic arthritis pain patients (N = 299)	Tangible social support	Arthritis pain	Depressive symptoms	1. Tangible social support interacted with arthritis pain to predict less	СН	Low	Buffering
Park et al ⁶⁹	Older adults	Social support	Pain intensity	Depressive symptoms	1. Social support interacted with pain intensity to predict depressive symptoms	CS	_	Buffering
Stress (functional Affleck et al ⁷⁰	disability) Patients with RA (N = 129)	Satisfaction with support	Functional disability	Psychosocial adjustment to illness	1. The positive relation between support satisfaction and psychosocial adjustment was increased when disability was included	CS	Medium	Buffering
Fitzpatrick et al ⁷¹	Patients with RA (N = 158)	Social relationship	Disability	Psychological well-being (depression; self-esteem)	 Social relationship had direct, favorable effect on psychological well-being without interacting with disability 	CS	Low	Main
Goldberg et al ⁷²	Male chronic pain patients (N = 105)	Spousal support	Pain-related interference	Depression	1. Spousal support interacted with pain-related interference to predict depression	CS	High	Buffering
Riemsma et al ⁵⁰	Patients with RA (N = 197)	Positive support	Physical function	Depression	1. Positive support predicted reduced depression without interacting with physical function	CS	High	Main
Cano et al ¹⁸	Chronic musculoskeletal pain patients (N = 110)	Marital satisfaction	Physical disability	Depressive symptoms	1. Marital satisfaction predicted reduced depressive symptoms without interacting with physical disability	CS	Low	Main
Strating et al ⁷³	Patients with RA (N = 129)	Emotional support satisfaction; satisfaction with social companionship	Impairment- disability	Distress	 Emotional support satisfaction predicted reduced distress without interaction with impairment-disability in short-term RA. Satisfaction with social companionship predicted reduced distress by interacting with impairment-disability in short-term RA. This interaction disappeared in long-term RA 	CH CH CH	Medium Medium Medium	Main Buffering Main
Kerns et al ⁷⁴	Chronic pain patients (N = 106)	Marital satisfaction	Spousal punishing response	Depressive symptoms	1. Marital satisfaction interacted with higher spousal punishing response to predict lower depressive sumptoms	CS	Low	Buffering
Revenson et al ⁶²	Patients with RA (N = 101)	Received positive support	Stressful response	Depression	 Received positive support interacted with high stressful response to predict reduced dopression 	CS	High	Buffering
Cano et al ⁷⁵	Chronic pain patients (N = 165)	Marital satisfaction	Negative spousal response	Depressive symptom	 Marital satisfaction mediated the positive relationship between negative spousal response and depressive symptom 	CS	Low	Buffering

(Continued)

Copyright © 2018 Wolters Kluwer Health, Inc. All rights reserved.

References	Sample	Type of Social Support	Cognitive/ Behavioral Variable	Pain Experience		Findings	Study Design (CS/CH)	Quality	Effect
Riemsma	Patients with R A	Positive support	Stressful	Depression	1	Positive support interacted	(CS)	High	Buffering
et al ⁵⁰	(N = 197)	i ositive support	response	Depression	1.	with stressful response to predict depression	es	mgn	Duntering
Cano et al ¹⁸	Chronic musculoskeletal pain patients (N = 110)	Marital satisfaction	Negative spousal response	Depressive symptoms	1.	Marital satisfaction predicted reduced depressive symptoms without interaction with negative spousal response	CS	Low	Main
Raichle et al ⁷⁶	Chronic pain patients (N = 94)	Marital satisfaction	Negative spousal response	Depression	1.	Marital satisfaction predicted reduced depression directly, without interacting with spousal negative response	CS	High	Main
Rosen et al ⁷⁷	Patients with provoked vestibulodynia (N = 175 couples)	Dyadic adjustment	Spousal negative response	Sexual satisfaction	1.	Dyadic adjustment mediated the adverse impact of negative spousal response on sexual satisfaction	CS	High	Buffering
Stress (other nega	tive life events) Married couples	Emotional	Negative life	Symptoms	1	Emotional support was	СН	High	Buffering
et al ⁷⁸	(N = 75)	support	events	(headache;	1.	negatively related to	CH	High	Buffering
				backache; flu; sore throat); mood	2.	hassle-next day symptoms association. Emotional support was negatively related to hassle-same day mood association.	СН СН	High High	Main Main
					3.	Emotional support was not related to hassle-same day symptoms association. Emotional support was not related to hassle-next			
Weinberger et al ⁷⁹	Patients with osteoarthritis (N = 439)	Social support	Negative life events	Functional status (psychological disability; physical disability; pain)	1.	day mood association Social support had direct, favorable effect on functional status, without interaction with daily hassle	CS	High	Main
Affleck et al ⁸⁰	Patients with RA (N = 74)	Social support	Negative life events	Mood disturbance	1.	Social support interacted with daily stressor to predict next day mood disturbance	СН	Medium	Buffering
Alonso and Coe ⁸¹	Healthy women (N = 184)	Access to support providers	Distress (anxiety; depression)	Menstrual pain	1.	Access to support providers interacted with distress to predict	CS	Low	Buffering
Pekkarinen et al ⁸²	Female geriatric nurses (N = 975)	Social support	Physical workload	Musculoskeletal symptoms	1.	Social support interacted with physical workload to predict musculoskeletal symptoms	CS	Medium	Buffering
Stress appraisal (1 Waltz et al ⁸³	perceived threat/copin Clinical pain patients (N = 234)	ng ability) Emotional support; social interaction	Psychological functioning (reduced RA helplessness; sense of mastery; self-esteem, etc.)	Pain	1.	Psychological functioning mediated the protective influence of emotional support on pain Psychological functioning mediated the protective influence of social interaction on	CH CH	High High	Buffering Buffering
Kerns et al ⁸⁴	Chronic pain patients (N = 234)	Pain-related support	Low self- appraised problem- solving competence	Depressive symptoms	1.	Pain-related support interacted with low problem-solving competence to predict depressive symptoms	CS	High	Buffering

TABLE 1. (continued)

(Continued)

TABLE 1. (continued)

References	Sample	Type of Social Support	Cognitive/ Behavioral Variable	Pain Experience	Findings	Study Design (CS/CH)	Quality	Effect
Holtzman and DeLongis ⁶⁶	Patients with RA (N=69)	Morning satisfaction with spousal response	Morning pain catastro- phizing	Evening negative affect; evening pain intensity	 Morning satisfaction with spousal response interacted with morning pain catastrophizing to predict evening negative effect. Morning satisfaction with spousal response decreased evening pain intensity without interaction with morning pain catastrophizing 	CS CS	Low Low	Buffering Main
Vlaeyen et al ⁵	Healthy participants (N = 149)	Social presence	Perceived threat of cold pain	Pain intensity; facial expression of pain	 Perceived threat mediated the inhibitory effect of social presence on pain intensity. Perceived threat mediated the inhibitory effect of social presence on facial expression of pain 	CS CS	Medium Medium	Buffering Buffering
Corley et al ⁸⁵	Healthy couples (N = 134 dyads)	Global relationship satisfaction; situational relationship satisfaction after 2-minute interaction	Threat manipula- tion of cold pain	Pain intensity	 Global relationship satisfaction predicted less pain without interacting with threat manipulation. Situational relationship satisfaction predicted less pain in the low threat condition 	CS CS	Low Low	Main Buffering
Active coping Manne and Zautra ⁴⁹	Women with RA (N = 103)	Spousal support	Coping with RA (information seeking; cognitive	Psychological adjustment	1. Adaptive coping mediated the positive relationship between spousal support and better psychological adjustment	CS	High	Buffering
Holtzman et al ¹¹	Patients with RA (N = 73)	Morning satisfaction with social support	Morning coping with RA (stoic distancing)	Evening pain severity	 Morning satisfaction with social support interacted with increased stoic distancing to predict reduced evening pain severity. 	CS	Low	Buffering
López- Martínez et al ²⁰	Chronic pain patient (N = 117)	Perceived social support	Active coping with pain	Depressed mood; pain intensity	 Perceived social support was related to reduced depressed mood without interacting with active coping response. Perceived social support was related to reduced pain intensity without interacting with active coping response 	CS CS	High High	Main Main

findings, 12/20 (60%) showed medium to high quality. Specifically, of the cross-sectional findings which reported the buffering effect (29/37), 20/29 (69%) were identified as medium to high quality. However, only 8 findings were from cohort studies which showed the buffering effect, and 6 of them (75%) showed medium to high quality. In the 20 findings showing the main effect, 15 of them (75%) were cross-sectional. But more than half (8/15, 53%) of them showed low quality. Only 5 findings were cohort studies, and all of them identified as medium to high quality. Full descriptions of study quality are detailed in the following sections. Details of quality assessment can be found at Supplementary Material S1 (Supplemental Digital Content 1, http://links.lww.com/CJP/A510).

The cognitive-behavioral studies focused on different aspects of stress that social support has beneficial effects on, that is, the overall adverse influence of stress, or a specific aspect (and potentially those that inform the former) including stress appraisal, and stress coping. Accordingly, studies were organized based on these stress categories.

Whether Social Support Buffers Stress to Reduce Pain

People with chronic pain experience a variety of stress events that impact on their quality of life in different ways⁵⁰ ("Stress" sections in Table 1).

First, physical suffering means the severity and frequency of painful episodes as well as related unpleasantness.⁸⁶ Physical suffering causes stress for individuals with chronic pain and is associated with increased functional impairment and depression.^{20,40} In this review, we identified 21 findings which examined, among patients with chronic pain, whether social support was able to decrease the adverse impact of physical suffering on other aspects of pain experience. Among them 8/21 (38%) showed that social support interacted with physical suffering to predict decreased depression,17,19,40,63,69 and emotional well-being.64,66 Another 4 findings (19%) showed a mediation role of social support, whereby the adverse influence of physical suffering was reduced on depression,^{65,68} and quality of life.⁶⁷ Moreover, 3 findings (14%) showed in mediation models that social support resulted in decreased physical suffering, which was in turn associated with reduced functional impairment and depression.²⁰ Further, among the 15 findings (71%) of buffering effect 10 (67%) showed medium to high quality. However, 6 findings (29%) showed a main effect where social support was linked to reduced depression^{18,50,62} and greater psychological functioning (mastery, self-efficacy, and self-esteem),40 without the interaction with pain severity. However, 3 of these (50%) reported low quality with inadequate description of inclusion/exclusion criteria, small sample size, insufficient assessment of variables, or no participation rate.

Functional disability is defined as the difficulty in performing an activity, for example, household activity and social activity.⁸⁷ It is a significant source of stress for people with chronic pain that is associated with greater emotional distress (anxiety and depression).⁵⁰ Eight findings were identified, which examined among patients with chronic pain whether or not social support can decrease the influence of functional disability on other pain experience. Three findings (38%) showed that social support interacted with functional disability to predict decreased emotional distress.^{70,72,73} All above findings (62%) reported the main effect, in which social support was negatively related to emotional distress, without interacting with functional disability.^{18,50,71,73} The main effect of support was often observed in long-term emotional distress (≥ 13 y),⁷³ and most of them (3/5) showed good quality.

Moreover, close others of patients with chronic pain sometimes respond to the displays of pain and suffering in negative ways, for example, expressing irritation or anger.87 Although ignoring pain expression is sometimes associated with better pain-related outcomes, this is typically perceived as a negative response that may make the individuals with pain feel more depressed.^{74,75} In total, we identified 7 findings which examined whether social support can buffer the influence of close other's negative response on pain experience in people with chronic pain. Five of these (71%)showed the buffering effect, in which social support decreased the adverse influence of negative responses on depression, or on sexual satisfaction in another study, either with an interaction, ^{50,62,74} or a mediation effect. ^{75,77} Moreover, 3 of these 5 findings (60%) showed good quality. However, another 2 findings (29%) showed the main effect in which social support was predictive of decreased depression without interacting with negative spousal response.^{18,76}

In addition, some studies examined whether social support can buffer the influence of other negative life events. Negative life events are undesirable life activities to which individuals need to substantially adjust their behavior.⁴⁷ These have been reported to aggravate physical symptoms and mood disturbance in both chronic pain,⁸⁰ and healthy populations.⁴⁸ Negative life events may take the form of specific distress, for example, workload⁸²; or cover multiple life domains.⁷⁸ Findings are inconsistent when negative life events are nonspecific. Three findings (3/8, 38%) showed that social support was able to interact with negative life events to predict decreased physical suffering or mood disturbance.^{78,80} But another 3 findings (3/8, 38%) only observed the main effect of support without significant interaction with negative life events.^{78,79} Moreover, the 6 findings were reported both in people with chronic pain,^{79,80} and in healthy populations who sometimes experienced symptoms like headache, backache, and shoulder pain.⁷⁸ All these 6 findings were showed to be of good quality. However, the buffering effect was consistently reported when specific negative life events were measured. Two findings (2/8, 24%) showed that social support interacted with emotional distress (anxiety and depression) or physical workload to predict less pain.^{81,82} Moreover, these findings included healthy participants experiencing menstrual pain,⁸¹ or some musculoskeletal symptoms.⁸² But only one of them showed good quality.⁸² The other study showed inadequate description of inclusion/exclusion criteria, inadequate report on demographics and the strength of effect, and no participation rate.⁸¹

Whether Social Support Changes Stress Appraisal to Reduce Pain

Here we identified 7 findings (78%) which showed that social support could help individuals with pain perceive the pain or related stress as less threatening or become more confident to cope with ("Stress appraisal" section in Table 1). Moreover, 5 findings were shown to be of good quality. Two findings have shown an interaction effect between social support and a stress appraisal variable in the prediction of decreased emotional well-being. Specifically, in pain patients who reported low social support, perceived incompetence to solve problem or high pain catastrophizing (helplessness, rumination, magnification) was associated with more depressive symptoms and higher negative effect. But these associations were decreased significantly in pain patients with high level of social support.^{66,84}

Another 2 findings showed with mediation models that emotional support and social interaction could decrease pain severity through psychological functioning in people with chronic pain. Higher emotional support or social interaction was associated with higher psychological functioning (self-esteem, sense of mastery), which in turn linked with lower pain severity.⁸³ In addition, an experimental study induced cold pain in which social presence (a female observer) helped healthy participants perceive cold pain as less threatening, which then resulted to lower pain intensity and less facial expression of pain despite the female observer was a stranger and had minimum verbal exchange with the participants.⁵ A recent experimental study also found that situational relationship satisfaction after 2-minute couple interaction was associated with lower induced cold pain when pain was perceived as threatening. But global relationship satisfaction was related to less pain irrespective of the level of threat.85 Another study also found the main effect of support in patient group, that morning satisfaction with spouse was related to less evening pain irrespective of morning pain catastrophizing.66

Whether Social Support Promotes Coping Strategies to Reduce Pain

Four findings were on the topic of whether social support modulates pain experience through active coping response, and all of them were conducted in patients with chronic pain ("Active coping" section in Table 1). In one finding, on mornings when patients were satisfied with social support, increased use of distancing (diverting attention away from pain) was associated with decreased evening pain severity.¹¹ This finding was categorized as low quality as

there was inadequate description of recruitment procedure and strength of effect (effect size). However, another finding showed quality results with mediation models whereby spousal support resulted to better psychological adjustment to arthritis pain through active coping responses, namely information seeking and cognitive reappraisal.⁴⁹ These 2 coping strategies describe the effort to search for advice about the illness and make it less distressing.⁸⁸

In contrast, a more recent finding showed with good quality that social support does not facilitate active coping responses to reduce pain intensity. In this study, perceived social support was related to decreased pain intensity and depressed mood without the interaction with active pain coping.²⁰ This study detailed multiple active coping responses, for example, engaging in physical therapy, clearing mind of bothersome thoughts, participating in leisure activities, distracting attention from the pain, etc.⁸⁹ Moreover, these sets of coping response could predict decreased depressed mood but not pain intensity.²⁰

Effect Size

Only a small proportion of studies provided available data to calculate effect size, whereas other studies provided the coefficient of social support variables (Supplementary Material S2, Supplemental Digital Content 1, http://links.lww.com/CJP/A510). Across the statistical models, social support had a small impact on pain experience in both the main^{20,71}) and the buffering model.^{68,75,82}

Neurobiological Studies Examining the Influence of Social Support on Pain

Four neurobiological studies have induced pain in healthy participants,^{24,28,30,90} and in 1 study in people with chronic pain.²⁶ Four studies have investigated the neural mechanisms of social support related to pain experience (Table 2). In 1 study, participants with a diagnosis of fibro-myalgia reported lower pain sensitivity and pain ratings to tactile stimulation when they were accompanied by their significant other compared with when they were alone. This study further found that primary somatosensory cortex activity in response to painful elbow stimulation (a tender point), measured using magnetoencephalography, was lower when the significant other was present.²⁶

In another study, married females underwent functional magnetic resonance imaging during the threat of electrical stimuli in 3 conditions: holding their husband's hand, holding a male stranger's hand and no hand-holding. Results showed that spousal and stranger hand-holding produced lower bodily arousal and brain activation in threat-related regions (ventral anterior cingulate cortex; supramarginal gyrus) relative to no hand-holding. Moreover, marital quality affected brain activation to electric stimuli during spousal hand-holding whereby higher marital quality predicted lower threat-related neural activation in anterior insula (AI), and hypothalamus.⁹⁰

In another study, female participants in a long-term romantic relationship were subject to thermal pain during functional magnetic resonance imaging scanning whereas they were shown pictures of their partner or control images (stranger, object).²⁸ Viewing partner pictures relative to control pictures produced less pain ratings, lower activation of pain-related regions (AI, dorsal anterior cingulate cortex [dACC]) but higher activation in safety-related region (ventromedial prefrontal cortex [VMPFC]). Further, greater activation of VMPFC in response to partner pictures was related to lower pain ratings as well as pain-related brain activation. Interestingly, greater VMPFC response to partner pictures covaried with longer length of relationship and greater perceived partner support.

In addition, reward-related neural processing is reported to underlie pain reduction.³⁰ In this study, viewing pictures of romantic partner relative to acquaintance produced lower thermal pain ratings in healthy participants. Imaging data further showed decreased pain-related activation (insula, thalamus), but increased activation of reward regions (OFC) in the viewing of romantic partner relative to acquaintance's picture. Moreover, pain relief during viewing pictures of romantic partner was also associated with increased reward activation (caudate nucleus, nucleus accumbens, OFC), but with decreased pain-related activation (dACC, AI). This pattern of neural response was not observed in a distraction task which also produced pain relief.

Social support was also found to reduce neuroendocrine and/or autonomic response to pain in 2 studies. Roberts et al²⁴ examined healthy participants' experience to cold pain in 1 of the 3 conditions: verbal social support, neutral nonsupport, alone. Compared with participants in neutral nonsupport and alone conditions, participants in the verbal support condition reported less pain, less task difficulty and tension, as well as lower blood pressure, heart rate, and cortisol response. Further analysis showed that perceived social support during cold pain was negatively associated with pain intensity, unpleasantness, and cortisol response.

DISCUSSION

Inspired by the evidence of support-assisted pain management therapies, the current review specifically explored the processes underlying the main and the buffering effect of social support on pain experience. Our review presented evidence supporting both the main and the buffering effect. However, the buffering effect is more often able to explain findings in studies that were deemed to be of higher quality. In this context, our results suggest that social support is associated with decreased pain experience through managing stress, namely, physical suffering, functional disability, stressful response of close other, and other negative life events. Findings further show potential processes associated with the buffering effect, that is, stress appraisal and coping. In this section, we first discuss why the buffering effect can more often describe the influence of support on pain reduction in comparison to the main effect. Then, cognitive, behavioral, and neurobiological processes are detailed that may underlie this buffering effect. Finally, we present a social-buffering model of pain reduction that binds together evidence across behavioral, neuroimaging, and physiological research fields.

The Buffering Effect Describes More Often the Influence of Social Support on Decreased Pain

The majority of our findings (37/57, 65%) show that the interaction of support and stress is likely associated with aspects of reduced pain experience.^{19,78} This finding is more often observed in cross-sectional studies (29/37, 78%) with good quality (20/29, 69%).^{69,82} There is also evidence (20/57, 35%) showing the main effect of social support.^{50,79} Although most of these findings (15/20, 75%) are cross-sectional, they were found to be of low quality.^{40,71} Our findings therefore suggest that in most of the cases social support may help patients cope with negative life events and manage pain experience.

	5	Type of Social	Pain Induction		
References	Sample	Support	Technique	Pain Experience	Findings
Montoya et al ²⁶	Fibromyalgia patients (N = 18); migraine patients (N = 18)	Presence of significant others; patient alone	Tactile stimulation	Pain sensitivity; pain rating; brain activity measured by magnetoencepha- lography	 Pain sensitivity and pain ratings were reduced in the presence of significant others compared with alone situation in fibromyalgia patients. Primary somatosensory cortex activity elicited by elbow stimulation was also reduced in fibromyalgia patients in the presence of their significant others acompared with alone acondition
Coan et al ⁹⁰	Married women (N=16)	Holding husband's hand; holding stranger's hand; no hand-holding	Electric stimuli	Rating of unpleasantness and body arousal; brain activation measured by fMRI	 Spousal hand-holding revealed reduced unpleasantness than stranger and no hand-holding. Spousal and stranger hand- holding revealed reduced body arousal than no hand-holding. Spousal and stranger hand- holding attenuated brain activation in threat-related regions relative to no hand- holding (vACC, PCC, supramarginal gyrus). Higher marital quality predicted less threat-related neural activation in spousal hand- holding (AL, hypothalamus)
Younger et al ³⁰	Healthy people in new romantic relationship (N=15)	Viewing picture of romantic partner; viewing picture of familiar acquaintance; a distracting task	Thermal pain	Pain rating; brain activation measured by fMRI	 Viewing pictures of romantic partner relative to acquaintance reduced self-reported pain. Viewing pictures of romantic partner relative to acquaintance increased reward neural activation (OFC); decreased pain- related activation (insula, thalamus). Pain relief was associated with increased reward neural activation (caudate head, nucleus accumbens, OFC); but with decreased pain- related activation (AI) in viewing
Eisenberger et al ²⁸	Female participants in long-term romantic relationship (N=21)	Picture of romantic partner; picture of stranger; picture of object	Heat stimulations	Pain rating; brain activation measured by fMRI	 Pictures of romanuc partner Viewing partner pictures led to reduced pain-related brain activation (dACC, AI) and increased safety-related brain activation (VMPFC). Greater VMPFC response to partner pictures was related to decreased pain ratings and attenuated pain-related brain activation (dACC, AI). Greater VMPFC response to partner pictures covaried with longer relationship lengths and greater perceived partner support
Roberts et al ²⁴	Healthy female undergraduates (N=76)	Verbal social support; neutral nonsupport; alone	Cold pressor task	Pain rating; physiological response (blood pressure, heart rate, cortisol); tension and task difficulty	 greater perceived partner support Social support condition attenuated blood pressure, heart rate, and cortisol reactivity, as well as reduced pain ratings, task difficulty, and tension compared with neutral nonsupport and alone conditions

AI indicates anterior insula; dACC, dorsal anterior cingulate cortex; fMRI, functional magnetic resonance imaging; OFC, orbitofrontal cortex; PCC, posterior cingulate cortex; vACC, ventral anterior cingulate cortex; VMPFC, ventromedial prefrontal cortex.

These results may be further explained by the fact that the buffering-effect hypothesis emphasizes more cognitive and behavioral changes that are critical in pain reduction. Compared with the main effect, which only suggests the sense of stability or predictability related to support, the buffering effect conceptualizes broader cognitive and behavioral processes including appraisal of life situations and coping resources, inhibition of maladaptive and/or facilitation of adaptive responses to life situations.^{39,47} The buffering-effect model thus recognizes that a broader range of cognitive and behavioral changes facilitated by supportive resources may help individuals with pain better cope with life situations and therefore manage pain experience. For instance, studies have shown that social support can help people with chronic pain redefine the situation they are in, distract attention from this illness; decrease thinking one cannot do anything to cope with pain, and reduce restrictions on social activities.11,20,49 Therefore, the buffering-effect hypothesis might be a more flexible theoretical account.

However, one should be careful to prefer the buffering over the main-effect perspective in the explanation of the benefits of support on pain experience. In this systematic review, we did not provide effect size for each study due to the diversity of study designs and statistics provided, that is, correlation, group contrast, regression, mediation model. Available data suggest that the analgesic influence of social support is small in supporting of both of the 2 models. Although relatively less supported by the literature, the main effect is still implicated in a number of studies. Moreover, some studies even showed both the main and the buffering effect in a single study.^{40,78} It suggests that the 2 theoretical accounts may be not mutually exclusive. Indeed the main and the buffering effect may depend on the type of measure of support.³⁹ A structural measure of support (eg, the number of close relationships) may influence health outcomes through the main effect, whereas a functional measure (eg, emotional support) may exert its impact mainly through the buffering effect on stress.⁴⁰ We therefore suggest, in the context of pain experience, that the buffering effect is more often able to account for the impact of social support on pain experience than the main effect.

We have also noted some factors that may bias the evidence in favor or against these 2 theoretical accounts. The accurate assessment of buffering effect requires that the choice of measures of stress is appropriate. As reviewed earlier, a wide variety of stressors (> 50 items) ranging from family, work, to outdoor activity, and economic situation have been described,^{78,79} for many of which a single type of support is not helpful. A direct, protective effect of social support on pain experience was thus observed. Moreover, the conceptualization and measure of social support are also critical. There is evidence suggesting that situational received support is able to buffer the influence of painrelated threat on pain intensity whereby global perceived social support has general analgesic effects irrespective of the level of threat.85 However, other studies have shown that the effects of situational received support are inconsistent,¹⁵ but global perceived social support is a more consistent predictor of health outcome.^{15,16,39} It therefore remains an open question whether a particular type of social support is related to decreased pain experience by acting upon one process rather than another.

In summary, the buffering effect is a more flexible theoretical account in the context of pain research, and thus may more often explain the influence of social support on pain experience. However, this does not mean that the main effect is inappropriate as of the lack of effect size. We have also discussed the potential factors that may result in the preference of one perspective over another in support and pain research. In the following section, we discuss which cognitive or behavioral mediators might underlie the buffering effect of support on pain experience.

The Buffering Effect on Pain Experience is Associated With Stress Appraisal and Coping

Our review also examines aspects through which the buffering effect may account for the association between social support and pain experience. When supportive resources are available, the reduction of pain experience is associated with decreased perceived threat of pain,^{5,85} decreased pain helplessness,⁸³ reduced pain catastrophizing, and perceived incompetence of problem solving,^{66,84} as well as by increased perceived mastery and self-esteem.⁸³ These findings were found to be of medium to high quality. They support the stress appraisal process as being one of the mediators of the buffering effect.³⁹

Social support may help appraise a potential stressor in more benign ways.⁴⁵ Literature surrounding support and threat has shown that social support helps individuals "calm down" when experiencing a potential stressor.91 For example, individuals demonstrate lower body arousal and perceived tension in the experience of painful stimuli when they hold their partner's hand or receive verbal support from confederate.^{24,90} Moreover, social support is suggested to supress heart rate increase and threat-related theta oscillation to upcoming painful stimulation, which are in turn associated with pain reduction.92 The findings therefore suggest that social support assists to perceiving pain and related stress as less threatening which in turn reduces pain. This argument is further corroborated by findings that social support fostered reinterpretation of uncomfortable feelings as neutral or comfortable sensations and produced higher tolerance to cold pain.^{22,23}

In addition, social support may enhance the perceived ability to cope with imposed demands. In one study, verbal social support from confederate reduced task difficulty in the experience of cold pain.²⁴ Another study showed that emotional support is associated with increased self-efficacy in the adjustment to diabetes.⁹³ In this review, findings show that social support is positively associated with perceived mastery and self-esteem in the experience of chronic pain. These psychological mediators further act to decrease pain severity.⁸³ Moreover, findings show that social support is able to buffer the adverse influence of perceived incompetence and helplessness on pain experience.^{66,84} These findings indicate that supportive resources contribute to the reduction of pain experience through bolstering perceived ability to cope with pain and related stress.

We also discuss evidence of the response regulation process of the buffering hypothesis. This line of research focuses on behavioral changes (action) to cope with pain, whereas stress appraisal specifically entails cognitive changes (perception) associated with social support. Specifically, 2 studies indicated that social support is associated with lower pain intensity and higher psychological adjustment by increased use of information seeking regarding treatments , cognitive reappraisal (ie, thinking of pain in more benign ways), and stoic distancing (ie, distracting attention away from pain).^{11,49} Other studies similarly showed that social support-assisted coping attempts (reinterpretation, distraction) and increased pain tolerance.^{22,23}

These studies suggest that pain experience was reduced via social support and its influence on coping responses. However, a more recent study indicated that the level of social support is associated with decreased pain intensity and depressed mood, independent of any interaction with coping response.²⁰ This discrepancy might be due to the manner in which coping responses were categorized. Although López-Martínez et al²⁰ used similar coping response criteria as former studies (eg, distracting attention from pain), they included broader categories of coping response (eg, engaging in physical therapy and leisure activities) and treated them as a single variable in the examination of buffering effect. It is likely that social support is not related to each of these coping strategies during the specific period of testing. Overall these findings provide evidence that stress coping may link social support to decreased pain experience, at least when supportive resources are able to prime a particular coping attempt.

Neurobiological Mechanisms Mediating the Process of Stress Appraisal

There is some preliminary neuroimaging evidence of mechanisms that may underlie the benefits of support on pain relief. In these studies, social support was manipulated in different ways: the presence of significant other, holding hand of significant other, and viewing pictures of significant other. In general, social support was associated with the reduction of pain rating, along with decreased threat-related and pain-related brain activation, for example, insula, anterior cingulate cortex (ACC), and hypothalamus^{26,28,30,90} (Table 2).

These studies indicate that the influence of social support on pain is mainly mediated by decreased neural response to threat cues. The neurological response elicited by painful stimuli is thought to embody not only a direct sensory pain response, but in addition a threat response that is registered across other cortical and subcortical regions. For example, the activation of ACC and insula is also known to occur in response to impending threat.⁹⁴ Moreover, it has been suggested that activation of the hypothalamus is relate to recruitment of the hypothalamus-pituitary-adrenal axis, which responds to a stressor and releases hormone cortisol for energy mobilization.⁹⁵ Given that activation of these areas is reduced by social support, social support may help to decrease the threatening nature of painful stimuli. This idea is further corroborated by the finding that social support is associated with less bodily arousal to painful stimuli, and that higher marital satisfaction predicts lower activation of insula and hypothalamus in response to painful stimuli.90

Similarly, social support may elicit neural responses related to a feeling of "safety" during the experience of pain. Eisenberger et al²⁸ report that viewing an image of one's partner while experiencing pain produces greater activity in the VMPFC, a brain node associated with tracking the safety value of a stimulus.^{96–98} Greater activity in VMPFC in response to partner's picture is associated with decreased activation of ACC and insula, as well as with longer relationship length and greater perceived partner support. Another study showed that self-reported social support is associated with increased gray matter volume in the posterior cingulate cortex,⁹⁹ a brain region that responds to safety cues along with VMPFC.⁹⁴ These results suggest that social support may signal safety and thereby reduce threat-related neural and autonomic responses.^{100–102} This is in line with

an earlier behavioral observation, which showed that social presence decreased pain by suppressing perceived threat of pain.⁵ Moreover, a recent study showed that images of a significant other (individuals from whom the participants perceive most support on a daily basis) represent prepared safety cues to inhibit fear-learning process.¹⁰³ This finding provides more evidence that social support could signal safety to suppress the threatening nature of pain.

The benefits of social support on pain have also been shown through physiological measures. An impending stressor can increase the response of the sympathetic nervous system and hypothalamus-pituitary-adrenal axis, which result to increases in heart rate, blood pressure, and cortisol level (see review in Muscatell and Eisenberger⁹⁵). Social support has been shown to suppress these physiological systems in the experience of a stressor.^{104–108} For example, lower level of blood pressure, heart rate, skin conductance, and cortisol response are observed when verbal social support or social presence is provided during the experience of thermal pain.^{24,109,110} These findings make sense given the threatening nature of painful stimuli and suggest that social support reduces the activation of physiological stress systems in response to pain stimuli.

MODEL AND SUMMARY OF FINDINGS

The evidence presented across behavioral-cognitive, neuroimaging, and physiological research lends significant support to the social-buffering model of pain reduction (Fig. 2). Evidence in favor of the buffering effect has been more often reported in social support and pain research. Although based on a relatively small number of studies, social support seems to prime safety-related brain activation (ie, VMPFC) when individuals are experiencing pain.^{28,100} This positive experience may assist with managing the threatening quality of pain and related stress (perceived threat, body arousal, and tension), and/or with perceived ability to cope with them (perceived mastery, self-esteem, and perceived competence). This stress appraisal process is also associated with decreased neural (ACC, AI, hypothalamus) and physiological (heart rate, skin conductance, blood pressure, cortisol) stress systems that are involved in the experience of threat and pain. This pattern of decreased cognitive-behavioral and neurobiological response to pain-related stress may in turn result in decreased pain experience.

It is worthwhile to note that social support may be associated with different stress appraisal processes, that is, perceived threat versus perceived coping ability, in acute and chronic pain. Pain-related cognitive patterns seem to be more strongly established in patients with chronic pain compared with transient pain evoked in healthy individuals.⁵² Moreover, these patterns of cognition include both decreased perceived threat of pain and increased perceived coping ability.66,83 Our review shows that, in people with chronic pain, social support is associated with both reduced perceived threat and increased perceived coping ability that can decrease pain. With regard to acute pain, the literature suggests that social support primarily inhibits perceived threat of pain.^{24,28} Therefore, this model may differ in acute versus chronic pain in how supportive resources can initiate the stress appraisal process.

Another limitation of this model is the difference in the nature of study designs. Studies in people with chronic pain almost exclusively examine potential associations between social support and pain, whereas experimental studies allow



FIGURE 2. Social-buffering model of pain reduction. Social support decreases pain experience through modulating pain-related stress. This effect is associated with the stress appraisal process. Specifically, supportive resources can be interpreted by individuals with pain as safety signals in the VMPFC, which in turn decrease the threatening quality of pain and/or increase perceived ability to cope with pain. Moreover, this stress appraisal process is mediated by decreased neural and physiological stress systems. Al indicates anterior insula; dACC, dorsal anterior cingulate cortex; GSR, galvanic skin response; HPA, hypothalamus-pituitary-adrenal axis; SNS, sympathetic nervous system; vACC, ventral anterior cingulate cortex; VMPFC, ventromedial prefrontal cortex.

manipulation of experimental conditions to generate stronger conclusions. However, there is a lack of experimental studies that could potentially generate stronger evidence in people with chronic pain. Only a single study was identified in which pain was experimentally induced in fibromyalgia patients. Nonetheless, consistent with findings in healthy individuals, social support reduced pain sensitivity, and threat-related and pain-related brain activation in the primary somatosensory cortex.²⁶

With regard to neurobiological mechanisms, the most consistent findings across studies are shown in Figure 2. However, one study provides an alternative explanation whereby social support could reduce pain via the activation of reward-related neural pathways (eg, caudate, nucleus accumbens, OFC).³⁰ Moreover, these brain areas are shown to have high densities of opioid receptors that have analgesic effects.^{111,112} Findings in this study are likely complementary to others^{28,90} as they focus on different brain areas that are involved in different mechanisms of analgesia. Moreover, differences in the nature of the romantic relationship might account for the inconsistency. Younger et al³⁰ recruited participants in the early stage of romantic relationship (first 9 months); whereas participants in the other 2 studies were in a long-term romantic relationship. One study suggests that long-term romantic relationship not only maintains the reward value from new love, but also involves neural systems implicated in attachment.¹¹³

Another notion is that the influence of social support on pain may be at least partially mediated by attentional distraction. Social support, regardless of whether in the form of social presence or picture imagery, could distract attention from painful stimuli and consequently reduce painrelated brain activation. This is difficult to rule out in some early studies that did not include a control condition.²⁶ However, more recent studies which include additional control conditions such as a distraction task and stranger condition, have confirmed and further detailed the influence of social support on pain experience.^{28,30,90} It has also been proposed that the benefits of support might be related to the attachment style of individuals with pain.¹¹⁴ Attachment style is associated with how the individual with pain interpret supportive resources. For example, some people prefer closeness but fear abandonment (attachment anxiety), whereas others find it difficult to trust and rely on others (attachment avoidance).¹¹⁵ This difference may then modulate the influences of social support and related neural and physiological response.^{110,114}

LIMITATIONS AND FUTURE DIRECTIONS

There are several limitations of this review. First, we were unable to examine the different types of social support, for example, emotional support (eg, love, caring), informational support (eg, advice, feedback), instrumental support (eg, materials, actions). The present review aimed to provide a general overview of the influence of support rather than to disentangle subtypes of social support. It was also not possible to definitively assign the reviewed supportive resources into either group as several covered multiple types of support, for example, social presence, positive support. Future studies could be designed to explore the role of different subtypes of social support in the experience of pain. Further, there are only a small number of studies that examine the neurobiological mechanisms of the buffering effect. This limits the conclusions that can be drawn at this stage and more studies are warranted. Moreover, the type of pain may influence results. Chronic pain was reported in the behavioral correlational studies, whereas acute pain was induced in the neurobiological studies. We then extended the evidence of neural mechanisms derived from acute pain to

explain the buffering effect in people with chronic pain. It is worthwhile to examine whether acute and chronic pain activate similar neural processes while supportive resources are available. Lastly, as pain itself is a stressor, it is somewhat artificial to dichotomize the presence of pain-related threat as a binary phenomenon. In this review, pain-related stress was mixed across binary and continuous variables. Future studies could consistently define and measure painrelated stress as a continuous variable.

CONCLUSIONS AND IMPLICATIONS

The benefits of social support on pain experience have been described in a large number of studies. Theoretical models suggest that social support is associated with decreased pain through its main effect or the buffering influence on stress.³⁹ Findings in the present review suggest that the buffering effect can more often describe the benefits of social support on pain experience. The evidence suggests that pain reduction is partially mediated by the process of support buffering the adverse influences of stress, through processes such as stress appraisal and active coping. Moreover, social support may serve as a safety signal to modulate the perception of threat, and thereby suppress the activation of neural and physiological stress systems in response to pain.

Findings from this review also offer suggestions for support-assisted pain management therapies. Recent guidelines for prescribing opioids for chronic pain suggested potential risks (ie, opioid use disorder), and recommended cognitive-behavioral therapy (CBT) for chronic pain.¹¹⁶

CBT has been shown to result in significant improvements in physical suffering and mood in chronic pain conditions.^{117–120} As a modified CBT, support (usually spousal)-assisted coping skills training has included training components like communication skills, behavioral rehearsal, and mutual goal-setting.^{34,35,121} Findings from initial trials suggest the potential of support therapies in decreasing pain and disability. Building on this, our review suggests the critical significance of perceived support and attachment. These psychological processes have been shown to buffer threat response and improve coping ability.^{15,100} Future trials could therefore further develop upon the role of support and attachment in therapeutic interventions, including the enhancement of support relationships and their dynamics.

REFERENCES

- 1. Merskey H, Bogduk N. *Classification of Chronic Pain*. Seattle: IASP Press; 1994.
- Hampton AJ, Hadjistavropoulos T, Gagnon MM, et al. The effects of emotion regulation strategies on the pain experience: a structured laboratory investigation. *Pain*. 2015;156:868–879.
- 3. Tracey I, Ploghaus A, Gati JS, et al. Imaging attentional modulation of pain in the periaqueductal gray in humans. *J Neurosci.* 2002;22:2748–2752.
- Villemure C, Bushnell CM. Cognitive modulation of pain: how do attention and emotion influence pain processing? *Pain*. 2002;95:195–199.
- Vlaeyen JW, Hanssen M, Goubert L, et al. Threat of pain influences social context effects on verbal pain report and facial expression. *Behav Res Ther.* 2009;47:774–782.
- Williams AcdC, Craig KD. Updating the definition of pain. Pain. 2016;157:2420–2423.
- Craig KD. Social communication model of pain. *Pain.* 2015;156: 1198–1199.
- Hadjistavropoulos T, Craig KD, Duck S, et al. A biopsychosocial formulation of pain communication. *Psychol Bull*. 2011;137:910–937.

- Brown JL, Sheffield D, Leary MR, et al. Social support and experimental pain. *Psychosom Med.* 2003;65:276–283.
- Gallant NL, Hadjistavropoulos T. Experiencing pain in the presence of others: a structured experimental investigation of older adults. J Pain. 2017;18:456–467.
- 11. Holtzman S, Newth S, Delongis A. The role of social support in coping with daily pain among patients with rheumatoid arthritis. *J Health Psychol*. 2004;9:677–695.
- Master SL, Eisenberger NI, Taylor SE, et al. A picture's worth partner photographs reduce experimentally induced pain. *Psychol Sci.* 2009;20:1316–1318.
- Wills TA. Social support and interpersonal relationships. In: Clark MS, ed. *Review of Personality and Social Psychology*. California: Sage Publications; 1991:265–289.
- Dunkel-Schetter C, Bennett TL. Differentiating the cognitive and behavioral aspects of social support. In: Sarason BR, Sarason IG, Pierce GR, eds. *Social Support: An Interactional View*. England: John Wiley; 1990:267–296.
- Uchino BN. Understanding the links between social support and physical health: a life-span perspective with emphasis on the separability of perceived and received support. *Perspect Psychol Sci.* 2009;4:236–255.
- Uchino BN, Bowen K, Carlisle M, et al. Psychological pathways linking social support to health outcomes: a visit with the "ghosts" of research past, present, and future. *Soc Sci Med.* 2012;74:949–957.
- Brown GK, Wallston KA, Nicassio PM. Social support and depression in rheumatoid arthritis: a one-year prospective study1. J Appl Soc Psychol. 1989;19:1164–1181.
- Cano A, Gillis M, Heinz W, et al. Marital functioning, chronic pain, and psychological distress. *Pain*. 2004;107:99–106.
- Lee JE, Kahana B, Kahana E. Social support and cognitive functioning as resources for elderly persons with chronic arthritis pain. *Aging Ment Health.* 2016;20:370–379.
- López-Martínez AE, Esteve-Zarazaga R, Ramírez-Maestre C. Perceived social support and coping responses are independent variables explaining pain adjustment among chronic pain patients. J Pain. 2008;9:373–379.
- Goldstein P, Shamay-Tsoory SG, Yellinek S, et al. Empathy predicts an experimental pain reduction during touch. *J Pain*. 2016;17:1049–1057.
- Jackson T. Interpersonal transactions and responses to cold pressor pain among Australian women and men. Sex Roles. 2007;56:55–62.
- Jackson T, Iezzi T, Chen H, et al. Gender, interpersonal transactions, and the perception of pain: an experimental analysis. J Pain. 2005;6:228–236.
- Roberts MH, Klatzkin RR, Mechlin B. Social support attenuates physiological stress responses and experimental pain sensitivity to cold pressor pain. *Ann Behav Med.* 2015;49: 557–569.
- Edwards R, Eccleston C, Keogh E. Observer influences on pain: an experimental series examining same-sex and oppositesex friends, strangers, and romantic partners. *Pain.* 2017;158: 846–855.
- Montoya P, Larbig W, Braun C, et al. Influence of social support and emotional context on pain processing and magnetic brain responses in fibromyalgia. *Arthritis Rheum.* 2004;50:4035–4044.
- Vervoort T, Caes L, Trost Z, et al. Social modulation of facial pain display in high-catastrophizing children: an observational study in schoolchildren and their parents. *Pain.* 2011;152: 1591–1599.
- Eisenberger NI, Master SL, Inagaki TK, et al. Attachment figures activate a safety signal-related neural region and reduce pain experience. *Proc Natl Acad Sci U S A*. 2011;108:11721–11726.
- Shaygan M, Böger A, Kröner-Herwig B. Valence and arousal value of visual stimuli and their role in the mitigation of chronic pain: what is the power of pictures? J Pain. 2016;18:124–131.
- Younger J, Aron A, Parke S, et al. Viewing pictures of a romantic partner reduces experimental pain: involvement of neural reward systems. *PLoS One*. 2010;5:e13309.

Copyright © 2018 Wolters Kluwer Health, Inc. All rights reserved.
- McClelland LE, McCubbin JA. Social influence and pain response in women and men. J Behav Med. 2008;31:413–420.
- Hurter S, Paloyelis Y, Williams ACdC, et al. Partners' empathy increases pain ratings: effects of perceived empathy and attachment style on pain report and display. J Pain. 2014;15:934–944.
- Modić Stanke K, Ivanec D. Social context of pain perception: the role of other people's presence and physical distance. *Rev Psychol.* 2010;17:69–74.
- 34. Abbasi M, Dehghani M, Keefe F, et al. Spouse-assisted training in pain coping skills and the outcome of multidisciplinary pain management for chronic low back pain treatment: a 1-year randomized controlled trial. *Eur J Pain*. 2012;16:1033–1043.
- Keefe FJ, Caldwell DS, Baucom D, et al. Spouse-assisted coping skills training in the management of knee pain in osteoarthritis: long-term followup results. *Arthritis Rheum*. 1999;12:101–111.
- Fordyce WE. Behavioral methods for chronic pain and illness. Pain. 1977;3:291–292.
- Sullivan MJ, Thorn B, Haythornthwaite JA, et al. Theoretical perspectives on the relation between catastrophizing and pain. *Clin J Pain*. 2001;17:52–64.
- Cano A, Williams ACdC. Social interaction in pain: reinforcing pain behaviors or building intimacy? *Pain*. 2010;149:9–11.
- Cohen S, Wills TA. Stress, social support, and the buffering hypothesis. *Psychol Bull*. 1985;98:310–357.
- Penninx BW, Van Tilburg T, Deeg DJ, et al. Direct and buffer effects of social support and personal coping resources in individuals with arthritis. Soc Sci Med. 1997;44:393–402.
- Penny KI, Purves AM, Smith BH, et al. Relationship between the Chronic Pain Grade and measures of physical, social and psychological well-being. *Pain*. 1999;79:275–279.
- 42. Liberati A, Altman DG, Tetzlaff J, et al. The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration. *Ann Intern Med.* 2009;151:65–94.
- Moher D, Liberati A, Tetzlaff J, et al. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *Ann Intern Med.* 2009;151:264–269.
- Blechman EA, Brownell KD. Behavioral Medicine and Women: A Comprehensive Handbook. New York, NY: Guilford Press; 1998.
- 45. Taylor SE. Social support: a review. In: Friedman HS, ed. Oxford Handbook of Health Psychology. New York, NY: Oxford University Press; 2011:189–214.
- Holmes TH, Rahe RH. The Social Readjustment Rating Scale. J Psychosom Res. 1967;11:213–218.
- Thoits PA. Social support as coping assistance. J Consult Clin Psych. 1986;54:416–423.
- DeLongis A, Coyne JC, Dakof G, et al. Relationship of daily hassles, uplifts, and major life events to health status. *Health Psychol.* 1982;1:119–136.
- Manne SL, Zautra AJ. Spouse criticism and support: their association with coping and psychological adjustment among women with rheumatoid arthritis. *J Pers Soc Psychol.* 1989;56: 608–617.
- Riemsma RP, Taal E, Wiegman O, et al. Problematic and positive support in relation to depression in people with rheumatoid arthritis. *J Health Psychol.* 2000;5:221–230.
- Berg CA, Upchurch R. A developmental-contextual model of couples coping with chronic illness across the adult life span. *Psychol Bull.* 2007;133:920–954.
- 52. Krahé C, Springer A, Weinman JA, et al. The social modulation of pain: others as predictive signals of salience a systematic review. *Front Hum Neurosci.* 2013;7:386.
- Vagnoli L, Caprilli S, Vernucci C, et al. Can presence of a dog reduce pain and distress in children during venipuncture? *Pain Manag Nurs.* 2015;16:89–95.
- 54. Eisenberger NI, Taylor SE, Gable SL, et al. Neural pathways link social support to attenuated neuroendocrine stress responses. *Neuroimage*. 2007;35:1601–1612.
- 55. Romano JM, Jensen MP, Turner JA, et al. Chronic pain patient-partner interactions: further support for a behavioral model of chronic pain. *Behav Ther*. 2000;31:415–440.

- Turk DC, Kerns RD, Rosenberg R. Effects of marital interaction on chronic pain and disability: examining the down side of social support. *Rehabil Psychol.* 1992;37:259–274.
- 57. Campbelll P, Wynne-Jones G, Dunn KM. Review: the influence of informal social support on risk and prognosis in spinal pain: a systematic review. *Eur J Pain.* 2011;15: 444.e1–444.e14.
- Campbell P, Wynne-Jones G, Muller S, et al. The influence of employment social support for risk and prognosis in nonspecific back pain: a systematic review and critical synthesis. *Int Arch Occup Environ Health.* 2013;86:119–137.
- Hoogendoorn WE, van Poppel MN, Bongers PM, et al. Systematic review of psychosocial factors at work and private life as risk factors for back pain. *Spine*. 2000;25:2114–2125.
- Mallen CD, Peat G, Thomas E, et al. Prognostic factors for musculoskeletal pain in primary care: a systematic review. Br J Gen Pract. 2007;57:655–661.
- Lakke SE, Soer R, Takken T, et al. Risk and prognostic factors for non-specific musculoskeletal pain: a synthesis of evidence from systematic reviews classified into ICF dimensions. *Pain*. 2009;147:153–164.
- Revenson TA, Schiaffino KM, Majerovitz SD, et al. Social support as a double-edged sword: the relation of positive and problematic support to depression among rheumatoid arthritis patients. *Soc Sci Med.* 1991;33:807–813.
- 63. Feldman SI, Downey G, Schaffer-Neitz R. Pain, negative mood, and perceived support in chronic pain patients: a daily diary study of people with reflex sympathetic dystrophy syndrome. J Consult Clin Psych. 1999;67:776–785.
- Telfair J, Gardner MM. African American adolescents with sickle cell disease: support groups and psychological wellbeing. J Black Psychol. 1999;25:378–390.
- Ferreira V, Sherman A. The relationship of optimism, pain and social support to well-being in older adults with osteoarthritis. *Aging Ment Health*. 2007;11:89–98.
- 66. Holtzman S, DeLongis A. One day at a time: the impact of daily satisfaction with spouse responses on pain, negative affect and catastrophizing among individuals with rheumatoid arthritis. *Pain.* 2007;131:202–213.
- Morgan MA, Small BJ, Donovan KA, et al. Cancer patients with pain: the spouse/partner relationship and quality of life. *Cancer Nurs.* 2011;34:13–23.
- Sturgeon JA, Dixon EA, Darnall BD, et al. Contributions of physical function and satisfaction with social roles to emotional distress in chronic pain: a Collaborative Health Outcomes Information Registry (CHOIR) Study. *Pain*. 2015;156:2627–2633.
- 69. Park J, Newman D, Engström G, et al. The moderating and covarying effects of social support and pain intensity on depressive symptomology among racially and ethnically diverse older adults. *Pain Manag.* 2016;7:19–32.
- Affleck G, Pfeiffer C, Tennen H, et al. Social support and psychosocial adjustment to rheumatoid arthritis quantitative and qualitative findings. *Arthritis Rheum.* 1988;1:71–77.
- Fitzpatrick R, Newman S, Lamb R, et al. Social relationships and psychological well-being in rheumatoid arthritis. *Soc Sci Med.* 1988;27:399–403.
- Goldberg GM, Kerns RD, Rosenberg R. Pain-relevant support as a buffer from depression among chronic pain patients low in instrumental activity. *Clin J Pain*. 1993;9:34–40.
- Strating MM, Suurmeijer TP, Van Schuur WH. Disability, social support, and distress in rheumatoid arthritis: results from a thirteenyear prospective study. *Arthritis Care Res.* 2006;55:736–744.
- Kerns RD, Haythornthwaite J, Southwick S, et al. The role of marital interaction in chronic pain and depressive symptom severity. J Psychosom Res. 1990;34:401–408.
- Cano A, Weisberg JN, Gallagher RM. Marital satisfaction and pain severity mediate the association between negative spouse responses to pain and depressive symptoms in a chronic pain patient sample. *Pain Med.* 2000;1:35–43.
- Raichle KA, Romano JM, Jensen MP. Partner responses to patient pain and well behaviors and their relationship to patient pain behavior, functioning, and depression. *Pain*. 2011;152:82–88.

- Rosen NO, Bergeron S, Lambert B, et al. Provoked vestibulodynia: mediators of the associations between partner responses, pain, and sexual satisfaction. *Arch Sex Behav.* 2013;42:129–141.
- DeLongis A, Folkman S, Lazarus RS. The impact of daily stress on health and mood: psychological and social resources as mediators. *J Pers Soc Psychol.* 1988;54:486–495.
- Weinberger M, Tierney WM, Booher P, et al. Social support, stress and functional status in patients with osteoarthritis. *Soc Sci Med.* 1990;30:503–508.
- Affleck G, Tennen H, Urrows S, et al. Person and contextual features of daily stress reactivity: individual differences in relations of undesirable daily events with mood disturbance and chronic pain intensity. J Pers Soc Psychol. 1994;66:329–340.
- Alonso C, Coe CL. Disruptions of social relationships accentuate the association between emotional distress and menstrual pain in young women. *Health Psychol.* 2001;20:411–416.
- Pekkarinen L, Elovainio M, Sinervo T, et al. Job demands and musculoskeletal symptoms among female geriatric nurses: the moderating role of psychosocial resources. J Occup Health Psych. 2013;18:211–219.
- Waltz M, Kriegel W, Bosch PVT. The social environment and health in rheumatoid arthritis: marital quality predicts individual variability in pain severity. *Arthritis Rheum*. 1998;11:356–374.
- Kerns RD, Rosenberg R, Otis JD. Self-appraised problem solving and pain-relevant social support as predictors of the experience of chronic pain. *Ann Behav Med.* 2002;24:100–105.
- Corley AM, Cano A, Goubert L, et al. Global and situational relationship satisfaction moderate the effect of threat on pain in couples. *Pain Med.* 2016;17:1664–1675.
- Meenan RF, Gertman PM, Mason JH, et al. The arthritis impact measurement scales. Further investigations of a health status measure. *Arthritis Rheum.* 1982;25:1048–1053.
- Kerns RD, Turk DC, Rudy TE. The West Haven-Yale Multidimensional Pain Inventory (WHYMPI). *Pain*. 1985;23:345–356.
- Felton BJ, Revenson TA. Coping with chronic illness: a study of illness controllability and the influence of coping strategies on psychological adjustment. J Consult Clin Psych. 1984;52: 343–353.
- Brown GK, Nicassio PM. Development of a questionnaire for the assessment of active and passive coping strategies in chronic pain patients. *Pain*. 1987;31:53–64.
- Coan JA, Schaefer HS, Davidson RJ. Lending a hand social regulation of the neural response to threat. *Psychol Sci.* 2006;17:1032–1039.
- Uchino BN. Social support and health: a review of physiological processes potentially underlying links to disease outcomes. J Behav Med. 2006;29:377–387.
- Che X, Cash R, Fitzgerald P, et al. The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat. J Pain. 2017;19:496–505.
- Kanbara S, Taniguchi H, Sakaue M, et al. Social support, selfefficacy and psychological stress responses among outpatients with diabetes in Yogyakarta, Indonesia. *Diabetes Res Clin Pr.* 2008;80:56–62.
- 94. Mobbs D, Yu R, Rowe JB, et al. Neural activity associated with monitoring the oscillating threat value of a tarantula. *Proc Natl Acad Sci U S A*. 2010;107:20582–20586.
- Muscatell KA, Eisenberger NI. A social neuroscience perspective on stress and health. *Soc Personal Psychol Compass.* 2012; 6:890–904.
- Milad MR, Quirk GJ, Pitman RK, et al. A role for the human dorsal anterior cingulate cortex in fear expression. *Biol Psychiatry*. 2007;62:1191–1194.
- Phelps EA, Delgado MR, Nearing KI, et al. Extinction learning in humans: role of the amygdala and vmPFC. *Neuron.* 2004;43:897–905.
- Quirk GJ, Garcia R, González-Lima F. Prefrontal mechanisms in extinction of conditioned fear. *Biol Psychiatry*. 2006;60:337–343.
- Che X, Wei D, Li W, et al. The correlation between gray matter volume and perceived social support: a voxel-based morphometry study. *Soc Neurosci.* 2014;9:152–159.

- Eisenberger NI, Cole SW. Social neuroscience and health: neurophysiological mechanisms linking social ties with physical health. *Nat Neurosci.* 2012;15:669–674.
- Onoda K, Okamoto Y, Nakashima Ki, et al. Decreased ventral anterior cingulate cortex activity is associated with reduced social pain during emotional support. *Soc Neurosci.* 2009;4:443–454.
- Wager TD, van Ast VA, Hughes BL, et al. Brain mediators of cardiovascular responses to social threat, part II: prefrontalsubcortical pathways and relationship with anxiety. *Neuroimage*. 2009;47:836–851.
- Hornstein EA, Fanselow MS, Eisenberger NI. A safe haven investigating social-support figures as prepared safety stimuli. *Psychol Sci.* 2016;27:1051–1060.
- Ditzen B, Heinrichs M. Psychobiology of social support: the social dimension of stress buffering. *Restor Neurol Neurosci*. 2014;32:149–162.
- Fontana AM, Diegnan T, Villeneuve A, et al. Nonevaluative social support reduces cardiovascular reactivity in young women during acutely stressful performance situations. J Behav Med. 1999;22:75–91.
- Heinrichs M, Baumgartner T, Kirschbaum C, et al. Social support and oxytocin interact to suppress cortisol and subjective responses to psychosocial stress. *Biol Psychiatry*. 2003;54:1389–1398.
- 107. Uchino BN, Cacioppo JT, Kiecolt-Glaser JK. The relationship between social support and physiological processes: a review with emphasis on underlying mechanisms and implications for health. *Psychol Bull*. 1996;119:488–531.
- Uchino B, Trettevik R, Kent DGR, et al. Social support, social integration, and inflammatory cytokines: a meta-analysis. *Health Psychol.* 2018;37:462–471.
- 109. Platow MJ, Voudouris NJ, Gilford N, et al. In-group reassurance in a pain setting produces lower levels of physiological arousal: direct support for a self-categorization analysis of social influence. *Eur J Soc Psychol.* 2007;37:649–660.
- 110. Sambo CF, Howard M, Kopelman M, et al. Knowing you care: effects of perceived empathy and attachment style on pain perception. *Pain*. 2010;151:687–693.
- Zubieta J-K, Smith YR, Bueller JA, et al. Regional mu opioid receptor regulation of sensory and affective dimensions of pain. *Science*. 2001;293:311–315.
- Wager TD, Scott DJ, Zubieta JK. Placebo effects on human μopioid activity during pain. *Proc Natl Acad Sci.* 2007;104: 11056–11061.
- Acevedo BP, Aron A, Fisher HE, et al. Neural correlates of long-term intense romantic love. Soc Cogn Affect Neurosci. 2012;7:145–159.
- 114. Krahé C, Paloyelis Y, Condon H, et al. Attachment style moderates partner presence effects on pain: a laser-evoked potentials study. Soc Cog Affect Neur. 2015;10:1030–1037.
- 115. Hazan C, Shaver P. Romantic love conceptualized as an attachment process. J Pers Soc Psychol. 1987;52:511–524.
- Dowell D, Haegerich TM, Chou R. CDC guideline for prescribing opioids for chronic pain—United States, 2016. *JAMA*. 2016;315:1624–1645.
- 117. Williams ACdC, Eccleston C, Morley S. Psychological therapies for the management of chronic pain (excluding headache) in adults. *Cochrane Database Syst Rev.* 2012: CD007407. Doi:10.1002/14651858.CD007407.pub3.
- Ehde DM, Dillworth TM, Turner JA. Cognitive-behavioral therapy for individuals with chronic pain: efficacy, innovations, and directions for research. *Am Psychol.* 2014;69:153–166.
- 119. Morley S, Eccleston C, Williams A. Systematic review and meta-analysis of randomized controlled trials of cognitive behaviour therapy and behaviour therapy for chronic pain in adults, excluding headache. *Pain*. 1999;80:1–13.
- 120. Butler AC, Chapman JE, Forman EM, et al. The empirical status of cognitive-behavioral therapy: a review of metaanalyses. *Clin Psychol Rev.* 2006;26:17–31.
- Keefe FJ, Caldwell DS, Baucom D, et al. Spouse-assisted coping skills training in the management of osteoarthritic knee pain. *Arthritis Rheum.* 1996;9:279–291.

Note:

1. There was an error in the text. 'Therefore, it remain unclear the cognitive and behavioural variables linking a general overview of social support to decreased pain experience' should be changed to 'Therefore, the cognitive and behavioural variables linking a general overview of social support to decreased pain experience remain unclear'. <u>(Page 39, Line 3)</u>

2. 'Higher emotional support or social interaction' should be changed to 'Higher emotional support or more pleasurable social life activity'. (*Page 45, Line 36*)

CHAPTER FOUR

The Threat of Pain

Manuscript

Che X., Cash R., Fitzgerald P., & Fitzgibbon B.M. (2018). The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat. *Journal of Pain, 19*, 496-505.

Preamble to empirical paper

Chapter 3 indicates that social support may reduce pain by modulating perceived threat of painful stimuli. However, previous studies have simultaneously assessed pain perception and the threat of pain, in which the buffering effect suffers from an indirect manner of examination (Eisenberger et al., 2011; Montoya et al., 2004; Younger et al., 2010).

This chapter aims to directly examine the buffering effect that may underpin the effect of social support on pain reduction. In the published manuscript, a novel experimental protocol is used to introduce 6-second continuous threat of pain leading up to painful stimulation in which the presence of social support is manipulated by holding the hand of a significant other. EEG and heart rate are used to evaluate neural and autonomic changes to the threat of pain. Neural and autonomic responses are analysed in the threat of pain phase and related to subsequent changes in pain perception. EEG source localization is performed to identify possible generators of neuronal oscillations and associate them to evidence from functional imaging.

This chapter significantly extends the literature by providing direct evidence that social support may reduce pain through buffering the autonomic and neurophysiological response to the threatening quality of noxious stimuli before the onset of painful stimuli. Moreover, findings in this chapter further indicate the particular significance of close relationship in reducing pain and related threat, and this effect is beyond a particular type of relationship but rather from a romantic partner, a family member, or a close friend.



RESEARCH EDUCATION TREATMENT ADVOCACY



The Journal of Pain, Vol **II**, No **II** (**II**), 2017: pp **II**-**II** Available online at www.jpain.org and www.sciencedirect.com

Original Report

The Social Regulation of Pain: Autonomic and Neurophysiological Changes Associated With Perceived Threat

Xianwei Che,* Robin Cash,* Paul Fitzgerald,*^{,†} and Bernadette M. Fitzgibbon*

*Monash Alfred Psychiatry Research Centre, The Alfred and Central Clinical School, Monash University, Melbourne, Australia. †Epworth Clinic, Epworth Healthcare, Camberwell, Victoria, Australia.

Abstract: The analgesic effect of social support is proposed as a function of social support modulating perceived threat of painful stimuli. In the current study, we directly examined the social buffering effect in the context of the threat of pain. Eighteen healthy participants were subjected to the threat of pain while they held the hand of a close other, a stranger, or not at all. Neural and autonomic responses were recorded using electroencephalogram and heart rate, respectively. Close other handholding reduced pain perception. This was accompanied by decreased heart rate and frontal theta oscillation (4–8 Hz) during the threat phase preceding painful stimulation. Interestingly, decreased heart rate and frontal theta in the close other hand-holding condition were uniquely associated with greater pain reduction during subsequent nociceptive stimulation. Neural changes were sourcelocalized to the insular cortex and the rostral-ventral portions of anterior cingulate cortex, regions involved in the processing of threat and pain. Together, our data build upon work to date linking social support to pain by showing autonomic and neurophysiological changes associated with pain reduction.

Perspective: Social support may reduce pain through buffering the autonomic and neurophysiological response to the threatening quality of noxious stimuli. Results implicate that in clinical settings the caregiver could help people with chronic pain reappraise pain and related conditions as less stressful.

© 2017 by the American Pain Society *Key words:* Social support, pain, stress, heart rate, theta oscillation.

S upportive relationships with others have substantial benefits for individuals.^{11,57} These benefits arise through forms of social support including the structure of one's social relationships (eg, having intimate relationships with others), or the explicit functions performed by others¹⁰ (eg, emotional, economical, informational support). In the context of pain research, literature has largely suggested that social support, in its

© 2017 by the American Pain Society

https://doi.org/10.1016/j.jpain.2017.12.007

various forms, has analgesic effects for pain patients^{7,33,36} as well as in experimental settings with healthy individuals.^{6,17,31,39} Thus, social connectedness may play an important role in the perception of pain.

Preliminary evidence in this area has indicated a "stressbuffering" process in which social support may reduce pain-related threat response that in turn decreases pain.^{25,51,60} For example, one study identified that perceived threat of pain mediated the influences of social support on pain reduction.⁵⁹ Several imaging studies have also suggested that social support may prime safety or reward themes that in turn result in the reduction of pain and related threat response.^{19,62} However, a limitation of current investigations into the buffering effect on pain is the simultaneous assessment of the painful sensation and threat of pain, in which the buffering effect suffers from an indirect manner of examination (eg, Eisenberger et al,¹⁹ and Younger et al⁶²). In addition, where the threat of pain has been specifically addressed, it has been assessed through a self-report dichotomous variable (ie,

Received August 4, 2017; Revised November 28, 2017; Accepted December 5, 2017.

This work was supported by the China Scholarship Council; the National Health and Medical Research Council Practitioner Fellowship (606907), and the National Health and Medical Research Council Early Career Fellowship. The authors have no conflicts of interest to declare. Supplementary data accompanying this article are available online at www.jpain.org and www.sciencedirect.com.

Address reprint requests to Xianwei Che, PhD candidate, Monash Alfred Psychiatry Research Centre, Monash University, Level 4, 607 St Kilda Road, Melbourne, Vic. 3004, Australia. E-mail: 1526-5900/\$36.00

2 The Journal of Pain

yes or no; eg, Corley et al,¹² and Vlaeyen et al⁵⁹). However, the social buffering effect involves not only the subjective, self-reported experience of threat or stress, but (albeit related) neural, cardiovascular, and neuroendocrine dimensions of perceived threat, which are closely associated with health outcomes.⁵⁶

In the current investigation, we build upon the literature of social support and pain to date by using electroencephalogram (EEG) and heart rate measures to assess neural and autonomic changes during the threat of pain and its relationship to pain reduction. To do so, we used a novel manipulation to introduce 6-second continuous threat of pain leading up to painful stimulation. It is important to note that EEG and heart rate offer high temporal resolution. In the present paradigm, this allowed us to reveal the dynamic increase in neural and autonomic response during pain-related threat, and crucially, directly show the buffering effect of social support on threat and pain experience as well as the graded influence of social support depending on the quality of the relationship (ie, close acquaintance vs stranger). Because EEG theta oscillation (4–8 Hz⁸) is suggested as one potential neural correlate of the perception of threatening cues,^{15,43} we hypothesized that social support would reduce theta activity and autonomic heart rate in the threat stage leading up to painful stimulation. Importantly, we anticipated that these changes during the threat of pain would be related to decreased pain ratings after noxious stimulation.

Methods

Subjects

Eighteen healthy adults participated in this study (8 male and 10 female, age range = 18–35 years, mean = 25.2, SD = 5.7). All participants were right-handed, not taking any medication, and had no history or current diagnosis of a neurological or a psychiatric disorder, assessed using the Mini International Neuropsychiatric Interview.⁵³ Involvement in the study required inclusion of a participant's close other (romantic partner, family member, or close friend). All study participants and their close other provided informed consent and the experiment was approved by the Alfred Hospital and Monash University Human Research and Ethics Committee. This study was conducted in accordance with the Declaration of Helsinki.

Experimental Design and Procedure

Noxious electrical stimuli were delivered under 3 conditions: during hand holding of 1) the subjects' close acquaintance ("close other" condition), 2) a stranger ("stranger" condition), and 3) in the absence of handholding ("no-holding" condition). The stranger was a staff member who was gender-matched to the close other (a total of 2 staff members only were used for this role). The close other/stranger was only present during the relevant experimental condition and was instructed not to talk or have any eye contact during the recording.

As shown in Fig 1, the experimental task consisted of 30 trials in each condition. Each trial started with the

l 6 0-2 1 8 6 Time(s)

Figure 1. Illustration of the experimental design.

presentation of a fixation cross for 1 second. This was followed by the presentation of numbers in center-screen counting down from 6 to 1, at the rate of 1 number per second. A blank screen was then shown for between 0 to 2 seconds, which was followed by a train of painful stimuli lasting for 1 second. This jittered interval was designed to control for the habituation to the impending stimuli. Immediately after the stimulation participants were asked to rate pain intensity and then pain unpleasantness aroused by the stimuli. Participants gave a verbal rating within 8 seconds (described in Trial-by-Trial Rating of Painful Sensation). The participant was allowed to relax for a further 6 seconds before commencement of the subsequent trial. The duration of each condition was approximately 10 minutes. The participant held the hand throughout the condition when relevant. The order of conditions were pseudorandomized and counterbalanced across participants.

Painful Stimuli

Before the start of each condition, individual current levels were determined. Small electrical stimuli were applied to the dorsum of the left little finger (Stimulus Isolator; ADInstruments, Sydney, Australia). Trains of fifty $500-\mu s$ (frequency = 50 Hz, total time = 1 second) stimuli were applied. After each train of stimuli participants gave a verbal pain rating between 0 and 10 (0 = no pain; 10 = worst pain imaginable). The calibration procedure started from a very low current, for which the participants could barely feel the stimulation, then the current was gradually increased. The calibration process was halted when participants rated the intensity as 7.26 We then repeated the current 10 more times to see if it can consistently create a painful feeling of 7 for at least 8 times. If not, the current was increased again until it did elicit a painful rating of 7 for 8 od 10 stimuli. This method of pain calibration has been used in previous studies, 32, 35, 52 and was chosen to avoid habituation to perceived threat of pain. The calibration procedure was performed in the absence of the close other or the stranger to avoid any confounding influence.

Trial-by-Trial Rating of Painful Sensation

To assess perceived pain after the offset of each train of stimuli, subjects provided pain intensity and unpleasantness ratings on a numerical scale ranging from 0 to 10 (pain intensity: no pain to worst pain imaginable; pain unpleasantness: not unpleasant to most unpleasant pain



Che et al

imaginable). To associate decreased neurophysiological threat response with pain relief (ie, a measurement of pain reduction induced by holding the hand of a close other or stranger compared with no-holding), we specifically computed the difference between the mean of pain intensity in no-holding trials and the mean of pain intensity in close other trials^{19,62} (or stranger trials) and then correlated to changes in heart rate and EEG theta activity.

Heart Rate Recording

Heart rate was monitored continuously during each of the 3 conditions using electrocardiogram (PowerLab/ 4SP; ADInstruments, Sydney, Australia). Three electrodes were administrated to the volar surface of bilateral forearms and the calf muscle of the right leg, respectively. Data were recorded with LabChart (version 5.5.6, ADInstruments) in the sampling rate of 1,000 Hz on a separate computer.

EEG Recording

Recordings took place in a sound-attenuated, temperature-controlled, and electrically shielded room. Subjects were seated in a slightly reclined chair with face approximately 50 cm from the computer monitor. Continuous EEG was recorded using a 64-channel Quickcap (Neuroscan Inc, Victoria, Australia) with CPZ as the reference electrode. Vertical electro-oculogram activity was monitored with electrodes attached above and below the left eye, and horizontal electro-oculogram activity was monitored with electrodes located at the outer canthus of both eyes. Data were sampled at 1,000 Hz with impedances below 5 k Ω throughout the testing.

Data Analysis

For the electrocardiogram data, interbeat interval series were derived using Pan-Tompkins algorithm that identifies the peak of the R wave as the fiducial point.⁴⁶ Artifacts were checked visually and edited as necessary according to published guidelines.³ Then interbeat interval series were transformed to beat-per-minute (BPM) series and baseline corrected (-500 to 0 ms, where time 0 represents the onset of the first countdown number) for each trial. Specifically, percent change of BPM was calculated using the Equation 1. This method is believed to control for individual differences in baseline heart rate, and capture the dynamics of event-related heart rate change in a short period.⁴ Percent change of BPM series were then averaged across trials for each participant in each condition, and area under the curve (AUC) was calculated with the linear trapezoidal rule to measure event-related heart rate change during the presentation of countdown numbers (0-6,000 ms).^{34,37} For 2 participants, heart rate recordings were unavailable due to technical issues.

> Percent change of BPM = (BPM at each time point - BPM mean [-500 to 0 ms])/ BPM mean (-500 to 0 ms)×100 (1)

The Journal of Pain 3

We then examined the correlation between heart rate change and pain relief in the close other (or stranger) compared with the no-holding condition. Because of the dynamic changes of heart rate, we were particularly interested in investigating the specific time windows in which those two would correlate. To this end, a sliding window method was used. The window length was set as 1,000 ms with 50% of overlapping. This was chosen as each countdown number was presented for 1,000 ms. In each window, heart rate reduction was computed as the difference of AUC between the no-holding and the close other (or stranger) condition. Pearson correlations were further computed to assess the relationship of heart rate reduction and pain relief.

Offline EEG data were preprocessed using customwritten scripts that implement functions from EEGLAB (version 13.6.5b; see Delorme and Makeig¹⁶) running under Matlab R2016b (The MathWorks, Inc, Natick, MA). Bad channels were first removed. Data were then filtered (Butterworth filter, band-pass = .5-100 Hz, bandstop = 48-52 Hz), referenced to the average reference, and corrected for stereotyped artifacts including eye blinks, lateral eye movements, muscle, and line noise using the FastICA algorithm.³⁰ Stereotyped artifacts were identified by visual inspection of the spatial and temporal representation of the independent components. Continuous data were then segmented into 7,000-ms nonoverlapping epochs spanning from 1,000 ms before to 6,000 ms after the onset of the first countdown number. Missing channels were interpolated, and epochs were inspected again to remove any anomalous activity in the signal.

Time–frequency representations were calculated with Hanning tapered "mtmconvol" method (7 cycles per time window), as implemented in FieldTrip toolbox.⁴⁵ This method can convolve dynamic EEG time–frequency data with a complex wavelet, and has the advantage that the temporal spread is fully confined to the time window of interest. We calculated power for frequencies ranging from 1 to 100 Hz in the time window of –1,000 to 6,000 ms. Power values were calculated for each trial, and averaged across trials for each subject in each condition. Single-trial baseline corrections were performed with an interval of –500 ms to 0 ms as the baseline.

We further assessed the relationship between power changes and pain relief between conditions. Specifically, power changes were calculated as the difference between no-holding and either close other or stranger conditions. A sliding window method was also adopted for the heart rate analysis, to capture the dynamic relationship between power changes and pain reduction.

Source Localization (Standardized Low Resolution Electromagnetic Tomography Algorithm)

Source localization was further performed to explore the possible generators of neuronal oscillations for which we observed significant differences across conditions. The standardized low resolution electromagnetic tomography

4 The Journal of Pain

algorithm (sLORETA) is frequently used to estimate possible generators of neuronal oscillations or evoked potentials.⁴⁷ It finds a unique inverse solution for the cortical source of scalp EEGs. Results of sLORETA have been validated using combined EEG-positron emission tomography and EEG-functional magnetic resonance imaging (fMRI) data.^{44,48}

In the current study, time-varying cross-spectra were calculated for single-trial data including baseline (-500 to 0 ms) and test intervals.^{29,42} Here we first limited the test intervals to the time periods in which differences in theta activity were observed between conditions (see the Results section). These steps were further supplemented by analyses in which the test intervals were extended to the entire presentation of countdown numbers (0–6,000 ms). Current source density of theta activity was estimated for cortical voxels. To align the source localization with the time–frequency analysis, event-related changes of the current source density for each time frame within the test interval were calculated as log event-locked deviations from baseline mean.

Statistical Analyses

Repeated measures 2-way analysis of variance (ANOVA) calculations were first performed with SPSS (version 22; IBM Corp, Armonk, NY) on pain intensity and pain unpleasantness to examine the habituation to the noxious stimuli. Pain intensity and pain unpleasantness were split to the first and second half, respectively (15 trials in each half), and condition (close other, stranger, no-holding), and time (first half, second half) were specified as the repeated measure factors. Post hoc t-tests were conducted with a Bonferroni correction with the α level set to .05.

Further, repeated measures 1-way ANOVAs were performed to examine the condition difference in current calibration, pain intensity, pain unpleasantness, and AUC of heart rate change, respectively. Condition (close other, stranger, no-holding) was specified as the repeated measure factor. Post hoc t-tests were conducted with a Bonferroni correction to further explore the significant main effects of condition, and the α level was set to .05.

For time-frequency data, differences between conditions were evaluated using a nonparametric clusterbased permutation test. This method can be used to compare conditions for significant spatiotemporal differences, which provides a straightforward way to solve the multiple comparisons problem.³⁸ In other words, this analysis identifies time periods and electrode clusters in which a given frequency bin differs across conditions. This method was applied to the time window of interest (0-6,000 ms) in all scalp channels. An observed test statistics value was considered in the cluster permutation if it was below the threshold of .05 in at least 2 of the neighboring channels.⁴⁵ Further, 5,000 iterations of trial randomization were carried out for generating the permutation distribution at a given frequency band. A threshold of .025 (2-tailed) was used for evaluating the electrodes that exhibit a significant difference in power. As our a priori hypothesis related to theta activity, statistical analysis was first carried out in theta band (4–8 Hz). Additional analyses were performed separately in other frequency bands (delta = 1–3 Hz, alpha = 9–12 Hz, beta = 13–30 Hz, gamma = 31-100 Hz).

For sLORETA statistical differences between conditions were calculated as images of voxel-by-voxel t values. The localization of differences in cortical activity was on the basis of the standardized electrical current density and resulted to 3-D t score images. In these images, cortical voxels of significant difference were identified using a nonparametric approach thresholded at .05 determined by 5,000 randomizations.⁴⁷

Results

Behavioral Results

Habituation Analysis

Two-way repeated measure ANOVA revealed a main effect of condition in pain intensity ($F_{2,34} = 6.19$, P < .05, partial eta squared $[\eta_{\rho}^2] = .27$). There was no main effect of time ($F_{1,17} = .16$, P = .69, $\eta_{\rho}^2 = .01$), or an interaction effect of Condition × Time ($F_{2,34} = 2.03$, P = .15, $\eta_{\rho}^2 = .11$). Similarly, we found a main effect of condition in pain unpleasantness ($F_{2,34} = 5.00$, P < .05, $\eta_{\rho}^2 = .23$). There was no main effect of time ($F_{1,17} = .14$, P = .71, $\eta_{\rho}^2 = .01$), or the interaction effect of Condition × Time ($F_{2,34} = 2.33$, P = .11, $\eta_{\rho}^2 = .12$). These data together confirmed the absence of habituation. Specific comparisons between conditions are detailed in 1-way ANOVA.

Current Calibration

A repeated measure ANOVA revealed that the main effect of condition was not significant in the current calibration ($F_{2,34} = .43$, P = .65, $\eta_p^2 = .03$), confirming that the participants received noxious stimuli of the same current level across 3 conditions (mean = 4.71 mA, SD = 1.56 across conditions).

Pain Intensity

A main effect of condition was found in pain intensity ($F_{2,34} = 6.24$, P < .05, $\eta_p^2 = .27$), with post hoc tests showing that pain intensity was lower in close other (95% confidence interval [CI] = -1.75 to -.13, P < .05) and stranger condition (95% CI = -1.26 to -.04, P < .05) relative to no-holding condition (Fig 2A). Close other and stranger condition revealed no differences in pain intensity (95% CI = -1.02 to .44, P = .92).

Pain Unpleasantness

Similarly, we found a main effect of condition in pain unpleasantness ($F_{2,34} = 4.96$, P < .05, $\eta_P^2 = .23$). Post hoc tests showed that pain unpleasantness was lower in the close other relative to the no-holding condition (95% CI = -1.51 to -.02, P < .05; Fig 2B), but no difference was found in the stranger versus no-holding condition (95% CI = -1.23

Che et al



Figure 2. Influence of social support on pain perception. Subjective ratings of (A) pain intensity and (B) pain unpleasantness (mean \pm standard error of the mean). The asterisk represents statistical significance (P < .05).

to .06, P = .10), or the close other versus stranger condition (95% Cl = -.81 to .44, P = .99).

Heart Rate Response and Relationship With Pain

Heart rate increased with the progression of countdown numbers in all conditions, although it appeared to taper off between 4,500 to 6,000 ms (Fig 3A). A main effect of condition was found in the AUC of heart rate change ($F_{2,28} = 3.19$, P < .05, $\eta_p^2 = .19$). Post hoc tests showed that close other condition produced less AUC than no-holding condition (95% CI = -7.88 to -.21, P < .05; Fig 3B), whereas other comparisons did not reach significance. One outlier was detected using GraphPad Prism (https://www.graphpad.com) thresholded at P < .05 and consequently removed.

Results also showed that heart rate decrease induced by close other hand-holding was associated with greater pain relief. Pearson correlation revealed that this relationship only reached significance in the late stage of countdown numbers presentation (3,500–6,000 ms; Fig 3C). The results remained consistent when absolute, rather than relative, heart rate measures were used for statistical analysis (see Supplementary Fig 1). In terms of the stranger condition, there was no significant correlation between heart rate reduction and pain relief.

Event-Related Time–Frequency Results and Relation to Pain

The whole-scalp cluster analysis on theta frequency revealed a period (3,300–4,310 ms) during which theta power was significantly decreased in the close other relative to no-holding condition (Fig 4A). The topography of this power decrease had a frontal, slightly right lateralized distribution (significant at AF4, F2, FZ, and FC2). A supplementary analysis showed that there was no difference in any other frequency bands between the close other and no-holding condition.

In terms of the relationship to pain relief (Fig 4B), modulation of theta power was associated with greater pain relief in the late stage of countdown numbers presentation (3,000–5,500 ms).



Figure 3. Heart rate change and relationship with pain relief. (A) Shows the event-related dynamic changes of heart rate with the progression of countdown numbers. (B) Shows the AUC of heart rate changes across conditions. Each column and error bar represent the mean and the standard error of the mean. The asterisk represents statistical significance (P < .05). (C) Shows the correlation between heart rate (HR) reduction (HR [no-holding condition] – HR [close other condition]) and pain relief (pain intensity [no-holding condition]). a.u., arbitrary unit.



Figure 4. Social support modulates EEG theta activity during threat of pain phase. (A) Time frequency plot showing the reduction in theta power in the close other relative to no-holding condition. Cluster analysis showed that this reduction in theta power reached significance at electrodes AF4, FZ, F2, and FC2 during 3,300 to 4,310 ms after the start of the threat phase. Areas with no significant differences were zeroed out. (B) There was a significant correlation (below the line; P < .05) between the reduction in theta power and subjective rating of pain relief in the same close other compared with no-holding condition. (C) Time frequency plot showing the electrodes (P1, P3, P5, PO3) and earlier time period (1,100–1,780 ms) in which theta power was significantly reduced in the stranger relative to no-holding condition.

6 The Journal of Pain

The nonparametric cluster-based permutation also identified an early period (1,100–1,780 ms) in which theta power was significantly decreased in the stranger relative to no-holding conditions. This power decrease was located over left parieto-occipital regions (P1, P3, P5, and PO3; Fig 4C). No other significant differences were observed in any other frequency bands between the stranger and no-holding condition. Moreover, this spatiotemporal modulation of theta power was not associated with pain relief.

Source Localization

Periods identified by the nonparametric clusterbased permutation as differing significantly between conditions were subjected to source localization procedures. In the time window that theta power difference was observed (3,300–4,310 ms), decreased source localized theta activity was found in the left pregenual anterior cingulate cortex (pgACC; Brodmann area [BA] 24, Montreal Neurological Institute [MNI] coordinates: X = -5, Y = 23, Z = 14, t = -3.47, P < .05), the bilateral subgenual anterior cingulate cortex (sgACC; BA 25, MNI coordinates: X = 0, Y = 1, Z = -5, t = -3.55, P < .05), and the right insula (BA 13, MNI coordinates: X = 45, Y = 10, Z = -5, t = -3.44, P < .05), in the close other relative to noholding condition (Fig 5).

When the test interval was extended to the whole stage of countdown numbers presentation (0–6,000 ms), decreased activity of the left pgACC (BA 32, MNI coordinates: X = -5, Y = 35, Z = 15, t = -4.37) was observed in the close other relative to no-holding condition (see Supplementary Fig 2).

In another time window (1,110–1,780 ms), no brain activation was found at P < .05, corrected for multiple



Figure 5. Source localization of EEG theta oscillation in the time window of 3,300 to 4,310 ms in close other versus no-holding condition. Results were shown at P < .05, multiple comparison corrected.

comparisons, in the stranger relative to no-holding condition. No brain activation was found when we extended the test interval to the entire presentation of countdown numbers (0–6,000 ms).

Discussion

Preliminary evidence has indicated a buffering process that may link social support to pain reduction.^{25,59,60} In the current study, we used a novel design to manipulate the threat of pain and characterize the autonomic and neurophysiological changes within the proposed buffering effect. Consistent with most of the literature, our results show that social support alleviated the experience of pain. In addition, our data provide novel evidence of the buffering effect in which social support reduced heart rate and fontal theta oscillations during the threat of pain (ie, before noxious stimulation), and that the magnitude of these changes were related to greater pain reduction. These findings provide evidence that the analgesic influence of social support may be driven by its role in the modulation of the threat of pain.

Previous studies have shown that social support is able to attenuate pain and physiological responses to painful stimuli, including heart rate,^{21,51} blood pressure,^{21,51} skin conductance,^{28,49} and cortisol levels.⁵¹ Building on these findings, our data show that hand-holding with a close acquaintance reduced pain perception (Figs 2A and 2B), and heart rate preceding delivery of painful stimuli that was seen in other conditions (Figs 3A and 3B). We further found that decreased autonomic response to the threat of pain was associated with greater pain reduction in the presence of social support. Therefore, by assessing the threat of pain before painful stimulation our results directly show the social-buffering effect through attenuated autonomic arousal to pain. Moreover, this was only observed in the close other but not in the stranger condition, which further suggests a modulatory influence of relationship quality in the buffering effect.

In addition, an interesting dynamic relationship was observed between the reduction in autonomic and subjective metrics of pain experience by social support (Fig 3C). Specifically, this relationship only reached significance in the late stage of presentation of countdown numbers (approximately 3.3-6 seconds) preceding the delivery of noxious stimuli. A well controlled methodology was used in the pain calibration to determine the intensity of painful stimulation. Countdown numbers were adopted to dynamically manipulate the threat of painful stimuli with the intention that individuals may experience accumulated stress with the impending threat of pain. This is supported by our results, which showed a dynamic increase in heart rate in anticipation of painful stimuli (Fig 3A). Unique to our study design, the buffering influence of social support on the autonomic response could be tracked in the lead up to the delivery of noxious stimuli. Furthermore, the data suggest that this change was not simply due to somatosensory distraction (as suggested by Eisenberger et al¹⁹), because stranger handholding did not decrease heart rate, which also created somatosensory distraction. Together, these findings

Che et al

suggest that the buffering influence of social support on pain experience may actually increase in value as pain threat increases. Thus, the analgesic effect of social support is associated with downregulating the threatening quality of painful stimuli.

We also showed decreased frontal theta activity during the threat of pain in the presence of close other handholding, which provides, to our knowledge, for the first time, the neurophysiological evidence of the social buffering effect in the context of pain. Previous studies have suggested that social support may prime safety or rewardrelated brain activation (eg, ventromedial prefrontal cortex) in the reduction of pain.^{18,19,62} One EEG study reported that frontal theta is related to the expression of fear,⁴³ whereas other studies suggested that this is associated with the experience of social distress (eq, social exclusion).^{13,58} Therefore, decreased frontal theta in our protocol may represent a lower level of stress aroused by impending painful stimuli. Importantly, decreased frontal theta in the close other condition selectively predicted greater pain reduction when the painful stimulation becomes more threatening (approximately 3-5.5 seconds; Fig 4B). This finding builds upon the behavioral and fMRI evidence in this area, by providing direct evidence for the social buffering hypothesis in which decreased frontal theta activity to the threat of pain is able to predict greater pain reduction.

However, there are other potential, although perhaps inter-related, interpretations. Beyond threat processing, frontal theta oscillation is also implicated in topdown control^{8,23,34} and behavioral adjustment to uncertain or aversive outcomes.^{9,16} In the context of the present results, an alternative explanation is therefore that frontal theta oscillations may also reflect the effort required to regulate the experienced distress. This may explain why heart rate tapers off in the end of stress manipulation when it is otherwise expected to be at its highest.

Finally we provide source localization analysis performed in the theta range to examine the potential neurological mediators generating the buffering effect. This procedure identified event-related changes of the current source density in the theta range that are presumed to account for the effects of social support observed as described previously. Specifically, decreased neural activity was found in the pgACC, sgACC, and insular cortex (anterior as well as posterior parts) in the close other compared with no-holding condition (Fig 5). Although EEG has the advantage of high temporal resolution, the results of EEG source localization are limited in terms of spatial resolution, especially compared with fMRI studies. Nonetheless, the results of EEG beam-forming were highly congruent with regions of neural activity identified in related fMRI works.9,19,62

It is acknowledged that the threat of pain may be highly connected to pain experience.⁵⁰ The current study induced threat of pain before painful stimulation, which was designed to directly examine the social buffering effect. However, this does not mean that they are independent of each other. Our data support this idea by showing high correlations between decreased neurophysiological threat response and pain reduction in the context of

The Journal of Pain 7

social support. Moreover, it was not our intention to show that heart rate and neural (EEG) activity are fully independent, because they are likely to be at least partially related via the hypothalamic-pituitary-adrenal axis. Instead, the present methodology allowed us to measure each independently and together they tell a story of closely inter-related autonomic and regionally specific neural changes associated with the social modulation of pain.

Close other hand-holding did not show any advantage over stranger hand-holding in decreasing pain or suppressing neurophysiological stress systems. This is different from the literature, which shows the effects of spousal hand-holding on pain reduction compared with stranger hand-holding.9,39 This finding could be interpreted to support the evidence that the quality of a relationship modulates the analgesic effects of social support.³¹ Indeed, several studies in this area manipulate social support through romantic relationships, whereas our participants' "close other" included not only romantic partners, but also friends and family members. This was done to be more reflective of the broad levels of social support in daily life, however, these types of relationships may not have the same salience as a romantic partner and may therefore have less of an effect on buffering the threat of pain.

Consistent with the literature,^{9,39} stranger handholding did not reduce pain unpleasantness (Fig 2B) or modulate the autonomic response in anticipation of painful stimuli (Fig 3B). However, pain intensity ratings were decreased in the stranger relative to no-holding condition (Fig 2A). This discrepancy in terms of the reduction in pain intensity but not unpleasantness underscores the multidimensional definition of pain experience (ie, somatosensory and affective-motivational dimension⁴⁰), and further research is warranted to explore the modulation of specific dimensions of pain by social support. Our findings suggest that stranger hand-holding does not necessarily provide social support to individuals in distress as it is not able to decrease the unpleasant feelings or physiological arousal. Nonetheless, the data indicate that some qualities of pain experience are reduced by stranger hand-holding and that a different mechanistic basis might underlie this effect.

To further explore this possibility, EEG oscillation was contrasted in the stranger with no hand-holding condition and significantly lower posterior theta was observed (eg, parietal, occipital regions) in the early presentation of countdown numbers (Fig 4C). A number of studies have shown the involvement of posterior theta in the processing of stimuli of high emotional arousal (eg, emotional faces), especially in the early stage of stimulus presentation.^{2,23} Further studies have reported an early increase in posterior theta power during selective processing of threatening cues.^{15,55,63} These studies suggest the critical role of posterior theta in selective attention for emotionally arousing and especially threatening cues. The reduction in posterior theta power in the present study may therefore reflect reduced processing of threatening stimuli, perhaps due to distraction or attentional processes associated with stranger hand-holding.

8 The Journal of Pain

Beyond the theoretical implications, our findings may also help understand and improve the role of social support in chronic pain patients. Our data may provide evidence in favor of support-assisted pain management therapies, in which the caregiver could help pain patients reappraise pain and related conditions as less stressful.^{1,24,27} Our results further emphasize the significance of relationship quality as well as the quality of social support in pain management. Factors influencing the quality of social support, such as frequent changes in caregivers in a pain management facility, could affect patient's pain outcomes. Moreover, although pain may not always be continuous, the threat of pain may continue between pain flares, and our data suggest that social support is likely to have a more prolonged and meaningful effect on patients' pain experience through modifying the threat of pain. However, it may be the case that our findings using a paradigm of acute pain in healthy participants do not generalize to chronic pain populations. In studies of people with chronic pain, cognitive responses to pain have been reported to be altered in chronic pain conditions compared with healthy individuals (ie, pain is perceived as more painful and threatening).^{14,22} It may be that there is a ceiling effect beyond which social support can cope. Moreover, although our behavioral outcomes assessed the effect of social support on perceived pain, social support may act on other aspects of the chronic pain experience such as functional disability and depression.^{5,61} Further research is therefore required to investigate the relationship and effect of social support in chronic pain patients.

There are several limitations in the study. Despite the significant results reported in this investigation, they are on the basis of a relatively small sample size. The extent to which the findings might therefore generalize to a larger and potentially more diverse sample is not clear and should be investigated in future studies. In addition, hand-holding was used to convey social support in this study. However, social support is a multidimensional

References

1. Abbasi M, Dehghani M, Keefe F, Jafari H, Behtash H, Shams J: Spouse-assisted training in pain coping skills and the outcome of multidisciplinary pain management for chronic low back pain treatment: A 1-year randomized controlled trial. Eur J Pain 16:1033-1043, 2012

2. Aftanas L, Varlamov A, Pavlov S, Makhnev V, Reva N: Affective picture processing: Event-related synchronization within individually defined human theta band is modulated by valence dimension. Neurosci Lett 303:115-118, 2001

3. Berntson GG, Thomas Bigger J, Eckberg DL, Grossman P, Kaufmann PG, Malik M, Nagaraja HN, Porges SW, Saul JP, Stone PH: Heart rate variability: Origins, methods, and interpretive caveats. Psychophysiology 34:623-648, 1997

4. Bradley MM, Codispoti M, Cuthbert BN, Lang PJ: Emotion and motivation I: Defensive and appetitive reactions in picture processing. Emotion 1:276, 2001

5. Brown GK, Wallston KA, Nicassio PM: Social support and depression in rheumatoid arthritis: A one-year prospective study. J Appl Soc Psychol 19:1164-1181, 1989

Social Regulation of Pain

construct that goes beyond physical contact. Indeed, a recent meta-analysis has shown that the influence of social support on pain is modulated by the subject's perception of the intention and capacity of the social partner to provide assistance, as well as the preexisting relationship between them.³¹ Further studies are therefore warranted to investigate the possible effects of social support across different contexts (ie, beyond handholding), as well as factors that may affect the influence of social support such as relationship quality. Finally, although we present analysis suggesting that the neural source of EEG activity in our data was congruent with brain regions involved in the processing of threat and pain,^{9,20,41,54} EEG source localization is limited in spatial resolution, particularly in the localization of deeper brain regions.⁴⁷ Future studies could consider examining this protocol with concurrent fMRI and EEG recording.

Overall, this study extends the work of previous studies suggesting that social support may be able to reduce pain through modifying perceived threat of pain.^{19,25,59,60} In this study, we provided a novel manipulation to induce a threat phase before the painful stimulation to directly examine the social buffering effect and the dynamics within this effect. Our findings show that social support decreased threat-related changes in heart rate and reduced frontal theta activity, with the magnitude of these changes associated with greater pain reduction. Furthermore, the social buffering effect was generally stronger and did not subside as the threat of pain increased. Our results thus provide evidence in support of the social buffering effect by showing the neural and physiological dimensions of this effect. Together, this study provides novel insights of the dynamics of the buffering effect of social support on pain.

Supplementary Data

Supplementary data related to this article can be found at https://doi.org/10.1016/j.jpain.2017.12.007.

6. Brown JL, Sheffield D, Leary MR, Robinson ME: Social support and experimental pain. Psychosom Med 65:276-283, 2003

7. Cano A, Gillis M, Heinz W, Geisser M, Foran H: Marital functioning, chronic pain, and psychological distress. Pain 107:99-106, 2004

8. Cavanagh JF, Shackman AJ: Frontal midline theta reflects anxiety and cognitive control: Meta-analytic evidence. J Physiol (Paris) 109:3-15, 2015

9. Coan JA, Schaefer HS, Davidson RJ: Lending a hand social regulation of the neural response to threat. Psychol Sci 17: 1032-1039, 2006

10. Cohen S, Underwood LG, Gottlieb BH (eds): Social Support Measurement and Intervention: A Guide for Health and Social Scientists. New York, NY, Oxford University Press, 2000

11. Cohen S, Wills TA: Stress, social support, and the buffering hypothesis. Psychol Bull 98:310, 1985

12. Corley AM, Cano A, Goubert L, Vlaeyen JW, Wurm LH: Global and situational relationship satisfaction moderate the

Che et al

effect of threat on pain in couples. Pain Med 17:1664-1675, 2016

13. Cristofori I, Moretti L, Harquel S, Posada A, Deiana G, Isnard J, Mauguière F, Sirigu A: Theta signal as the neural signature of social exclusion. Cereb Cortex 23:2437-2447, 2012

14. Crombez G, Hermans D, Adriaensen H: The emotional stroop task and chronic pain: What is threatening for chronic pain sufferers? Eur J Pain 4:37-44, 2000

15. DeLaRosa BL, Spence JS, Shakal SK, Motes MA, Calley CS, Calley VI, Hart J, Kraut MA: Electrophysiological spatiotemporal dynamics during implicit visual threat processing. Brain Cogn 91:54-61, 2014

16. Delorme A, Makeig S: EEGLAB: An open source toolbox for analysis of single-trial EEG dynamics including independent component analysis. J Neurosci Methods 134:9-21, 2004

17. Edwards R, Eccleston C, Keogh E: Observer influences on pain: An experimental series examining same-sex and opposite-sex friends, strangers, and romantic partners. Pain 158:846-855, 2017

18. Eisenberger NI, Cole SW: Social neuroscience and health: Neurophysiological mechanisms linking social ties with physical health. Nat Neurosci 15:669-674, 2012

19. Eisenberger NI, Master SL, Inagaki TK, Taylor SE, Shirinyan D, Lieberman MD, Naliboff BD: Attachment figures activate a safety signal-related neural region and reduce pain experience. Proc Natl Acad Sci U S A 108:11721-11726, 2011

20. Fan J, Gu X, Liu X, Guise KG, Park Y, Martin L, de Marchena A, Tang CY, Minzenberg MJ, Hof PR: Involvement of the anterior cingulate and frontoinsular cortices in rapid processing of salient facial emotional information. Neuroimage 54:2539-2546, 2011

21. Fishman E, Turkheimer E, DeGood DE: Touch relieves stress and pain. J Behav Med 18:69-79, 1995

22. Gatchel RJ, Peng YB, Peters ML, Fuchs PN, Turk DC: The biopsychosocial approach to chronic pain: Scientific advances and future directions. Psychol Bull 133:581, 2007

23. González-Roldan AM, Martínez-Jauand M, Muñoz-García MA, Sitges C, Cifre I, Montoya P: Temporal dissociation in the brain processing of pain and anger faces with different intensities of emotional expression. Pain 152:853-859, 2011

24. Guillory J, Chang P, Henderson CR Jr, Shengelia R, Lama S, Warmington M, Jowza M, Waldman S, Gay G, Reid MC: Piloting a text message-based social support intervention for patients with chronic pain: Establishing feasibility and preliminary efficacy. Clin J Pain 31:548-556, 2015

25. Holtzman S, DeLongis A: One day at a time: The impact of daily satisfaction with spouse responses on pain, negative affect and catastrophizing among individuals with rheumatoid arthritis. Pain 131:202-213, 2007

26. Ibinson JW, Vogt KM, Taylor KB, Dua SB, Becker CJ, Loggia M, Wasan AD: Optimizing and interpreting insular functional connectivity maps obtained during acute experimental pain: The effects of global signal and task paradigm regression. Brain Connect 5:649-657, 2015

27. Keefe FJ, Caldwell DS, Baucom D, Salley A, Robinson E, Timmons K, Beaupre P, Weisberg J, Helms M: Spouse-assisted coping skills training in the management of knee

pain in osteoarthritis: Long-term followup results. Arthritis Rheum 12:101-111, 1999

28. Kleck RE, Vaughan RC, Cartwright-Smith J, Vaughan KB, Colby CZ, Lanzetta JT: Effects of being observed on expressive, subjective, and physiological responses to painful stimuli. J Pers Soc Psychol 34:1211, 1976

29. Knyazev G, Slobodskoj-Plusnin J, Bocharov A: Eventrelated delta and theta synchronization during explicit and implicit emotion processing. Neuroscience 164:1588-1600, 2009

30. Korhonen RJ, Hernandez-Pavon JC, Metsomaa J, Mäki H, Ilmoniemi RJ, Sarvas J: Removal of large muscle artifacts from transcranial magnetic stimulation-evoked EEG by independent component analysis. Med Biol Eng Comput 49: 397-407, 2011

31. Krahé C, Springer A, Weinman JA, Fotopoulou A: The social modulation of pain: Others as predictive signals of salience–a systematic review. Front Hum Neurosci 7:386, 2013

32. Lapate RC, Lee H, Salomons TV, van Reekum CM, Greischar LL, Davidson RJ: Amygdalar function reflects common individual differences in emotion and pain regulation success. J Cog Neurosci 24:148-158, 2012

33. Lee JE, Kahana B, Kahana E: Social support and cognitive functioning as resources for elderly persons with chronic arthritis pain. Aging Ment Health 20:370-379, 2016

34. Logier R, Jeanne M, Tavernier B: Pain/analgesia evaluation using heart rate variability analysis. Conf Proc IEEE Eng Med Biol Soc 4303-4306, 2006

35. Longe S, Wise R, Bantick S, Lloyd D, Johansen-Berg H, McGlone F, Tracey I: Counter-stimulatory effects on pain perception and processing are significantly altered by attention: An fMRI study. Neuroreport 12:2021-2025, 2001

36. López-Martínez AE, Esteve-Zarazaga R, Ramírez-Maestre C: Perceived social support and coping responses are independent variables explaining pain adjustment among chronic pain patients. J Pain 9:373-379, 2008

37. Mager DE, Wan R, Brown M, Cheng A, Wareski P, Abernethy DR, Mattson MP: Caloric restriction and intermittent fasting alter spectral measures of heart rate and blood pressure variability in rats. FASEB J 20:631-637, 2006

38. Maris E, Oostenveld R: Nonparametric statistical testing of EEG-and MEG-data. J Neurosci Methods 164:177-190, 2007

39. Master SL, Eisenberger NI, Taylor SE, Naliboff BD, Shirinyan D, Lieberman MD: A picture's worth partner photographs reduce experimentally induced pain. Psychol Sci 20: 1316-1318, 2009

40. Melzack R, Casey KL: Sensory, motivational and central control determinants of pain, in Kenshalo DR (ed): The Skin Senses. Springfield, C C Thomas, 1968, pp 423-443

41. Mobbs D, Marchant JL, Hassabis D, Seymour B, Tan G, Gray M, Petrovic P, Dolan RJ, Frith CD: From threat to fear: The neural organization of defensive fear systems in humans. J Neurosci 29:12236-12243, 2009

42. Moont R, Crispel Y, Lev R, Pud D, Yarnitsky D: Temporal changes in cortical activation during conditioned pain modulation (CPM), a LORETA study. Pain 152:1469-1477, 2011

10 The Journal of Pain

43. Mueller EM, Panitz C, Hermann C, Pizzagalli DA: Prefrontal oscillations during recall of conditioned and extinguished fear in humans. J Neurosci 34:7059-7066, 2014

44. Mulert C, Jäger L, Schmitt R, Bussfeld P, Pogarell O, Möller HJ, Juckel G, Hegerl U: Integration of fMRI and simultaneous EEG: Towards a comprehensive understanding of localization and time-course of brain activity in target detection. Neuroimage 22:83-94, 2004

45. Oostenveld R, Fries P, Maris E, Schoffelen JM: FieldTrip: Open source software for advanced analysis of MEG, EEG, and invasive electrophysiological data. Comput Intell Neurosci 156869:2011, 2011

46. Pan J, Tompkins WJ: A real-time QRS detection algorithm. IEEE Trans Biomed Eng 230-236, 1985

47. Pascual-Marqui RD: Standardized low-resolution brain electromagnetic tomography (sLORETA): technical details. Methods Find Exp Clin Pharmacol 24:5-12, 2002

48. Pizzagalli DA, Greischar LL, Davidson RJ: Spatio-temporal dynamics of brain mechanisms in aversive classical conditioning: High-density event-related potential and brain electrical tomography analyses. Neuropsychologia 41:184-194, 2003

49. Platow MJ, Voudouris NJ, Gilford N, Jamieson R, Najdovski L, Papaleo N, Pollard C, Terry L: In-group reassurance in a pain setting produces lower levels of physiological arousal: Direct support for a self-categorization analysis of social influence. Eur J Soc Psychol 37:649-660, 2007

50. Reicherts P, Wiemer J, Gerdes A, Schulz SM, Pauli P, Wieser MJ: Anxious anticipation and pain: The influence of instructed vs conditioned threat on pain. Soc Cogn Affect Neurosci 12:544-554, 2017

51. Roberts MH, Klatzkin RR, Mechlin B: Social support attenuates physiological stress responses and experimental pain sensitivity to cold pressor pain. Ann Behav Med 49:557-569, 2015

52. Salomons TV, Johnstone T, Backonja MM, Davidson RJ: Perceived controllability modulates the neural response to pain. J Neurosci 24:7199-7203, 2004 53. Sheehan D, Lecrubier Y: (eds): MINI SCREEN 5.0. 0. Tampa, University of South Florida, 2006

54. Straube T, Schmidt S, Weiss T, Mentzel HJ, Miltner WH: Dynamic activation of the anterior cingulate cortex during anticipatory anxiety. Neuroimage 44:975-981, 2009

55. Sun J, Sun B, Wang B, Gong H: The processing bias for threatening cues revealed by event-related potential and event-related oscillation analyses. Neuroscience 203:91-98, 2012

56. Uchino BN: Social support and health: A review of physiological processes potentially underlying links to disease outcomes. J Behav Med 29:377-387, 2006

57. Uchino BN: Understanding the links between social support and physical health: A life-span perspective with emphasis on the separability of perceived and received support. Perspect Psychol Sci 4:236-255, 2009

58. van Noordt SJ, White LO, Wu J, Mayes LC, Crowley MJ: Social exclusion modulates event-related frontal theta and tracks ostracism distress in children. Neuroimage 118:248-255, 2015

59. Vlaeyen JW, Hanssen M, Goubert L, Vervoort T, Peters M, van Breukelen G, Sullivan MJ, Morley S: Threat of pain influences social context effects on verbal pain report and facial expression. Behav Res Ther 47:774-782, 2009

60. Waltz M, Kriegel W, Bosch PV: The social environment and health in rheumatoid arthritis: Marital quality predicts individual variability in pain severity. Arthritis Rheum 11: 356-374, 1998

61. Weinberger M, Tierney WM, Booher P, Hiner SL: Social support, stress and functional status in patients with osteo-arthritis. Soc Sci Med 30:503-508, 1990

62. Younger J, Aron A, Parke S, Chatterjee N, Mackey S: Viewing pictures of a romantic partner reduces experimental pain: Involvement of neural reward systems. PLoS One 5:e13309, 2010

63. Zhang D, Wang L, Luo Y, Luo Y: Individual differences in detecting rapidly presented fearful faces. PLoS One 7:e49517, 2012

Note:

1. There was an error in the text. 'od' should be 'of'. (Page 58, Line 40)

CHAPTER FIVE

Neural Oscillations and Parasympathetic Activity

Manuscript

Che X., Cash R., Fitzgerald P., & Fitzgibbon B.M. Distinct neural dynamics and autonomic activity in response to viewing a romantic other compared to a stranger during pain: An EEG and HRV investigation. Paper under review.

Preamble to empirical paper

Chapter 3 and preliminary fMRI evidence suggest that social support may reduce pain by priming feelings of attachment and/or reward (Eisenberger et al., 2011; Younger et al., 2010). Social support may also modulate the processing of nociceptive information (e.g. ACC, insular cortex) as revealed by fMRI evidence (Eisenberger et al., 2011). However, fMRI has limited temporal resolution and EEG may reveal new information on how social support may modify the gating and integration of nociceptive information. It is also unknown whether social support can act through the parasympathetic system to affect pain, in addition to the autonomous evidence.

This chapter aims to investigate the role of social support in modulating pain perception as well as associated neural dynamics and parasympathetic activity. A tonic pain protocol is introduced which induces prolonged pain during which social support is manipulated by visual representation of an intimate other. Continuous EEG and ECG are recorded during pain delivery to evaluate the dynamics in neural oscillations and HF-HRV respectively. This chapter extends the literature by demonstrating the role of social support in modulating the gating and integration of nociceptive information as well as in priming regulatory control over bodily arousal to pain.

The organisation of this chapter is in keeping with the requirements for submission to the journal. However, where appropriate, minor formatting changes have been made to improve consistency with other sections of this thesis. The study and related data presented here were from the first session of a three-session data collection protocol presented in chapter six. In this chapter, EEG and ECG data are exclusively looked at to investigate the role of social support in modulating pain perception as well as associated neural dynamics and parasympathetic activity.

Title Page

Distinct neural dynamics and autonomic activity in response to viewing a romantic other

compared to a stranger during pain: An EEG and HRV investigation

Xianwei Che^{* 1}, Robin Cash¹, Paul B Fitzgerald^{1, 2}, Bernadette M. Fitzgibbon¹

¹Monash Alfred Psychiatry Research Centre (MAPrc), The Alfred and Central Clinical School, Monash

University, Melbourne, Australia

²Epworth Centre for Innovation in Mental Health, Epworth, VIC, Australia

*Address correspondence to:

Xianwei Che

Monash Alfred Psychiatry Research Centre (MAPrc)

Monash University

Level 4, 607 St Kilda Road, Melbourne VIC 3004, Australia

Fax:

Email:

Original	lartic	le
0		

Conflicts of Interest and Source of Funding:

This work was supported by the China Scholarship Council (201506990016), the NHMRC Practitioner Fellowship (606907), and the NHMRC Early Career Fellowship. The authors declare no conflicts of interest.

Significance:

Social support may modulate the gating and integration of nociceptive information.

Abstract

Background: This study was designed to investigate the effects of social support through the visual representation of an intimate other on the subjective experience of pain and measured associated neural and autonomic changes.

Methods: Healthy participants (N = 23) were subjected to cold pain during which they viewed an image of their romantic partner or a stranger. Neural activity and heart rate variability (HRV) were evaluated using electroencephalogram (EEG) and electrocardiography (ECG) respectively. **Results**: Compared to the stranger image, viewing a romantic partner was associated with reduced pain intensity ratings and increased perceived support. We also identified distinct neural oscillations in the romantic partner relative to stranger condition, which is characterized by increased frontocentral alpha activity and decreased central gamma activity. The ECG data also demonstrated increased high-frequency HRV in the romantic partner relative to the stranger condition. **Conclusions**: Social support from viewing an intimate other is sufficient to reduce pain. This effect may be associated with social support modulating the gating and integration of nociceptive information. Our data further suggest that social support analgesia may also be associated with increased regulatory control over bodily arousal to pain.

Significance:

Social support may modulate the gating and integration of nociceptive information.

Keywords: romantic relationship; social support; pain; alpha oscillation; gamma oscillation; HRV

Introduction

The experience of pain can be reduced by social support (Brown et al., 2003; Master et al., 2009; Goldstein et al., 2016), which refers to received support from others and the perceived availability of supportive resources (Dunkel-Schetter and Bennett, 1990; Uchino et al., 2012). However, simply having supportive resources is not necessarily sufficient to reduce pain (Krahé et al., 2013). Indeed, a recent meta-analysis by our group has demonstrated that the physical presence of a significant may not in itself reduce pain (Che et al., 2018). Rather, social support tends to have the most effect on pain reduction when the pain recipient perceives the intentions of the social partner as helpful (e.g. view or touch a romantic partner) (Master et al., 2009; Younger et al., 2010; Eisenberger et al., 2011). However, the latter argument is only based on a limited number of studies, and more research is needed to better understand the importance of how social support is delivered, and by whom, in social support analgesia (Che et al., 2018).

There is also a lack of evidence on the mechanisms related to the influence of social support on pain. Painful stimuli carried by afferent C fibre and Aδ nociceptors terminate in the dorsal horn, with the information being further transmitted to the thalamus and higher-order brain regions (e.g. anterior cingulate cortex, ACC) (Craig, 2003). This spino-thalamo-cortical circuit also regulates the balance of the sympathetic ('SNS') and parasympathetic ('PNS') nervous system (Cortelli et al., 2013). In social support literature, one proposed mechanism is the capacity of social support to prime feelings of attachment which buffers pain-related responses (e.g. ACC) (Younger et al., 2010; Eisenberger et al., 2011). However, there is no evidence surrounding the dynamic neurophysiological changes related to the social modulation of pain. Research using electroencephalography (EEG) indicates that pain is encoded by suppressed alpha (8-12 Hz) and increased gamma (31-100 Hz) activity. These features are thought to underlie early nociceptive attention and the subsequent subjective experience of pain respectively (Peng et al., 2014; Nickel et al., 2017). At the peripheral level, social support is suggested to buffer sympathetic arousal to pain (Sambo et al., 2010; Roberts et al., 2015; Che et al., 2017). While studies have explored sympathetic arousal, the PNS response

involved in the regulatory control over arousal remains relatively unexplored (Thayer et al., 2009) and may therefore provide a means by which social support can increase physiological control over pain. Heart rate variability (HRV), especially high-frequency HRV (HF-HRV), is thought to reflect bodily control over sympathetic arousal (Thayer et al., 2012; Holzman and Bridgett, 2017).

In this investigation, we aimed to elucidate the effects of intimate social support on pain experience. Using EEG and HRV, we further assessed the neurophysiological and autonomic changes associated with the social modulation of pain. Healthy individuals were subjected to 3-minutes cold pain while viewing an image of a romantic partner or a stranger. This was done to coincide with the functional imaging evidence which also used the image-viewing protocol (Younger et al., 2010; Eisenberger et al., 2011). We hypothesized that a romantic partner image would result in decreased self-reported pain as well as decreased gamma oscillations, and increased alpha activity and HF-HRV. Together, this investigation will provide evidence of the effects of social support on self-report response to pain, as well as information on neural integration processes and bodily arousal that may be associated with these effects.

Methods

Participants

A group of healthy, pain-free, right-handed adults participated in this study (8 males and 15 females, age range: 21-36 years, Mean=26.26, SD=4.24). The sample size was comparable to studies which investigated the neural mechanisms associated with effects of social support on pain (Younger et al., 2010; Eisenberger et al., 2011). Exclusion criteria included use of psychoactive medication, or a history or current diagnosis of a psychiatric disorder, as assessed by the Mini International Neuropsychiatric Interview (MINI) (Sheehan and Lecrubier, 2001). Participation in the study required involvement in a relationship reported as romantic. Participation did not require a pre-requisite measure of relationship quality, although this information was recorded. All study participants provided informed consent and the experiment was approved by the Alfred Hospital and Monash

University Human Research and Ethics Committee. This study was conducted in accordance with the Declaration of Helsinki.

Experimental Design and Procedure

Participants recruited to this study underwent a three-session data collection protocol belonging to a more extensive experiment. The study presented here was from the first session only where EEG and ECG were recorded. Following consent, participants first completed questionnaires before set-up of EEG and ECG. Participants were asked to complete a number of questionnaires, this included the Dyadic Adjustment Scale (DAS) (Spanier, 1976) which is the only one included in this investigation. During 3 minute pain stimulation trials, participants viewed an image of either a stranger or their romantic partner, and rated pain at 20 second intervals.

Dyadic Adjustment in the Relationship

The Dyadic Adjustment Scale (DAS) was the only questionnaire used in this study (Spanier, 1976). DAS is a widely used measure of quality of romantic relationships (Coan et al., 2006). The 32item DAS consists of four correlated subscales (i.e. Dyadic Consensus, Dyadic Satisfaction, Dyadic Cohesion, and Affectional Expression). The total DAS score is recommended which has a theoretical range of 0 to 151.

Attachment Image

Participants were asked to take a digital image of their romantic partner, consistent with a template image (i.e. the gender matched stranger image), and provide it for the study. Specifically, the image should show a natural smile that comforts them, include only the head and shoulders of the romantic partner and be taken against a plain white background. Images were then edited by the researchers (GNU Image Manipulation Program, GIMP, V2.8,) to have the fixed size (17.5 cm × 17.5 cm), content (from the top of head to shoulders), and matched for luminance using the "auto adjust color levels" function (Parsons et al., 2013). Two images, one male and one female in early adulthood, served as the stranger images.

Tonic Pain Protocol

Prior to, and between experimental trials, participants were required to insert their dominant hand into a bucket of warm water (40 °C \pm 0.2 °C) for 2 minutes and relax for another 5 minutes. This approach was used to adjust and control baseline hand temperature (Hadjileontiadis, 2015). Next, participants were asked to hold a bottle filled with ice with the dominant hand for 3 minutes (i.e. a single trial). Participants were told to press a button with the non-dominant hand in the keypad to start the trial. The button press started the presentation of an image (Presentation, Version 17.0, Neurobehavioral Systems, Berkeley, CA), which was either their romantic partner or a stranger matched to the partner's gender (Eisenberger et al., 2011). These two conditions were performed separately in a single testing session, and the order of conditions were pseudorandomized and counterbalanced across participants. The 3 minute trail was divided into 20second blocks (i.e. total 9 blocks per trial) during which the participants viewed the image for 16 seconds and then rated pain intensity within 4 seconds. The participants were asked to rate "pain intensity at the moment" on a scale of 0-10 (0 = no pain; 10 = worst pain imaginable), by pressing the corresponding button on a keypad. At the end of the trial, participants were asked to rate their overall perceived support from the image by pressing the corresponding button (0 = not at all; 10 =extremely high).

Pain Stimulation

In accordance with the work by Hadjileontiadis (2015), participants were asked to hold a 0.5L plastic bottle with iced water (-1 °C \pm 0.2 °C) with the dominant hand for 3 minutes (Fig. 1a). A pillow was provided on which participants rested their hands to avoid hand movement. Participants were asked to put the volar surface of his/her dominant hand on the surface of the bottle, and not to squeeze or avoid it, to minimize the variability of touching. Different ice bottles were used in different experimental trials for consistency.

EEG Recording

EEG recordings took place in a temperature-controlled, sound-attenuated, and electrically shielded room. Participants were seated in a slightly reclined chair with their faces approximately

0.5m from the computer monitor. A 64 channel Quickcap (NeuroScan Inc., Australia) was used to record continuous EEG with CPZ as the reference electrode. Two electrodes were attached above and below the left eye respectively to monitor Vertical electrooculogram (EOG) activity. Horizontal EOG activity was monitored with electrodes located at the outer canthus of both eyes. Data were sampled at 1000 Hz with impedances below 5 k Ω throughout the testing.

ECG Recording to Assess HRV

Continuous ECG was recorded simultaneously with EEG recording (Chouchoul et al., 2011; Ahn et al., 2016), using the same NeuroScan Acquire system. Two Ag/AgCl electrodes were attached to the left and right wrist of the participant respectively, which shared the reference electrode (i.e. CPZ) with the EEG recording. Data were also sampled at 1000 Hz.

Data Analysis

Offline EEG data were preprocessed using custom-written scripts that implement functions from EEGLAB (version 13.6.5b) (Delorme and Makeig, 2004) running under Matlab R2017b (The MathWorks, Inc.). Data from malfunctioning channels (no more than three) were visually inspected and removed. This number never exceeded three electrodes. Butterworth filters (band-pass: 0.5-100Hz; band-stop notch filter: 48-52 Hz) were then applied to the data (Selesnick and Burrus, 1998). Continuous data were segmented to retain only the image viewing stage (i.e. 16 seconds in each block, in total 9 in each trial). Data were further segmented into 1 second non-overlapping epochs in order to examine the dynamic changes in oscillatory activity (Peng et al., 2014; Schulz et al., 2015). Segmented data were re-referenced to the average signal, and the fast independent component analysis algorithm (FastICA) was employed to remove stereotyped artefacts including eye blinks, lateral eye movements, muscle, and line noise. (Korhonen et al., 2011). Stereotyped artefacts were identified by visual inspection of the temporal and spatial representation of the independent components. Data was interpolated for excluded channels, and epochs were inspected again to remove any anomalous activity.

Running title: social support modulation of pain

EEG frequency representations were calculated with Multitaper Method Fast Fourier Transform ('mtmfft'), as implemented in FieldTrip toolbox, in the range of 0-100Hz (Oostenveld et al., 2011). Power spectra were calculated for each epoch, and averaged across epochs for each minute (i.e. first, second, third) in each condition. This was done to investigate the dynamic changes in pain-related oscillations (Peng et al., 2014). Power spectra were also averaged across epochs for the entire 3 minute of pain induction in a separate analysis.

For the ECG data, inter-beat-interval (IBI) series were derived by using the Pan-Tompkins algorithm which identifies the peak of the R wave as the fiducial point (Pan and Tompkins, 1985). Artefacts were checked visually and edited as necessary according to published guidelines (Berntson et al., 1997). Thereafter, IBI series were linearly interpolated to obtain evenly sampled signals and detrended using a highpass filter with the cutoff frequency of 0.02 Hz (Terkelsen et al., 2005; Peng et al., 2015). In order to investigate the dynamic changes in HF-HRV and coincide with the analysis of EEG, IBI series for each minute were subjected to power spectral analysis using discrete Fourier transformation (DeBoer et al., 1987; Akselrod, 1988). HF-HRV was expressed as the relative value of high frequency component (0.15-0.4 Hz) in proportion to the total power minus the very low frequency component (0.0-04 Hz) (Cardiology, 1996). Relative values of HF-HRV are designed to emphasize the controlled and balanced behavior of the sympathetic and parasympathetic branch of the autonomic nervous system (Cardiology, 1996). This measure was also used in a recent study which examined the dynamics of HF-HRV in tonic pain (Peng et al., 2015).

Statistical Analyses

In order to examine social support and time effects on pain, a repeated measures two-way ANOVA was performed in SPSS (version 23; IBM Corp, Armonk, NY). Condition (partner, stranger) and time (9 time points at 20 seconds interval) were specified as the two repeated measures factors. Post-hoc pairwise comparisons were conducted to further explore the significant main and interaction effects, with a Bonferroni correction and α -level set to 0.05.

Running title: social support modulation of pain

We also evaluated the difference between conditions of overall perceived support using a paired sample T-test. Bivariate correlation analyses were further performed to investigate the associations between pain intensity changes and perceived support (and relationship quality). Specifically, changes in pain intensity were calculated using area under the curve (AUC) of pain intensity with the linear trapezoidal rule. The AUC approach was employed as it provides a summary measure of the pain dynamics across a specific time window in which overall perceived support (i.e. 3-minute) or neurophysiological (i.e. 1-minute or 3-minute window) data were analyzed.

For EEG data, non-parametric cluster-based permutation tests were performed in order to identify changes in neural oscillatory activity related to the influence of social support on pain. This approach provides a straightforward way to solve the problem of multiple comparisons (Maris and Oostenveld, 2007). This method was applied to the 1 second time windows across all channels. An observed test statistics value was considered in the cluster permutation if it was below the threshold of 0.05 in at least 2 of the neighboring channels (Oostenveld et al., 2011). We performed 5000 iterations of trial randomization for generating the permutation distribution at a given frequency band. A threshold of 0.025 (two-tailed) was then adopted to evaluate the electrodes that exhibit significant difference in power. According to our a priori hypothesis, statistical analyses were firstly carried out in delta (1-3 Hz), alpha (8-12 Hz), and gamma (31-100Hz) band. Additional analyses were performed separately in other frequency bands (i.e. theta: 4-7 Hz; beta: 13-30 Hz).

HF-HRV data were examined to investigate the peripheral changes in the social support influence over pain. In a repeated measures two-way ANOVA, condition (partner, stranger) and time (3 time points at 1 minute interval) were specified as the two repeated measures factors. Post-hoc pairwise comparisons were performed with Bonferroni correction, and the α -level was set to 0.05.

A series of correlation analyses were performed to examine the associations between changes in neurophysiological responses and pain intensity. We also investigated the associations between neurophysiological changes and perceived support (and relationship quality).

Results

Pain Intensity Ratings

As shown in the Fig. 1b, pain intensity demonstrated a pattern of ascending, stabilization, and a desensitization response especially in the control condition. These patterns could be roughly delineated by the 1st, 2nd and 3rd minutes. The ANOVA revealed a significant main effect of condition for pain intensity ($F_{1,22} = 10.71$, P = 0.003, $\eta_p^2 = 0.33$). Post-hoc pairwise comparison indicated that pain intensity was lower ($P_{Bonf} = 0.003$) in the romantic partner compared to stranger condition (Fig. 1b). The ANOVA also revealed a main effect of time ($F_{8,176} = 5.21$, P = 0.016, $\eta_p^2 = 0.19$). Post-hoc pairwise comparisons showed that, across conditions, pain intensity kept increasing ($P_{Bonf} < 0.05$) till the end of the second minute (20-120 second) compared to the start of pain induction.

-----Please insert Fig. 1 here-----

Overall Perceived Support and Relationship Quality

A paired sample T-test confirmed that overall perceived support was higher in the romantic partner compared to the stranger condition ($t_{22} = 9.22$, P = 0.0001) (Fig. 1c). Further analysis showed that higher perceived support was associated with greater pain reduction throughout the course of pain in the romantic partner relative to stranger condition (P = 0.006) (Fig. 1d).

Mean relationship quality scores were 116.68 (SD = 10.28). We did not observe an association between partner relationship quality and pain intensity ratings (P > 0.05).

EEG Analysis

In the first minute of pain induction, no differences were found between the romantic partner and stranger condition.

In the second minute, when pain sensation has stabilized, cluster-based permutation testing revealed a significant increase of alpha power in the romantic partner relative to stranger condition

(Fig. 2a). The topography of this power increase had a fronto-central, left-lateralized distribution (significant at F5, FC3, FC5 and C3) (Fig. 2b). Alpha power changes did not correlate with the changes in pain intensity or perceived support or relationship quality ($P_s > 0.05$). No difference was found in any other frequency bands.

-----Please insert Fig. 2 here-----

In the third minute of pain induction, characterized by apparent desensitization, cluster analysis identified a significant decrease in gamma power in the romantic partner relative to stranger condition (Fig. 2c). This power decrease was located over the central electrodes (significant at CZ, C2, CP1, and CP2) (Fig. 2d). Separate analyses confirmed that this result was not influenced by the 50 Hz notch filter. The decrease in gamma power was associated with greater pain reduction during the third minute of pain in the romantic partner relative to stranger condition (P = 0.04) (Fig. 2e). Gamma changes were not associated with perceived support or relationship quality ($P_s > 0.05$)

The cluster analysis also identified two clusters in the delta band that significantly differed between the romantic partner and stranger condition during the third minute of pain. In one cluster, partner condition showed decreased delta power over the frontal electrodes (significant at AF3, AF4, F3, F1, FZ, F2, and FCZ) (Fig. 3a, 3b). Romantic partner condition also demonstrated reduced delta power over the parietal electrodes (significant at CP5, CP3, CP1, CP2, CP4, CP6, P5, P3, P1, PZ, P2, P6, PO3, and POZ) (Fig. 3c, 3d). Changes in frontal or parietal delta power across conditions were not associated with pain intensity changes or perceived support or relationship quality (P_s > 0.05). No further difference was found in other frequency bands.

-----Please insert Fig. 3 here-----

The supplementary cluster analysis revealed no difference in any frequency bands when data were averaged across the entire period of pain induction.

High-frequency Heart Rate Variability

A two-way ANOVA revealed a condition × time interaction effect in HF-HRV ($F_{2,44} = 3.72$, P = 0.04, $\eta_p^2 = 0.15$). To investigate this interaction effect, two one-way ANOVAs were performed across time windows within each condition, and paired T-tests were conducted across conditions for each time window. A significant main effect was found in the romantic partner condition ($F_{2,44} = 4.75$, P = 0.02, $\eta_p^2 = 0.18$), with post-hoc pairwise comparisons showing that HF-HRV was significantly higher in the second than the first minute of pain induction ($P_{Bonf} = 0.03$). No significant main effect was observed in the stranger condition (P > 0.05). Across conditions, paired T-tests showed a significantly higher HF-HRV in the romantic partner relative to the stranger condition only in the second minute of pain induction ($P_{Bonf} = 0.03$) (Fig. 4). Changes in HF-HRV were not correlated with perceived support or relationship quality ($P_s > 0.05$).

-----Please insert Fig. 4 here------

Discussion

The present study was designed to characterize the neurophysiological and autonomic mechanisms associated with the effects of social support on pain experience. Consistent with previous evidence (Master et al., 2009; Eisenberger et al., 2011), our behavioral results indicated that pain perception was reduced when viewing an image of a romantic partner. The benefits of social support on pain perception were associated with distinct oscillatory responses between conditions, with the romantic partner image demonstrating higher alpha activity during the apparent pain stabilization stage (2nd minute) followed by lower gamma and delta oscillations in the late desensitization period (3rd minute) of pain induction. Interestingly, a positive relationship was

observed between decreased gamma power and pain reduction in the romantic partner relative to stranger image. Our results also indicated that romantic partner image was associated with higher HF-HRV during tonic pain. Together, this work provides several lines of evidence demonstrating that social support modulates the dynamic neural, autonomic and subjective response to pain.

Initial evidence suggests that social support by viewing or touching an intimate other may have an analgesic influence (Krahé et al., 2013; Che et al., 2018). This argument is based on limited evidence (Master et al., 2009; Younger et al., 2010; Eisenberger et al., 2011) which needs to be further examined. We presented an experimental pain protocol, which is seemingly characterized by an ascending, stabilization, and desensitization phase (Fig. 1b). Moreover, pain was reduced in the social support condition, and this effect was most prominent after the initial increase in pain. Our findings provide further evidence of the analgesic influence of intimate social support, as well as a unique contribution of social support during the ongoing induction of pain.

As expected, the ratings of perceived support were higher in the romantic partner relative to stranger condition (Fig. 1c). Furthermore, pain reduction was greatest in those individuals who reported the greatest level of perceived support (Fig. 1d), further substantiating the moderating influence of social support on pain experience. Surprisingly, relationship quality was found to have no impact on pain perception or on the neurophysiological changes in this study. It may be that perceived support provides a metric that is more directly relevant to moderating pain experience while relationship quality encompasses a more complex constellation of features (e.g. dyadic satisfaction and affectional expression) (Spanier, 1976). Nonetheless, the result was somewhat unexpected as relationship quality (e.g. dyadic satisfaction) was found to buffer brain activations (e.g. insula, hypothalamus) to pain-related threat (Coan et al., 2006). It is possible that social support variants (e.g. hand-holding, visual presentation) may modulate the influence of relationship quality on the neurobiological activation to pain.

We explored the neural changes associated with and potentially mediating the influence of social support on tonic pain. Following an initial increase in perceived pain, the stabilization phase

(2nd minute) displayed significantly higher frontocentral alpha power in the romantic partner compared to stranger condition (Fig. 2a, 2b). Given that pain has been shown to reduce alpha power (Huber et al., 2006; Dowman et al., 2008; Nir et al., 2012; Shao et al., 2012; Gram et al., 2015; Nickel et al., 2017), the increased alpha power in the romantic partner condition is consistent with a reduction in subjective pain at the cortical level. Moreover, pain is proposed to suppress alpha power to open the 'gate' and permit the relevant nociceptive inputs to be actively processed in highorder brain areas (Ploner et al., 2006; Jensen and Mazaheri, 2010; Hauck et al., 2015). Decreased alpha is therefore suggested as a mechanism of attention driven by nociceptive inputs (Chang et al., 2002; Ohara et al., 2004; Hauck et al., 2015). Indeed, contralateral alpha power was found to be negatively associated with the stimulus intensity of pain which directly modulated the attention level (Nickel et al., 2017). Moreover, cognitive distraction restored the alpha suppression to pain (Peng et al., 2014). Conversely, increased alpha activity is commonly associated with cortical "deactivation" or inhibition and is sometimes referred to as an 'idling' state (Jensen and Mazaheri, 2010; Cash et al., 2017). The present results therefore suggest that alpha power may be also involved in social support modulating the gating of nociceptive transmission.

In addition, the romantic partner condition was associated with reduced central gamma power during the late stage of pain induction (Fig. 2c, 2d). Our data demonstrated that decreased gamma power was correlated with greater pain reduction (Fig 2e). This is in line with previous research which indicated that pain enhances gamma oscillatory activity, primarily over frontal and central regions (Schulz et al., 2015; Nickel et al., 2017). Moreover, these changes were found to be independent of stimulus saliency and the site of noxious stimulation (Schulz et al., 2015; Nickel et al., 2017). Increased gamma activity is therefore thought to represent the integration of nociceptive inputs with higher-order cognitive and affective functions, thus mediating the subjective experience of pain (Zhang et al., 2012; Peng et al., 2014; Hauck et al., 2015). Taken together, higher alpha power in the social support condition may reflect increased gating of nociceptive information, while

reduced central gamma activity may indicate reduced transmission and integration of nociceptive information for cognitive processing, thus modulating the subjective experience of pain.

Several studies have also demonstrated that pain increases delta activity (Le Pera et al., 2000; Chang et al., 2002; Huber et al., 2006; Hauck et al., 2015; Ploner et al., 2017). Similarly, we observed decreased delta activity in the frontal and parietal cortices in the romantic partner relative to the stranger condition during pain induction (Fig 3a-d). Interestingly, one study found that frontal delta power was increased during attachment image-viewing, but this was in the absence of pain induction (Başar et al., 2008). It is possible that the delta increase associated with viewing a romantic partner is offset by the dramatic delta decrease related to lower level of pain. Our results are therefore not necessarily to rule out the possibility of perceived attachment (i.e. increased delta) in the social modulation of tonic pain, but delta changes may rather reflect a combination of both perceived attachment and pain. Future research may therefore wish to isolate the contributions of perceived attachment and pain to delta changes by including more control conditions (e.g. supportno pain condition).

We also observed a more rapid and larger increase in HF-HRV in the romantic partner condition beyond the cortical activation (Fig. 4). During exposure to pain, HF-HRV returns to baseline after an initial decrease, which suggests regulatory control over body arousal (Treister et al., 2012; Peng et al., 2015). Increased HF-HRV in our data is potentially indicative of a higher level of physiological control in response to pain. Our findings therefore extend the evidence of social support-assisted management of sympathetic arousal (Sambo et al., 2010; Roberts et al., 2015), by directly providing evidence of the parasympathetic activity that is associated with the physiological control over pain. Moreover, higher HF-HRV in social support condition is consistent with the increased alpha and decreased gamma activity. Painful stimuli can disrupt the homeostasis of human body mediated by the spino-thalamo-cortical circuit (Craig, 2003), which is associated with increased sympathetic and suppressed parasympathetic activity (Cortelli et al., 2013). Our results therefore provide comprehensive evidence that social support can modulate not only the gating and

integration of nociceptive information in the cortical level, but the regulatory control over bodily arousal in the peripheral level.

Experimental pain protocols, as employed in this study, may improve our understanding of the role of social factors in pain experience, a key component in the biopsychosocial model which views pain as the result of the dynamic interactions among physiologic, psychological, and social factors (Gatchel, 2004; Gatchel et al., 2007). Our findings suggest the impact of a social context on the neurobiological processing of pain. Our data may also provide information on how the types of interactions within relationships may impact upon pain, although more work is needed in this area. The intimacy model, for example, argues that verbal communication between couples may serve to enhance intimacy and adaptive coping responses to pain (Cano and Williams, 2010). Using a nonverbal visual presentation of social support, our results suggest the benefits on decreasing pain and possibly building intimacy (as indicated by increased perceived support). Beyond theoretical implications, this study further informs the therapeutic benefits of social support in the management of pain. Shaygan and colleagues (Shaygan et al., 2016) demonstrated that viewing images of a significant other mitigated pain in chronic pain conditions. Although conducted in a group of healthy people, our findings add to this work suggesting that the analgesic effect seen in Shaygan et al study (Shaygan et al., 2016) could be associated with the modulation of pain processing in the cortical and peripheral level. These results may therefore inform to optimize the partner-assisted pain management strategies which lack empirical evidence surrounding the contexts and mechanisms of the pain-relieving effect of social support (Keefe et al., 1996, 1999; Abbasi et al., 2012).

There are several limitations of the study. We did not require a certain level of quality in the romantic relationship of our participants. This may limit the replications of previous findings among early stage relationships (Younger et al., 2010) or in studies which only included relationships where there was a high level of satisfaction (Coan et al., 2006). Nonetheless, this should have been at least partially captured by our measure of relationship quality and therefore can be included when

comparing studies of this kind. We also did not control for attraction arousal of the two stranger images, which should be recorded in future research. Our study design also induced pain to the dominant hand in this study as opposed to the non-dominant hand in some studies (Younger et al., 2010; Eisenberger et al., 2011), which may further limit comparability to other work. It is also important to note the limitation of translating our findings to clinical groups as we presented findings in healthy participants. As the central nervous system undergoes significant change in chronic pain (Apkarian et al., 2011), neural responses to social support may have a differential effect as seen in this study. Finally, while we presented a sample size comparable to the literature on this topic, it was relatively small with a restricted age range, and we are therefore not able to compare age or sex differences in the analgesic effect of intimate relationship which has been mainly observed in females (Master et al., 2009; Eisenberger et al., 2011).

Future directions to investigate the mechanisms related to the influence of social support on pain could expand on neurophysiological and peripheral responses as were investigated in this study. This includes the assessment of the neuroendocrine responses which may reveal valuable information on the potential mechanisms associated with social support and pain. Social interaction with an intimate other may result in the release of oxytocin (Holt-Lunstad et al., 2008) which has analgesic effects (Paloyelis et al., 2016), given recent evidence that oxytocin may enhance the analgesic influence of social support (Kreuder et al., 2018). Moreover, other candidate hormones may be also considered as of their involvement in the processing of social intimacy and pain (e.g. βendorphin (Keverne et al., 1989; Pearce et al., 2017), glucocorticoids (Wittig et al., 2016)). In addition, other psychological processes may be also related to the experience of social support and therefore modulate pain experience, for example, the feeling of attachment and attraction (Eisenberger et al., 2011), self-esteem and perceived competence (Che et al., 2018), which were not included in this study.

To conclude, our findings suggest that social support, as provided by visual representation of a romantic partner, is sufficient to significantly reduce neural, autonomic and subjective dimensions

of pain experience. In the context of previous findings, our data suggest that this reduction of subjective pain experience may be associated with the gating and reduced integration of neural signals associated with pain. Our data further suggest that this reduction may also be associated with increased regulatory control over bodily arousal to pain. Together these findings enhance our understanding of the neural and autonomic factors associated with the analgesic effects of social support.
Acknowledgements:

We wish to thank all the participants.

Author Contributions:

XC, RC, and BF contributed to the experimental design, data collection and analysis, and manuscript

preparation. PF contributed to experimental design and manuscript preparation.

References:

- Abbasi, M., Dehghani, M., Keefe, F., Jafari, H., Behtash, H., & Shams, J. (2012). Spouse-assisted training in pain coping skills and the outcome of multidisciplinary pain management for chronic low back pain treatment: A 1-year randomized controlled trial. *European Journal of Pain, 16*, 1033-1043.
- Ahn, S., Nguyen, T., Jang, H., Kim, J. G., & Jun, S. C. (2016). Exploring neuro-physiological correlates of drivers' mental fatigue caused by sleep deprivation using simultaneous EEG, ECG, and fNIRS data. *Frontiers in Human Neuroscience, 10*, e219.
- Akselrod, S. (1988). Spectral analysis of fluctuations in cardiovascular parameters: a quantitative tool for the investigation of autonomic control. *Trends in Pharmacological Sciences, 9*, 6-9.
- Apkarian, A. V., Hashmi, J. A., & Baliki, M. N. (2011). Pain and the brain: specificity and plasticity of the brain in clinical chronic pain. *Pain*, *152*, S49-S64.
- Başar, E., Schmiedt-Fehr, C., Öniz, A., & Başar-Eroğlu, C. (2008). Brain oscillations evoked by the face of a loved person. *Brain Research*, *1214*, 105-115.
- Berntson, G. G., Thomas Bigger, J., Eckberg, D. L., Grossman, P., Kaufmann, P. G., Malik, M., . . . Stone, P. H. (1997). Heart rate variability: origins, methods, and interpretive caveats. *Psychophysiology*, *34*, 623-648.
- Brown, J. L., Sheffield, D., Leary, M. R., & Robinson, M. E. (2003). Social support and experimental pain. *Psychosomatic Medicine*, *65*, 276-283.
- Cano, A., & Williams, A. C. d. C. (2010). Social interaction in pain: Reinforcing pain behaviors or building intimacy? *Pain, 149*, 9-11.
- Cardiology, T. F. o. t. E. S. o. (1996). Heart rate variability: standards of measurement, physiological interpretation, and clinical use. *Circulation*, *93*, 1043-1065.
- Cash, R. F., Noda, Y., Zomorrodi, R., Radhu, N., Farzan, F., Rajji, T. K., . . . Blumberger, D. M. (2017). Characterization of Glutamatergic and GABA A-Mediated Neurotransmission in Motor and

Dorsolateral Prefrontal Cortex Using Paired-Pulse TMS–EEG. *Neuropsychopharmacology, 42,* 502-511.

- Chang, P. F., Arendt-Nielsen, L., & Chen, A. C. (2002). Dynamic changes and spatial correlation of EEG activities during cold pressor test in man. *Brain Research Bulletin, 57*, 667-675.
- Che, X., Cash, R., Chung, S. W., Fitzgerald, P., & Fitzgibbon, B. M. (2018). Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis. *Neuroscience & Biobehavioral Reviews*, *92*, 437-452.
- Che, X., Cash, R., Fitzgerald, P., & Fitzgibbon, B. M. (2017). The Social Regulation of Pain: Autonomic and Neurophysiological Changes Associated with Perceived Threat. *The Journal of Pain*, *19*, 496-505.
- Che, X., Cash, R., Ng, S. K., Fitzgerald, P., & Fitzgibbon, B. M. (2018). A Systematic Review of the Processes Underlying the Main and the Buffering Effect of Social Support on the Experience of Pain. *The Clinical Journal of Pain*, *34*, 1061-1076.
- Chouchoul, F., Pichotl, V., Perchetl, C., Legrainl, V., Garcia-Larreal, L., Rochel, F., & Bastujil, H. (2011). Autonomic pain responses during sleep: a study of heart rate variability. *European Journal of Pain, 15*, 554-560.
- Coan, J. A., Schaefer, H. S., & Davidson, R. J. (2006). Lending a hand social regulation of the neural response to threat. *Psychological Science*, *17*, 1032-1039.
- Cortelli, P., Giannini, G., Favoni, V., Cevoli, S., & Pierangeli, G. (2013). Nociception and autonomic nervous system. *Neurological Sciences, 34*, 41-46.
- Craig, A. (2003). Pain mechanisms: labeled lines versus convergence in central processing. *Annual Review of Neuroscience, 26*, 1-30.
- DeBoer, R., Karemaker, J. M., & Strackee, J. (1987). Hemodynamic fluctuations and baroreflex sensitivity in humans: a beat-to-beat model. *American Journal of Physiology-Heart and Circulatory Physiology, 253,* H680-H689.

- Delorme, A., & Makeig, S. (2004). EEGLAB: an open source toolbox for analysis of single-trial EEG dynamics including independent component analysis. *Journal of Neuroscience Methods, 134*, 9-21.
- Dowman, R., Rissacher, D., & Schuckers, S. (2008). EEG indices of tonic pain-related activity in the somatosensory cortices. *Clinical Neurophysiology*, *119*, 1201-1212.
- Dunkel-Schetter, C., & Bennett, T. L. (1990). Differentiating the cognitive and behavioral aspects of social support. In: B. R. Sarason, I. G. Sarason & G. R. Pierce, (Eds.), Wiley Series on Personality Processes. Social Support: An Interactional View (pp. 267-296). Oxford: John Wiley.
- Eisenberger, N. I., Master, S. L., Inagaki, T. K., Taylor, S. E., Shirinyan, D., Lieberman, M. D., & Naliboff, B. D. (2011). Attachment figures activate a safety signal-related neural region and reduce pain experience. *Proceedings of the National Academy of Sciences, 108*, 11721-11726.
- Gatchel, R. J. (2004). Comorbidity of chronic pain and mental health disorders: the biopsychosocial perspective. *American Psychologist, 59*, 795-805.
- Gatchel, R. J., Peng, Y. B., Peters, M. L., Fuchs, P. N., & Turk, D. C. (2007). The biopsychosocial approach to chronic pain: scientific advances and future directions. *Psychological Bulletin, 133*, 581-624.
- Goldstein, P., Shamay-Tsoory, S. G., Yellinek, S., & Weissman-Fogel, I. (2016). Empathy predicts an experimental pain reduction during touch. *The Journal of Pain, 17*, 1049-1057.
- Gram, M., Graversen, C., Olesen, S., & Drewes, A. (2015). Dynamic spectral indices of the
 electroencephalogram provide new insights into tonic pain. *Clinical Neurophysiology*, *126*, 763-771.
- Hadjileontiadis, L. J. (2015). EEG-Based Tonic Cold Pain Characterization Using Wavelet Higher Order Spectral Features. *IEEE Transactions on Biomedical Engineering*, *62*, 1981-1991.

- Hauck, M., Lorenz, J., Domnick, C., Gerloff, C., & Engel, A. K. (2015). Top-down and bottom-up modulation of pain-induced oscillations. *Frontiers in Human Neuroscience*, *9*, e375.
- Holt-Lunstad, J., Birmingham, W. A., & Light, K. C. (2008). Influence of a "warm touch" support enhancement intervention among married couples on ambulatory blood pressure, oxytocin, alpha amylase, and cortisol. *Psychosomatic Medicine, 70*, 976-985.
- Holzman, J. B., & Bridgett, D. J. (2017). Heart Rate Variability Indices as Bio-Markers of Top-Down Self-Regulatory Mechanisms: A Meta-Analytic Review. *Neuroscience & Biobehavioral Reviews*, 74, 233-255.
- Huber, M., Bartling, J., Pachur, D. v., Woikowsky-Biedau, S., & Lautenbacher, S. (2006). EEG responses to tonic heat pain. *Experimental Brain Research*, *173*, 14-24.
- Jensen, O., & Mazaheri, A. (2010). Shaping functional architecture by oscillatory alpha activity: gating by inhibition. *Frontiers in Human Neuroscience*, *4*, e186.
- Keefe, F. J., Caldwell, D. S., Baucom, D., Salley, A., Robinson, E., Timmons, K., . . . Helms, M. (1996).
 Spouse-assisted coping skills training in the management of osteoarthritic knee pain.
 Arthritis & Rheumatology, 9, 279-291.
- Keefe, F. J., Caldwell, D. S., Baucom, D., Salley, A., Robinson, E., Timmons, K., . . . Helms, M. (1999).
 Spouse-assisted coping skills training in the management of knee pain in osteoarthritis: long-term followup results. *Arthritis & Rheumatism*, *12*, 101-111.
- Keverne, E. B., Martensz, N. D., & Tuite, B. (1989). Beta-endorphin concentrations in cerebrospinal fluid of monkeys are influenced by grooming relationships. *Psychoneuroendocrinology*, 14, 155-161.
- Korhonen, R. J., Hernandez-Pavon, J. C., Metsomaa, J., Mäki, H., Ilmoniemi, R. J., & Sarvas, J. (2011).
 Removal of large muscle artifacts from transcranial magnetic stimulation-evoked EEG by
 independent component analysis. *Medical & Biological Engineering & Computing, 49*, 397-407.

- Krahé, C., Springer, A., Weinman, J. A., & Fotopoulou, A. (2013). The social modulation of pain: others as predictive signals of salience–a systematic review. *Frontiers in Human Neuroscience 7*, e386.
- Kreuder, A. K., Wassermann, L., Wollseifer, M., Ditzen, B., Eckstein, M., Stoffel-Wagner, B., . . .
 Scheele, D. (2018). Oxytocin enhances the pain-relieving effects of social support in romantic couples. *Human Brain Mapping*, *40*, 242-251.
- Le Pera, D., Svensson, P., Valeriani, M., Watanabe, I., Arendt-Nielsen, L., & Chen, A. C. (2000). Longlasting effect evoked by tonic muscle pain on parietal EEG activity in humans. *Clinical Neurophysiology*, *111*, 2130-2137.
- Maris, E., & Oostenveld, R. (2007). Nonparametric statistical testing of EEG-and MEG-data. *Journal of Neuroscience Methods, 164*, 177-190.
- Master, S. L., Eisenberger, N. I., Taylor, S. E., Naliboff, B. D., Shirinyan, D., & Lieberman, M. D. (2009). A picture's worth: partner photographs reduce experimentally induced pain. *Psychological Science*, *20*, 1316-1318.
- Nickel, M. M., May, E. S., Tiemann, L., Schmidt, P., Postorino, M., Dinh, S. T., . . . Ploner, M. (2017). Brain oscillations differentially encode noxious stimulus intensity and pain intensity. *Neuroimage, 148*, 141-147.
- Nir, R.-R., Sinai, A., Moont, R., Harari, E., & Yarnitsky, D. (2012). Tonic pain and continuous EEG: prediction of subjective pain perception by alpha-1 power during stimulation and at rest. *Clinical Neurophysiology*, 123, 605-612.
- Ohara, S., Crone, N., Weiss, N., & Lenz, F. (2004). Attention to a painful cutaneous laser stimulus modulates electrocorticographic event-related desynchronization in humans. *Clinical Neurophysiology*, *115*, 1641-1652.
- Oostenveld, R., Fries, P., Maris, E., & Schoffelen, J.-M. (2011). FieldTrip: open source software for advanced analysis of MEG, EEG, and invasive electrophysiological data. *Computational Intelligence and Neuroscience, e*156869. doi: 10.1155/2011/156869

- Paloyelis, Y., Krahé, C., Maltezos, S., Williams, S., Howard, M., & Fotopoulou, A. (2016). The Analgesic Effect of Oxytocin in Humans: A Double-Blind, Placebo-Controlled Cross-Over Study Using Laser-Evoked Potentials. *Journal of Neuroendocrinology, 28*, e12347.
- Pan, J., & Tompkins, W. J. (1985). A real-time QRS detection algorithm. *IEEE Transactions on Biomedical Engineering*, 230-236.
- Parsons, C. E., Young, K. S., Mohseni, H., Woolrich, M. W., Thomsen, K. R., Joensson, M., . . . Kringelbach, M. L. (2013). Minor structural abnormalities in the infant face disrupt neural processing: a unique window into early caregiving responses. *Social Neuroscience*, *8*, 268-274.
- Pearce, E., Wlodarski, R., Machin, A., & Dunbar, R. I. (2017). Variation in the β-endorphin, oxytocin, and dopamine receptor genes is associated with different dimensions of human sociality. *Proceedings of the National Academy of Sciences*, *114*, 5300-5305.
- Peng, R.-C., Yan, W.-R., Zhou, X.-L., Zhang, N.-L., Lin, W.-H., & Zhang, Y.-T. (2015). Time-frequency analysis of heart rate variability during the cold pressor test using a time-varying autoregressive model. *Physiological Measurement, 36*, 441-452.
- Peng, W., Hu, L., Zhang, Z., & Hu, Y. (2014). Changes of spontaneous oscillatory activity to tonic heat pain. *PLoS ONE, 9*, e91052.
- Ploner, M., Gross, J., Timmermann, L., Pollok, B., & Schnitzler, A. (2006). Oscillatory activity reflects the excitability of the human somatosensory system. *Neuroimage*, *32*, 1231-1236.

Ploner, M., Sorg, C., & Gross, J. (2017). Brain rhythms of pain. Trends Cogn Sci, 21(2), 100-110.

- Roberts, M. H., Klatzkin, R. R., & Mechlin, B. (2015). Social support attenuates physiological stress responses and experimental pain sensitivity to cold pressor pain. *Annals of Behavioral Medicine, 49*, 557-569.
- Sambo, C. F., Howard, M., Kopelman, M., Williams, S., & Fotopoulou, A. (2010). Knowing you care: Effects of perceived empathy and attachment style on pain perception. *Pain*, *151*, 687-693.

- Schulz, E., May, E. S., Postorino, M., Tiemann, L., Nickel, M. M., Witkovsky, V., . . . Ploner, M. (2015). Prefrontal gamma oscillations encode tonic pain in humans. *Cerebral Cortex, 25*, 4407-4414.
- Selesnick, I. W., & Burrus, C. S. (1998). Generalized digital Butterworth filter design. *IEEE Transactions on Signal Processing, 46*, 1688-1694.
- Shao, S., Shen, K., Yu, K., Wilder-Smith, E. P., & Li, X. (2012). Frequency-domain EEG source analysis for acute tonic cold pain perception. *Clinical Neurophysiology*, *123*, 2042-2049.
- Shaygan, M., Böger, A., & Kröner-Herwig, B. (2016). Valence and Arousal Value of Visual Stimuli and Their Role in the Mitigation of Chronic Pain: What Is the Power of Pictures? *The Journal of Pain*, 18, 124-131.
- Sheehan, D., & Lecrubier, Y. (2001). MINI SCREEN 5.0. 0/English version/DSM-IV July/1/06. Florida: University of South Florida-Tampa.
- Spanier, G. B. (1976). Measuring dyadic adjustment: New scales for assessing the quality of marriage and similar dyads. *Journal of Marriage and the Family, 38*, 15-28.
- Terkelsen, A. J., Mølgaard, H., Hansen, J., Andersen, O. K., & Jensen, T. S. (2005). Acute pain increases heart rate: differential mechanisms during rest and mental stress. *Autonomic Neuroscience: Basic and Clinical, 121*, 101-109.
- Thayer, J. F., Åhs, F., Fredrikson, M., Sollers, J. J., & Wager, T. D. (2012). A meta-analysis of heart rate variability and neuroimaging studies: implications for heart rate variability as a marker of stress and health. *Neuroscience & Biobehavioral Reviews, 36*, 747-756.
- Thayer, J. F., Hansen, A. L., Saus-Rose, E., & Johnsen, B. H. (2009). Heart rate variability, prefrontal neural function, and cognitive performance: the neurovisceral integration perspective on self-regulation, adaptation, and health. *Annals of Behavioral Medicine*, *37*, 141-153.
- Treister, R., Kliger, M., Zuckerman, G., Aryeh, I. G., & Eisenberg, E. (2012). Differentiating between heat pain intensities: the combined effect of multiple autonomic parameters. *Pain, 153,* 1807-1814.

- Uchino, B. N., Bowen, K., Carlisle, M., & Birmingham, W. (2012). Psychological pathways linking social support to health outcomes: A visit with the "ghosts" of research past, present, and future. *Social Science & Medicine, 74*, 949-957.
- Wittig, R. M., Crockford, C., Weltring, A., Langergraber, K. E., Deschner, T., & Zuberbühler, K. (2016). Social support reduces stress hormone levels in wild chimpanzees across stressful events and everyday affiliations. *Nature Communications, 7*, e13361.
- Younger, J., Aron, A., Parke, S., Chatterjee, N., & Mackey, S. (2010). Viewing pictures of a romantic partner reduces experimental pain: Involvement of neural reward systems. *PLoS One, 5*, e13309.
- Zhang, Z., Hu, L., Hung, Y. S., Mouraux, A., & Iannetti, G. (2012). Gamma-band oscillations in the primary somatosensory cortex—a direct and obligatory correlate of subjective pain intensity. *Journal of Neuroscience, 32*, 7429-7438.

Figure legends:

Fig. 1. Pain delivery, rating and overall perceived support. (a) Experimental set-up, showing pain stimulus (ice-filled water bottle) and pain intensity rating using a key-pad. (b) Dynamic changes in pain intensity during painful stimulation. Data points indicate group mean pain intensity. Each one minute interval is shaded separately. (c) The partner condition showed higher overall perceived support. (d) Higher perceived support was associated with greater pain reduction in the partner condition (n = 23). PT and ST denote the partner and stranger image-viewing condition respectively. Error bars represent standard error of the mean (SEM). *** denotes P < 0.001.

Fig. 2. Neural oscillatory activity during the stabilization (2^{nd} minute) and desensitization (3^{rd} minute) phase of pain. (a) Frequency spectrum indicating that alpha power was significantly greater in the partner compared to stranger image-viewing condition during pain stabilization. (b) The increase in alpha power reached significance at fronto-central electrodes F5, FC3, FC5 and C3. (c) Frequency spectrum showing that gamma power was significantly lower in the partner compared to stranger image-viewing condition. (d) The decrease in gamma power reached significant at electrodes CZ, C2, CP1, and CP2. (e) Reduced gamma power from the significant channels was associated with greater pain reduction in the partner condition (n = 23). PT and ST denote the partner and stranger image-viewing condition respectively. The shaded area represents the SEM. X indicates *P* < 0.05.

Fig. 3. Partner image-viewing decreases delta oscillations in the pain desensitization phase (3rd minute). (a) – (b) Decrease in delta power at frontal electrodes AF3, AF4, F3, F1, FZ, F2, and FCZ. (c) – (d) Decrease in delta power at posterior electrodes CP5, CP3, CP1, CP2, CP4, CP6, P5, P3, P1, PZ, P2, P6, PO3, and POZ. The shaded area represent the SEM. X indicates *P* < 0.05.

Fig. 4. Changes in the high-frequency heart rate variability across the pain induction. Error bars represent the SEMs. n.u. denotes the normalized units. * denotes P < 0.05.









CHAPTER SIX

Neural Plasticity and Connectivity

Manuscript

Che X., Cash R., Chung S., Bailey N., Fitzgerald P., & Fitzgibbon B.M. The medial prefrontal cortex as a flexible hub mediating behavioural as well as local and distributed neural effects of social support on pain: a theta burst stimulation and TMS-EEG Study. Paper under review.

Preamble to empirical paper

Preliminary evidence suggests a key role of the medial prefrontal cortex (mPFC) in the analgesic influence of social support (Eisenberger et al., 2011; Younger et al., 2010). This is suggested to be related to the role of the mPFC in signalling safety (Delgado et al., 2006; Phelps et al., 2004; Quirk et al., 2006). However, to date a causal influence of the mPFC in facilitating the processing of social support and its impact on pain has not been demonstrated. Further, Chapter 5 demonstrates a role of social support in modulating neural oscillations supporting the processing of nociceptive information, but the brain-wide dynamics that may underpin the influence of social support on pain remain unknown.

In this chapter, TMS was used to investigate the causal role of the mPFC in orchestrating the behavioural and brain connectivity effects of social support on pain. Activity of the mPFC is either increased or decreased (or no change) across sessions using different TBS protocols. Pain perception and EEG connectivity are evaluated before and after the delivery of TBS. Using TMS-EEG, we further aim to explore the neural plasticity changes over the prefrontal cortex and its association with social modulation of pain.

This chapter demonstrates the role of the mPFC in causally modulating pain perception and network configuration dependent on social support contexts.

The organisation of this chapter is in keeping with the requirements for submission to the journal. However, where appropriate, minor formatting changes have been made to improve consistency with other sections of this thesis. *7. Manuscript Click here to view linked References

Title Page

The Medial Prefrontal Cortex as a Flexible Hub Mediating Behavioral as well as Local and

Distributed Neural Effects of Social Support on Pain: a Theta Burst Stimulation and TMS-EEG Study

Xianwei Che^{*1}, Robin Cash¹, Sung Wook Chung¹, Neil Bailey^{1, 2}, Paul B Fitzgerald^{1, 2}, Bernadette M.

Fitzgibbon¹

¹Monash Alfred Psychiatry Research Centre (MAPrc), The Alfred and Central Clinical School, Monash

University, Melbourne, Australia

²Epworth Centre for Innovation in Mental Health, Epworth, VIC, Australia

*Address correspondence to:

Xianwei Che

Monash Alfred Psychiatry Research Centre (MAPrc)

Monash University

Level 4, 607 St Kilda Road, Melbourne VIC 3004, Australia

Disclosures: This work was supported by the China Scholarship Council; the NHMRC Practitioner Fellowship (606907), and the NHMRC Early Career Fellowship. The authors declare no competing financial interests.

Abstract

Increasing evidence points to an analgesic influence of social support, in which the medial prefrontal cortex (mPFC) is thought to play a key role. Transcranial Magnetic Stimulation (TMS) has the capacity to causally modulate brain activity. This study was designed to investigate the potential role of mPFC in orchestrating the behavioral and neural effects of social support on pain. Twenty-three healthy participants underwent a three-session cross-over, single-blinded, sham-controlled protocol in which they received Theta Burst Stimulation (TBS) (facilitatory intermittent TBS, suppressive continuous TBS, or Sham) delivered to the mPFC. In each session, participants underwent cold pain while viewing an image of a romantic partner or a stranger. Effects of TBS to the mPFC were assessed using a measure of pain perception, neural activity and network connectivity using electroencephalography (EEG) and TMS-EEG. In the stranger condition, pain experience increased following iTBS. This was associated with increased connectivity between central regions and frontoparietal regions. In contrast, in the romantic partner condition, iTBS increased connectivity only between frontal and occipital regions and did not modulate pain experience. In line with recent studies, neither cTBS nor Sham stimulation elicited neural or behavioral changes. Together these findings suggest that the mPFC has the capacity to causally modulate pain-related information integration and network configuration in a context-dependent manner.

Keywords: social support; pain; theta burst stimulation; TMS-EEG; connectivity

1. Introduction

Pain experience is modulated by a variety of social contexts in daily life (for review see Krahé et al., 2013). Emerging evidence has suggested an analgesic influence of social support (Che et al., 2017; Goldstein et al., 2016), which commonly refers to received support from others and the perceived availability of supportive resources (Dunkel-Schetter and Bennett, 1990). However, recent systematic reviews, including one meta-analysis from our group, have demonstrated that this effect is not universal and highlighted the role of contextual factors (Che et al., 2018; Krahé et al., 2013). Indeed, social support from a significant other can prime intimacy and reduce pain, but the presence of another person in general may not have an overall effect (for a review see Che et al., 2018).

Advancing our understanding of the role of social support in pain therefore requires improved knowledge of the mechanisms that mediate the impact of social support on pain. Initial evidence suggests a role of social support in priming feelings of attachment, as supported by increased brain activity (blood-oxygen-level dependent imaging, BOLD) in the medial prefrontal cortex (mPFC) (Eisenberger et al., 2011; Younger et al., 2010). This is consistent with the proposed role of the mPFC in processing social intimacy (Gamond and Cattaneo, 2016; Gamond et al., 2017). Moreover, the mPFC is a key node of the default mode network (DMN), which is associated with self-report perceived social support (Che et al., 2014) and mind wandering away from pain (Kucyi et al., 2013). However, given methodological limitations of the techniques used, this initial evidence was unable to assess the causal relationships of the mPFC in facilitating the processing of social support and its impact on pain. Moreover, the brain-wide dynamics that may underpin the influence of social support on pain remain unknown.

Transcranial Magnetic Stimulation (TMS) is a non-invasive approach which can modulate brain activity, and thereby provide unique insights into brain-behavior relationships. Theta Burst Stimulation (TBS) is one of the most established repetitive TMS protocols and has the capacity to modulate neural excitability (Huang et al., 2005; Ni et al., 2014) and network connectivity (Iwabuchi et al., 2017; Rastogi et al., 2017). More specifically, TBS can increase or decrease cortical excitability

depending on whether intermittent (iTBS) or continuous (cTBS) stimulation, respectively, is employed (Huang et al., 2005). Thus, TBS provides a means to causally modulate the mPFC and its associated networks and explore the role of this brain region in the effects of social support on pain. In addition, changes in neural activity induced by TBS can be measured using single-pulse TMS and concurrent electroencephalography (TMS-EEG) (Cash et al., 2017; Chung et al., 2017). Of note, recent studies suggest that iTBS may be more effective than cTBS in modulating brain activity when applied to the prefrontal cortex (Berlim et al., 2017; Chung et al., 2017).

Changes in EEG activity and connectivity following TBS may help to reveal regional and network-based neural mechanisms that mediate the influence of social support on pain. Although the frequency dynamics of the experience of pain are not entirely understood (Ploner et al., 2017), with each of the bands implicated (e.g. for alpha see Furman et al., 2018), most of the studies link subjective pain ratings with increased gamma activity (31-100 Hz) in central and frontal cortex (Nickel et al., 2017; Schulz et al., 2015; Zhang et al., 2012). To our knowledge, only one EEG study has explored phase synchronization in response to painful stimuli, which found increased theta range (4-7 Hz) connectivity between central and parietal regions (Taesler and Rose, 2016). Additionally, only one study has looked at EEG connectivity in the context of social influence on pain, in which increased alpha band (8-12 Hz) connectivity was suggested to underlie the effects of social touch on pain reduction (Goldstein et al., 2018).

This study was designed to investigate the role of the mPFC in mediating the effects of social support on pain. Participants underwent a three-session cross-over, single-blinded, sham-controlled protocol in which they received either iTBS, cTBS, or Sham stimulation. A pain task was performed before and after TBS during which participants were subjected to cold pain while viewing an image of either a romantic partner or a stranger (Eisenberger et al., 2011). We hypothesized that iTBS would increase neural activity in the mPFC (assessed using TMS-EEG) and that in the romantic partner condition, iTBS would increase social support related alpha connectivity and reduce pain related gamma power, theta connectivity, and pain sensation. In contrast, we hypothesized no

changes in the stranger condition. We also anticipated less pronounced and potentially opposite changes in the sham and cTBS sessions.

2. Methods

2.1. Participants

Twenty-three healthy right-handed adults participated in this study. Three participants withdrew after the first session. Data from 20 participants were therefore analyzed (see Table 1). Exclusion criteria included a history or current diagnosis of a psychiatric disorder, or use of psychoactive medication, as assessed by the Mini International Neuropsychiatric Interview (MINI) (Sheehan and Lecrubier, 2001). All participants identified as being in a romantic relationship. An informed consent was provided by all participants, and the experiment was approved by the Alfred Hospital and Monash University Human Research and Ethics Committee. This study was conducted in accordance with the Declaration of Helsinki.

2.2. Experimental Design and Procedure

Participants underwent a three-session data collection protocol with each session at least 72 hours apart to avoid any potential carry-over effects (Fig. 1). The experimental procedures were the same in each of the three sessions except for the type of TBS protocol (i.e. iTBS, cTBS or Sham), the order of which was counterbalanced across participants. During the first session participants completed several self-report questionnaires (see Table 1). In each session, participants first underwent a 3-minute pain protocol with EEG recorded to assess neural oscillations and connectivity. In the pain protocol, participants viewed an image of either a stranger or their romantic partner (in total 2 separate trials), and rated pain at 20-second intervals for the three minutes. Next, single-pulse TMS was delivered to the mPFC and cortical responses were recorded using concurrent EEG. Participants then received either iTBS, cTBS or Sham stimulation to the mPFC, which was followed by a repeat of the pain and TMS-EEG protocol described above.

------ Please insert Fig. 1 here ------

2.3. Social support or Stranger Attachment Image

Consistent with a template image (i.e. the gender-matched stranger image), a digital image of the romantic partner was taken by the participants and provided for the study. Specifically, the image was requested to be against a plain white background, demonstrating a natural smile and including only the head and shoulders. One of the researchers (XC) then edited the images (GNU Image Manipulation Program, GIMP, V2.8,) to have the fixed size (17.5 cm × 17.5 cm), content (from the top of head to shoulders), and matched for luminance using the "auto adjust color levels" function (Parsons et al., 2013). Two images, one male and one female in early adulthood, served as the stranger images.

2.4. Pain Protocol

In order to control baseline hand temperature (Hadjileontiadis, 2015), participants were asked to insert their dominant hand into a bucket of warm water (40 °C \pm 0.2 °C) for 2 minutes and relax for another 5 minutes prior to and between experimental trials. Participants were then asked to hold a bottle filled with iced water (-1 °C \pm 0.2 °C) for 3 minutes (i.e. a single trial) with the dominant hand (Hadjileontiadis, 2015). A trial was started by pressing a button with the non-dominant hand in the keypad. The button press started the presentation of an image (Presentation, Version 17.0, Neurobehavioral Systems, Berkeley, CA), which was either their romantic partner or a stranger matched to the partner's gender (Eisenberger et al., 2011). These two conditions were performed separately in two trials, and the order of conditions was pseudorandomized and counterbalanced across participants. A 3-minute trial was divided into nine 20-second blocks during which the participants viewed the image for 16 seconds and then rated pain intensity within 4 seconds. The participants were asked to rate "pain intensity at the moment" on a scale of 0-10 (0 = no pain; 10 = worst pain imaginable), by pressing the corresponding button on a keypad. At the end

of the trial, participants rated their "overall perceived support" from the image by pressing the corresponding button (0 = not at all; 10 = extremely high). A pillow was provided on which participants rested the back of their hands to avoid hand movement. Participants were told to put the volar surface of the dominant hand on the surface of the bottle, and not to avoid or squeeze it, to minimize the variability of touching. Fresh ice bottles were used for each experimental trial for consistency.

2.5. EEG Recordings

EEG recordings took place in a temperature-controlled, sound-attenuated, and electrically shielded room. Participants were seated in a slightly reclined chair with their faces approximately 0.5m from the computer monitor. A 64-channel EEG cap (NeuroScan Inc., Australia) was used to record continuous EEG with CPZ and FPZ as the reference and ground electrode respectively. Two electrodes were attached above and below the left eye to monitor vertical electrooculogram (EOG) activity. Horizontal EOG activity was monitored with electrodes located at the outer canthus of both eyes. EEG impedances were kept below 5 k Ω throughout the experiment.

For EEG recordings during the pain protocol, EEG signals were filtered (0.05–200 Hz) and sampled at 1,000 Hz. For TMS-EEG recordings, EEG signals were amplified (1,000×) and low-pass filtered (DC—2,000 Hz) with a high acquisition rate of 10,000 Hz. As the TMS click sound might contaminate the EEG signals (Nikouline et al., 1999), participants were asked to listen to white noise through intra-auricular earphones (Etymotic Research, ER3-14A, USA) during TMS-EEG recordings (Fuggetta et al., 2005). The sound level was adjusted such that each individual reported that they could no longer hear single-pulse TMS at 110% resting motor threshold (RMT).

2.6. Transcranial Magnetic Stimulation

Both single-pulse TMS and TBS were delivered using a figure-of-eight MagVenture B-65 fluidcooled coil (MagVenture A/S, Denmark) in a biphasic mode. In the determination of the RMT, stimuli

were applied to the left motor cortex with the coil positioned at 45° angle relative to midline (Chung et al., 2018a; Chung et al., 2018b). The RMT was determined with the EEG cap on, as the minimum stimulus intensity required to elicit at least three out of five motor evoked potentials (MEPs) >0.05 mV in amplitude in the relaxed first dorsal interosseous muscles (Conforto et al., 2004). Prefrontal TMS was then administered, with the coil centred at the F1 electrode. This electrode was selected as it is positioned over the superior frontal gyrus (Koessler et al., 2009), and increased left mPFC activity is associated with social support effects on pain (Eisenberger et al., 2011; Younger et al., 2010). Moreover, the coil was positioned at 90° angle relative to midline (handle pointing left, see Fig. 2a) in order to target the mPFC (Downar et al., 2012; Salomons et al., 2014). The edge of the TMS coil was marked on the EEG cap for consistent re-positioning of the coil within and between sessions (Rogasch et al., 2013). The TMS-evoked potentials (TEPs) were recorded using EEG during singlepulse TMS (105 pulses, 4 s interval ± 10% jitter) at 110% RMT, delivered before and after TBS. TBS consisted of a burst of 3 pulses given at 50Hz repeated every 5Hz, where (1) iTBS involved a 2s train of TBS repeated every 10s for a total of 192s, (2) cTBS without any break/interruption for a total of 40s, or (3) Sham – the iTBS protocol was administrated using a MagVenture Placebo B-65 coil which has a sound level identical to the B-65 coil. TBS was delivered with the intensity of 70% RMT (Goldsworthy et al., 2012), with a total of 600 pulses.

In order to validate the target site, electric field simulations for the cooled B-65 TMS coil (90° angle) were performed using the SimNIBS modelling environment, which utilizes a finite element model of brain current flow based on an MRI derived template head model (Windhoff et al., 2013). Visualization of the electrical fields was performed using the Gmsh mesh generator (Geuzaine and Remacle, 2009).

2.7. EEG Data Analysis

EEG data during the pain protocol were preprocessed offline using custom-written scripts that implement functions from EEGLAB (version 13.6.5b) (Delorme and Makeig, 2004) running under

Matlab R2017b (The MathWorks, Inc.). Data from malfunctioning channels were visually inspected and removed. Butterworth filters (band-pass: 0.5-100Hz; band-stop notch filter: 48-52 Hz) were then applied to the data (Selesnick and Burrus, 1998). Continuous data were segmented to retain only the image viewing stage, i.e. 16 seconds in each block, in total 9 in each three-minute pain stimulation trial. Data were further segmented into 1-second non-overlapping epochs to remove any contaminated data (Peng et al., 2014; Schulz et al., 2015). Segmented data were re-referenced to the average reference, and the fast independent component analysis algorithm (FastICA) was used to remove stereotyped artefacts, e.g. eye blinks, lateral eye movements, muscle, and line noise (Korhonen et al., 2011). Stereotyped artefacts were identified by visual inspection of the temporal and spatial representation of the independent components. Missing channels were then interpolated, and epochs were inspected again to remove any anomalous activity in the signal.

EEG frequency representations were calculated with the Multitaper Method Fast Fourier Transform ('mtmfft'), as implemented in FieldTrip toolbox, in the range of 0.5-100Hz (Oostenveld et al., 2011). Power spectra were calculated for each epoch, and then averaged across epochs in each condition.

EEG connectivity was calculated between each electrode using the debiased estimator of the weighted phase lag index (wPLI) based on the frequency representations obtained above. The wPLI is considered as a conservative measure of phase synchronization, which is suggested to be robust against volume conduction, non-brain related artifacts and common reference artifacts (Vinck et al., 2011). The measure also has good test-retest reliability (Hardmeier et al., 2014). For each frequency, the wPLI provided a value of coherence for each electrode pair. Connectivity values were then averaged in the frequency domains of interest, i.e. theta (4-7 Hz), or alpha (8-12 Hz). Exploratory analyses were also performed in other frequency bands (i.e. delta: 1-3 Hz; beta: 13-30 Hz; gamma: 31-100 Hz).

2.8. TMS-EEG Data Analysis

TMS-EEG data were preprocessed as previously described (Chung et al., 2017). Specifically, data were epoched around the TMS pulses (–1,000 to 1,000 ms), baseline corrected (–500 to –50 ms), and then the large magnetic pulses were removed and interpolated (–5 to 15 ms). Data were down-sampled to 1,000 Hz and epochs containing excessive noise and/or disconnected electrodes were removed during manual inspection. Prior to independent component analysis (ICA) based artifact rejection, the epoched data were concatenated across the two time-points (Pre and Post) to avoid bias in component rejection. Two rounds of FastICA were performed using semi-automated component classification algorithm (Rogasch et al., 2017). The first ICA was performed to remove large TMS-evoked muscle artefacts and decay artefact (Rogasch et al., 2014). All data were bandpass (1–80 Hz) and band-stop filtered (line noise removal, 48–52 Hz), and epochs were visually inspected again to remove any anomalous activity. The second round of ICA was used to remove other non-neural artefacts, e.g. eye blinks, saccadic movement, persistent muscle activity, decay artefact and electrode noises. Removed channels were then interpolated. Finally, data were re-referenced to common average and segregated into original time-point blocks (Pre and Post) and epochs averaged for each condition.

2.9. Source Estimation

Cortical sources of the TEPs were estimated using Brainstorm (Tadel et al., 2011). TMS-EEG data were co-registered with the template brain model (i.e. ICBM 152). The forward model used the Symmetric Boundary Element Method implemented in OpenMEEG software (Gramfort et al., 2010), and the inverse model used the computation of minimum norm estimations (MNEs) with dipole orientations constrained to be normal to the cortex (Lin et al., 2006). Differences in estimation were calculated using absolute subtraction.

2.10. Statistical Analyses

Statistical analyses were performed on the primary outcome measures (self-reported pain ratings, TEPs, EEG power and connectivity).

Pain intensity ratings across 3-minute blocks were summarized by calculating the area under the curve (AUC). The effects of TBS and time as well as their interaction were analyzed using repeated measures two-way ANOVAs in SPSS (version 23; IBM Corp, Armonk, NY). The partner and stranger condition were examined separately. TBS (iTBS, cTBS or Sham) and time (Pre, Post) were specified as the two repeated measures factors. Post-hoc pairwise comparisons were conducted to further explore the significant main and interaction effects, with the α -level set to 0.05 and Bonferroni corrected.

For TEPs, statistical analyses were conducted using cluster-based permutation statistics at a global scalp level (Maris and Oostenveld, 2007). The cluster-based permutation test provides a straightforward way to solve the problem of multiple comparisons across space (EEG channels) and time (Maris and Oostenveld, 2007). Statistics were performed on a priori peak of interest (i.e. N100: 90–130 ms), which is considered to be the most prominent and robust TMS-EEG component for the exploration of TMS induced plasticity changes (Chung et al., 2015; Nikulin et al., 2003). Exploratory analyses were also performed on other peaks: N40 (30–50 ms), P60 (50–80 ms), and P200 (160–240 ms), which have been commonly observed following prefrontal stimulation (Chung et al., 2017; Rogasch et al., 2014). Paired T-tests were first made across time point (Post vs Pre) for each TBS condition ('within-comparison'). Comparisons between TBS conditions ('between-comparison') were then performed using delta score of each TBS condition (Δ = Post–Pre). An observed test statistics value was considered in the cluster permutation if it was below the threshold of 0.05 in at least 2 of the neighboring channels (Oostenveld et al., 2011). We performed 5000 iterations of trial randomization for generating the permutation distribution, controlling for multiple comparisons across space (*P* < 0.025; two-tailed test).

For EEG power changes, the same cluster-based permutation statistics were used to identify differences in neural oscillatory activity. This method was applied to each 1-second time window

across all channels. Comparisons were made between Pre- and Post-stimulation for each TBS (iTBS, cTBS or Sham) and image (partner, stranger) condition. According to our a priori hypothesis, statistical analyses were firstly carried out in alpha (8-12 Hz), theta (4-7 Hz) and gamma (31-100Hz) bands. Additional exploratory analyses were performed separately in other frequency bands (i.e. delta: 1-3 Hz; beta: 13-30 Hz).

Connectivity wPLI values were statistically tested using the network-based statistic (NBS) toolbox (Zalesky et al., 2010). The NBS is a non-parametric statistical method which uses cluster analysis to perform null hypothesis testing across networks of values from pairs of potentially connected nodes (Zalesky et al., 2010). It is robust against unequal sample sizes and controls for the family-wise error rate (Zalesky et al., 2010). Paired T-tests were performed to examine significant connectivity changes from Pre- to Post-stimulation for each TBS (iTBS, cTBS, Sham) condition. Statistical comparisons were made using 5000 permutations, with a primary threshold for electrode pairs set at P < 0.005 to ensure only robust differences in connectivity between electrode pairs would be compared at the cluster level (Bailey et al., 2018). The secondary threshold for family-wise corrected cluster null hypothesis testing was P < 0.025 (two-tailed).

Correlation analysis was performed to examine brain-behaviour relationships between TBSinduced changes in pain ratings and significant TBS-induced changes in EEG power and TEP amplitude.

In the supplementary analysis, a repeated measures two-way ANOVA was performed on overall perceived support to examine the effects of TBS (iTBS, cTBS or Sham) and time (Pre, Post) as well as the interaction effect.

3. Results

The effectiveness of TBS over the mPFC was initially validated before the investigation of outcome measures. Fig. 1b shows the electric field distribution in the cortical grey matter for the cooled B-65 coil. The TMS coil effectively targeted the left prefrontal cortex, with maximum field

strengths occurring around the left mPFC. Moreover, single-pulse TMS over left mPFC resulted in a series of negative and positive peaks including N40, P60, N100 and P200 (*Supplementary Material S1*). Consistent with other TMS-EEG studies in the prefrontal cortex (Chung et al., 2018b; Hill et al., 2017), each peak showed a distinctive pattern in scalp topography and source estimation.

------ Please insert Fig. 2 here ------

3.1. Effects of TBS on Pain Intensity Ratings

In the partner condition, the ANOVA revealed no effect of TBS protocol, time, or their interaction on pain ratings ($P_s > 0.05$), suggesting that pain ratings were the same across stimulation type and time window (Fig. 3a-d). In the stranger condition (Fig 3e-h), the ANOVA revealed an interaction effect of TBS protocol and time on pain ratings ($F_{2,38} = 3.51$, P = 0.04, $\eta_p^2 = 0.16$). Posthoc tests were performed to investigate this interaction. Two one-way ANOVAs (one for Pre, one for Post) were first performed across TBS conditions, which were followed by three paired T-tests (one for each TBS condition) conducted across time windows. One-way ANOVAs revealed no effect of TBS on pain ratings in either time window ($P_s > 0.05$), suggesting that pain ratings were the same across TBS conditions in both Pre- and Post-stimulation (Fig. 3e-h). Across time windows, paired T-tests showed a significant increase in pain ratings from Pre- to Post-iTBS ($P_{Bonf} = 0.003$) (Fig. 3e and Fig. 3h). No difference was observed in the cTBS (Fig. 3f) or Sham (Fig. 3g) condition ($P_{Bonf} > 0.05$).

------ Please insert Fig. 3 here -----

3.2. Plastic Effects of TBS on TEPs

Cluster-based permutation tests revealed that the amplitude of the primary TEP component of interest, N100, was significantly increased from Pre- to Post-iTBS ($P_{corrected} = 0.011$) (Fig. 4a). Moreover, the topography of this change was mainly distributed around the frontal regions where iTBS was delivered (Fig. 4d *left panel*). In line with recent evidence (Chung et al., 2017), there were no changes in N100 amplitude following cTBS (Fig. 4b) or Sham (Fig. 4c) ($P_{corrected} > 0.05$). Changes in N100 amplitudes were not associated with increased pain in the stranger condition from Pre- to Post-iTBS (P > 0.05).

These findings were supported by secondary analyses in which N100 amplitude Pre- to Poststimulation was compared across all TBS conditions using permutation tests. Relative to sham, N100 changes from Pre- to Post-stimulation were significantly larger in the iTBS condition ($P_{corrected} =$ 0.006). Moreover, changes in N100 were mainly distributed around the fronto-central regions (Fig. 4d, *right panel*). No difference was found in the iTBS versus cTBS or the cTBS versus Sham comparisons.

----- Please insert Fig. 4 here ------

3.3. Effects of TBS on EEG Oscillations in the Pain Protocol

In the partner condition, cluster-based permutation statistics revealed a significant increase in gamma power from Pre- to Post-iTBS ($P_{corrected} = 0.024$). The topography of this power increase had a fronto-central, right-lateralized distribution (significant at F4, F6, FC4, FC6, and C4) (Fig. 5a). The increase in gamma power was associated with larger N100 amplitude from Pre- to Post-iTBS (r = -0.49, P = 0.03) (Fig. 5a). No correlation was found between gamma changes and pain rating changes (P > 0.05). No difference was found in any other frequency bands.

------ Please insert Fig. 5 here ------

In the stranger condition, we also found increased gamma power from Pre- to Post-iTBS ($P_{corrected} = 0.019$). Increased gamma in the stranger condition was mainly distributed in the left central-parietal regions (significant at T7, C5, C3, CP5, P7, P5 and O1) (Fig. 5b). However, no correlation was found between gamma changes and N100 amplitude or pain ratings changes ($P_s > 0.05$). No difference was found in any other frequency bands.

Following sham stimulation, alpha activity was greater in both the partner and stranger condition. Moreover, in both the partner ($P_{corrected} = 0.006$) and stranger ($P_{corrected} = 0.021$) condition, increased alpha was mainly distributed in the central regions (Fig. 5c, and Fig. 5d).

3.4. Effects of TBS on EEG Connectivity in the Pain Protocol

3.5. Supplementary Analysis

In the partner condition, the ANOVA revealed a main effect of time on overall perceived support ($F_{1,19} = 6.54$, P = 0.019, $\eta_p^2 = 0.26$), with post-hoc test showing that overall perceived support decreased from Pre- to Post-stimulation across TBS conditions ($P_{Bonf} = 0.019$) (*Supplementary Material S2*). No effect was observed in the stranger condition ($P_s > 0.05$).

4. Discussion

Previous studies have indicated that the mPFC plays a key role in mediating the analgesic effects of social support (Eisenberger et al., 2011; Younger et al., 2010). Using TBS, this study was designed to modulate mPFC activity and related networks mediating pain experience in the presence or absence of visual social support. Our data indicate that while iTBS elicited typical changes in TMS-EEG measures of neural activity at rest, it led to differential effects across social support conditions during pain. Crucially, these data suggest that iTBS differentially influenced network connectivity and behavior across social support conditions. In line with other recent studies, cTBS and Sham stimulation did not elicit prominent changes. In sum, these findings provide novel evidence to support the role of mPFC as a hub mediating the analgesic effects of social support on pain processing. Firstly, it is important to examine the efficacy of TBS conditions in modulating neural excitability, prior to considering behavioral and other effects. The neuroplastic effects of TBS delivered over non-motor regions such as the mPFC is most commonly quantified using TEPs. TEPs are suggested to reflect the shifts in the inhibition-excitation balance in cortical circuits following a single TMS pulse (Du et al., 2018; Rogasch and Fitzgerald, 2013). In this instance, we focused on N100 amplitude which is considered to be the most robust TEP component with the greatest signal-to-noise ratio (for discussion, see Cash et al., 2017; Noda et al., 2016). Furthermore, prior studies have identified N100 as being the component most reliably modulated by TBS (Chung et al., 2017; Chung et al., 2018a). In agreement with previous studies, N100 amplitude was increased by iTBS (Fig. 4a), however there were no changes with cTBS or Sham stimulation (Fig. 4b-c). The absence of TEP changes with cTBS is in line with recent evidence which also observed more robust effects of iTBS compared to cTBS in modulating activity in prefrontal cortex (Chung et al., 2017; Chung et al., 2018a).

iTBS delivered to the mPFC modulated the influence of social support on pain experience in a context-dependent fashion. Specifically, viewing a stranger was associated with more pain following mPFC-iTBS whereas pain did not increase in the romantic partner condition. This selective and differential influence of mPFC stimulation on pain experience according to social support condition was not evident in the Sham or cTBS conditions. Studies have shown a dual, opposing roles for the mPFC in pain (for a review see Ong et al., 2018), in which mPFC may serve to suppress pain via its connection with the periaqueductal gray (PAG) (An et al., 1998; Kucyi et al., 2013), but induces the persistency and chronification of pain via the corticostriatal projection (Baliki et al., 2012; Vachon-Presseau et al., 2016). Moreover, the mPFC has been found to mediate the selective attention to pain (Peyron et al., 1999), and as a key node of the DMN, it has been associated with rumination about pain (Kucyi et al., 2014). Effectively, our findings may indicate that mPFC-iTBS actually increased pain experience, but that social support was sufficient to buffer this change. The

effects of social support on pain may be further explained by neural and network related changes as described next.

Previous research has established that pain and pain modulation are associated with distinct changes in neural activity, particularly in the gamma band (Peng et al., 2014). Our research supports these findings, showing that modulating mPFC activity using iTBS increased fronto-central gamma activity in the romantic partner condition. Furthermore, these changes correlated with plastic changes in TEP amplitude following iTBS (Fig. 5a). In contrast, the stranger condition was associated with increased central-parietal gamma activity post iTBS (Fig. 5b). Gamma band activity is suggested to encode subjective pain experience (Nickel et al., 2017; Zhang et al., 2012). However, the increase in fronto-central gamma activity in the social support condition may otherwise reflect the role of gamma band activity in other higher-order cognitive processes, such as attention and the episodic memory retrieval (Canolty et al., 2006; Keizer et al., 2010; Marshall et al., 2015). One possibility is that the increase in fronto-central gamma activity following iTBS is related to attentional processing of visual social support and the retrieval of memory related to the romantic partner. The increase in gamma activity proximal to the stimulation site following excitatory stimulation at the prefrontal cortex is consistent with recent findings (Noda et al., 2017), however our data are the first to indicate a relationship between plastic changes in N100 amplitude and gamma activity linked to social support.

In the stranger condition in which pain experience increased following iTBS, gamma activity was increased in central-parietal regions contralateral to the somatic site of pain delivery (Fig. 5b). This finding is consistent with previous studies which indicated gamma band activity in central-parietal cortical regions associated with somatosensory perception and pain integration (Hauck et al., 2015; Ploner et al., 2017; Zhang et al., 2012).

One of the most interesting aspects of our findings relates to the differential effects of mPFC TBS on network connectivity across different social support conditions. Connectivity across distributed neural regions typically manifests across lower frequencies, in particular the theta and

alpha bands (Sauseng et al., 2010; Sauseng and Klimesch, 2008). In the romantic partner condition, iTBS increased alpha band connectivity between frontal regions surrounding the stimulation site and occipital regions involved in visual processing (Fig. 6a). This finding is consistent with previous evidence identifying a specific role of alpha band connectivity in mediating social support analgesia (Goldstein et al., 2018). In particular, social support by touching a romantic partner reduced pain experience, and this effect was associated with increased synchronization of the couple's alpha waves (Goldstein et al., 2018). Beyond this evidence, alpha band coherence is suggested to support complex social interactions, e.g. imitation (Dumas et al., 2010), social coordination (Tognoli et al., 2007). Our finding extends the literature by demonstrating that alpha coupling between frontal and occipital regions may serve to process visually presented social support information.

In contrast, iTBS over the mPFC led to distributed changes in theta connectivity in the stranger condition. Specifically, connectivity was increased between left central regions associated with sensory processing of the pain stimulus, frontal regions near the TBS stimulation site and parietal regions likely involved in the integration of pain-related information (Fig. 6b). Previous evidence has identified a specific role of theta connectivity in relation to pain experience in which painful stimuli increased theta connectivity between central cortex and the parietal and frontal cortices in healthy individuals (Taesler and Rose, 2016). Moreover, we found increased coupling between temporal (as indicated by T7 electrode) and frontal regions, in which the temporal cortex plays a key role in understanding the intention of others (Allison et al., 2000; Pelphrey et al., 2004). In sum, it appears that stimulation of the mPFC using iTBS is able to enhance the neural signatures associated with social support and pain experience in a context-specific fashion. Furthermore, these findings provide evidence that the mPFC has the capacity to orchestrate and modulate distributed effects associated with social support and pain across distributed regions.

In contrast to iTBS, cTBS did not demonstrate any changes in cortical plasticity (Fig. 4b), pain experience (Fig. 3b, Fig. 3f) or neural activity. This is consistent with the literature which suggests that cTBS is less effective in modulating brain activity compared to iTBS (Berlim et al., 2017; Chung et

al., 2017). However, we did observe an increase in alpha power following Sham stimulation independent of the image type (Fig. 5c, Fig. 5d). Pain has been shown to suppress alpha activity which is related to nociceptive attention (Nickel et al., 2017; Peng et al., 2014). Conversely, increased alpha activity is associated with cortical "deactivation" and is sometimes referred to as an 'idling' state (Cash et al., 2017; Jensen and Mazaheri, 2010). Increased alpha activity post Sham stimulation therefore may suggest less attention driven by the painful stimuli or the social support information as the participants knew what to expect from Pre- to Post-stimulation.

Findings of the current study improves our understanding of the mechanisms that may mediate the influence of social support on pain. Using TBS, activity of the mPFC was temporarily modulated which allowed the investigatation of its causal role in the processing of social support as well as the influence on pain experience. Moreover, TMS-EEG has the capacity to assess neural plasticity changes caused by TBS. Combination of these techniques would supplement in generating stronger conclusions surrounding social support and pain. Furthermore, the current study provides evidence beyond local brain activation, which extends the literature to understand the neural networks and dynamics surrounding the impact of social support on pain. Overall, our findings may help to understand the potential therapeutic influence of social support in the management of pain whereby empirical evidence is limited surrounding the social contexts and mechanisms of the painrelieving effect of social support (Keefe et al., 1996, 1999).

We acknowledge some limitations with the current study. We only used one image of the romantic partner whereby a consistent reduction in perceived social support across sessions was evident, independent of TBS condition (*Supplementary Material S2*). Habituation to the image may have influenced the capacity to elicit reductions in pain experience in the romantic partner condition following iTBS. Future studies would benefit from using multiple images that have the same valence to the support recipient. Secondly, the TMS coil was positioned based on F1 electrode location. Electrode-based positioning is commonly employed in TMS-EEG research (Cash et al., 2017; Chung et al., 2018b) and similar to the F3 approach used in the treatment of clinical disorders (Beam et al.,
2009). However, it is not as accurate as MRI-guided neuronavigation which was not feasible in the present study. Moreover, we only targeted the left mPFC based on fMRI evidence identifying the left mPFC activation in social support analgesia (Eisenberger et al., 2011; Younger et al., 2010). Future studies would provide interesting evidence by targeting both left and right mPFC.

In conclusion, this study used TBS alongside EEG and TMS-EEG to investigate the mechanisms underlying the influence of social support on pain. Our data demonstrated neural plasticity changes in the mPFC, which was associated with increased pain experience at the subjective and neural level in the context of viewing a stranger. In contrast, social support was associated with distinct neural activity and connectivity that could be modulated by plasticity changes in the mPFC activity. As such, it seems that stimulation of the mPFC can flexibly elicit context-dependent behavioral and brain connectivity changes surrounding social support and pain. Overall, our findings demonstrate the key role of the mPFC in social modulation of pain and potentially as a network hub regulating local and distributed brain activities.

Author contributions:

XC, RC, and BF contributed to the experimental design, data collection and analysis, and manuscript preparation. PF contributed to the experimental design and manuscript preparation. SC and NB contributed to the data analysis and manuscript preparation.

References:

Allison, T., Puce, A., McCarthy, G., 2000. Social perception from visual cues: role of the STS region. Trends in Cognitive Sciences 4, 267-278.

An, X., Bandler, R., Öngür, D., Price, J., 1998. Prefrontal cortical projections to longitudinal columns in the midbrain periaqueductal gray in macaque monkeys. Journal of Comparative Neurology 401, 455-479.

Bailey, N., Hoy, K., Rogasch, N., Thomson, R., McQueen, S., Elliot, D., Sullivan, C., Fulcher, B.,
Daskalakis, Z., Fitzgerald, P., 2018. Responders to rTMS for depression show increased frontomidline theta and theta connectivity compared to non-responders. Brain Stimulation 11, 190-203.
Baliki, M.N., Petre, B., Torbey, S., Herrmann, K.M., Huang, L., Schnitzer, T.J., Fields, H.L., Apkarian,
A.V., 2012. Corticostriatal functional connectivity predicts transition to chronic back pain. Nature Neuroscience 15, 1117-1119.

Beam, W., Borckardt, J.J., Reeves, S.T., George, M.S., 2009. An efficient and accurate new method for locating the F3 position for prefrontal TMS applications. Brain Stimulation 2, 50-54.

Berlim, M.T., McGirr, A., dos Santos, N.R., Tremblay, S., Martins, R., 2017. Efficacy of theta burst stimulation (TBS) for major depression: an exploratory meta-analysis of randomized and sham-controlled trials. Journal of Psychiatric Research 90, 102-109.

Canolty, R.T., Edwards, E., Dalal, S.S., Soltani, M., Nagarajan, S.S., Kirsch, H.E., Berger, M.S., Barbaro, N.M., Knight, R.T., 2006. High gamma power is phase-locked to theta oscillations in human neocortex. Science 313, 1626-1628.

Cash, R.F., Noda, Y., Zomorrodi, R., Radhu, N., Farzan, F., Rajji, T.K., Fitzgerald, P.B., Chen, R., Daskalakis, Z.J., Blumberger, D.M., 2017. Characterization of Glutamatergic and GABA A-Mediated Neurotransmission in Motor and Dorsolateral Prefrontal Cortex Using Paired-Pulse TMS–EEG. Neuropsychopharmacology 42, 502-511. Che, X., Cash, R., Chung, S.W., Fitzgerald, P., Fitzgibbon, B.M., 2018. Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis. Neuroscience & Biobehavioral Reviews 92, 437-452.

Che, X., Cash, R., Fitzgerald, P., Fitzgibbon, B.M., 2017. The Social Regulation of Pain: Autonomic and Neurophysiological Changes Associated with Perceived Threat. The Journal of Pain 19, 496-505. Che, X., Zhang, Q., Zhao, J., Wei, D., Li, B., Guo, Y., Qiu, J., Liu, Y., 2014. Synchronous activation within the default mode network correlates with perceived social support. Neuropsychologia 63, 26-33. Chung, S.W., Lewis, B.P., Rogasch, N.C., Saeki, T., Thomson, R.H., Hoy, K.E., Bailey, N.W., Fitzgerald, P.B., 2017. Demonstration of short-term plasticity in the dorsolateral prefrontal cortex with theta burst stimulation: A TMS-EEG study. Clinical Neurophysiology 128, 1117-1126.

Chung, S.W., Rogasch, N.C., Hoy, K.E., Fitzgerald, P.B., 2015. Measuring brain stimulation induced changes in cortical properties using TMS-EEG. Brain Stimulation 8, 1010-1020.

Chung, S.W., Rogasch, N.C., Hoy, K.E., Sullivan, C.M., Cash, R.F., Fitzgerald, P.B., 2018a. Impact of different intensities of intermittent theta burst stimulation on the cortical properties during TMS-EEG and working memory performance. Human Brain Mapping 39, 783-802.

Chung, S.W., Sullivan, C.M., Rogasch, N.C., Hoy, K.E., Bailey, N.W., Cash, R.F., Fitzgerald, P.B., 2018b. The effects of individualised intermittent theta burst stimulation in the prefrontal cortex: A TMS-EEG study. Human Brain Mapping. doi: 10.1002/hbm.24398

Conforto, A.B., Z'Graggen, W.J., Kohl, A.S., Rösler, K.M., Kaelin-Lang, A., 2004. Impact of coil position and electrophysiological monitoring on determination of motor thresholds to transcranial magnetic stimulation. Clinical Neurophysiology 115, 812-819.

Delorme, A., Makeig, S., 2004. EEGLAB: an open source toolbox for analysis of single-trial EEG dynamics including independent component analysis. Journal of Neuroscience Methods 134, 9-21. Downar, J., Sankar, A., Giacobbe, P., Woodside, B., Colton, P., 2012. Unanticipated rapid remission of refractory bulimia nervosa, during high-dose repetitive transcranial magnetic stimulation of the dorsomedial prefrontal cortex: a case report. Frontiers in Psychiatry 3, e30.

Du, X., Rowland, L.M., Summerfelt, A., Wijtenburg, S.A., Chiappelli, J., Wisner, K., Kochunov, P., Choa, F.-S., Hong, L.E., 2018. TMS evoked N100 reflects local GABA and glutamate balance. Brain Stimulation 11, 1071-1079.

Dumas, G., Nadel, J., Soussignan, R., Martinerie, J., Garnero, L., 2010. Inter-brain synchronization during social interaction. PLoS One 5, e12166.

Dunkel-Schetter, C., Bennett, T.L., 1990. Differentiating the cognitive and behavioral aspects of social support. In: Sarason, B.R., Sarason, I.G., Pierce, G.R. (Eds.), Wiley Series on Personality Processes. Social Support: An Interactional View. John Wiley, Oxford, pp. 267-296.

Eisenberger, N.I., Master, S.L., Inagaki, T.K., Taylor, S.E., Shirinyan, D., Lieberman, M.D., Naliboff,

B.D., 2011. Attachment figures activate a safety signal-related neural region and reduce pain experience. Proceedings of the National Academy of Sciences 108, 11721-11726.

Fuggetta, G., Fiaschi, A., Manganotti, P., 2005. Modulation of cortical oscillatory activities induced by varying single-pulse transcranial magnetic stimulation intensity over the left primary motor area: a combined EEG and TMS study. Neuroimage 27, 896-908.

Furman, A.J., Meeker, T.J., Rietschel, J.C., Yoo, S., Muthulingam, J., Prokhorenko, M., Keaser, M.L., Goodman, R.N., Mazaheri, A., Seminowicz, D.A., 2018. Cerebral peak alpha frequency predicts individual differences in pain sensitivity. Neuroimage 167, 203-210.

Gamond, L., Cattaneo, Z., 2016. The dorsomedial prefrontal cortex plays a causal role in mediating in-group advantage in emotion recognition: a TMS study. Neuropsychologia 93, 312-317. Gamond, L., Ferrari, C., La Rocca, S., Cattaneo, Z., 2017. Dorsomedial prefrontal cortex and cerebellar contribution to in-group attitudes: a transcranial magnetic stimulation study. European Journal of Neuroscience 45, 932-939.

Geuzaine, C., Remacle, J.F., 2009. Gmsh: A 3-D finite element mesh generator with built-in pre-and post-processing facilities. International journal for numerical methods in engineering 79, 1309-1331. Goldstein, P., Shamay-Tsoory, S.G., Yellinek, S., Weissman-Fogel, I., 2016. Empathy predicts an experimental pain reduction during touch. The Journal of Pain 17, 1049-1057.

Goldstein, P., Weissman-Fogel, I., Dumas, G., Shamay-Tsoory, S.G., 2018. Brain-to-brain coupling during handholding is associated with pain reduction. Proceedings of the National Academy of Sciences 115, E2528-E2537.

Goldsworthy, M.R., Pitcher, J.B., Ridding, M.C., 2012. The application of spaced theta burst protocols induces long-lasting neuroplastic changes in the human motor cortex. European Journal of Neuroscience 35, 125-134.

Gramfort, A., Papadopoulo, T., Olivi, E., Clerc, M., 2010. OpenMEEG: opensource software for quasistatic bioelectromagnetics. Biomedical Engineering Online 9, e45.

Hadjileontiadis, L.J., 2015. EEG-Based Tonic Cold Pain Characterization Using Wavelet Higher Order Spectral Features. IEEE Transactions on Biomedical Engineering 62, 1981-1991.

Hardmeier, M., Hatz, F., Bousleiman, H., Schindler, C., Stam, C.J., Fuhr, P., 2014. Reproducibility of functional connectivity and graph measures based on the phase lag index (PLI) and weighted phase lag index (wPLI) derived from high resolution EEG. PLoS One 9, e108648.

Hauck, M., Lorenz, J., Domnick, C., Gerloff, C., Engel, A.K., 2015. Top-down and bottom-up modulation of pain-induced oscillations. Frontiers in Human Neuroscience 9, e375.

Hill, A.T., Rogasch, N.C., Fitzgerald, P.B., Hoy, K.E., 2017. Effects of prefrontal bipolar and highdefinition transcranial direct current stimulation on cortical reactivity and working memory in healthy adults. Neuroimage 152, 142-157.

Huang, Y.-Z., Edwards, M.J., Rounis, E., Bhatia, K.P., Rothwell, J.C., 2005. Theta burst stimulation of the human motor cortex. Neuron 45, 201-206.

Iwabuchi, S.J., Raschke, F., Auer, D.P., Liddle, P.F., Lankappa, S.T., Palaniyappan, L., 2017. Targeted transcranial theta-burst stimulation alters fronto-insular network and prefrontal GABA. Neuroimage 146, 395-403.

Jensen, O., Mazaheri, A., 2010. Shaping functional architecture by oscillatory alpha activity: gating by inhibition. Frontiers in human neuroscience 4, e186.

Keefe, F.J., Caldwell, D.S., Baucom, D., Salley, A., Robinson, E., Timmons, K., Beaupre, P., Weisberg, J., Helms, M., 1996. Spouse-assisted coping skills training in the management of osteoarthritic knee pain. Arthritis & Rheumatology 9, 279-291.

Keefe, F.J., Caldwell, D.S., Baucom, D., Salley, A., Robinson, E., Timmons, K., Beaupre, P., Weisberg, J., Helms, M., 1999. Spouse-assisted coping skills training in the management of knee pain in osteoarthritis: long-term followup results. Arthritis & Rheumatism 12, 101-111.

Keizer, A.W., Verment, R.S., Hommel, B., 2010. Enhancing cognitive control through neurofeedback:
A role of gamma-band activity in managing episodic retrieval. Neuroimage 49, 3404-3413.
Koessler, L., Maillard, L., Benhadid, A., Vignal, J.P., Felblinger, J., Vespignani, H., Braun, M., 2009.
Automated cortical projection of EEG sensors: anatomical correlation via the international 10–10
system. Neuroimage 46, 64-72.

Korhonen, R.J., Hernandez-Pavon, J.C., Metsomaa, J., Mäki, H., Ilmoniemi, R.J., Sarvas, J., 2011. Removal of large muscle artifacts from transcranial magnetic stimulation-evoked EEG by independent component analysis. Medical & Biological Engineering & Computing 49, 397-407.

Krahé, C., Springer, A., Weinman, J.A., Fotopoulou, A., 2013. The social modulation of pain: others as predictive signals of salience–a systematic review 7, e386.

Kucyi, A., Moayedi, M., Weissman-Fogel, I., Goldberg, M.B., Freeman, B.V., Tenenbaum, H.C., Davis, K.D., 2014. Enhanced medial prefrontal-default mode network functional connectivity in chronic pain and its association with pain rumination. Journal of Neuroscience 34, 3969-3975.

Kucyi, A., Salomons, T.V., Davis, K.D., 2013. Mind wandering away from pain dynamically engages antinociceptive and default mode brain networks. Proceedings of the National Academy of Sciences 110, 18692–18697.

Lin, F.H., Belliveau, J.W., Dale, A.M., Hämäläinen, M.S., 2006. Distributed current estimates using cortical orientation constraints. Human Brain Mapping 27, 1-13.

Maris, E., Oostenveld, R., 2007. Nonparametric statistical testing of EEG-and MEG-data. Journal of Neuroscience Methods 164, 177-190.

Marshall, T.R., O'Shea, J., Jensen, O., Bergmann, T.O., 2015. Frontal eye fields control attentional modulation of alpha and gamma oscillations in contralateral occipitoparietal cortex. Journal of Neuroscience 35, 1638-1647.

Ni, Z., Gunraj, C., Kailey, P., Cash, R.F., Chen, R., 2014. Heterosynaptic modulation of motor cortical plasticity in human. Journal of Neuroscience 34, 7314-7321.

Nickel, M.M., May, E.S., Tiemann, L., Schmidt, P., Postorino, M., Dinh, S.T., Gross, J., Ploner, M., 2017. Brain oscillations differentially encode noxious stimulus intensity and pain intensity. NeuroImage 148, 141-147.

Nikouline, V., Ruohonen, J., Ilmoniemi, R.J., 1999. The role of the coil click in TMS assessed with simultaneous EEG. Clinical Neurophysiology 110, 1325-1328.

Nikulin, V.V., Kičić, D., Kähkönen, S., Ilmoniemi, R.J., 2003. Modulation of electroencephalographic responses to transcranial magnetic stimulation: evidence for changes in cortical excitability related to movement. European Journal of Neuroscience 18, 1206-1212.

Noda, Y., Cash, R.F., Zomorrodi, R., Dominguez, L.G., Farzan, F., Rajji, T.K., Barr, M.S., Chen, R., Daskalakis, Z.J., Blumberger, D.M., 2016. A combined TMS-EEG study of short-latency afferent inhibition in the motor and dorsolateral prefrontal cortex. Journal of Neurophysiology 116, 938-948. Noda, Y., Zomorrodi, R., Saeki, T., Rajji, T.K., Blumberger, D.M., Daskalakis, Z.J., Nakamura, M., 2017. Resting-state EEG gamma power and theta–gamma coupling enhancement following high-frequency left dorsolateral prefrontal rTMS in patients with depression. Clinical Neurophysiology 128, 424-432. Ong, W.-Y., Stohler, C.S., Herr, D.R., 2018. Role of the Prefrontal Cortex in Pain Processing. Molecular Neurobiology. doi: 10.1007/s12035-018-1130-9

Oostenveld, R., Fries, P., Maris, E., Schoffelen, J.-M., 2011. FieldTrip: open source software for advanced analysis of MEG, EEG, and invasive electrophysiological data. Computational Intelligence and Neuroscience 156869. doi: 10.1155/2011/156869

Parsons, C.E., Young, K.S., Mohseni, H., Woolrich, M.W., Thomsen, K.R., Joensson, M., Murray, L., Goodacre, T., Stein, A., Kringelbach, M.L., 2013. Minor structural abnormalities in the infant face

disrupt neural processing: a unique window into early caregiving responses. Social Neuroscience 8, 268-274.

Pelphrey, K.A., Morris, J.P., Mccarthy, G., 2004. Grasping the intentions of others: the perceived intentionality of an action influences activity in the superior temporal sulcus during social perception. Journal of Cognitive Neuroscience 16, 1706-1716.

Peng, W., Hu, L., Zhang, Z., Hu, Y., 2014. Changes of spontaneous oscillatory activity to tonic heat pain. PLoS One 9, e91052.

Peyron, R., García-Larrea, L., Grégoire, M.-C., Costes, N., Convers, P., Lavenne, F., Mauguière, F., Michel, D., Laurent, B., 1999. Haemodynamic brain responses to acute pain in humans: sensory and attentional networks. Brain 122, 1765-1780.

Ploner, M., Sorg, C., Gross, J., 2017. Brain rhythms of pain. Trends in cognitive sciences 21, 100-110.
Rastogi, A., Cash, R., Dunlop, K., Vesia, M., Kucyi, A., Ghahremani, A., Downar, J., Chen, J., Chen, R.,
2017. Modulation of cognitive cerebello-cerebral functional connectivity by lateral cerebellar
continuous theta burst stimulation. NeuroImage 158, 48-57.

Rogasch, N.C., Fitzgerald, P.B., 2013. Assessing cortical network properties using TMS–EEG. Human Brain Mapping 34, 1652-1669.

Rogasch, N.C., Sullivan, C., Thomson, R.H., Rose, N.S., Bailey, N.W., Fitzgerald, P.B., Farzan, F., Hernandez-Pavon, J.C., 2017. Analysing concurrent transcranial magnetic stimulation and electroencephalographic data: a review and introduction to the open-source TESA software. Neuroimage 147, 934-951.

Rogasch, N.C., Thomson, R.H., Daskalakis, Z.J., Fitzgerald, P.B., 2013. Short-latency artifacts associated with concurrent TMS–EEG. Brain Stimulation 6, 868-876.

Rogasch, N.C., Thomson, R.H., Farzan, F., Fitzgibbon, B.M., Bailey, N.W., Hernandez-Pavon, J.C., Daskalakis, Z.J., Fitzgerald, P.B., 2014. Removing artefacts from TMS-EEG recordings using independent component analysis: importance for assessing prefrontal and motor cortex network properties. Neuroimage 101, 425-439. Salomons, T.V., Dunlop, K., Kennedy, S.H., Flint, A., Geraci, J., Giacobbe, P., Downar, J., 2014. Restingstate cortico-thalamic-striatal connectivity predicts response to dorsomedial prefrontal rTMS in major depressive disorder. Neuropsychopharmacology 39, 488-498.

Sauseng, P., Griesmayr, B., Freunberger, R., Klimesch, W., 2010. Control mechanisms in working memory: a possible function of EEG theta oscillations. Neuroscience & Biobehavioral Reviews 34, 1015-1022.

Sauseng, P., Klimesch, W., 2008. What does phase information of oscillatory brain activity tell us about cognitive processes? Neuroscience & Biobehavioral Reviews 32, 1001-1013.

Schulz, E., May, E.S., Postorino, M., Tiemann, L., Nickel, M.M., Witkovsky, V., Schmidt, P., Gross, J., Ploner, M., 2015. Prefrontal gamma oscillations encode tonic pain in humans. Cerebral Cortex 25,

4407-4414.

Selesnick, I.W., Burrus, C.S., 1998. Generalized digital Butterworth filter design. IEEE Transactions on Signal Processing 46, 1688-1694.

Sheehan, D., Lecrubier, Y. (2001/2006). M.I.N.I SCREEN 5.0.0 / English version / DSM-IV July/1/06 Florida: University of South Florida-Tampa.

Tadel, F., Baillet, S., Mosher, J.C., Pantazis, D., Leahy, R.M., 2011. Brainstorm: a user-friendly application for MEG/EEG analysis. Computational Intelligence and Neuroscience 2011, e879716. Taesler, P., Rose, M., 2016. Prestimulus theta oscillations and connectivity modulate pain perception. Journal of Neuroscience 36, 5026-5033.

Tognoli, E., Lagarde, J., DeGuzman, G.C., Kelso, J.S., 2007. The phi complex as a neuromarker of human social coordination. Proceedings of the National Academy of Sciences 104, 8190-8195. Vachon-Presseau, E., Tétreault, P., Petre, B., Huang, L., Berger, S.E., Torbey, S., Baria, A.T., Mansour, A.R., Hashmi, J.A., Griffith, J.W., 2016. Corticolimbic anatomical characteristics predetermine risk for chronic pain. Brain 139, 1958-1970. Vinck, M., Oostenveld, R., Van Wingerden, M., Battaglia, F., Pennartz, C.M., 2011. An improved index of phase-synchronization for electrophysiological data in the presence of volume-conduction, noise and sample-size bias. Neuroimage 55, 1548-1565.

Windhoff, M., Opitz, A., Thielscher, A., 2013. Electric field calculations in brain stimulation based on finite elements: an optimized processing pipeline for the generation and usage of accurate individual head models. Human Brain Mapping 34, 923-935.

Younger, J., Aron, A., Parke, S., Chatterjee, N., Mackey, S., 2010. Viewing pictures of a romantic partner reduces experimental pain: Involvement of neural reward systems. PLoS One 5, e13309. Zalesky, A., Fornito, A., Bullmore, E.T., 2010. Network-based statistic: identifying differences in brain networks. Neuroimage 53, 1197-1207.

Zhang, Z., Hu, L., Hung, Y.S., Mouraux, A., Iannetti, G., 2012. Gamma-band oscillations in the primary somatosensory cortex—a direct and obligatory correlate of subjective pain intensity. Journal of Neuroscience 32, 7429-7438.

Figure legends:

Fig. 1. Experimental procedure.

Fig. 2. TMS delivery and electric field modelling of the study. (a) Both single-pulse and TBS were delivered to the F1 electrode in the 10-20 system. (b) Distribution of the induced normalized electric field (norm E) with maximal activation at the left mPFC.

Fig. 3. Pain ratings modulated by different TBS protocols. In the romantic partner condition, (a) – (c) show no effects of iTBS, cTBS, or Sham stimulation on pain ratings across the 3-minute pain task. (d) represents the area-under-curves (AUCs) of pain ratings across the pain task. In the stranger condition, (e) – (g) show the pain ratings across the pain task and (g) represents the AUC of pain dynamics. Stranger condition was associated with more pain from Pre- to Post-iTBS. ** indicates P < 0.01.

Fig. 4. Modulation of cortical activity assessed via TEPs following different TBS protocols. Grand average TEP waveforms from the three fronto-central electrodes (FC1, FCZ and FC2) for (a) iTBS, (b) cTBS and (c) Sham conditions. iTBS resulted in a larger N100 amplitude from Pre- to Poststimulation. (d) Scalp maps represent the comparison between iTBS-induced N100 from Pre- to Poststimulation (*left panel*), and between iTBS-induced N100 change and Sham-induced N100 change (*right panel*). X indicates P < 0.05, * indicates P < 0.01.

Fig. 5. Modulation of EEG oscillations by different TBS protocols. (a) iTBS increased fronto-central gamma power in the romantic partner condition (significant at F4, F6, FC4, FC6, and C4) (*left and middle panel*). Increased gamma activity was associated with larger N100 amplitude from Pre- to Post-iTBS (*right panel*). (b) iTBS increased central-parietal gamma power in the stranger condition (significant at T7, C5, C3, CP5, P7, P5 and O1). (c) – (d) show the increased central alpha power in the romantic partner and stranger condition respectively from Pre- to Post-Sham stimulation. Power is expressed as $10*\log_{10} (\mu V^2/Hz)$. X indicates *P* < 0.05, * indicates *P* < 0.01.

Fig. 6. EEG connectivity modulated by TBS protocols. (a) iTBS increased fronto-occipital alpha connectivity in the romantic partner condition. (b) In the stranger condition, iTBS increased central-

frontal and central-parietal theta connectivity. Large dots highlight the significant electrodes, and the color and thickness of the lines indicate the T statistics.

Table 1. Descriptive characteristics of the sample

	Sample size	Mean	Standard deviation
Gender	8 (M), 12 (F)	-	-
Age	20	26.45	4.54
BDI	20	2.2	2.46
STAY-Y2		36.5	6.47
MSPSS		6.03	0.82
PCS		12.9	9.39
DAS		117.55	10.15
ECR-R: anxiety		2.26	0.82
ECR-R: avoidance		2.13	0.77

M-male; F-female; SD-Standard Deviation; BDI-The Beck Depression Inventory; STAY-Y2-The State-Trait Anxiety Inventory-trait; MSPSS-The Multidimensional Scale of Perceived Social Support; PCS-The Pain Catastrophizing Scale; DAS-Dyadic Adjustment Scale; ECR-R-The Experiences in Close Relationships-Revised







Fig. 4 Click here to download 9. Figure: Fig. 4.pdf



Fig. 5 Click here to download 9. Figure: Fig. 5.pdf









All data and Matlab codes are available upon request from the corresponding author.

CHAPTER SEVEN

General Discussion

This thesis examined the role of social support in pain experience and the underpinning neurophysiological mechanisms. The main findings are summarized below, followed by a discussion of potential implications. Limitations of the research and future directions for the field are also presented.

7.1. Summary of Main Findings

7.1.1. Chapter 2: Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis

Chapter 2 provided a systematic review and meta-analysis investigating the social context in which social support may have an analgesic effect and the magnitude of the effect. The results indicated that the presence of another person without any form of interaction, regardless of a stranger or a significant other, was not sufficient to modulate pain perception (e.g. pain ratings, tolerability, or threshold). However, the presence of a significant other was related to increased facial expression of pain, which was consistently observed across various close relationship pairings (i.e. a parent, romantic partner, and a family member). In contrast, when social support was provided actively, analgesic effects were observed. Specifically, verbal communication of support decreased pain experience. The majority of evidence for this effect were derived from verbal support from a stranger. Moderate to large analgesic effects were also seen in response to social touch or visual representation within intimate relationships. This was not observed from strangers. This review also

provided the first attempt to identify the influence of social support on pain-related physiological arousal. Although social support overall tended to reduce physiological arousal, results were highly limited by the small number of studies in this area. Finally, in an overall assessment of the literature, publication bias for pain-related arousal was observed but not for pain experience.

The primary finding of this investigation is that the effects of social support on pain are contextdependent and differ in magnitude. Not only does the relationship with the person providing support matter, so does the delivery of social support. In particular, social support may have the most analgesic impact when it is clearly communicated (e.g. verbal communication) or explicitly expressed by a significant other (e.g. touch, visual representation). Together with the evidence from a systematic review (Krahé et al., 2013), these findings may address the discrepancy whereby social support shows mixed effects on pain (Brown et al., 2003; Gallant and Hadjistavropoulos, 2017; Karmann et al., 2014; Modić Stanke and Ivanec, 2010; Vervoort et al., 2011). In addition, the findings presented within this chapter on the magnitude of the effects outline potential directions to develop social support strategies in the management of pain (e.g. social touch, verbal communication).

7.1.2. Chapter 3: A systematic review of the processes underlying the main and the buffering effect of social support on the experience of pain

In Chapter 3, we systematically reviewed the psychological and behavioural processes linking social support to pain reduction within two primary models of general health literature, i.e. the main and the buffering effect hypothesis. We also systematically reviewed the neurobiological mechanisms through which social support may reduce pain experience. Overall, the findings indicated that social support may reduce pain through buffering the adverse influence of pain-related stress. Further breakdown of the buffering effect suggested that the social-buffering effect derived from social support reappraising pain-related stress and facilitating coping attempts. This review also summarised the neurobiological evidence which, while few in number, indicated reduced activity

within neural (e.g. ACC, hypothalamus, insular cortex) and physiological (e.g. heart rate, skin response, blood pressure) stress systems to pain in the context of social support. Finally, this review presented an integrated model, i.e. the social-buffering model, which summarised the behavioural and neurobiological literature to account for potential pathways through which social support ma have an analgesic influence.

This review is the first to present an integrative perspective of the processes and mechanisms by which social support may reduce pain. Previous studies have demonstrated separate evidence surrounding psychological (Holtzman and DeLongis, 2007; Vlaeyen et al., 2009) and behavioural processes (Holtzman et al., 2004; López-Martínez et al., 2008) as well as neural (Eisenberger et al., 2011; Younger et al., 2010) and physiological responses (Roberts et al., 2015) associated with the analgesic influence of social support.

However, most of the studies included in this review were cross-sectional correlational studies, in which pain experience and the threatening quality of pain were mixed and could be interchanged. Therefore, in combination with Chapter One, this review indicated the need for more investigation into the mechanistic evidence surrounding social support and pain. This leads to the following empirical studies within this thesis to investigate the mechanisms underpinning the influence of social support on pain, with key findings outlined below.

7.1.3. Chapter 4: The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat

Chapter 4 directly investigated the social-buffering hypothesis by introducing threat of pain before the onset of painful stimuli. This was the first study to separate threat of pain from pain experience. Using EEG and ECG, we recorded neural and autonomic response to the threat of pain in a six second period prior to the administration of a painful stimulus. Participants underwent this paradigm either

holding the hand of a significant other, a stranger, or no hand-holding. The behavioural results demonstrated that social support, as manipulated by holding hand of a significant other, decreased pain perception compared to no hand-holding. We also identified that holding hand of a significant other resulted in decreased heart rate and frontal theta activity (4-8 Hz) in the threat-of-pain phase. Using correlation analysis, we found that decreased heart rate and frontal theta activity were associated with less pain perception in the significant other compared with no-holding condition. Significant neural changes in frontal theta activity were further localized to the insular cortex and the rostral-ventral portions of anterior cingulate cortex which were commonly observed in the processing of pain and stress. These findings support the buffering effect whereby attenuated threat response to pain can predict pain reduction in the context of social support. Notably, holding hand of a stranger had no overall effect on pain perception or autonomic response but decreased posterior theta activity compared to no hand-holding.

The study is significant as it directly demonstrates mechanistic processes involved in the buffering effect through which social support may act to reduce pain. Specifically, social support may decrease neural (i.e. frontal theta) and autonomic (i.e. heart rate) threat response to pain which is further associated with pain reduction. The results of this study also indicated the significance of relationship quality in the social modulation of pain as holding hand of a stranger had no overall effect on pain perception or autonomic response.

7.1.4. Chapter 5: Distinct neural dynamics and autonomic activity in response to viewing a romantic other compared to a stranger during pain: An EEG and HRV investigation

Chapter 5 presented a study investigating the neural dynamics and parasympathetic activity associated with the effects of social support on pain. This extended Chapter 4 by providing evidence on how social support may modulate neural oscillations and parasympathetic activity supporting the processing of pain and threat. In this study, social support was manipulated by visual representation

of an intimate other or a stranger while participants underwent cold pain over several minutes. The results indicated that social support reduced pain perception, with the effect being most prominent in the phases when pain experience stabilises and then desensitises. Further, participants reported higher overall perceived support in the social support condition, which was associated with lower pain perception. The neural oscillation data revealed higher frontocentral alpha activity (8-12 Hz) and lower central gamma activity (31-100 Hz) in the social support compared to the stranger condition. A correlation was observed between decreased pain and reduced central gamma activity in the social support was associated with higher HF-HRV compared to the stranger condition.

The results of this study are significant as they demonstrate oscillatory changes supporting the role of social support in modulating the processing of nociceptive information. Previous studies have suggested that social support may buffer pain-related neural systems, such as the ACC and insular cortex (Eisenberger et al., 2011; Younger et al., 2010). Using EEG, this study demonstrated increased alpha but decreased gamma activity in the social support condition, which are thought to underlie early nociceptive attention and the subsequent subjective experience of pain respectively (Nickel et al., 2017; Peng et al., 2014). Social support may therefore buffer the gating and integration of painful information as relevant to the changes in ACC and insular cortex observed in earlier fMRI studies.

The results of this study also identified increased PNS activity in the social support condition. The literature to date has focused on the SNS whereby social support may be able to decrease sympathetic arousal to pain (Roberts et al., 2015; Sambo et al., 2010). This study provided the first line of evidence that social support may increase parasympathetic activity to pain which is believed to be associated with the regulatory control over sympathetic arousal (Thayer et al., 2009). Overall, the neural dynamics and PNS activity add a novel understanding to how social support may reduce pain.

7.1.5. Chapter 6: The medial prefrontal cortex as a flexible hub mediating behavioural as well as local and distributed neural effects of social support on pain: a theta burst stimulation and TMS-EEG Study Chapter 6 contained a study to explore the causal role of the mPFC in orchestrating the behavioural and neural connectivity effects of social support on pain. The mPFC is thought to process social intimacy (Gamond and Cattaneo, 2016; Gamond et al., 2017) and is positively associated with perceived social support (Che et al., 2014). In this study, state-of-the-art neurophysiological methods were applied to modulate the activity of the mPFC, as well as explore changes in frequency dynamics and neural connectivity. Participants underwent a three-session protocol in which the activity in the mPFC was either increased (iTBS), decreased (cTBS), or not changed (Sham). Pain in the context of social support was manipulated both Pre- and Post-stimulation using the same protocol as Chapter 5. The results indicated that iTBS over the mPFC modulated pain perception and neural connectivity in a context-dependent manner. Specifically, in the stranger condition, iTBS increased pain perception and neural connectivity between central regions and frontoparietal regions from Pre- to Post-stimulation. In contrast, in the social support condition, iTBS did not modulate pain perception and increased connectivity only between frontal and occipital regions from Pre- to Post-stimulation. In line with these changes, iTBS resulted in a larger N100 amplitude around the frontal cortex as an index of neuroplasticity. However, neither cTBS nor Sham stimulation elicited neural or behavioural changes.

Compared to previous studies exploring brain activity supporting social support effects on pain (Eisenberger et al., 2011; Younger et al., 2010), this study directly modulated the activity of the mPFC using TBS. In contrast to the null hypothesis, iTBS produced increased pain perception in the stranger condition, which may be associated with the role of the mPFC in inffering the help motivation of another person (Harmer et al., 2001; Schuwerk et al., 2014). However, iTBS over the mPFC did not reduce pain perception in the social support condition. This study was also able to

provide novel evidence on brain-wide dynamics, whereby increased alpha band connectivity after iTBS may underpin the processing of social support (Goldstein et al., 2018) in the romantic partner condition and increased theta band connectivity may support pain-related information integration in the stranger condition (Taesler and Rose, 2016). Overall, these findings suggest that the mPFC has the capacity to causally modulate the influence of social support on pain perception and associated neural coupling in a context-dependent manner.

In addition, this study indicted a larger N100 amplitude following iTBS, which demonstrated neuroplasticity changes related to social support effects on pain. The N100 may reflect the shifts in the inhibition-excitation balance in cortical circuits (Du et al., 2018; Rogasch and Fitzgerald, 2013), although its specific neurophysiological functioning remains to be established.

7.2. Implications

There are a number of key implications arising from the findings of this thesis. These findings may theoretically improve our understanding of the social context in which social support may have an analgesic influence and the underlying mechanisms. This thesis also introduced novel neurophysiological techniques that may be used to explore the mechanistic evidence surrounding social support and pain. Indeed, the work presented here represents the first application of EEG, TBS and TMS-EEG to investigate neural oscillations, neural connectivity, and neuroplasticity underpinning the influence of social support on pain. While not directly explored in this thesis, the findings from this body of work may help optimise the use of social support strategies in the management of pain by better understanding the contexts and mechanisms of social support in pain.

7.2.1. Understanding the Effects and Mechanisms of Social Support on Pain

This thesis has contributed to understanding the complex influence of social support on pain. The literature to date has suggested mixed effects of social support on pain (Brown et al., 2003; Gallant and Hadjistavropoulos, 2017; Vervoort et al., 2011). Chapter 2 (Study One) explored these effects and demonstrated them to be context-dependent that how support is delivered (i.e. actively or passively) matteres as dose who is providing the support. Moreover, this study quantified these effects using a meta-analysis. Overall, social support may have the strongest protective effect on pain when it is clearly expressed through verbal communication or explicitly delivered (e.g. touch, visual representation) by an intimate other. This conclusion is further supported by the empirical studies in the subsequent chapters (Chapter 4 and 5).

The findings from the thesis also improve our understanding of the mechanisms through which social support can decrease pain. Chapter 3 (Study Two) presented an integrated perspective, the social-buffering model of pain reduction, which indicated that social support may reduce pain through modifying the behavioural and neurobiological stress systems. Chapter 4 (Study Three) directly demonstrated the buffering hypothesis in which social support reduced pain by buffering the neural and autonomic responses to the threat of pain. In Chapter 5 (Study Four), social support was found to modulate the gating (i.e. alpha power) and integration (i.e. gamma power) of nociceptive information. Changes in alpha and gamma activity are suggested to be generated from the sensorimotor cortex and prefrontal cortex respectively which encode stimulus intensity and sensory integration of pain (Nickel et al., 2017; Schulz et al., 2015). Therefore, Chapter 5 added further evidence to the social-buffering model in which reduced gating and integration of nociception may be closely related to brain activities encoding pain and threat. In Chapter 5, social support was also associated with increased bodily control over pain-related arousal (i.e. HF-HRV), which is consistent with the attenuated neurobiological stress systems in the social-buffering model. In addition, Chapter 6 (Study Five) demonstrated a causal role of the mPFC in orchestrating neural

networks supporting the processing of social support (i.e. alpha coupling) and pain (i.e. theta coupling). This indicates the relevance of the mPFC in priming feelings of attachment and buffering pain and related threat (Eisenberger and Cole, 2012; Eisenberger et al., 2011). Overall, the results of these series of findings indicate a mechanistic pathway in which social support may suppress neurophysiological response to pain that possibly arising from its priming influence on perceived attachment.

Beyond the social context, findings in this thesis also extend the neural mechanisms of pain. The insular cortex and ACC are suggested to represent interoception and to produce behavioural drive respectively in the 'interoception' perspective (Craig, 2002, 2003a, b). These two regions are also the key nodes of the 'salience network' (Legrain et al., 2011; Seeley et al., 2007). As shown in Chapter 4 (Study Three), our finding demonstrates that ACC and insula activity may be represented as EEG theta oscillation in the context of upcoming painful stimuli. Therefore, this result may add neurophysiological evidence to the pain mechanisms.

7.2.2. Novel Neuroimaging Techniques to Reveal the Mechanistic Evidence

Another implication of this thesis is the use of novel neuroimaging techniques to investigate complex human experiences, in this case how social support may influence pain experience from a mechanistic perspective. Compared to functional imaging and neurophysiological recordings, TBS is capable of modulating brain activity (Chung et al., 2016; Chung et al., 2017; Huang et al., 2005) and therefore provides a window to investigate the brain-behavioural relationships. In Chapter 6 (Study Five), activity of the mPFC was modulated using TBS and the results demonstrated a causal role of the mPFC in mediating the behavioural and network changes related to social support and pain. Furthermore, neuroplasticity changes caused by TBS were assessed using TMS-EEG in this study. By analysing TEPs and TMS-evoked oscillations TMS-EEG can provide valuable information on neuroplasticity changes underpinning behavioural performances (Chung et al., 2018a; Chung et al.,

2018b; Hill et al., 2017). Interestingly, Chapter 6 found a positive relationship between a larger N100 amplitude and frontal gamma activity following iTBS. This finding provides direct evidence that neuroplasticity changes induced by iTBS are associated with brain activity involved in the social modulation of pain. Overall, these techniques advanced our understanding of how the mPFC may mediate the influence of social support on pain. This work extends the previous work using fMRI (Eisenberger et al., 2011; Younger et al., 2010) by causally modulating the activity in the mPFC and measuring changes in neural coupling and neuroplasticity. To our knowledge, this is the first study to use TBS and TMS-EEG in the social modulation of pain and we expect it to lead to future TMS and TMS-EEG studies on this topic due to its ability to provide rich information on causality and neuroplasticity.

Further, this thesis simultaneously assessed neural and autonomic responses which is important to understand the integration of the CNS and ANS as well as their relevance in pain and social support. Chapter 4 (Study Three) indicated attenuated theta activity and a smaller heart rate acceleration evoked by the threat of pain under social support. In Chapter 5 (Study Four), social support decreased the gating (i.e. alpha power) and integration (i.e. gamma power) of painful information but increased bodily control over pain-related arousal (i.e. HF-HRV). Although the direct association between neural and autonomic responses were not found, these studies provided comprehensive evidence on how social support can help manage the neural and physiological systems mediating pain and related threat.

In addition, brain-wide dynamics were evaluated in Chapter 6 (Study Five), which demonstrated network configurations potentially supporting the processing of social support and pain. This is highly novel as it provides a network perspective to account for the social modulation of pain beyond the separate brain activations.

7.2.3. Translational Implications for Social Support Strategies in the Management of Pain In the past three decades, studies included a spouse in the management of chronic pain (Abbasi et al., 2012; Keefe et al., 2004; Keefe et al., 1996, 1999; Martire et al., 2003; Martire et al., 2007, 2008; Moore and Chaney, 1985; a family member in Radojevic et al., 1992). These programs usually included coping skills training (e.g. distraction, activity-based skills, and cognitive restructuring) and couple skills training aimed to supplement the effects of coping skills. Most of the studies indicated that adding a spouse resulted in less pain and/or increased psychological adjustment to pain (Abbasi et al., 2012; Keefe et al., 2004; Keefe et al., 1996, 1999; Martire et al., 2003; Martire et al., 2008; Radojevic et al., 1992), with a few studies demonstrating no add-on effects of a spouse (Martire et al., 2007; Moore and Chaney, 1985). The findings from this thesis may have implications for improving the integration of a romantic partner, or alternative a close other, in pain management.

A key finding of the thesis was that social support may reduce pain through modifying the neural and physiological stress systems to pain and related threat (Chapter 3 to 5). This finding indicates the critical role of social support in helping people with chronic pain reappraise the threatening quality of pain and the perceived efficacy to cope with pain. Indeed, in the spouse-assisted pain management programs, patients who had increased self-efficacy over the course of the program had significant improvement in pain, physical disability, and psychological adjustment (Keefe et al., 1996). Moreover, the modulating impact of self-efficacy on these outcomes was observed at 6month and 12-month follow-up (Keefe et al., 1999). Therefore, social support strategies may have better outcomes whereby the support provider can help people with chronic pain feel less threatened by the chronic condition and enhance their perceived competence and efficacy to cope with pain.

Another important finding of the thesis was the significance of intimate relationships in pain reduction (Chapter 2, 5, and 6). An intimate relationship is suggested to increase emotional

communication between the couple (Goldstein et al., 2017) and to prime feelings of attachment and reward (Eisenberger et al., 2011; Younger et al., 2010). Keefe and colleagues (1996, 1999) demonstrated that the efficacy of a spouse-assisted pain management program was largely determined by the changes in marital adjustment over the program, whereby patients showed increased marital satisfaction had much better outcomes than others. Therefore, social support therapies may wishto include an intimate other where possible. Moreover, perceived intimacy could be an area of therapeutic focus in the spouse-assisted pain management programs, potentially through mutual activities and verbal communication and validation (Keefe et al., 1996, 1999). That is, including a focus on improving relationship quality as part of the pain-management program.

The findings from this thesis also indicated the importance of how social support is presented. Specifically, conveying social support in an explicit manner (e.g. through verbal support) may reduce pain. In contrast, the mere presence of a significant other may not be sufficient to reduce pain and may even increase the facial expression of pain. This may be caused by the pain sufferer being unclear of the social partner's intention and ability to help (Krahé et al., 2013), or the presence of a significant other enhancing pain behaviours in order to seek empathy (Williams, 2002). These findings suggest that supportive others in pain management should provide support in a clear manner. It also suggests that the mere presence of a significant other should be avoided, or behaviours and interactions addressed, where it results in increased pain behaviours.

In addition to spouse-assisted programs, this thesis also supports the efficacy of self-help support groups in the management of chronic pain. One study revealed increased functional ability and decreased recourse to health professionals following a self-help support group, which was characterized by participating group outdoor activities, sharing knowledge on the chronic pain condition, and developing mutual support between group members (Subramaniam et al., 1999). Chapter 2 (Study One) indicated verbal support, even from a stranger, can communicate support

and reduce pain perception. Chapter 5 and 6 further suggested the significance of intimate relationship in decreasing pain and nociceptive information integration. These findings collaborate the self-help support groups in promoting verbal communication and building close relationships between group members.

This thesis also suggests the use social support strategies in pain-related procedures. There is evidence that gentle social touch can effectively reduce pain and crying in newborns undergoing a procedure (e.g. heel lance procedure) (Gray et al., 2000; Herrington and Chiodo, 2014). Our results demonstrated moderate to larger effects of social support on pain by social touch. In addition to skin-to-skin touch, our findings also indicate other social support variants, e.g. verbal support, which may be used to manage pain in other age groups (e.g. children, adults) and procedures (e.g. dental procedures).

7.3. Methodological Considerations

There are several important considerations which apply across the empirical studies in this thesis. Awareness of these will be important for future investigations to explore the effects and mechanisms of social support on pain. For limitations specific to individual empirical studies, the reader is directed to the relevant thesis chapters.

7.3.1. EEG Analysis

EEG has been extensively used as a neuroimaging technique. Adequate offline data cleaning steps are essential for the removal of various, often large-amplitude artefacts in EEG data. There are different pipelines to clean EEG data and a semi-automatic pipeline running under MATLAB (The MathWorks Inc, Natick, MA) was used in this thesis. An important limitation of this pipeline is that components representing artefact must be manually selected for removal by the experimenter, which could potentially lead to a level of subjectivity. We have tried to concatenate the data from different experimental conditions to reduce the subjectivity. There are also fully-automatic cleaning pipelines (Bigdely-Shamlo et al., 2015; Winkler et al., 2011), which in a way can control the subjectivity but the results may be influenced by the embedded algorithm and thresholding. As EEG analysis techniques continue to evolve, it would be essential to develop more objective frameworks for cleaning and analysing EEG data.

7.3.2. Sampling Considerations

All of the empirical studies in this thesis recruited healthy participants from the local universities and neighbourhoods as approved by the Alfred Hospital and Monash University Human Resources and Ethics Committee. This may limit the translation of our findings to chronic pain populations. Both behavioural patterns and the central nervous system undergoes significant change in chronic pain (Apkarian et al., 2011; Lumley et al., 2011). Therefore, behavioural and neural responses to social support in clinical groups may have a differential effect as seen in this thesis. There are several studies investigating the role of social support in chronic pain in experimental settings. One study found that the presence of a romantic partner reduced pain and related somatosensory activity in fibromyalgia (Montoya et al., 2004). Another study demonstrated decreased daily pain in the visual representation of a significant other among people with chronic pain (Shaygan et al., 2017). These studies provide valuable information on the effects and mechanisms of social support in chronic pain and more studies of this are needed.

The sample sizes used in the present studies were also relatively small (N = 18-23). A limitation in Chapter 4 was that a power analysis was not performed. For the studies described in Chapters 5 and 6, we performed a power analysis using G*Power (<u>http://www.gpower.hhu.de/en.html</u>). The analyses yielded a sample size of 24, which should provide a power of greater than 95% to identify a large effect size at an alpha value of 0.05 when using a Repeated Measures ANOVA. Further, the samples in the thesis have a relatively narrow age range, mainly consist of young adults. This may limit the generation of the conclusions to other age ranges, e.g. older adults or children. In addition, the sample sizes did not allow the examination of sex differences in the social modulation of pain. Sex effects may be of note, given research suggesting female pain recipients may benefit from social support more than men (Jackson et al., 2005), as well as biological sex differences which may influence mechanisistic processes. Future studies might wish to use a larger sample with wider age range, capable of exploring sex effects.

7.3.3. Social Support Paradigms

It is noted that both social touch and primed social support were manipulated in this thesis, as they both have been demonstrated to have a protective effect on pain experience (Chapter 2, Study One) and are both ecologically valid. However, it is possible that different social support paradigms have different effects, as our systematic and meta-analysis review revealed (Chapter 2). In the work of this thesis, the paradigms differed due to differences in the requirements of the study. For chapters five and six, participants were required to attend for three sessions. Thus, we chose to use primed social support to increase the likelihood we would be able to recruit volunteers. Human touch communicates distinct emotions (Hertenstein et al., 2006) and thus it provides an excellent protocol to investigate distinct relationships. Indeed, social touch by a romantic partner may promote intimacy (Goldstein et al., 2017) while by a stranger may be somewhat socially uncomfortable (Krahé et al., 2013). But it remains to be determined the shared and distinct effects between social touch by a romantic partner and by an acquaintance. It is also noted that no hand-holding was used as a control condition which may differ from social touch in sensory input and attention and thus holding an object would be considered in future studies. In addition, socially desirable effects need to be considered when participants reported pain experience in the presence of a significant other. Similarly, primed social support was found to reduce pain (Chapter 5, Study Four) and this effect was
suggested to arise from its role in priming feelings of attachment (Chapter 6, Study Five) and modulating the transmission and integration of nociceptive information (Chapter 5, Study Four). This paradigm was used in this thesis to be compatible with previous functional imaging studies (Eisenberger et al., 2011; Younger et al., 2010). But this paradigm is slightly different from the proposed spousal pain therapies whereby a spouse is present. Findings from this paradigm indicate the potential benefits of a mental imaginary component in spousal pain coping therapies. These findings also have direct implications for coping with procedure pain, e.g. a dental procedure.

The majority of social support protocols in the literature have targeted on emotional support. Other types of social support, such as the informational and instrumental support, have been less investigated. It is possible that different types of social support may have distinct effects on pain experience (Cohen and Wills, 1985; DeLongis et al., 1988; Lee et al., 2016), and thus a more comprehensive understanding on the impact of multiple forms social support types is needed. We also used different types of close relationships in chapter four. It is possible that different types of close relationships have different effects (Chapter 2) and more research is required into this area.

7.3.4. Relationship Quality

The empirical studies did not require a certain level of relationship quality between the pain recipient and the support provider for eligibility. This may limit the comparability of our findings to other studies which only included early stage romantic relationship (Younger et al., 2010) or highly satisfied relationship (Coan et al., 2006). Nonetheless, we demonstrated significant pain reduction in Chapter 3 and 5. Moreover, we have recorded the relationship quality in the last two empirical studies but found no association between relationship quality and pain or neurophysiological changes. In contrast, we did observe a positive relationship between overall perceived support and pain reduction in Chapter 5. Whilst relationship quality was more general, overall perceived support

was measured during the pain task and specific to the ongoing pain. It is possible that situational (e.g. online social support) and global (e.g. relationship quality) social support has distinct roles in pain experience (Corley et al., 2016). Future studies are needed to further examine the distinct impact of situational and global support on pain.

7.3.5. Pain Induction

It is noted that bottled iced water was used to induce ongoing pain in chapter 5 and chapter 6, compared to previous studies in which cold pressor test (CPT) was used (Brown et al., 2003; Edwards et al., 2017). A recent study demonstrated the effectiveness of bottle iced water in inducing cold pain (Hadjileontiadis, 2015). We chose this method as it can induce ongoing cold pain that is tolerable to the participants for the duration of the experimental conditions. The CPT was problematic in this context as participants may need to withdraw their hand prior to the completion of the condition. However, one limitation of the bottled iced water was condensation on the surface of the bottle. To control for this, we used a fresh bottle before each condition and precisely controlling the timing to take out the bottle from the freezer to minimise condensation.

7.3.6. Potential Additional Modulating Factors

Pain is influenced by a wide range of psychological and social factors that are integrated, along with the sensory input, to create a pain experience. Current evidence has suggested the particular relevance of several personality and social factors in the social modulation of pain, including adult attachment style, pain catastrophizing, and the motivation to help by the social partner. Adult attachment style describes individual differences in representational models of close relationships (Bowlby, 1969). Sambo and colleagues (2010) found that individuals with high attachment avoidance reported more pain in the presence relative to the absence of a stranger. Avoidant individuals tend to mistrust social relationships (Bowlby, 1969; Feeney and Noller, 1990; Main, 2000), thus the

presence of a total stranger may be associated with increased anxiety and pain perception (Sambo et al., 2010). In a following study, avoidant individuals reported more pain and N2 and P2 laserevoked potentials (LEPs) in the presence relative to the absence of a romantic partner. These findings further corroborated the modulating role of attachment style and suggested mechanistic evidence in which attachment avoidance may modulate the perceived salience of painful stimuli (Krahé et al., 2015).

Another important factor that may influence the impact of social support on pain is pain catastrophizing. Pain catastrophizing is defined as 'an exaggerated negative mental set brought to bear during actual or anticipated painful experience' (Sullivan et al., 2001). In the presence of a stranger, high pain catastrophisers express more pain behaviours (e.g. facial displays, vocalizations) and utilise fewer coping strategies that might minimise pain (Sullivan et al., 2004). In contrast, another study found no difference in pain expression among high-catastrophizing children when alone or in the presence of a parent. Instead, low-catastrophizing children expressed more pain in the presence of a parent (Vervoort et al., 2011). Pain catastrophizing may be also relevant to the role of social support in chronic pain conditions. In a group of chronic pain population, pain catastrophizing was found to interact with social support in the prediction of emotional health (Holtzman and DeLongis, 2007). Therefore, while the influence of pain catatrophisation has not shown consistent effects, the modulating role of pain catastrophizing warrants further investigations. In addition, there are other pain-related cognitions that may also have an impact, such as helplessness, fear of movement/(re)inury, and sense of mastery (Bunketorp et al., 2006; Waltz et al., 1998).

A line of research indicates motivation of the social partner to help the person in pain to be of particular importance in determining the effects of social support on pain experience. The spouses of people with chronic pain may provide help for autonomous or volitional motives (e.g. enjoyment,

full commitment) or rather for controlled or pressured (e.g. avoiding guilt and criticism) motives (Kindt et al., 2016). Studies have largely indicated that social support may have protective effects on emotional health of people with chronic pain only when the help is out of volitional motives (Kindt et al., 2018a; Kindt et al., 2018b). Moreover, this effect is suggested to be accounted for by the improvements in the need satisfaction of the pain sufferer (Kindt et al., 2015; Kindt et al., 2016). Future studies may wish to further characterise the role of helping motivation in social support effects on pain, and the influence of this on mechanistic findings.

7.4. Future Directions

The present findings provide the initial groundwork for a number of future explorations. In addition to large sample replication studies of those presented here, this also includes evaluating the synchronisation between the pain recipient and the support provider using hypescanning, investigating the role of oxytocin in the social modulation of pain, social network analysis, using TMS and TMS-EEG in social support and pain, and investigating the role of helping motivation of the social partner.

7.4.1. Using Hypescanning to Reveal the Synchronisation between Social Partners

Recent advances in neuroimaging techniques allow to simultaneously monitor the brain or body activity of several persons engaged in an interpersonal mutual exchange ('hypescanning') (Babiloni and Astolfi, 2014; Dumas, 2011; Dumas et al., 2011). Due to its high temporal resolution, EEG has been largely used to evaluate synchronised brain activity using hypescanning systems (Wang et al., 2018). In a recent study, Goldstein and colleagues (2018) found increased between-partner alpha band coupling while individuals held hand of a romantic partner during pain administration. Moreover, increased coupling in this network was associated with less pain and increased

empathetic accuracy of the romantic partner. This study provides the first line of evidence that brain-to-brain coupling is involved in the analgesic influence of social support. In another study, the authors from the same group demonstrated decreased pain and increased heart rate coupling between romantic couples during social touch and pain administration. Beyond pain research, hypescanning has been used in a variety of social interactions, e.g. imitation (Dumas et al., 2010), social coordination (Tognoli et al., 2007). This technique will provide valuable information on how synchronised activity between the pain recipient and the social partner may mediate the effects of social support on pain.

7.4.2. Oxytocin in the Social Modulation of Pain

The hypothalamic peptide oxytocin has been largely linked to positive social interactions (Chen et al., 2011; Holt-Lunstad et al., 2008; Light et al., 2005) and has analgesic effects (González-Hernández et al., 2014; Paloyelis et al., 2016). Further, intranasal administration of oxytocin enhanced the buffering effect of social support on cortisol response to a psychosocial stressor (Heinrichs et al., 2003). Only recently has one study directly explored the role of oxytocin in the analgesic effect of social support (Kreuder et al., 2018). In this study, intranasal administration of oxytocin increased the analgesic effect of social support, which was associated with a stronger decrease in anterior insular, a larger increase in the middle frontal gyrus, and a higher functional coupling between these two regions. This study firstly demonstrates that oxytocin may increase the benefits of social support on pain and that this effect may be mediated by brain activity involved in cognitive control and pain inhibition. Future studies are needed to further reveal how oxytocin can mediate the analgesic influence of social support. In addition, oxytocin may be administrated in partner-assisted pain management therapies to potentally increase the benefits of social support.

7.4.3. Social Network Analysis

Social network analysis (SNA) is an emerging technique in sociology which may serve as a valuable tool in the investigation of social support and pain. SNA characterizes networked structures in terms of nodes (e.g. a person within the network) and links (e.g. relationships or interactions) as well as a series of other characteristics (e.g. bridge, centrality) (Otte and Rousseau, 2002). Using SNA, a recent study demonstrated a positive association between pain tolerance of an individual and the pain tolerance of his or her friends. Specifically, for every one second increase in friends' average pain tolerance, the expected pain tolerance of the individual increased by 0.21 second (Engebretsen et al., 2018). Using this technique a variety of questions could be addressed surrounding social support and pain. For example, from which types of social support can a certain pain sufferer benefit the most? How the strength of relationship can affect pain perception? What are the changes in social networks and social relationships as a result of pain? It is expected to see more SNA investigations in social support and pain, providing rich information in the broad impact of an individual's social network. This has the potential to provide novel information outside of the targeted and specific relationship pairings currently used within this field (e.g. romantic other, family etc).

7.5. Conclusions

Social support plays a role in pain experience, however, the influence of social support on pain is context-dependent. The findings presented in this thesis contribute to our understanding of this effect. To our knowledge, this thesis provides the first meta-analysis of the magnitude of effects of different social support contexts, and it provides the first EEG, TBS and TMS-EEG investigation of potential underlying mechanisms. Overall, the findings presented here support that social support generally has a protective effect on pain, particularly when it is clearly expressed by a significant other. In addition, the analgesic influence of social support may be associated with the role of social support in buffering neurophysiological stress systems to pain, impacting on the integration and

gating of pain signals. Finally, we provide a world-first study, directly modulating the activity of the medial prefrontal cortex, and identifying the role of this region in the social-buffering effect through orchestrating the neural networks underpinning social support and pain. It is hoped that this body of work will contribute to the understanding of the effects and potential mechanisms of social support in pain which may ultimately have therapeutic implications and improve the lives of those living with pain.

REFERENCES

Abbasi, M., Dehghani, M., Keefe, F., Jafari, H., Behtash, H., Shams, J., 2012. Spouse-assisted training in pain coping skills and the outcome of multidisciplinary pain management for chronic low back pain treatment: A 1-year randomized controlled trial. European Journal of Pain 16, 1033-1043.

Albe-Fessar, D., Berkley, K., Kruger, L., Ralston III, H., Willis Jr, W., 1985. Diencephalic mechanisms of pain sensation. Brain Research Reviews 9, 217-296.

Amanzio, M., Benedetti, F., 1999. Neuropharmacological dissection of placebo analgesia: expectation-activated opioid systems versus conditioning-activated specific subsystems. Journal of

Neuroscience 19, 484-494.

Apkarian, A.V., Bushnell, M.C., Treede, R.-D., Zubieta, J.-K., 2005. Human brain mechanisms of pain perception and regulation in health and disease. European Journal of Pain 9, 463-484.

Apkarian, A.V., Hashmi, J.A., Baliki, M.N., 2011. Pain and the brain: specificity and plasticity of the brain in clinical chronic pain. Pain 152, S49-S64.

Avenanti, A., Bueti, D., Galati, G., Aglioti, S.M., 2005. Transcranial magnetic stimulation highlights the sensorimotor side of empathy for pain. Nature Neuroscience 8, 955-960.

Babiloni, F., Astolfi, L., 2014. Social neuroscience and hyperscanning techniques: past, present and future. Neuroscience & Biobehavioral Reviews 44, 76-93.

Baliki, M.N., Chialvo, D.R., Geha, P.Y., Levy, R.M., Harden, R.N., Parrish, T.B., Apkarian, A.V., 2006.
Chronic pain and the emotional brain: specific brain activity associated with spontaneous
fluctuations of intensity of chronic back pain. Journal of Neuroscience 26, 12165-12173.
Baliki, M.N., Geha, P.Y., Apkarian, A.V., Chialvo, D.R., 2008. Beyond feeling: chronic pain hurts the
brain, disrupting the default-mode network dynamics. Journal of Neuroscience 28, 1398-1403.
Bantick, S.J., Wise, R.G., Ploghaus, A., Clare, S., Smith, S.M., Tracey, I., 2002. Imaging how attention
modulates pain in humans using functional MRI. Brain 125, 310-319.

Başar, E., Schmiedt-Fehr, C., Öniz, A., Başar-Eroğlu, C., 2008. Brain oscillations evoked by the face of a loved person. Brain Research 1214, 105-115.

Bender, S., Basseler, K., Sebastian, I., Resch, F., Kammer, T., Oelkers-Ax, R., Weisbrod, M., 2005. Electroencephalographic response to transcranial magnetic stimulation in children: Evidence for giant inhibitory potentials. Annals of Neurology 58, 58-67.

Benedetti, F., Arduino, C., Amanzio, M., 1999. Somatotopic activation of opioid systems by targetdirected expectations of analgesia. Journal of Neuroscience 19, 3639-3648.

Bernardes, S.F., Forgeron, P., Fournier, K., Reszel, J., 2017. Beyond solicitousness: a comprehensive review on informal pain-related social support. Pain 158, 2066-2076.

Bigdely-Shamlo, N., Mullen, T., Kothe, C., Su, K.-M., Robbins, K.A., 2015. The PREP pipeline:

standardized preprocessing for large-scale EEG analysis. Frontiers in Neuroinformatics 9, e16.

Bonnard, M., Spieser, L., Meziane, H., De Graaf, J., Pailhous, J., 2009. Prior intention can locally tune inhibitory processes in the primary motor cortex: direct evidence from combined TMS-EEG.

European Journal of Neuroscience 30, 913-923.

Bootsma, M., Swenne, C.A., Van Bolhuis, H.H., Chang, P.C., Cats, V.M., Bruschke, A., 1994. Heart rate and heart rate variability as indexes of sympathovagal balance. American Journal of Physiology-Heart and Circulatory Physiology 266, H1565-H1571.

Bowlby, J., 1969. Attachment, Vol. 1 of Attachment and loss. New York: Basic Books.

Brooks, J.C., Nurmikko, T.J., Bimson, W.E., Singh, K.D., Roberts, N., 2002. fMRI of thermal pain: effects of stimulus laterality and attention. NeuroImage 15, 293-301.

Brown, J.L., Sheffield, D., Leary, M.R., Robinson, M.E., 2003. Social support and experimental pain, Psychosomatic Medicine 65, 276-283.

Bunketorp, L., Lindh, M., Carlsson, J., Stener-Victorin, E., 2006. The perception of pain and painrelated cognitions in subacute whiplash-associated disorders: its influence on prolonged disability. Disability and rehabilitation 28, 271-279. Bushnell, M.C., Čeko, M., Low, L.A., 2013. Cognitive and emotional control of pain and its disruption in chronic pain. Nature Reviews Neuroscience 14, 502-511.

Buzsaki, G., 2006. Rhythms of the Brain. England: Oxford University Press.

Buzsáki, G., 2004. Large-scale recording of neuronal ensembles. Nature Neuroscience 7, 446-451.

Buzsáki, G., Draguhn, A., 2004. Neuronal oscillations in cortical networks. Science 304, 1926-1929.

Cannon, W.B., 1939. The wisdom of the body. England: Norton & Co.

Cano, A., Williams, A.C.d.C., 2010. Social interaction in pain: Reinforcing pain behaviors or building intimacy? Pain 149, 9-11.

Cardiology, 1996. Heart rate variability: standards of measurement, physiological interpretation, and clinical use. Circulation 93, 1043-1065.

Cash, R.F., Noda, Y., Zomorrodi, R., Radhu, N., Farzan, F., Rajji, T.K., Fitzgerald, P.B., Chen, R., Daskalakis, Z.J., Blumberger, D.M., 2017. Characterization of Glutamatergic and GABA A-Mediated Neurotransmission in Motor and Dorsolateral Prefrontal Cortex Using Paired-Pulse TMS–EEG. Neuropsychopharmacology 42, 502-511.

Cauda, F., Torta, D.M., Sacco, K., Geda, E., D'Agata, F., Costa, T., Duca, S., Geminiani, G., Amanzio, M., 2012. Shared "core" areas between the pain and other task-related networks. PLoS One 7, e41929.

Chang, P.F., Arendt-Nielsen, L., Chen, A.C., 2002. Dynamic changes and spatial correlation of EEG activities during cold pressor test in man. Brain Research Bulletin 57, 667-675.

Che, X., Zhang, Q., Zhao, J., Wei, D., Li, B., Guo, Y., Qiu, J., Liu, Y., 2014. Synchronous activation within the default mode network correlates with perceived social support. Neuropsychologia 63, 26-33. Chen, F.S., Kumsta, R., von Dawans, B., Monakhov, M., Ebstein, R.P., Heinrichs, M., 2011. Common oxytocin receptor gene (OXTR) polymorphism and social support interact to reduce stress in humans. Proceedings of the National Academy of Sciences 108, 19937-19942. Chung, S.W., Hill, A.T., Rogasch, N.C., Hoy, K.E., Fitzgerald, P.B., 2016. Use of theta-burst stimulation in changing excitability of motor cortex: a systematic review and meta-analysis. Neuroscience & Biobehavioral Reviews 63, 43-64.

Chung, S.W., Lewis, B.P., Rogasch, N.C., Saeki, T., Thomson, R.H., Hoy, K.E., Bailey, N.W., Fitzgerald, P.B., 2017. Demonstration of short-term plasticity in the dorsolateral prefrontal cortex with theta burst stimulation: A TMS-EEG study. Clinical Neurophysiology 128, 1117-1126.

Chung, S.W., Rogasch, N.C., Hoy, K.E., Fitzgerald, P.B., 2018a. The effect of single and repeated prefrontal intermittent theta burst stimulation on cortical reactivity and working memory. Brain Stimulation 11, 566-574.

Chung, S.W., Rogasch, N.C., Hoy, K.E., Sullivan, C.M., Cash, R.F., Fitzgerald, P.B., 2018b. Impact of different intensities of intermittent theta burst stimulation on the cortical properties during TMS-EEG and working memory performance. Human Brain Mapping 39, 783-802.

Coan, J.A., Schaefer, H.S., Davidson, R.J., 2006. Lending a hand social regulation of the neural response to threat. Psychological Science 17, 1032-1039.

Cohen, S., Wills, T.A., 1985. Stress, social support, and the buffering hypothesis. Psychological Bulletin 98, 310-357.

Corley, A.M., Cano, A., Goubert, L., Vlaeyen, J.W., Wurm, L.H., 2016. Global and situational relationship satisfaction moderate the effect of threat on pain in couples. Pain Medicine 17, 1664-1675.

Cortelli, P., Giannini, G., Favoni, V., Cevoli, S., Pierangeli, G., 2013. Nociception and autonomic nervous system. Neurological Sciences 34, 41-46.

Craig, A.D., 2002. How do you feel? Interoception: the sense of the physiological condition of the body. Nature Reviews Neuroscience 3, 655-666.

Craig, A.D., 2003a. A new view of pain as a homeostatic emotion. Trends in Neurosciences 26, 303-307.

Craig, A.D., 2003b. Pain mechanisms: labeled lines versus convergence in central processing. Annual Review of Neuroscience 26, 1-30.

Delgado, M.R., Olsson, A., Phelps, E.A., 2006. Extending animal models of fear conditioning to humans. Biological Psychology 73, 39-48.

DeLongis, A., Folkman, S., Lazarus, R.S., 1988. The impact of daily stress on health and mood: psychological and social resources as mediators. Journal of Personality and Social Psychology 54, 486-495.

Derbyshire, S.W., 2000. Exploring the pain "neuromatrix". Current Review of Pain 4, 467-477.

Du, X., Rowland, L.M., Summerfelt, A., Wijtenburg, S.A., Chiappelli, J., Wisner, K., Kochunov, P., Choa,

F.-S., Hong, L.E., 2018. TMS evoked N100 reflects local GABA and glutamate balance. Brain Stimulation 11, 1071-1079.

Dumas, G., 2011. Towards a two-body neuroscience. Communicative & Integrative Biology 4, 349-352.

Dumas, G., Lachat, F., Martinerie, J., Nadel, J., George, N., 2011. From social behaviour to brain synchronization: review and perspectives in hyperscanning. IRBM 32, 48-53.

Dumas, G., Nadel, J., Soussignan, R., Martinerie, J., Garnero, L., 2010. Inter-brain synchronization during social interaction. PLoS One 5, e12166.

Dunckley, P., Aziz, Q., Wise, R.G., Brooks, J., Tracey, I., Chang, L., 2007. Attentional modulation of visceral and somatic pain. Neurogastroenterology & Motility 19, 569-577.

Dunkel-Schetter, C., Bennett, T.L., 1990. Differentiating the cognitive and behavioral aspects of social support, in: Sarason, B.R., Sarason, I.G., Pierce, G.R. (Eds.), Social Support: An Interactional View. John Wiley, Oxford, pp. 267–296.

Edwards, R., Eccleston, C., Keogh, E., 2017. Observer influences on pain: an experimental series examining same-sex and opposite-sex friends, strangers, and romantic partners. Pain 158, 846–855. Eippert, F., Bingel, U., Schoell, E.D., Yacubian, J., Klinger, R., Lorenz, J., Büchel, C., 2009a. Activation of the opioidergic descending pain control system underlies placebo analgesia. Neuron 63, 533-543. Eippert, F., Finsterbusch, J., Bingel, U., Büchel, C., 2009b. Direct evidence for spinal cord involvement in placebo analgesia. Science 326, 404-404.

Eisenberger, N.I., Cole, S.W., 2012. Social neuroscience and health: neurophysiological mechanisms linking social ties with physical health. Nature Neuroscience 15, 669-674.

Eisenberger, N.I., Master, S.L., Inagaki, T.K., Taylor, S.E., Shirinyan, D., Lieberman, M.D., Naliboff,

B.D., 2011. Attachment figures activate a safety signal-related neural region and reduce pain

experience. Proceedings of the National Academy of Sciences 108, 11721-11726.

Engebretsen, S., Frigessi, A., Engø-Monsen, K., Furberg, A.-S., Stubhaug, A., de Blasio, B.F., Nielsen,

C.S., 2018. The peer effect on pain tolerance. Scandinavian Journal of Pain 18, 467-477.

Farzan, F., Barr, M.S., Hoppenbrouwers, S.S., Fitzgerald, P.B., Chen, R., Pascual-Leone, A., Daskalakis,

Z.J., 2013. The EEG correlates of the TMS-induced EMG silent period in humans. NeuroImage 83, 120-134.

Feeney, J.A., Noller, P., 1990. Attachment style as a predictor of adult romantic relationships. Journal of Personality and Social Psychology 58, 281-291.

Fordyce, W., 1977. Behavioral methods for chronic pain and illness. St. Louis: Mosby

Frankenstein, U., Richter, W., McIntyre, M., Remy, F., 2001. Distraction modulates anterior cingulate gyrus activations during the cold pressor test. Neuroimage 14, 827-836.

Fuster, J.n.M., 1997. Network memory. Trends in Neurosciences 20, 451-459.

Gallant, N.L., Hadjistavropoulos, T., 2017. Experiencing pain in the presence of others: A structured experimental investigation of older adults. The Journal of Pain 18, 456-467.

Gamond, L., Cattaneo, Z., 2016. The dorsomedial prefrontal cortex plays a causal role in mediating in-group advantage in emotion recognition: a TMS study. Neuropsychologia 93, 312-317.

Gamond, L., Ferrari, C., La Rocca, S., Cattaneo, Z., 2017. Dorsomedial prefrontal cortex and cerebellar contribution to in-group attitudes: a transcranial magnetic stimulation study. European Journal of Neuroscience 45, 932-939.

Geha, P.Y., Baliki, M.N., Harden, R.N., Bauer, W.R., Parrish, T.B., Apkarian, A.V., 2008. The brain in chronic CRPS pain: abnormal gray-white matter interactions in emotional and autonomic regions. Neuron 60, 570-581.

Goldstein, P., Shamay-Tsoory, S.G., Yellinek, S., Weissman-Fogel, I., 2016. Empathy predicts an experimental pain reduction during touch. The Journal of Pain 17, 1049-1057.

Goldstein, P., Weissman-Fogel, I., Dumas, G., Shamay-Tsoory, S.G., 2018. Brain-to-brain coupling during handholding is associated with pain reduction. Proceedings of the National Academy of Sciences 115, E2528-E2537.

Goldstein, P., Weissman-Fogel, I., Shamay-Tsoory, S.G., 2017. The role of touch in regulating interpartner physiological coupling during empathy for pain. Scientific reports 7, e3252.

González-Hernández, A., Rojas-Piloni, G., Condés-Lara, M., 2014. Oxytocin and analgesia: future trends. Trends in Pharmacological Sciences 35, 549-551.

Gray, L., Watt, L., Blass, E.M., 2000. Skin-to-skin contact is analgesic in healthy newborns. Pediatrics 105, e14-e14.

Hadjileontiadis, L.J., 2015. EEG-Based Tonic Cold Pain Characterization Using Wavelet Higher Order Spectral Features. IEEE Transactions on Biomedical Engineering 62, 1981-1991.

Haggard, P., Iannetti, G.D., Longo, M.R., 2013. Spatial sensory organization and body representation in pain perception. Current Biology 23, R164-R176.

Hampton, A.J., Hadjistavropoulos, T., Gagnon, M.M., Williams, J., Clark, D., 2015. The effects of emotion regulation strategies on the pain experience: a structured laboratory investigation. Pain 156, 868-879.

Harmer, C., Thilo, K., Rothwell, J., Goodwin, G., 2001. Transcranial magnetic stimulation of medial– frontal cortex impairs the processing of angry facial expressions. Nature Neuroscience 4, 17-18. Hauck, M., Lorenz, J., Domnick, C., Gerloff, C., Engel, A.K., 2015. Top-down and bottom-up modulation of pain-induced oscillations. Frontiers in Human Neuroscience 9, e375.

Heinrichs, M., Baumgartner, T., Kirschbaum, C., Ehlert, U., 2003. Social support and oxytocin interact to suppress cortisol and subjective responses to psychosocial stress. Biological Psychiatry 54, 1389-1398.

Herrington, C.J., Chiodo, L.M., 2014. Human touch effectively and safely reduces pain in the newborn intensive care unit. Pain Management Nursing 15, 107-115.

Hertenstein, M.J., Keltner, D., App, B., Bulleit, B.A., Jaskolka, A.R., 2006. Touch communicates distinct emotions. Emotion 6, 528-533.

Hill, A.T., Rogasch, N.C., Fitzgerald, P.B., Hoy, K.E., 2017. Effects of prefrontal bipolar and highdefinition transcranial direct current stimulation on cortical reactivity and working memory in healthy adults. Neuroimage 152, 142-157.

Hirsch, L., Brenner, R., 2011. Atlas of EEG in critical care. Oxford: John Wiley & Sons.

Hodges, P.W., Tucker, K., 2011. Moving differently in pain: a new theory to explain the adaptation to pain. Pain 152, S90-S98.

Holt-Lunstad, J., Birmingham, W.A., Light, K.C., 2008. Influence of a "warm touch" support enhancement intervention among married couples on ambulatory blood pressure, oxytocin, alpha amylase, and cortisol. Psychosomatic Medicine 70, 976-985.

Holtzman, S., DeLongis, A., 2007. One day at a time: The impact of daily satisfaction with spouse responses on pain, negative affect and catastrophizing among individuals with rheumatoid arthritis. Pain 131, 202-213.

Holtzman, S., Newth, S., Delongis, A., 2004. The role of social support in coping with daily pain among patients with rheumatoid arthritis. Journal of Health Psychology 9, 677-695.

House, J.S., 1983. Work stress and social support. Addison-Wesley Series on Occupational Stress. Huang, Y.-Z., Edwards, M.J., Rounis, E., Bhatia, K.P., Rothwell, J.C., 2005. Theta burst stimulation of the human motor cortex. Neuron 45, 201-206.

Huber, M., Bartling, J., Pachur, D.v., Woikowsky-Biedau, S., Lautenbacher, S., 2006. EEG responses to tonic heat pain. Experimental Brain Research 173, 14-24.

IASP, 1994. PART III pain terms, a current list with definitions and notes on usage. Classification of chronic pain-descriptions of chronic pain syndromes and definitions of pain terms, 207-214. Ingvar, M., Hsieh, J., 1999. The image of pain. Textbook of Pain 4, 215-233.

Jackson, T., Iezzi, T., Chen, H., Ebnet, S., Eglitis, K., 2005. Gender, interpersonal transactions, and the perception of pain: An experimental analysis. The Journal of Pain 6, 228-236.

Jensen, M.P., Karoly, P., 1991. Control beliefs, coping efforts, and adjustment to chronic pain. Journal of Consulting and Clinical Psychology 59, 431-438.

Jensen, M.P., Turner, J.A., Romano, J.M., 2001. Changes in beliefs, catastrophizing, and coping are associated with improvement in multidisciplinary pain treatment. Journal of Consulting and Clinical Psychology 69, 655-662.

Karmann, A.J., Lautenbacher, S., Bauer, F., Kunz, M., 2014. The influence of communicative relations on facial responses to pain: does it matter who is watching? Pain Research & Management 19, 15-22.

Keefe, F.J., Blumenthal, J., Baucom, D., Affleck, G., Waugh, R., Caldwell, D.S., Beaupre, P., Kashikar-Zuck, S., Wright, K., Egert, J., 2004. Effects of spouse-assisted coping skills training and exercise training in patients with osteoarthritic knee pain: a randomized controlled study. Pain 110, 539-549.
Keefe, F.J., Caldwell, D.S., Baucom, D., Salley, A., Robinson, E., Timmons, K., Beaupre, P., Weisberg, J., Helms, M., 1996. Spouse-assisted coping skills training in the management of osteoarthritic knee pain. Arthritis & Rheumatology 9, 279-291.

Keefe, F.J., Caldwell, D.S., Baucom, D., Salley, A., Robinson, E., Timmons, K., Beaupre, P., Weisberg, J., Helms, M., 1999. Spouse-assisted coping skills training in the management of knee pain in osteoarthritis: long-term followup results. Arthritis & Rheumatism 12, 101-111.

Kindt, S., Vansteenkiste, M., Brenning, K., Goubert, L., 2018a. The Effects of Partners' Helping Motivation on Chronic Pain Patients' Functioning Over Time. The Journal of Pain 20, 348-357.

Kindt, S., Vansteenkiste, M., Josephy, H., Bernardes, S.F., Goubert, L., 2018b. Helping your partner with chronic pain: the importance of helping motivation, received social support, and Its timeliness. Pain Medicine 20, 77-89.

Kindt, S., Vansteenkiste, M., Loeys, T., Cano, A., Lauwerier, E., Verhofstadt, L.L., Goubert, L., 2015. When is helping your partner with chronic pain a burden? The relation between helping motivation and personal and relational functioning. Pain Medicine 16, 1732-1744.

Kindt, S., Vansteenkiste, M., Loeys, T., Goubert, L., 2016. Helping motivation and well-being of chronic pain couples: A daily diary study. Pain 157, 1551-1562.

Koenig, J., Jarczok, M.N., Ellis, R.J., Hillecke, T., Thayer, J.F., 2014. Heart rate variability and experimentally induced pain in healthy adults: a systematic review. European Journal of Pain 18, 301-314.

Kong, J., Jensen, K., Loiotile, R., Cheetham, A., Wey, H.-Y., Tan, Y., Rosen, B., Smoller, J.W., Kaptchuk, T.J., Gollub, R.L., 2013. Functional connectivity of the frontoparietal network predicts cognitive modulation of pain. Pain 154, 459-467.

Krahé, C., Paloyelis, Y., Condon, H., Jenkinson, P.M., Williams, S.C., Fotopoulou, A., 2015. Attachment style moderates partner presence effects on pain: a laser-evoked potentials study. Social Cognitive and Affective Neuroscience 10, 1030–1037.

Krahé, C., Springer, A., Weinman, J.A., Fotopoulou, A., 2013. The social modulation of pain: others as predictive signals of salience–a systematic review. Frontiers in Human Neuroscience 7, e386. Kreuder, A.K., Wassermann, L., Wollseifer, M., Ditzen, B., Eckstein, M., Stoffel-Wagner, B., Hennig,

J., Hurlemann, R., Scheele, D., 2018. Oxytocin enhances the pain-relieving effects of social support in

romantic couples. Human Brain Mapping 40, 242-251.

Lapate, R.C., Lee, H., Salomons, T.V., van Reekum, C.M., Greischar, L.L., Davidson, R.J., 2012. Amygdalar function reflects common individual differences in emotion and pain regulation success. Journal of Cognitive Neuroscience 24, 148-158. Le Pera, D., Svensson, P., Valeriani, M., Watanabe, I., Arendt-Nielsen, L., Chen, A.C., 2000. Longlasting effect evoked by tonic muscle pain on parietal EEG activity in humans. Clinical Neurophysiology 111, 2130-2137.

Le Van Quyen, M., Bragin, A., 2007. Analysis of dynamic brain oscillations: methodological advances. Trends in Neurosciences 30, 365-373.

Lee, J.E., Kahana, B., Kahana, E., 2016. Social support and cognitive functioning as resources for elderly persons with chronic arthritis pain. Aging & Mental Health 20, 370-379.

Lee, M.C., Mouraux, A., Iannetti, G.D., 2009. Characterizing the cortical activity through which pain emerges from nociception. Journal of Neuroscience 29, 7909-7916.

Legrain, V., Iannetti, G.D., Plaghki, L., Mouraux, A., 2011. The pain matrix reloaded: a salience detection system for the body. Progress in Neurobiology 93, 111-124.

Light, K.C., Grewen, K.M., Amico, J.A., 2005. More frequent partner hugs and higher oxytocin levels are linked to lower blood pressure and heart rate in premenopausal women. Biological Psychology 69, 5-21.

Loeser, J.D., 2001. Bonica's management of pain. Philadelphia: Lippincott Williams & Wilkins.

Loggia, M.L., Mogil, J.S., Bushnell, M.C., 2008. Experimentally induced mood changes preferentially affect pain unpleasantness. The Journal of Pain 9, 784-791.

López-Martínez, A.E., Esteve-Zarazaga, R., Ramírez-Maestre, C., 2008. Perceived social support and coping responses are independent variables explaining pain adjustment among chronic pain patients. The Journal of Pain 9, 373-379.

Lumley, M.A., Cohen, J.L., Borszcz, G.S., Cano, A., Radcliffe, A.M., Porter, L.S., Schubiner, H., Keefe, F.J., 2011. Pain and emotion: a biopsychosocial review of recent research. Journal of Clinical Psychology 67, 942-968.

Maeda, F., Keenan, J.P., Tormos, J.M., Topka, H., Pascual-Leone, A., 2000. Interindividual variability of the modulatory effects of repetitive transcranial magnetic stimulation on cortical excitability. Experimental Brain Research 133, 425-430. Main, M., 2000. The organized categories of infant, child, and adult attachment: Flexible vs. inflexible attention under attachment-related stress. Journal of the American Psychoanalytic Association 48, 1055-1096.

Mäki, H., Ilmoniemi, R.J., 2010. The relationship between peripheral and early cortical activation induced by transcranial magnetic stimulation. Neuroscience Letters 478, 24-28.

Martire, L.M., Schulz, R., Keefe, F., Starz, T., Osial Jr, T., Dew, M., Reynolds III, C., 2003. Feasibility of a dyadic intervention for management of osteoarthritis: a pilot study with older patients and their spousal caregivers. Aging & Mental Health 7, 53-60.

Martire, L.M., Schulz, R., Keefe, F.J., Rudy, T.E., Starz, T.W., 2007. Couple-oriented education and support intervention: Effects on individuals with osteoarthritis and their spouses. Rehabilitation Psychology 52, 121-132.

Martire, L.M., Schulz, R., Keefe, F.J., Rudy, T.E., Starz, T.W., 2008. Couple-oriented education and support intervention for osteoarthritis: Effects on spouses' support and responses to patient pain. Families, Systems, & Health 26, 185-195.

Martucci, K.T., Ng, P., Mackey, S., 2014. Neuroimaging chronic pain: what have we learned and where are we going? Future Neurology 9, 615-626.

Master, S.L., Eisenberger, N.I., Taylor, S.E., Naliboff, B.D., Shirinyan, D., Lieberman, M.D., 2009. A picture's worth: partner photographs reduce experimentally induced pain. Psychological Science 20, 1316-1318.

McClelland, L.E., McCubbin, J.A., 2008. Social influence and pain response in women and men. Journal of Behavioral Medicine 31, 413-420.

Melzack, R., 1999. From the gate to the neuromatrix. Pain 82, S121-S126.

Melzack, R., Casey, K.L., 1968. Sensory, motivational and central control determinants of pain: a new conceptual model. In D. R. Kenshalo (Ed.), The skin senses (pp. 423-443). Springfield, IL: C C Thomas.

Modić Stanke, K., Ivanec, D., 2010. Social context of pain perception: the role of other people's presence and physical distance. Review of Psychology 17, 69-74.

Montoya, P., Larbig, W., Braun, C., Preissl, H., Birbaumer, N., 2004. Influence of social support and emotional context on pain processing and magnetic brain responses in fibromyalgia. Arthritis & Rheumatology 50, 4035-4044.

Moore, J.E., Chaney, E.F., 1985. Outpatient group treatment of chronic pain: effects of spouse involvement. Journal of Consulting and Clinical Psychology 53, 326-334.

Morton, D.L., Sandhu, J.S., Jones, A.K., 2016. Brain imaging of pain: state of the art. Journal of Pain Research 9, 613-624.

Mouraux, A., Iannetti, G.D., 2018. The search for pain biomarkers in the human brain. Brain 141, 3290-3307.

Ni, Z., Gunraj, C., Kailey, P., Cash, R.F., Chen, R., 2014. Heterosynaptic modulation of motor cortical plasticity in human. Journal of Neuroscience 34, 7314-7321.

Nickel, M.M., May, E.S., Tiemann, L., Schmidt, P., Postorino, M., Dinh, S.T., Gross, J., Ploner, M.,

2017. Brain oscillations differentially encode noxious stimulus intensity and pain intensity.

NeuroImage 148, 141-147.

Noda, Y., Cash, R.F., Zomorrodi, R., Dominguez, L.G., Farzan, F., Rajji, T.K., Barr, M.S., Chen, R., Daskalakis, Z.J., Blumberger, D.M., 2016. A combined TMS-EEG study of short-latency afferent inhibition in the motor and dorsolateral prefrontal cortex. Journal of Neurophysiology 116, 938-948. Ossipov, M.H., Dussor, G.O., Porreca, F., 2010. Central modulation of pain. The Journal of Clinical Investigation 120, 3779-3787.

Ossipov, M.H., Morimura, K., Porreca, F., 2014. Descending pain modulation and chronification of pain. Current opinion in Supportive and Palliative Care 8, 143-151.

Otte, E., Rousseau, R., 2002. Social network analysis: a powerful strategy, also for the information sciences. Journal of Information Science 28, 441-453.

Paloyelis, Y., Krahé, C., Maltezos, S., Williams, S., Howard, M., Fotopoulou, A., 2016. The Analgesic Effect of Oxytocin in Humans: A Double-Blind, Placebo-Controlled Cross-Over Study Using Laser-Evoked Potentials. Journal of Neuroendocrinology 28, 1-10.

Peng, W., Hu, L., Zhang, Z., Hu, Y., 2014. Changes of spontaneous oscillatory activity to tonic heat pain. PLoS One 9, e91052.

Petrovic, P., Kalso, E., Petersson, K.M., Ingvar, M., 2002. Placebo and opioid analgesia--imaging a shared neuronal network. Science 295, 1737-1740.

Petrovic, P., Petersson, K.M., Ghatan, P., Stone-Elander, S., Ingvar, M., 2000. Pain-related cerebral activation is altered by a distracting cognitive task. Pain 85, 19-30.

Peyron, R., García-Larrea, L., Grégoire, M.-C., Costes, N., Convers, P., Lavenne, F., Mauguière, F., Michel, D., Laurent, B., 1999. Haemodynamic brain responses to acute pain in humans: sensory and attentional networks. Brain 122, 1765-1780.

Phelps, E.A., Delgado, M.R., Nearing, K.I., LeDoux, J.E., 2004. Extinction learning in humans: role of the amygdala and vmPFC. Neuron 43, 897-905.

Platow, M.J., Voudouris, N.J., Gilford, N., Jamieson, R., Najdovski, L., Papaleo, N., Pollard, C., Terry, L., 2007. In-group reassurance in a pain setting produces lower levels of physiological arousal: direct support for a self-categorization analysis of social influence. European Journal of Social Psychology 37, 649-660.

Ploghaus, A., Tracey, I., Gati, J.S., Clare, S., Menon, R.S., Matthews, P.M., Rawlins, J.N.P., 1999. Dissociating pain from its anticipation in the human brain. Science 284, 1979-1981.

Ploner, M., Lee, M.C., Wiech, K., Bingel, U., Tracey, I., 2010. Flexible cerebral connectivity patterns subserve contextual modulations of pain. Cerebral Cortex 21, 719-726.

Ploner, M., Sorg, C., Gross, J., 2017. Brain rhythms of pain. Trends in cognitive sciences 21, 100-110. Price, D.D., Milling, L.S., Kirsch, I., Duff, A., Montgomery, G.H., Nicholls, S.S., 1999. An analysis of factors that contribute to the magnitude of placebo analgesia in an experimental paradigm. Pain 83, 147-156. Quirk, G.J., Garcia, R., González-Lima, F., 2006. Prefrontal mechanisms in extinction of conditioned fear. Biological Psychiatry 60, 337-343.

Radojevic, V., Nicassio, P.M., Weisman, M.H., 1992. Behavioral intervention with and without family support for rheumatoid arthritis. Behavior Therapy 23, 13-30.

Roberts, M.H., Klatzkin, R.R., Mechlin, B., 2015. Social support attenuates physiological stress responses and experimental pain sensitivity to cold pressor pain. Annals of Behavioral Medicine 49, 557-569.

Rodriguez-Raecke, R., Niemeier, A., Ihle, K., Ruether, W., May, A., 2009. Brain gray matter decrease in chronic pain is the consequence and not the cause of pain. Journal of Neuroscience 29, 13746-13750.

Rogasch, N.C., Daskalakis, Z.J., Fitzgerald, P.B., 2012. Mechanisms underlying long-interval cortical inhibition in the human motor cortex: a TMS-EEG study. Journal of Neurophysiology 109, 89-98. Rogasch, N.C., Daskalakis, Z.J., Fitzgerald, P.B., 2015. Cortical inhibition of distinct mechanisms in the dorsolateral prefrontal cortex is related to working memory performance: a TMS–EEG study. Cortex 64, 68-77.

Rogasch, N.C., Fitzgerald, P.B., 2013. Assessing cortical network properties using TMS–EEG. Human Brain Mapping 34, 1652-1669.

Rogasch, N.C., Thomson, R.H., Farzan, F., Fitzgibbon, B.M., Bailey, N.W., Hernandez-Pavon, J.C., Daskalakis, Z.J., Fitzgerald, P.B., 2014. Removing artefacts from TMS-EEG recordings using independent component analysis: importance for assessing prefrontal and motor cortex network properties. Neuroimage 101, 425-439.

Royl, M., Lebuisl, A., Peretzl, I., Rainvillel, P., 2011. The modulation of pain by attention and emotion: a dissociation of perceptual and spinal nociceptive processes. European journal of pain 15, e1-e10. Salomons, T.V., Johnstone, T., Backonja, M.-M., Davidson, R.J., 2004. Perceived controllability modulates the neural response to pain. Journal of Neuroscience 24, 7199-7203.

Salomons, T.V., Johnstone, T., Backonja, M.-M., Shackman, A.J., Davidson, R.J., 2007. Individual differences in the effects of perceived controllability on pain perception: critical role of the prefrontal cortex. Journal of Cognitive Neuroscience 19, 993-1003.

Sambo, C.F., Howard, M., Kopelman, M., Williams, S., Fotopoulou, A., 2010. Knowing you care: Effects of perceived empathy and attachment style on pain perception. Pain 151, 687-693. Sauseng, P., Klimesch, W., 2008. What does phase information of oscillatory brain activity tell us about cognitive processes? Neuroscience & Biobehavioral Reviews 32, 1001-1013.

Schulz, E., May, E.S., Postorino, M., Tiemann, L., Nickel, M.M., Witkovsky, V., Schmidt, P., Gross, J., Ploner, M., 2015. Prefrontal gamma oscillations encode tonic pain in humans. Cerebral Cortex 25, 4407-4414.

Schuwerk, T., Schecklmann, M., Langguth, B., Döhnel, K., Sodian, B., Sommer, M., 2014. Inhibiting the posterior medial prefrontal cortex by rTMS decreases the discrepancy between self and other in theory of mind reasoning. Behavioural Brain Research 274, 312-318.

Seeley, W.W., Menon, V., Schatzberg, A.F., Keller, J., Glover, G.H., Kenna, H., Reiss, A.L., Greicius,

M.D., 2007. Dissociable intrinsic connectivity networks for salience processing and executive control. Journal of Neuroscience 27, 2349-2356.

Seminowicz, D., Mikulis, D., Davis, K., 2004. Cognitive modulation of pain-related brain responses depends on behavioral strategy. Pain 112, 48-58.

Shaygan, M., Böger, A., Kröner-Herwig, B., 2017. Valence and arousal value of visual stimuli and their role in the mitigation of chronic pain: What is the power of pictures? The Journal of Pain 18, 124-131.

Siebner, H., Rothwell, J., 2003. Transcranial magnetic stimulation: new insights into representational cortical plasticity. Experimental Brain Research 148, 1-16.

Subramaniam, V., Stewart, M.W., Smith, J.F., 1999. The development and impact of a chronic pain support group: a qualitative and quantitative study. Journal of Pain and Symptom Management 17, 376-383.

Sullivan, M.J., Thorn, B., Haythornthwaite, J.A., Keefe, F., Martin, M., Bradley, L.A., Lefebvre, J.C.,

2001. Theoretical perspectives on the relation between catastrophizing and pain. The Clinical Journal of Pain 17, 52-64.

Sullivan, M.J.L., Adams, H., Sullivan, M.E., 2004. Communicative dimensions of pain catastrophizing: Social cueing effects on pain behaviour and coping. Pain 107, 220-226.

Taesler, P., Rose, M., 2016. Prestimulus theta oscillations and connectivity modulate pain perception. Journal of Neuroscience 36, 5026-5033.

Thayer, J.F., Hansen, A.L., Saus-Rose, E., Johnsen, B.H., 2009. Heart rate variability, prefrontal neural function, and cognitive performance: the neurovisceral integration perspective on self-regulation, adaptation, and health. Annals of Behavioral Medicine 37, 141-153.

Thayer, J.F., Yamamoto, S.S., Brosschot, J.F., 2010. The relationship of autonomic imbalance, heart rate variability and cardiovascular disease risk factors. International Journal of Cardiology 141, 122-131.

Tiemann, L., Schulz, E., Gross, J., Ploner, M., 2010. Gamma oscillations as a neuronal correlate of the attentional effects of pain. Pain 150, 302-308.

Tognoli, E., Lagarde, J., DeGuzman, G.C., Kelso, J.S., 2007. The phi complex as a neuromarker of human social coordination. Proceedings of the National Academy of Sciences 104, 8190-8195.

Tracey, I., 2008. Imaging pain. British Journal of Anaesthesia 101, 32-39.

Tracey, I., Becerra, L., Chang, I., Breiter, H., Jenkins, L., Borsook, D., González, R.G., 2000. Noxious hot and cold stimulation produce common patterns of brain activation in humans: a functional magnetic resonance imaging study. Neuroscience Letters 288, 159-162.

Tracey, I., Mantyh, P.W., 2007. The cerebral signature for pain perception and its modulation. Neuron 55, 377-391.

Tracey, I., Ploghaus, A., Gati, J.S., Clare, S., Smith, S., Menon, R.S., Matthews, P.M., 2002. Imaging attentional modulation of pain in the periaqueductal gray in humans. Journal of Neuroscience 22, 2748-2752.

Turner, R.J., 1983. Direct, indirect, and moderating effects of social support on psychological distress and associated conditions. In H. Kaplan (Ed.), Psychosocial Stress: Trends in Theory and Research (pp. 105-155). New York: Academic Press.

Uchino, B.N., 2006. Social support and health: a review of physiological processes potentially underlying links to disease outcomes. Journal of Behavioral Medicine 29, 377-387.

Uchino, B.N., 2009. Understanding the links between social support and physical health: A life-span perspective with emphasis on the separability of perceived and received support. Perspectives on Psychological Science 4, 236-255.

Uchino, B.N., Cacioppo, J.T., Kiecolt-Glaser, J.K., 1996. The relationship between social support and physiological processes: a review with emphasis on underlying mechanisms and implications for health. Psychological Bulletin 119, 488-531.

Valet, M., Sprenger, T., Boecker, H., Willoch, F., Rummeny, E., Conrad, B., Erhard, P., Tolle, T.R., 2004. Distraction modulates connectivity of the cingulo-frontal cortex and the midbrain during pain—an fMRI analysis. Pain 109, 399-408.

Varela, F., Lachaux, J.-P., Rodriguez, E., Martinerie, J., 2001. The brainweb: phase synchronization and large-scale integration. Nature Reviews Neuroscience 2, 229-239.

Vervoort, T., Caes, L., Trost, Z., Sullivan, M., Vangronsveld, K., Goubert, L., 2011. Social modulation of facial pain display in high-catastrophizing children: An observational study in schoolchildren and their parents. Pain 152, 1591-1599.

Villemure, C., Bushnell, M.C., 2009. Mood influences supraspinal pain processing separately from attention. Journal of Neuroscience 29, 705-715.

Villemure, C., Slotnick, B.M., Bushnell, M.C., 2003. Effects of odors on pain perception: deciphering the roles of emotion and attention. Pain 106, 101-108.

Vlaeyen, J.W., Hanssen, M., Goubert, L., Vervoort, T., Peters, M., van Breukelen, G., Sullivan, M.J., Morley, S., 2009. Threat of pain influences social context effects on verbal pain report and facial expression. Behaviour Research and Therapy 47, 774-782.

Wager, T.D., Rilling, J.K., Smith, E.E., Sokolik, A., Casey, K.L., Davidson, R.J., Kosslyn, S.M., Rose, R.M., Cohen, J.D., 2004. Placebo-induced changes in FMRI in the anticipation and experience of pain. Science 303, 1162-1167.

Wagner, T., Valero-Cabre, A., Pascual-Leone, A., 2007. Noninvasive human brain stimulation. Annual Review of Biomedical Engineering 9, 527-565.

Waltz, M., Kriegel, W., Bosch, P.V.T., 1998. The social environment and health in rheumatoid arthritis: marital quality predicts individual variability in pain severity. Arthritis & Rheumatism 11, 356-374.

Wang, M.-Y., Luan, P., Zhang, J., Xiang, Y.-T., Niu, H., Yuan, Z., 2018. Concurrent mapping of brain activation from multiple subjects during social interaction by hyperscanning: a mini-review. Quantitative Imaging in Medicine and Surgery 8, 819-837.

Wiech, K., Kalisch, R., Weiskopf, N., Pleger, B., Stephan, K.E., Dolan, R.J., 2006. Anterolateral prefrontal cortex mediates the analgesic effect of expected and perceived control over pain. Journal of Neuroscience 26, 11501-11509.

Wiech, K., Tracey, I., 2009. The influence of negative emotions on pain: behavioral effects and neural mechanisms. Neuroimage 47, 987-994.

Williams, A.C.d.C., 2002. Facial expression of pain: an evolutionary account. Behavioral and Brain Sciences 25, 439-455.

Williams, A.C.d.C., Craig, K.D., 2016. Updating the definition of pain. Pain 157, 2420-2423.

Wills, T.A., 1991. Social support and interpersonal relationships, in: Clark, M.S. (Ed.), Review of personality and social psychology. Sage Publications, Thousand Oaks, CA, pp. 265-289.

Winkler, I., Haufe, S., Tangermann, M., 2011. Automatic classification of artifactual ICA-components for artifact removal in EEG signals. Behavioral and Brain Functions 7, e30.

Yantis, S., 2008. The neural basis of selective attention: Cortical sources and targets of attentional modulation. Current Directions in Psychological Science 17, 86-90.

Younger, J., Aron, A., Parke, S., Chatterjee, N., Mackey, S., 2010. Viewing pictures of a romantic partner reduces experimental pain: Involvement of neural reward systems. PLoS One 5, e13309. Zhang, Z., Hu, L., Hung, Y.S., Mouraux, A., Iannetti, G., 2012. Gamma-band oscillations in the primary somatosensory cortex—a direct and obligatory correlate of subjective pain intensity. Journal of Neuroscience 32, 7429-7438.

Zubieta, J.-K., Bueller, J.A., Jackson, L.R., Scott, D.J., Xu, Y., Koeppe, R.A., Nichols, T.E., Stohler, C.S., 2005. Placebo effects mediated by endogenous opioid activity on μ-opioid receptors. Journal of Neuroscience 25, 7754-7762.

APPENDIX A

Supplemental Material for Research Chapters

The following is the supplemental material for the study entitled "Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and metaanalysis" which forms Chapter Two of this thesis.

Download history Clear history

Tutorial

PubMed Home More Resources Help

PubMed Advanced Search Builder

Filters activated: Journal Article, Full text, Publication date to 2017/08/31, Humans, English. Clear all

The following terms were not found in PubMed: interpersonal[MeSH Major Topic], attachment[MeSH Major Topic].

Quoted phrase not found.

<u>Edit</u>						
Builder						
All Fields	\$	•			Show in	dex list
AND V All Fields	3	•			Show in	dex list

History

Search	Add to builder	Query	Items found	Time
<u>#13</u>	Add	Search (((((((()[qain[MeSH Major Topic]) AND interpersonal[MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01"[PDat] : "2017/08/31"[PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND "social interaction"[MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01"[PDat] : "2017/08/31"[PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND "social modulation"[MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01"[PDat] : "2017/08/31" [PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND "social context" [MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01"[PDat] : "2017/08/31" [PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND "social context" [MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01"[PDat] : "2017/08/31"[PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND attachment[MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01"[PDat] : "2017/08/31"[PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND "social influence"[MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01" [PDat] : "2017/08/31"[PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND "social touch"[MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01" [PDat] : "2017/08/31"[PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND empathy[MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01" [PDat] : "2017/08/31"[PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND menathy[MeSH Major Topic] AND Journal Article[ptyp] AND full text[sb] AND ("0001/01/01" [PDat] : "2017/08/31"[PDat]) AND Humans[Mesh] AND English[Iang])) OR ((pain[MeSH Major Topic]) AND "social support"[MeSH Major Topic] AND Journal	349	20:53:57
<u>#12</u>	Add	Search (pain[MeSH Major Topic]) AND "social presence"[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>0</u>	20:53:13
<u>#11</u>	<u>Add</u>	Search (pain[MeSH Major Topic]) AND "social support"[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>152</u>	20:52:39
<u>#10</u>	<u>Add</u>	Search (pain[MeSH Major Topic]) AND empathy[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>198</u>	20:51:45
<u>#9</u>	Add	Search (pain[MeSH Major Topic]) AND "social touch"[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	Q	20:50:25
<u>#8</u>	Add	Search (pain[MeSH Major Topic]) AND "social influence"[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>0</u>	20:50:03

Advanced search - PubMed - NCBI

Search	Add to builder	Query	Items found	Time
<u>#7</u> .	Add	Search (pain[MeSH Major Topic]) AND attachment[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>0</u>	20:49:42
<u>#6</u>	Add	Search (pain[MeSH Major Topic]) AND "social context"[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>0</u>	20:49:16
<u>#5</u>	Add	Search (pain[MeSH Major Topic]) AND "social modulation"[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>0</u>	20:48:50
<u>#4</u>	Add	Search (pain[MeSH Major Topic]) AND "social interaction"[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>0</u>	20:48:21
<u>#1</u>	Add	Search (pain[MeSH Major Topic]) AND interpersonal[MeSH Major Topic] Filters: Journal Article; Full text; Publication date to 2017/08/31; Humans; English	<u>0</u>	20:45:23

To search an exact word(s) use quotation marks, e.g. "hospital" finds hospital; hospital (no quotation marks) finds hospital and hospitals; pay finds paid, pays, paying, payed)

Add to top		View fewer lines
#1 pain:ti,ab,kw and "interpersonal":ti,ab,kw	<u>216</u>	
#2 pain:ti,ab,kw and "social support":ti,ab,kw in Trials	275	
#3 "pain":ti,ab,kw and "social presence":ti,ab,kw in Trials	<u>0</u>	
#4 "pain":ti,ab,kw and "social interaction":ti,ab,kw in Trials	<u>294</u>	
#5 "pain":ti,ab,kw and "social modulation":ti,ab,kw in Trials	Q	
#6 "pain":ti,ab,kw and "social context":ti,ab,kw in Trials	Z	
#7 "pain":ti,ab,kw and "attachment":ti,ab,kw in Trials	177	
#8 "pain":ti,ab,kw and "social influence":ti,ab,kw in Trials	Z	
#9 "pain":ti,ab,kw and "social touch":ti,ab,kw in Trials	<u>0</u>	
#10 "pain":ti,ab,kw and "empathy":ti,ab,kw in Trials	55	
#11 #1 or #2 or #3 or #4 or #5Earc#6ras #7 sear#8 or #9 or #10	988	
<	•	
Clear Strategy		— Hicklicht ombo
Search Help		lines
Save strategy		
Strategy Name	Save Strategy	
Comments		

\bigcirc	الم ال					(ک). ۱	Wolters Kluwer
O\	DIV			My Account	Support & Training	Monash University Library Ask a Monast	Help Logoff h University Librariar
Searc	h	ournals Books Multimedia My Workspace OvidMD EBP Tools Mobile					
Basic Enter k (* or \$ f	1 R eyword for trunc	Find Citation Search Tools Search Fields Advanced Search Multi-Field Search ource selected Hide Chance mbase 1974 to 2017 September 14 phrase ion) Search Include Multimedia Map Term to Subject Heading Ovid Full Text Available English Language Publication Year Publication Types Source Types Conference Abstracts MEDLINE Publication Limits Edit Limits					
:	Search	listory (21)				View Sa	aved
	# 🔺	Searches	Results	Туре	Actions	Annotations	
	1	(pain and interpersonal).mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word]	6213	Advanced	Display Results More	•	Contract
	2	limit 1 to (ovid full text available and human and english language and embase and article and journal)	185	Advanced	Display Results More	÷ 🗘	
	3	(pain and "social support").mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word]	4059	Advanced	Display Results More	•	
	4	limit 3 to (ovid full text available and human and english language and embase and article and journal)	206	Advanced	Display Results More	ə 🖓	
	5	(pain and "social presence").mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word]	7	Advanced	Display Results More	÷ 🗘	
	6	limit 5 to (ovid full text available and human and english language and embase and article and journal)	1	Advanced	Display Results More	ə 🖓	
	7	(pain and "social interaction").mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word]	3903	Advanced	Display Results More	÷ 🖓	
	8	limit 7 to (ovid full text available and human and english language and embase and article and journal)	156	Advanced	Display Results More	e 🗘	
	9	(pain and "social modulation").mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word]	9	Advanced	Display Results More	;	
	10	limit 9 to (ovid full text available and human and english language and embase and article and journal)	2	Advanced	Display Results More	•	
	11	(pain and "social context").mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word]	227	Advanced	Display Results More	• 🖓	
	12	- limit 11 to (ovid full text available and human and english language and embase and article and journal)	21	Advanced	Display Results More	e 🗘	
	13	(pain and attachment).mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word]	1835	Advanced	Display Results More	,	
	13 14	(pain and attachment).mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word] limit 13 to (ovid full text available and human and english language and embase and article and journal)	1835 92	Advanced Advanced	Display Results More	»	

15/09/2017

Ovid: Search Form

16 limit 15 to (ovid full t	ext available and human and english language and embase and article and journal)	5	Advanced	Display Results More	
17 (pain and "social tou	ch").mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading	6	Advanced	Display Results More	\Box
18 limit 17 to (ovid full t	ext available and human and english language and embase and article and journal)	1	Advanced	Display Results More	
 19 (pain and empathy). 	mp. [mp=title, abstract, heading word, drug trade name, original title, device manufacturer, drug manufacturer, device trade name, keyword, floating subheading word]	1548	Advanced	Display Results More	\Box
20 limit 19 to (ovid full t	ext available and human and english language and embase and article and journal)	51	Advanced	Display Results More	
21 2 or 4 or 6 or 8 or 10	or 12 or 14 or 16 or 18 or 20	649	Advanced	Display Results More	
					~
Save Remove Combine w	ith: AND OR				
Save All Edit Create RSS	View Saved				
Options	Print Email Export	+ My Projects	Keep Selected	To search Open Access o	content on Ovid, go to Basic Search.
 Search Information 	All Range Clear 20 Per Page V 1 Go				Next >
You searched:					
2 or 4 or 6 or 8 or 10 or 12 or 14 or 16 or 18 or 20	1. Development and Validation of Quality Criteria for Providing Patient- and Family-centered Injury Care.				Ovid Full Text
Secret forms used	Boyd J.M., Burton R., Butler B.L., Dyer D., Evans D.C., Felteau M., Gruen R.L., Jaffe K.M., Kortbeek J., Lang E., Lougheed V., Moore L., Narciso M., Oxland P., Ri	/ara F.P., Rober	ts D., Sarakbi D., V	Vine K., Stelfox H.T.	Abstract Reference Complete Reference
attachment	Armais of Surgery. 200 (2) (pp 207-290), 2017. Date of Publication. 01 Aug 2017.				
context					Similar
empathy	Publisher Lippincott Williams and Wilkins (E-mail: kathiest.clai@apta.org)				Sind Citing Articles
influence interaction					
interpersonal	Abstract 🛛 📆 Article as PDF (309KB) 🛛 🔜 + My Projects 🛛 🔜 + Annotate				
modulation					
pain					
social	2. Validation of the social provisions scale in people with multiple sclerosis.				Ovid Full Text
support	Chiu CY., Motl R.W., Ditchman N.				Abstract Reference
touch	Rehabilitation Psychology. 61 (3) (pp 297-307), 2016. Date of Publication: 01 Aug 2016.				Complete Reference
Search Returned:	[Article]				
649 text results	Publisher				Grind Similar
Sort By:	American Psychological Association Inc. (E-mail: journals@apa.org)				Sind Citing Articles
- •	Abstract 💭 Acticle on DDE (166KD) 🔤 - My Projecto 📃 - Appetato				
Customize Display					
▼ Filter By	3. The empathy impulse: A multinomial model of intentional and unintentional empathy for pain.				Ovid Full Text
Add to Soarah History	Cameron C.D., Spring V.L., Todd A.R.				Abstract Reference
Auu to Search History	Emotion. 17 (3) (pp 395-411), 2017. Date of Publication: 01 Apr 2017.				Complete Reference
Selected Only (0)	[Article]				
Years	Publisher				Similar
All reals	American Psychological Association Inc. (E-mail: journals@apa.org)				Find Citing Articles
Past 3 years					
Past 5 years	Abstract 📆 Article as PDF (350KB) 🛛 🔜 + My Projects 🛛 🔜 + Annotate				
Specific Year Range					
Subject	A Fasters influencing amount of weakly exercise time in colorectal concer our dyors				
Author	4. ractors innuencing annum of weekig exercise time in colorectal cancer survivors.				Ovid Full Text
Journal	Circu TJ., Lai TH., Lin BK., Liang JI., Shun SC.				Complete Reference
Publication Type	Cancer Nursing. 40 (3) (pp 201-208), 2017. Date of Publication: 01 May 2017.				

Ovid®					•	WOLLEIS KLUWEI
Ovid			My Account	t Support & Training	Monash University Librar Ask a Mona	y Help Logoff ish University Libraria
Search Journals	s Books Multimedia My Workspace OvidMD EBP Tools Mobile					
Desis Oserah Fire	d Otherse L Osciela L Osciela L Advanced Osciela L Malli Field Osciela					
Basic Search Find	d Citation Search Tools Search Fields Advanced Search Multi-Field Search					
PsycINF	50 1806 to September Week 2 2017					
Enter keyword or phrase	e • Keyword • Author • Title • Journal					
(* or \$ for truncation)	(Search)					
	Limits (close) Include Multimedia Map Term to Subject Heading					
	Full Text PsycARTICLES Journals All Journals Latest Lindate Human Epolish Language					
	Abstracts Test DOI Open Access					
	Impact Statement					
	Publication Year -					
	Publication Types					
	0100 Journal 0110 Peer Peviewed Journal					
	0120 Non-Peer-Reviewed Journal			1	103	
				1	195	
	0200 Book					
	Additional Limits Edit Limits					
	Additional Limits Edit Limits					
Search History	Additional Limits Edit Limits				View S	Saved
Search History	Additional Limits Edit Limits y (21) ches	Results	Туре	Actions	View S Annotations	Saved
Search History # Searc 1 (pain a	Additional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures]	Results 1771	Type Advanced	Actions Display Results More	View S Annotations	Saved
Search History # ▲ Search 1 (pain and and and and and and and and and an	Additional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal")	Results 1771 176	Type Advanced Advanced	Actions Display Results More Display Results More	View S Annotations	Gaved
Search History # ▲ Search 1 (pain and and and and and and and and and an	Additional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures]	Results 1771 176 1362	Type Advanced Advanced Advanced	Actions Display Results More Display Results More Display Results More	View S Annotations	Saved Contract
Search History #▲ Search 1 (pain and and and and and and and and and an	Additional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal")	Results 1771 176 1362 138	Type Advanced Advanced Advanced Advanced	Actions Display Results More Display Results More Display Results More Display Results More	View S	Saved Contract
Search History	Additional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures]	Results 1771 176 1362 138 9	Type Advanced Advanced Advanced Advanced Advanced	Actions Display Results More Display Results More Display Results More Display Results More Display Results More	View S	Saved Contract
Search History Image: Imag	Additional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal")	Results 1771 176 1362 138 9 3	Type Advanced Advanced Advanced Advanced Advanced Advanced	Actions Display Results More	View S	Saved Contract
Search History	Additional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal")	Results 1771 176 1362 138 9 3 295	Type Advanced Advanced Advanced Advanced Advanced Advanced Advanced	Actions Display Results More	View S	Saved Contract
Search History # A Search 1 (pain a) 2 limit 1 3 (pain a) 4 limit 3 5 (pain a) 6 limit 5 7 (pain a) 8 limit 7	Additional Limits Edit Limits Additional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal")	Results 1771 176 1362 138 9 3 295 22	Type Advanced Advanced Advanced Advanced Advanced Advanced Advanced Advanced	Actions Display Results More Display Results More	View S	Saved Contract
Search History # A Search 1 (pain a 2 limit 1 3 (pain a 4 limit 3 5 (pain a 6 limit 3 7 (pain a 8 limit 7 9 (pain a	Additional Limits Edit Limits (21) Ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] ito (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] ito (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures]	Results 1771 176 1362 138 9 3 295 22 8	Type Advanced Advanced Advanced Advanced Advanced Advanced Advanced Advanced Advanced	Actions Display Results More	View S	Saved Contract
Search History #▲ Search 1 (pain and 2 limit 1 3 (pain and 4 limit 3 5 (pain and 6 limit 3 6 limit 3 9 (pain and 9 (pain and 9 (pain and 10 limit 7	Additional Limits Edit Limits Additional Limits Edit Limits Additional Limits Edit Limits (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	Results 1771 176 1362 138 9 3 295 22 8 1 1	Type Advanced	Actions Display Results More	View S	Saved Contract
Search History # A Search 1 (pain a) 2 limit 1 3 (pain a) 4 limit 3 5 (pain a) 6 limit 5 7 (pain a) 8 limit 7 9 (pain a) 10 limit 9 11 (pain a)	Additional Limits Edit Limits Additional Limits Edit Limits Additional Limits Edit Limits Additional Limits Edit Limits ad ditional Limits Edit Limits ad disocial support), mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction"), mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction"), mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social modulation"), mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social modulation", mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social modulation", mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures]	Results 1771 176 1362 138 9 3 295 222 8 1 19	Type Advanced	Actions Display Results More Display Results More	View S	Saved Contract
Search History # A Search 1 (pain a) 2 limit 1 3 (pain a) 4 limit 3 5 (pain a) 6 limit 3 7 (pain a) 8 limit 7 9 (pain a) 10 limit 9 11 (pain a) 12 limit 1	Additional Limits Edit Limits Addition Limits Addition Pathodit Limits Addition Pathodit Addit Contents, key concepts, original title, tests & measures] Addition (Iuli text and human and english language and "0110 peer-reviewed journal") And "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] Addition Addition").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] Addition Addition").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] Addition Addition").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] Addition Addition").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] Additional Limits Addition").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] Additional Context").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] Additional Context").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] Additional Context").mp. [mp=title, abstract, heading word, tab	Results 1771 176 1362 138 9 3 295 222 8 1192 30	Type Advanced	Actions Display Results More Display Results More	View S	Saved Contract
Search History #▲ Search 1 (pain and 2 limit 1 3 (pain and 4 limit 3 5 (pain and 6 limit 5 7 (pain and 8 limit 5 9 (pain and 10 limit 5 11 (pain and 12 limit 1 13 (pain and	y (21) Ches dddttional Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social modulation").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and attachuman, a	Results 1771 176 1362 138 9 3 295 222 8 1 192 30 619	Type Advanced	Actions Display Results More	View 5	Saved Contract
Search History # A Search 1 (pain a) 2 limit 1 3 (pain a) 4 limit 3 5 (pain a) 6 limit 5 7 (pain a) 8 limit 7 9 (pain a) 10 limit 9 11 (pain a) 12 limit 10 13 (pain a) 14 limit 13	y (21) Ches dddttonal Limits Edit Limits y (21) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social modulation").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social modulation").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and attachment).mp. [m	Results 1771 176 1362 138 9 3 295 222 8 1 192 30 619 48	Type Advanced	Actions Display Results More	View 5	Saved Contract
Search History # A Search 1 (pain a) 2 limit 1 3 (pain a) 4 limit 3 5 (pain a) 6 limit 3 6 limit 3 7 (pain a) 9 (pain a) 10 limit 3 11 (pain a) 12 limit 11 13 (pain a) 14 limit 13 15 (pain a)	ddttional Limits Edit Limits ddttional Limits Edit Limits ddttional Limits Edit Limits ddttional Limits Edit Limits g(2) ches and interpersonal).mp. [mp=tite, abstract, heading word, table of contents, key concepts, original tite, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=tite, abstract, heading word, table of contents, key concepts, original tite, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence").mp. [mp=tite, abstract, heading word, table of contents, key concepts, original tite, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=tite, abstract, heading word, table of contents, key concepts, original tite, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction").mp. [mp=tite, abstract, heading word, table of contents, key concepts, original tite, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social modulation").mp. [mp=tite, abstract, heading word, table of contents, key concepts, original tite, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context").mp. [mp=tite, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context").mp. [mp=tite, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and tachement).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and	Results 1771 1776 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 1362 192 30 619 48 24	Type Advanced	Actions Display Results More	View 5	Saved Contract
Search History # A Search 1 (pain a) 1 (pain a) 2 limit 1 3 (pain a) 4 limit 3 5 (pain a) 6 limit 5 7 (pain a) 8 limit 7 9 (pain a) 10 limit 9 11 (pain a) 12 limit 11 13 (pain a) 14 limit 13 15 (pain a) 16 limit 15	deditional Limits Edit Limits deditional Limits Edit Limits deditional Limits Edit Limits deditional Limits Edit Limits y(2) ches and interpersonal).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social support").mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social presence".mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction",mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social interaction",mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context", mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and "social context", mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and atachment).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language and "0110 peer-reviewed journal") and atachment).mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] to (full text and human and english language	Results 1771 1776 1362 1383 9 3 295 222 8 1192 30 619 48 24 5	Type Advanced Advanced	Actions Display Results More	View 5	Saved Contract

15/09/2017

Ovid: Search Form

18 limit 17 to (full text a	nd human and english language and "0110 peer-reviewed journal") 0 Advanced Save More	
 19 (pain and empathy). 	mp. [mp=title, abstract, heading word, table of contents, key concepts, original title, tests & measures] 902 Advanced Display Results More	
20 limit 19 to (full text a	nd human and english language and "0110 peer-reviewed journal") 80 Advanced Display Results More 📿	
21 2 or 4 or 6 or 8 or 10	2 or 12 or 14 or 16 or 18 or 20 430 Advanced Display Results More	
Save Remove Combine w	vith: AND OR	
Save All Edit Create RSS	S View Saved	
Options	✓ Print Email Export + My Projects → Keep Selected To search Open Access content on O	vid, go to <u>Basic Search</u> .
 Search Information 	All Range Clear 20 Per Page V 1 Go	Next >
You searched:		
2 or 4 or 6 or 8 or 10 or 12 or 14 or 16 or 18 or 20	1. Health care providers' judgments in chronic pain: The influence of gender and trustworthiness. [References].	vid Full Text
	Schafer, Grainne; Prkachin, Kenneth M; Kaseweter, Kimberley A; de C Williams, Amanda C.	bstract Reference
Search terms used:	Pain. Vol.157(8), 2016, pp. 1618-1625.	omplete Reference
attachment	[Journal; Peer Reviewed Journal]	
empathy	Year of Publication	Find Similar
influence		The oning Autores
interaction	Publication Month/Season	
modulation		
pain	Abstract 📆 Article as PDF (249KB) 🛛 😹 + My Projects 🔜 + Annotate	
presence		
support		
touch	2. The impact of mood on empathy for pain: Evidence from an EEG study. [References].	bstract Reference
Search Returned:	Li, Xiang; Meng, Xianxin; Li, Hong; Yang, Jiemin; Yuan, Jiajin.	omplete Reference
430 text results	Psychophysiology. Vol.54(9), 2017, pp. 1311-1322.	
Sort By:	[Journal; Peer Reviewed Journal]	Find Similar
- •	Year of Publication	Find Citing Articles
Customize Display	2017 Rublication Month/Season	back for full text
	Sep	
 Filter By 		
Add to Search History		
Selected Only (0)		
Years	3. Culture moderates children's responses to ostracism situations. [References].	vid Full Text
All Years	Over, Harriet; Uskul, Ayse K.	bstract Reference
Current year	Journal of Personality and Social Psychology. Vol.110(5), 2016, pp. 710-724.	omplete Reference
Past 3 years	[Journal; Peer Reviewed Journal]	
Past 5 years	Year of Publication	Find Similar
Specific Year Range	2016	Find Citing Articles
Subject	Publication Month/Season	
Author	May	
Publication Type	Abstract 📆 Article as PDF (398KB) 🐻 + My Projects 🔜 + Annotate	
My Projects	 4. Affective empathy, cognitive empathy and social attention in children at high risk of criminal behaviour. [References]. 	

Supplementary Material S2: Methodological study appraisal

Each criterion was judged as either:

"Successfully fulfilled" (1);

"Partially fulfilled (0.5);

"Not fulfilled" (0).

1. Reported blinding processes of study purpose to participants.

2. Similar gender distribution across all condition/group.

© As long as the support and no-support condition have the same distribution.

3. Specification of stimulus location.

4. Specification of physical type of stimulus.

5. Report on psychophysical method of threshold determination.

© 0.5 was given to studies using cold pain as the temperature was consistent.

6. To which extent the study population represents the true population.

© 0.5 was given to studies having 1 SD of the sample size generated from this meta-analysis (55.16±34.75), which means at least 21 participants.

© 0.5 was given to studies having at least two categories of age range below. For studies that reported mean and SD of age, one SD was used to revert the data.

Early adult: 18-25 years;

Middle adult: 26-60;

Old adult: 61 and above
Study	Sample	1	2	3	4	5	6	Total
								score
Borsook & MacDonald	Healthy	1	1	1	1	1	0.5	5.5
(2010)	narticinants (N							
	=45)							
Brown et al. (2003)	Healthy	1	1	1	1	0.5	0.5	5
	participants							
	(N=101)							
Edwards et al. (2017)	Experiment 1:	0	1	1	1	0.5	0.5	4
	Healthy							
	participants							
	(N=48)							
	Experiment 2:	0	1	1	1	0.5	0.5	4
	Healthy							
	participants							
	(N=48)							
	Fun avin ant 2			1	1	0.5	0.5	
	Experiment 3:	0			Ţ	0.5	0.5	4
	nedicity							
	(N=48)							
Fisenberger et al. (2011)	Healthy	0	1	1	1	1	0	4
	participants	0	-	-	-	-	Ū	
	(N= 17)							
Fishman et al. (1995)	Healthy	0	1	1	1	0.5	0.5	4
	participants							
	(N=60)							
Gallant et al. (2017)	Healthy	0	1	1	1	1	0.5	4.5
	participants (N =							
	97)							
Goldstein et al. (2016) ^a	Healthy	0	1	1	1	1	0.5	4.5
	participants							
	(N= 23)							

Jackson et al. (2005) ^a	Healthy	0	1	1	1	0.5	1	4.5
	participants							
	(N= 88)							
Karmann et al. (2014)	Healthy	1	1	1	1	1	1	6
	participants							
	(N=126)							
Kleck et al. (1976)	Experiment 1:	0	1	1	1	1	0	4
	Healthy							
	participants							
	(N= 20)							
	Experiment 2:	0	1	1	1	1	0.5	4.5
	Healthy male							
	participants							
	(N= 40)							
Krahé et al. (2015)	Healthy	1	1	1	1	1	0.5	5.5
	participants							
	(N= 39)							
Master et al. (2009)	Healthy	0	1	1	1	1	0	4
	participants							
	(N= 25)							
McClelland & McCubbin	Healthy	0	1	1	1	0.5	0.5	4
2008)	participants							
	(N=68)							
Modić et al. (2010)	Healthy	1	1	1	1	0.5	1	5.5
	participants							
	(N= 43)							
Montoya et al. (2004)	Fibromyalgia	0	1	1	1	1	0.5	4.5
	patients (FIB)							
	(N= 16);							
	Migraine patients							
	(MIG)							
	(N= 16)							
Platow et al. (2007)	Healthy	1	1	1	1	0	1	5
	participants							
	(N= 54)							
Roberts et al. (2015)	Healthy	1	1	1	1	0.5	0.5	5
	participants							
	(N= 76)							

Sambo et al. (2010)	Healthy	0	1	1	1	1	0.5	4.5
	participants							
	(N= 30)							
Sullivan et al. (2004)	Healthy	1	1	1	1	0.5	0.5	5
	participants (N							
	=64)							
Vervoort et al. (2011)	Healthy children	1	1	1	1	0.5	0.5	5
	and parents (N=							
	38)							
Vlaeyen et al. (2009)	Healthy	1	1	1	1	0.5	1	5.5
	participants							
	(N= 149)							

Supplementary Material S3: Forest plot of the Hedge's adjusted g analysis for social presence on behavioural pain outcomes.

Author: Contrast: Outcome							Hedge's g (95% CI)	Weight %
Brown et al. (2003):3:1	Pain intensity	-	 				-0.55 (-1.43: 0.33)	0.36%
Edwards et al. (2017):1:1	· · · · · · · · · · · · · · · · · · ·						0.01 (-0.55; 0.58)	0.87%
Edwards et al. (2017):2:1			•				0.08 (-0.49; 0.65)	0.87%
Edwards et al. (2017):3:1			•				0.04 (-0.52; 0.61)	0.87%
Edwards et al. (2017):4:1			•				0.02 (-0.55; 0.58)	0.87%
Edwards et al. (2017):5:1			Þ				-0.04 (-0.6; 0.53)	0.87%
Edwards et al. (2017):6:1			l •				-0.01 (-0.57; 0.56)	0.87%
Gallant et al. (2017):1:1			I •				-0.01 (-0.31; 0.29)	3.13%
Gallant et al. (2017):2:1			<u> </u>				-0.1 (-0.4: 0.2)	3.13%
Goldstein et al. (2016):3:1							-0.66 (-1.26; -0.07)	0.79%
Karmannetal (2014):1:1			<u> </u>				-0.13 (-0.38: 0.12)	4.58%
Karmannetal (2014):2:1							-0.09 (-0.34: 0.16)	4.58%
Kleck et al. (1976):1:1							-0.24 (-0.86: 0.38)	0.72%
Kleck et al. (1976):2:1							-0.56(-1:-0.11)	1.4%
Krabé et al (2015):1:1							0.1 (-0.35:0.54)	1 42%
Krabé et al (2015):2:1				_			0.02 (-0.42: 0.45)	1 42%
McClelland & MarCubbin (2008):1:1							0.24(-0.23:0.72)	1 25%
Modić Stanke & Jyanes (2010):1:1							0.16(-0.43:0.75)	0.8%
Modić Stanke & Junes (2010) 2:1							0.17(-0.43:0.78)	0.75%
Monteur et al. (2004):1:1			-				0.22 (1.02:0.27)	0.70%
Roberts et al. (2004).1.1							-0.52 (-1.02, 0.57)	0.57%
Sombo et al. (2010):1:1			-				0.05 (0.45, 0.45)	1.09%
Sambo et al. (2010).1.1							0.00 (-0.43, 0.56)	1.05%
Sambo et al. (2010):2:1			1.	_			-0.01 (-0.52; 0.5)	1.09%
Sullivanet al. (2004):1:1							0.04 (-0.45; 0.53)	1.16%
vervoort et al. (2011):1:1	-						-0.31 (-0.76; 0.14)	1.57%
Vlaeyen et al. (2009):1:1							-0.24 (-0.56; 0.08)	2.69%
Subtotal		<	P				-0.09 (-0.18, -0.004)	38.44%
Gallant et al. (2017):1:2	Pain unpleasantness		<u> </u>				-0.08 (-0.38; 0.22)	3.13%
Gallant et al. (2017):2:2			₩				-0.16 (-0.46; 0.13)	3.12%
Modić Stanke & Ivane (2010):1:2							0.26 (-0.33; 0.85)	0.79%
Modić Stanke & Ivane (2010):2:2			H •				0.09 (-0.51; 0.7)	0.76%
Roberts et al. (2015):2:2				-			-0.11 (-0.67; 0.44)	0.91%
Subtotal		<					0.07 (0.25 0.11)	0 749/
Subtotal	Facial annual states		Π				-0.07 (-0.25, 0.11)	0./176
Gallant et al. (2017):1:3	racial expression of pain	-	•	_			0.2 (-0.1; 0.5)	3.11%
Gallant et al. (2017):2:3							-0.23 (-0.53; 0.07)	3.11%
Karmannetal. (2014):1:2							0.17 (-0.08; 0.42)	4.57%
Karmannetal. (2014):2:2							-0.03 (-0.28; 0.22)	4.59%
Kleck et al. (1976):1:2							-0.37 (-0.99; 0.26)	0.72%
Kleck et al. (1976):2:2							-0.54 (-0.98; -0.09)	1.4%
Sullivan et al. (2004):1:2			•				0.33 (-0.16; 0.83)	1.14%
Vervoort et al. (2011):1:2			•				0.36 (-0.09; 0.82)	1.36%
Vlaeyen et al. (2009):1:2							-0.01 (-0.33; 0.31)	2.71%
Subtotal		<	\sim				0.005 (-0.16, 0.17)	22.71%
Edwards et al. (2017)-1-2	Pain tolerance						-0.22 (-0.78: 0.35)	0.87%
Edwards et al. (2017):2:2							0.12(0.7:0.42)	0.07%
Edwards et al. (2017):2:2							0.28(0.95:0.28)	0.000
Edwards et al. (2017):4:2	_						-0.21 (-0.78: 0.36)	0.87%
Edwards et al. (2017).4.2		· · · ·					0.11(0.08,0.45)	0.07%
Edwards et al. (2017):5:2							0.1(0.55:0.47)	0.07%
Collect at al (2017)-0.2							0.05(0.00,0.77)	0.0/70
Gallant et al. (2017):1:4							0.05(-0.25; 0.55)	3.53%
McClolland & MacCubbia (2000)-1-2							0.09(0.22:0.24)	3.3370
Madiá Stanka 8, kursa (2010) 1.2				-			0.05 (-0.56; 0.50)	1.2570
Would stanke & Wane (2010):1:3			II ⁻ .				0.05(-0.54; 0.64)	0.8%
Modic Stanke & Ivane (2010):2:3							0.15(-0.41, 0.8)	0.76%
Subtotal		<	\Rightarrow				-0.04 (-0.17, 0.09)	15.09%
Edwards et al. (2017):1:3	Pain threshold		11	-			-0.1 (-0.67; 0.47)	0.87%
Edwards et al. (2017):2:3			41	_			-0.06 (-0.63; 0.5)	0.87%
Edwards et al. (2017):3:3			H				-0.16 (-0.73; 0.41)	0.87%
Edwards et al. (2017):4:3			<u> </u>	-			-0.11 (-0.68; 0.45)	0.87%
Edwards et al. (2017):5:3			 •				-0.03 (-0.6; 0.53)	0.87%
Edwards et al. (2017):6:3			14				-0.01 (-0.57; 0.56)	0.87%
Gallant et al. (2017):1:5			4∔				-0.06 (-0.35; 0.22)	3.53%
Gallant et al. (2017):2:5			+⊢				-0.22 (-0.5; 0.07)	3.51%
Modić Stanke & Ivane (2010):1:4			1+				0 (-0.59; 0.6)	0.8%
Modić Stanke & Ivane (2010):2:4			•				0.04 (-0.57; 0.64)	0.76%
Montoya et al. (2004):1:2			H				-0.1 (-0.6; 0.39)	1.16%
Subtotal		_	₩				-0.1 (-0.24, 0.04)	14.98%
Countly a nin			I				0.05 (0.44 . 0.004)	4000/
synthesis		*	-				-0.06 (-0.11, -0.004)	100%
	2 45 1					4.5		
	-2 -1.5 -1	-0.5	U	0.5	1	1.5		
	(Decrease pain)	Eff	ect size	(Incr	eased pain)			

Supplementary Material S4: Forest plot of the Hedge's adjusted g analysis for gender difference in the social support effect on behavioural pain outcomes.



The following is the supplemental material for the study entitled "A Systematic Review of the Processes Underlying the Main and the Buffering Effect of Social Support on the Experience of Pain", which forms Chapter Three of this thesis.

Supplementary Material S1

Quality assessment checklist (CS Cross-section, CH Cohort):

1: Does the study have a clear defined research objective? CH/CS

2: Does the study clearly describe the recruitment procedure? CH/CS

3: Does the study adequately describe the inclusion/exclusion criteria? CH/CS

4: Does the study report on the population parameters and demographics? CH/CS

5: Does the study report participation rates and provide evidence of comparisons of responders and

non-responders to the recruitment? CH/CS

6: Does the study include the sufficient assessment of social support and pain experience? Criteria-

Higher quality where measure is validated or measures at least two dimensions. CH/CS

7: Does the study adequately report on the strength of effect (e.g. ways of calculating effect size,

reporting of confidence intervals)? CH/CS

8: Does the study use multivariate analysis? CH/CS

9: Is the study sample size appropriate for the analysis used? CH/CS. The number of cases in the final

multivariable model was at least 10 times the number of independent variables in the analysis.

10: Do the authors report on the limitations of their study? CH/CS

11: Does the study report attrition rates and provide evidence of comparisons of responders and non-responders? CH

12: Does the study report an attrition rate<20 %? CH

13: Does the study have a follow up time period>6 months? CH

Quality assessment results:

Reference	Scori	ing cri	iteria	for q	uality	asse	ssme	nt						Score
	All de	esigns	6								Coho	rt		%
	1	2	3	4	5	6	7	8	9	10	11	12	13	
Affleck (1988)	Y	Y	Ν	Y	Y	Y	Y	Y	Y	Ν	N/A	N/A	N/A	80
Fitzpatrick (1988)	Y	Y	Ν	Y	Ν	Y	Ν	Y	Y	Y	N/A	N/A	N/A	70
Manne (1989)	Y	Y	Y	Y	Y	Y	Ν	Y	Y	Y	N/A	N/A	N/A	90
Kerns (1990)	Y	Ν	Ν	Y	Ν	Y	Y	Y	Y	Y	N/A	N/A	N/A	70
Weinberger (1990)	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	N/A	N/A	N/A	100
Revenson (1991)	Y	Y	Y	Y	Ν	Y	Y	Y	Y	Y	N/A	N/A	N/A	90
Goldberg (1993)	Y	Y	Y	Y	Ν	Y	Y	Y	Y	Y	N/A	N/A	N/A	90
Penninx (1997)	Y	Y	Ν	Y	Ν	Ν	Y	Y	Y	Y	N/A	N/A	N/A	70
Telfair (1999)	Y	Y	Y	Y	Ν	Ν	Y	Y	Y	Y	N/A	N/A	N/A	80
Cano (2000)	Y	Ν	Ν	Y	Ν	Y	Y	Y	Y	Y	N/A	N/A	N/A	70
Riemsma (2000)	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	N/A	N/A	N/A	100
Alonso (2001)	Y	Y	Ν	Ν	Ν	Y	Ν	Y	Y	Y	N/A	N/A	N/A	60
Kerns (2002)	Y	Y	Y	Y	Ν	Y	Y	Y	Y	Y	N/A	N/A	N/A	90
Cano (2004)	Y	Y	Y	Y	Ν	Y	Ν	Y	Ν	Y	N/A	N/A	N/A	70
Holtzman (2004)	Y	Ν	Y	Y	Y	Y	Ν	Y	Y	Ν	N/A	N/A	N/A	70
Ferreira (2007)	Y	Y	Ν	Y	Y	Y	Y	Y	Y	Y	N/A	N/A	N/A	90
Holtzman (2007)	Y	Y	Y	Y	Ν	Ν	Y	Y	Ν	Y	N/A	N/A	N/A	70
Lopez-Martinez (2008)	Y	Y	Y	Y	Y	Ν	Y	Y	Y	Y	N/A	N/A	N/A	90
Vlaeyen (2009)	Y	Y	Y	Y	Ν	Ν	Y	Y	Y	Y	N/A	N/A	N/A	80
Morgan (2011)	Y	Ν	Y	Y	Ν	Y	Y	Y	Y	Y	N/A	N/A	N/A	80
Raichle (2011)	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	N/A	N/A	N/A	100
Pekkarinen (2013)	Y	Y	Ν	Y	N	Y	Y	Y	Y	Y	N/A	N/A	N/A	80
Rosen (2013)	Y	Y	Y	Y	Ν	Y	Y	Y	Y	Y	N/A	N/A	N/A	90
Sturgeon (2015)	Y	Y	Ν	Y	Ν	Y	Y	Y	Y	Y	N/A	N/A	N/A	80
Corley (2016)	Y	Y	Y	Y	Ν	Ν	Ν	Y	Y	Y	N/A	N/A	N/A	70

Park (2016)	Y	Y	Y	Y	Ν	Ν	Y	Y	Y	Y	N/A	N/A	N/A	80
Delongis (1988)	Y	Y	Y	Y	Y	Y	Y	N	Y	Y	N	Y	Y	85
Brown (1989)	Y	Y	Y	Y	Ν	Y	Y	Y	Y	Y	Ν	Ν	Y	77
Affleck (1994)	Y	Y	Y	Y	Y	Y	Ν	Y	Y	Y	Ν	Y	Ν	77
Waltz (1998)	Y	Y	Ν	Y	Y	Y	Y	Y	Y	Y	Ν	Y	Y	85
Feldman (1999)	Y	Y	Ν	Y	Ν	Ν	Ν	Y	Y	Y	Y	Y	Ν	62
Strating (2006)	Y	Ν	Y	Y	Ν	Y	Y	Y	Y	Y	Y	Ν	Y	77
Lee (2016)	Y	Y	Y	Y	Ν	Y	Ν	Y	Y	Y	Ν	Ν	Y	69
Overall total %	100	85	67	97	33	76	73	97	94	94	29	57	71	

Y yes, N no, N/A not applicable

Supplementary Material S2

Reference	Findings	Effect	Effect size statistics	Effect size	
				proximity	
Stress (physical s	uffering)				
Brown et al.	1. The negative correlation of	Buffering	Differences of R ² in multiple	0.055 (wave 1	
(1989) [16]	emotional support and depression was		regression (Free Statistics	and 2	
	strengthened when pain was included.		Calculators)	averaged)	
	2. Moderating effect of emotional	Main	Support and pain effect were	n/a	
	support was not found over a 6-month		reported together.		
	period.				
Revenson et al.	1. Received positive support predicted	Main	Multiple regression	β = 0.250	
(1991) [66]	reduced depression without interacting				
	with pain severity.				
Penninx et al.	1. Presence of a partner had direct,	Main	Multiple regression (no	β = 0.165	
(1997) [39]	favourable effect on psychological		disease and severe arthritis		
	functioning without interacting with		averaged)		
	arthritis.				
	2. Close social relationships had direct,	Main	Multiple regression (no	β = 0.100	
	favourable effect on psychological		disease and severe arthritis		
	functioning without interacting with		averaged)		
	arthritis.				
	3. Having diffuse social relationships	Buffering	Multiple regression (no	$\Delta \beta = 0.100$	
	interacted with arthritis pain to predict		disease and severe arthritis		
	depressive symptoms in severe		difference)		
	arthritis.				
	4. Emotional support interacted with	Buffering	Multiple regression (no	Δβ=0.008	
	arthritis pain to predict depressive		disease and severe arthritis		
	symptoms in severe arthritis.		difference)		

Feldman et al.	1. Previous day' support interacted	Buffering	Multiple regression	B= 0.02
(1999) [59]	with previous day' pain to predict			
	present day' depressed mood.			
Telfair and	1. Group satisfaction interacted with	Buffering	ANOVA (common variance	0.237
Gardner (1999)	high pain to predict high psychological		was set to 0.5 in WebPower)	
[61]	wellbeing.			
Riemsma et al.	1. Positive support predicted reduced	Main	Multiple regression	β = 0.150
(2000) [49]	depression without interacting with			
	pain.			
Cano et al. (2004)	1. Marital satisfaction predicted	Main	Multiple regression	β = 0.210
[17]	reduced depressive symptoms without			
	interaction with pain.			
Ferreira and	1. Perceived social support mediated	Buffering	Differences of R ² in multiple	0.147
Sherman (2007)	the impact of pain on depressive		regression (Free Statistics	
[63]	symptoms.		Calculators)	
Holtzman and	1. Morning satisfaction with spousal	Buffering	Multiple regression	b = 0.030
DeLongis (2007)	response interacted with morning pain			
DeLongis (2007) [62]	response interacted with morning pain severity to predict evening pain			
DeLongis (2007) [62]	response interacted with morning pain severity to predict evening pain catastrophizing.			
DeLongis (2007) [62] López-Martínez et	response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated	Buffering	Path analysis	0.029 (path
DeLongis (2007) [62] López-Martínez et al. (2008) [16]	 response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated the relationship between perceived 	Buffering	Path analysis	0.029 (path coefficient)
DeLongis (2007) [62] López-Martínez et al. (2008) [16]	 response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated the relationship between perceived social support and decreased 	Buffering	Path analysis	0.029 (path coefficient)
DeLongis (2007) [62] López-Martínez et al. (2008) [16]	response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated the relationship between perceived social support and decreased depression.	Buffering	Path analysis	0.029 (path coefficient)
DeLongis (2007) [62] López-Martínez et al. (2008) [16]	 response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated the relationship between perceived social support and decreased depression. 2. Decreased pain intensity mediated 	Buffering Buffering	Path analysis Path analysis	0.029 (path coefficient) 0.031 (path
DeLongis (2007) [62] López-Martínez et al. (2008) [16]	response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated the relationship between perceived depression. 2. Decreased pain intensity mediated the relationship between perceived	Buffering Buffering	Path analysis Path analysis	0.029 (path coefficient) 0.031 (path coefficient)
DeLongis (2007) [62] López-Martínez et al. (2008) [16]	response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated the relationship between perceived depression. 2. Decreased pain intensity mediated the relationship between perceived social support and decreased	Buffering Buffering	Path analysis Path analysis	0.029 (path coefficient) 0.031 (path coefficient)
DeLongis (2007) [62] López-Martínez et al. (2008) [16]	response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated the relationship between perceived social support and decreased depression. 2. Decreased pain intensity mediated the relationship between perceived social support and decreased	Buffering Buffering	Path analysis Path analysis	0.029 (path coefficient) 0.031 (path coefficient)
DeLongis (2007) [62] López-Martínez et al. (2008) [16]	response interacted with morning pain severity to predict evening pain catastrophizing. 1. Decreased pain intensity mediated the relationship between perceived social support and decreased depression. 2. Decreased pain intensity mediated the relationship between perceived social support and decreased functional impairment. 3. Decreased pain intensity mediated	Buffering Buffering Buffering	Path analysis Path analysis Path analysis	0.029 (path coefficient) 0.031 (path coefficient) 0.019 (path

social support and increased functional

status.

Morgan et al.	1. Partner support mediated the	Buffering	Path analysis	0.070 (path
(2011) [65]	negative impact of pain on quality of			coefficient)
	life.			
Sturgeon et al.	1. Satisfaction with social roles	Buffering	Path analysis	0.114 (path
(2015) [64]	mediated the negative impact of pain			coefficient)
	intensity on depression.			
	2. Satisfaction with social roles	Buffering	Path analysis	0.081 (path
	mediated the negative impact of pain			coefficient)
	intensity on anger.			
Lee et al. (2016)	1. Tangible social support interacted	Buffering	Multiple regression	β = 0.040
[18]	with arthritis pain to predict less			
	depressive symptoms.			
Park et al. (2016)	1. Social support interacted with pain	Buffering	Multiple regression	β = 0.040
[60]	intensity to predict depressive			
	symptoms.			

Stress (functional disability)

Affleck et al.	1. The positive relation between	Buffering	Differences of R ² in multiple	0.143
(1988) [68]	support satisfaction and psychosocial		regression (Free Statistics	
	adjustment was increased when		Calculators)	
	disability was included.			
Fitzpatrick et al.	1. Social relationship had direct,	Main	ANOVA (support sum of	0.202
(1988) [71]	favourable effect on psychological well-		square divided by total sum	
	being without interacting with		of square)	
	disability.			
Goldberg et al.	1. Spousal support interacted with	Buffering	Multiple regression	β = 0.800
(1993) [69]	pain-related interference to predict			
	depression.			

Riemsma et al.	1. Positive support predicted reduced	Main	Multiple regression	β = 0.140
(2000) [49]	depression without interacting with			
	physical function.			
Cano et al. (2004)	1. Marital satisfaction predicted	Main	Multiple regression	β = 0.210
[17]	reduced depressive symptoms without			
	interacting with physical disability.			
Strating et al.	1. Emotional support satisfaction	Main	Multiple regression	β = 0.170
(2006) [70]	predicted reduced distress without			
	interaction with impairment-disability			
	in short-term RA.			
	2. Satisfaction with social	Bufferina	Multiple regression	β = 0.220
	companionship predicted reduced	Dujjernig		
	distress by interacting with			
	impairment-disability in short-term RA.		Multiple regression	β = 0.110
	3. This interaction disappeared in long-	Main		
	term RA.			

Stress (stressful response of close other)

Kerns et al. (1990)	1. Marital satisfaction interacted with	Buffering	Multiple regression (only Δ	n/a
[72]	higher spousal punishing response to		R^2 , no β or separate R^2)	
	predict lower depressive symptoms.			
Revenson et al.	1. Received positive support interacted	Buffering	Multiple regression (no β)	n/a
(1991) [66]	with high stressful response to predict			
	reduced depression.			
Cano et al. (2000)	1. Marital satisfaction mediated the	Buffering	Path analysis (gender	0.120 (path
[73]	positive relationship between negative		averaged)	coefficient)
	spousal response and depressive			
	symptom.			

Riemsma et al.	1. Positive support interacted with	Buffering	Multiple regression	β = 0.113
(2000) [49]	stressful response to predict			
	depression.			
Cano et al. (2004)	1. Marital satisfaction predicted	Main	Multiple regression	β = 0.210
[17]	reduced depressive symptoms without			
	interaction with negative spousal			
	response.			
Raichle et al.	1. Marital satisfaction predicted	Main	Multiple regression	β = 0.430
(2011) [75]	reduced depression directly, without			
	interacting with spousal negative			
	response.			
Rosen et al.	1. Dyadic adjustment mediated the	Buffering	Path analysis	0.104 (path
(2013) [74]	adverse impact of negative spousal			coefficient)
	response on sexual satisfaction.			
Stress (other negati	ve life events)			
DeLongis et al.	1. Emotional support was negatively	Buffering	Multiple regression	β = 0.120
(1988) [78]	related to hassle-next day symptoms			
	association.			
	2. Emotional support was negatively	Buffering	Multiple regression	β = 0.200
	related to hassle-same day mood			
	association.			
	3. Emotional support was not related			
	to hassle-same day symptoms	Main	Multiple regression (no β or	n/a
	association.		separate R ²)	
	4. Emotional support was not related			
	to hassle-next day mood association.	Main	Multiple regression (no β or	n/a
			separate R ²)	
Weinberger et al.	1. Social support had direct, favourable	Main	Multiple regression (no β or	n/a
(1990) [79]	effect on functional status, without		separate R ²)	
	interaction with daily hassle.			

Affleck et al.	1. Social support interacted with daily	Buffering	Multiple regression	β = 0.410
(1994) [76]	stressor to predict next-day mood			
	disturbance.			
Alonso and Coe	1. Access to support providers	Buffering	ANOVA (only F and p value)	n/a
(2001) [80]	interacted with distress to predict			
	menstrual pain.			
Pekkarinen et al.	1. Social support interacted with	Buffering	Logistic regression	0.49
(2013) [77]	physical workload to predict			
	musculoskeletal symptoms.			
Stress appraisal (pe	rceived threat/ coping ability)			
Waltz et al. (1998)	1. Psychological functioning mediated	Buffering	Path analysis (no path	n/a
[82]	the protective influence of emotional		coefficient)	
	support on pain.			
	2. Psychological functioning mediated	Buffering	Path analysis (no path	n/a
	the protective influence of social		coefficient)	
	interaction on pain.			
Kerns et al. (2002)	1. Pain-related support interacted with	Buffering	Differences of R ² in multiple	0.042
[81]	low problem-solving competence to		regression (Free Statistics	
	predict depressive symptoms.		Calculators)	
Holtzman and	1. Morning satisfaction with spousal	Buffering	Multiple regression	b = 0.090
DeLongis (2007)	response interacted with morning pain			
[62]	catastrophizing to predict evening			
	negative affect.			
	2. Morning satisfaction with spousal			
	response decreased evening pain	Main	Multiple regression	b = 0.220
	intensity without interaction with			
	morning pain catastrophizing.			
Vlaeyen et al.	1. Perceived threat mediated the	Buffering	Path analysis	0.098 (path
(2009) [6]	inhibitory effect of social presence on			coefficient)
	pain intensity.			

Buffering

Path analysis

	2. Perceived threat mediated the			0.066 (path
	inhibitory effect of social presence on			coefficient)
	facial expression of pain.			
Corley et al.	1. Global relationship satisfaction	Main	Quadratic modelling (no	n/a
(2016) [83]	predicted less pain without interacting		effect size)	
	with threat manipulation.			
	2. Situational relationship satisfaction	Pufforing	Quadratic modelling (no	n/a
	predicted less pain in the low threat	Dujjernig	effect size)	
	condition.			
Active coping				
Manne and	1. Adaptive coping mediated the	Buffering	Path analysis	0.121 (path
Zautra (1989) [48]	positive relationship between spousal			coefficient)
	support and better psychological			
	adjustment.			
Holtzman et al.	1. Morning satisfaction with social	Buffering	Multiple regression	β = 0.080
(2004) [10]	support interacted with increased stoic			
	distancing to predict reduced evening			
	pain severity.			
López-Martínez et	1. Perceived social support was related	Main	Path analysis	0.120 (path
al. (2008) [19]	to reduced depressed mood without			coefficient)
	interacting with active coping			
	response.			
	2. Perceived social support was related	Main	Path analysis	0.110 (path
	to reduced pain intensity without			coefficient)
	interacting with active coping			
	response.			

Note: buffering indicates the buffering effect; main means the main effect. Effect size proximity is shown with the absolute value. β is standardized coefficient, while B and b are unstandardized coefficients. R² is the goodness-of-fit measure for the model.

The following is the supplemental material for the study entitled *"The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat"*, which forms Chapter Four of this thesis.



Supplementary Material S1. Absolute heart rate changes and relationship with pain relief. (a) Shows the event-related dynamic changes of heart rate with the progression of countdown numbers. (b) Shows the area under the curve (AUC) of heart rate changes across conditions. Each column and error bar represent the mean and the SEM. Asterisk represents statistical significance (p<0.05). n.s means non-significant. (c) Shows the correlation between heart rate reduction (HR (no-holding) – HR (close other)) and pain relief (Pain intensity (no-holding) – Pain intensity (close other)).



Supplementary Material S2. Source localization of EEG theta oscillation in the entire presentation of countdown numbers (0-6000 ms) in the close other versus no-holding condition.

The following is the supplemental material for the study entitled *"The Medial Prefrontal Cortex as a Flexible Hub Mediating Behavioural as well as Local and Distributed Neural Effects of Social Support on Pain: a Theta Burst Stimulation and TMS-EEG Study"*, which forms Chapter Six of this thesis.

Supplementary Material S1: A series of negative and positive peaks resulted from single-pulse TMS over the mPFC.

(a) Butterfly plot of all electrodes with peaks of interest shown in text. The red line indicates the waveform obtained from the fronto-central electrode FCZ for graphical representation. (b) Voltage distribution and (c) Minimum Norm Estimates (MNEs) of the source level activity at the cortex for each peak.



Supplementary Material S2: Overall perceived support as a function of TBS and time.

(a)- (b) Overall perceived support decreased from Pre- to Post-stimulation across TBS conditions



APPENDIX B

Poster Abstracts

Australian Pain Society, Sydney, Australia, 2018

Che, X., Cash, R., Fitzgerald, P., &, Fitzgibbon B. The social regulation of pain: autonomic and neurophysiological changes associated with perceived threat.

Background and Aims:

There is evidence to suggest that social support reduces the intensity of pain. More recent studies have suggested that this effect could be mediated by the social-buffering effect on perceived threat of pain. However, previous studies have ultimately combined painful sensations with perceived threat of pain, which makes it difficult to directly examine this idea. In this study we therefore used a unique study design to tease apart the influence of social support on the threat of pain, as opposed to actual pain, to more directly investigate the social-buffering effect of pain and related autonomic and neural responses.

Methods:

Eighteen healthy participants (8 males and 10 females, age range: 18-35 years, Mean=25.2, *SD*=5.7) were included in the study. Subjects were asked to hold the hand of a close other, or a stranger or not at all while they experienced the threat of pain. This was induced by presenting countdown numbers from 6 to 1 (0-6s) which was followed by a painful stimulus. During this time, neural and autonomic responses were recorded using electroencephalogram (EEG) and heart rate respectively.

217

Results:

One-way ANOVAs revealed that close other hand-holding reduced pain intensity (95% CI = [-1.75, -.13], p < .05) and pain unpleasantness (95% CI = [-1.26, -.04], p < .05) compared to no hand-holding. This was accompanied by decreased heart rate (95% CI = [-7.88, -.21], p < .05), and frontal theta oscillation (4-8Hz, significant at AF4, F2, FZ, and FC2) (p < .05) in the later stage preceding painful stimulation (3-6s in heart rate and 3.3-4.3s in frontal theta respectively) in the close other versus no-holding condition. Interestingly, decreased heart rate and frontal theta were uniquely associated with greater pain reduction in the later stage leading up to painful stimulation (3.5-6s in heart rate and 3-5.5s in frontal theta respectively, $p_s < .05$). Neural changes in close other hand-holding were further source localized to the insular cortex and the rostral-ventral portions of the anterior cingulate cortex, regions involved in the processing of threat and pain.

Conclusions:

The results of the present study demonstrate the analgesic effect of social support. Our data further suggest that this effect could be mediated by the changes in autonomic and neurophysiological responding to the threat of nociceptive stimuli. Overall, our findings present evidence that social support may decrease pain through a buffering effect on the threatening quality of pain.

World Congress on Pain, Boston, USA, 2018

Che, X., Cash, R., Chung, S., Fitzgerald, P., &, Fitzgibbon B. (2018). Investigating the influence of social support on experimental pain and related physiological arousal: A systematic review and meta-analysis.

Aim of Investigation:

Under the right conditions, having supportive relationships with others can reduce pain and related physiological arousal. However, some studies have shown no effect or even a negative effect of social support on pain and arousal. A recent systematic review has further suggested possible covariates of this influence, for example, social support variations (e.g. physical presence, social touch), and social relationships (e.g. significant other, stranger). In the current study, a meta-analysis was therefore conducted to quantify the complex and multivariate influence of social support on pain and related arousal.

Methods:

A comprehensive electronic literature search was performed in PubMed, PsycINFO, The Cochrane Library and EMBASE to the end of August 2017. The keywords used for the search were 'pain' AND ('interpersonal' OR 'social support' OR 'social presence' OR 'social interaction' OR 'social modulation' OR 'social context' OR 'attachment' OR 'social influence' OR 'social touch' OR 'empathy'). Two reviewers (XC and SC) independently assessed the search results against the inclusion criteria: (1) pain was induced in experimental settings; and (2) both the presence and absence of social support was manipulated; and (3) behavioural pain outcome was assessed; and (4) data were provided that enable analysis and estimation of effect size; and (5) it was published in a peer-reviewed journal in English. Behavioural pain outcomes included pain intensity, pain unpleasantness, facial expression of pain, pain threshold, and pain tolerance. Physiological outcomes included heart rate, blood pressure (both diastolic and systolic), skin response, and cortisol levels. Data were extracted to calculate

219

effect size using Hedge's adjusted g (standardised mean difference, SMD). Test of heterogeneity (I^2 statistics), and publication bias (selectivity funnel plot, Begg-Mazumdar Kendall τ , Egger's regression test) were also performed.

Results:

A total of 2416 studies were identified in a systematic search, among which 20 were eligible for the meta-analysis. Our results showed that the mere presence of another person was not sufficient to decrease pain perception (with no SMD \ge 0.2, and $p \le$ 0.05). However, the presence of a significant other resulted in an increase of facial expression of pain (SMD = 0.21, 95% CI: [0.03, 0.38], p = 0.02), and the presence of a stranger decreased physiological arousal (SMD = -0.31, 95% CI: [-0.48, -0.13], p = 0.0004). Meanwhile, verbal support, mainly from a stranger, decreased both pain (SMD = -0.69, 95% CI: [-1.30, -0.08], p = 0.03) and arousal (SMD = -0.99, 95% CI: [-1.29, -0.69], p = 0.00001). Our data also showed the analgesic effect of intimate relationships occurred in response to touching (SMD = -0.95, 95% CI: [-1.37, -0.52], p = 0.00001), as well as viewing a romantic other (SMD = -0.60, 95% CI: [-1.04, -0.16], p = 0.008). Finally, we found evidence of publication bias for pain-related arousal ($\tau = -0.32$, p = 0.03; Egger's t = -3.25, p = 0.05), but not for behavioural pain outcomes ($\tau = -0.14$, p = 0.07; Egger's t = -0.72, p = 0.08).

Conclusions:

In this study we systematically quantified the influence of social support on pain and related arousal. Our results suggest that the mere presence of another person, although it may suppress physiological arousal, is not sufficient to reduce pain. However, social support is shown to decrease pain when there is active engagement of social support, e.g. verbal communication, hand-holding. Findings further highlight the significance of intimate relationships in emotion communication and pain reduction. Our results may therefore provide insights for the support therapies.

220

APPENDIX C

Additional Published Manuscript

Mindfulness (2018) 9:1522-1528 https://doi.org/10.1007/s12671-018-0900-9

ORIGINAL PAPER



Self-compassion Modulates Heart Rate Variability and Negative Affect to Experimentally Induced Stress

Xi Luo^{1,2} · Lei Qiao^{3,4} · Xianwei Che⁵

Published online: 17 February 2018 © Springer Science+Business Media, LLC, part of Springer Nature 2018

Abstract

Self-compassion has increasingly been recognized to buffer stress and promote emotional health. However, few studies have examined the influences of self-compassion on physiological stress response. The current study aimed to investigate the impact of self-reported self-compassion on physiological stress response and negative affect induced in a laboratory setting. Healthy male participants (N = 34, Asians) were grouped into high (N = 17, age: mean = 19.65, SD = 0.59) or low (N = 17, age: mean = 19.71, SD = 0.82) self-compassion groups based on the Self-Compassion Scale. They were subjected to the Trier Social Stress Test, with electrocardiography recorded and negative affect assessed by the Positive and Negative Affect Schedule. Results demonstrated that self-compassionate individuals showed higher vagally mediated heart rate variability (vmHRV) at baseline (CI = [0.30, 0.91], p = 0.01). Interestingly, self-compassionate individuals demonstrated higher vmHRV to an acute stressor after an anticipated decrease in vmHRV (CI = [0.02, 0.67], p = 0.04). Moreover, self-compassionate individuals reported less negative affect in response to stress (CI = [-8.29, -0.42], p = 0.03). Our results demonstrate the role of self-compassion in the flexible adjustment of physiological and psychological responses to stress.

Keywords Self-compassion · Stress · Emotion · Heart rate variability



ORIGINAL RESEARCH published: 14 August 2018 doi: 10.3389/fnhum.2018.00328



The Motivation-Based Promotion of Proactive Control: The Role of Salience Network

Lei Qiao^{1,2}, Lei Xu³, Xianwei Che⁴, Lijie Zhang², Yadan Li⁵, Gui Xue^{2,6}, Hong Li¹ and Antao Chen²*

¹College of Psychology and Sociology, Shenzhen University, Shenzhen, China, ²Key Laboratory of Cognition and Personality of Ministry of Education, Faculty of Psychology, Southwest University, Chongqing, China, ⁹Key Laboratory for Neuroinformation, Center for Information in Medicine, University of Electronic Science and Technology of China, Chengdu, China, ⁴Monash Alfred Psychiatry Research Centre, The Alfred and Central Clinical School, Monash University, Melbourne, VIC, Australia, ⁵MOE Key Laboratory of Modern Teaching Technology, Shaanxi Normal University, Xi'an, China, ⁶State Key Laboratory of Cognitive Neuroscience and Learning & IDG/McGovern Institute for Brain Research, Beijing Normal University, Beijing, China

It has been shown that reward motivation can facilitate proactive control, a cognitive control mode that is characterized of prior preparation and sustained holding of the goal-relevant information in working memory. However, it remains to be established the neural networks that may be involved in this promotion effect. In this study, participants underwent the AX-Continuous Performance Task (AX-CPT) that measures relative proactive control during functional magnetic resonance imaging (fMRI) scanning. We employed independent component analysis to decompose multiple brain networks and identified the task related network. Results showed that the salience network (SN) was engaged in the AX-CPT protocol. Importantly, our data demonstrated that reward modulated the association between task engagement of SN and proactive control, whereby the positive correlation was particularly observed in the reward condition. Moreover, reward modulated task engagement of the SN in a proactive manner, which may contribute to the behavioral proactive performance. Overall, our data suggest the involvement of SN in the reward facilitation effect of proactive control.

OPEN ACCESS

Edited by:

Carol Seger, Colorado State University, United States

Reviewed by: Mattie Tops,

VU University Amsterdam, VU University Amsterdam, Netherlards Debbie Yee, Washington University in St. Louis,

Keywords: dual-mechanism of control, independent component analysis, AX-Continuous Performance Task, salience network, proactive control, reactive control, reward



OPEN ACCESS

Citation: Li B, Cheng G, Zhang D, Wei D, Qiao L, Wang X, et al. (2016) Regional Brain Responses Are Blased Toward Infant Facial Expressions Compared to Adult Facial Expressions in Nulliparous Women. PLoS ONE 11(12): e0166860. doi:10.1371/journal.pone.0166860

Editor: Cosimo Urgesi, Universita degli Studi di Udine, ITALY

Received: January 28, 2016

Accepted: November 5, 2016

Published: December 15, 2016

Copyright: © 2016 Li et al. This is an open access article distributed under the terms of the <u>Creative</u> <u>Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data necessary for replicating the study's findings are within the paper and available from the figshare repository (DDI:<u>10.6084/m9.figshare.4109643</u>).

Funding: The authors received no specific funding for this work. RESEARCH ARTICLE

Regional Brain Responses Are Biased Toward Infant Facial Expressions Compared to Adult Facial Expressions in Nulliparous Women

Bingbing Li^{1,2}e, Gang Cheng^{3e}, Dajun Zhang^{1,2}*, Dongtao Wei^{1,4}, Lei Qiao^{1,4}, Xiangpeng Wang^{1,4}, Xianwei Che⁵

1 School of Psychology, Southwest University, Chongqing, China, 2 Center for Mental Health Education, Southwest University, Chongqing, China, 3 School of Educational Science, Guizhou Normal University, Guiyang, China, 4 Key Laboratory of Cognition and Personality (SWU), Ministry of Education, Chongqing, China, 5 Monash Alfred Psychiatry Research Centre, The Alfred and Central Clinical School, Monash University, Melbourne, Australia

Chese authors contributed equally to this work.

Abstract

Recent neuroimaging studies suggest that neutral infant faces compared to neutral adult faces elicit greater activity in brain areas associated with face processing, attention, empathic response, reward, and movement. However, whether infant facial expressions evoke larger brain responses than adult facial expressions remains unclear. Here, we performed event-related functional magnetic resonance imaging in nulliparous women while they were presented with images of matched unfamiliar infant and adult facial expressions (happy, neutral, and uncomfortable/sad) in a pseudo-randomized order. We found that the bilateral fusiform and right lingual gyrus were overall more activated during the presentation of infant facial expressions compared to adult facial expressions. Uncomfortable infant faces compared to sad adult faces evoked greater activation in the bilateral fusiform gyrus, precentral gyrus, postcentral gyrus, posterior cingulate cortex-thalamus, and precuneus. Neutral infant faces activated larger brain responses in the left fusiform gyrus compared to neutral adult faces. Happy infant faces compared to happy adult faces elicited larger responses in areas of the brain associated with emotion and reward processing using a more liberal threshold of p < 0.005 uncorrected. Furthermore, the level of the test subjects' Interest-In-Infants was positively associated with the intensity of right fusiform gyrus response to infant faces and uncomfortable infant faces compared to sad adult faces. In addition, the Perspective Taking subscale score on the Interpersonal Reactivity Index-Chinese was significantly correlated with precuneus activity during uncomfortable infant faces compared to sad adult faces. Our findings suggest that regional brain areas may bias cognitive and emotional responses to infant facial expressions compared to adult facial expressions among nulliparous women, and this bias may be modulated by individual differences in Interest-In-Infants and perspective taking ability.



M



Yonghong Zhang¹¹, Xi Luo¹¹, Xianwei Che² and Wenjie Duan^{3*}

¹ School of Cultural and Social Development, Southwest University, Chongqing, Chine, ² Monash Altred Psychiatry Research Centre, Central Clinical School and the Altred, Monash University, Melbourne, VIC, Australia, ³ Department of Sociology, Wuhan University, Wuhan, China

The literature has shown that self-compassion is a protective factor of an individual's

OPEN ACCESS

Edited by:

Alexander V. Libin, US Department of Veleran Atfairs and Georgefown University, USA

Reviewed by:

Michelle Dow Keawphalouk, Harvard University and Massachusetts institute of Technology, USA Giorgio Bartolotti, Fondazione Salvatore Maugeri (IRCCS), Italy

> *Correspondence: Wenjle Duan

[†]These authors have contributed equally to this work and joint emotional response to chronic stress. However, this stress-buffering effect has not been completely analyzed in individuals who report significantly high academic stress. The present study explored the role of self-compassion in a group of undergraduate students who experience chronic academic stress. A total of 208 undergraduate students who were preparing for the Postgraduate Entrance Examination (PEE) were recruited and completed the Self-Compassion Scale, Adolescent Self-Rating Life Event Check List, and Positive and Negative Affect Schedule. Differences analysis confirmed that the participants reported significantly higher academic stress than their peers who were not preparing for PEE. Self-compassion positively related to positive affect but negatively related to negative affect and learning stress. Further analysis showed that self-compassion negatively mediated the relationship between chronic academic stress and negative affect. Findings imply that self-compassion-centered interventions can be developed in the educational context to assist students cope with chronic academic stress.

Keywords: self-compassion, chronic academic stress, stress-buffering effect, emotional response, mediation

frontiers in Psychology Biological Psychology 112 (2015) 20-26



Contents lists available at ScienceDirect **Biological Psychology**

journal homepage: www.elsevier.com/locate/biopsycho

Habitual suppression relates to difficulty in regulating emotion with cognitive reappraisal



BIOLOGICAL PSYCHOLOGY

Xianwei Che^{a,b,d}, Xi Luo^c, Dandan Tong^{a,b}, Bernadette M. Fitzgibbon^d, Juan Yang^{a,b,*}

^a Key Laboratory of Cognition and Personality (SWU), Ministry of Education, Chongqing 400715, China
^b Department of Psychology, Southwest University, Chongqing 400715, China

Faculty of Culture and Social Development Studies, Southwest University, Chongqing 400715, China ^d Monash Alfred Psychiatry Research Centre, The Alfred and Central Clinical School, Monash University, Melbourne, Australia

ARTICLE INFO

Article history: Received 6 February 2015 Received in revised form 23 September 2015 Accepted 28 September 2015 Available online 22 October 2015

Keywords: Habitual suppression Cognitive reappraisal Affective disorder Gender difference

ABSTRACT

One common strategy to cope with the difficulties of daily life is suppression. Habitual users of sup-pression tend to suppress their feelings rather than expressing them. Although this strategy may reduce outward response to emotion, it is not thought to lessen induced negative affect. Moreover, it remains unclear whether people with high suppression scores can reduce negative affect through cognitive reappraisal. In the present study, twenty-nine healthy participants differing in suppression scores were directed to reappraise aversive stimuli during functional magnetic resonance imaging (fMRI). Results showed that higher suppression scores correlated with decreased response of dorsomedial prefrontal cortex (dmPFC) during cognitive reappraisal. Further, high suppression scores related to enhanced negative affect to stimuli with greater negative affect correlating with decreased dmPFC response during cognitive reappraisal. This study suggests that people with high suppression scores experience difficulty in reducing negative affect through cognitive reappraisal and implicates neurobiological processes that may underlie this difficulty.

© 2015 Elsevier B.V. All rights reserved.